ED 271 517	TM 860 447
AUTHOR TITLE	Rock, Donald A.; And Others Factors Associated with Decline of Test Scores of High School Seniors, 1972 to 1980. A Study of Excellence in High School Education: Educational Policies, School Quality, and Student Outcomes. Contractor Report.
INSTITUTION	Educational Testing Service, Princeton, N.J.
SPONS AGENCY Report no Pub date	Center for Statistics (OERI/ED), Washington, DC. CS-85-217 Dec 85
CONTRACT	OE-300-83-0247
NOTE	284p.
PUB TYPE	Reports - Research/Technical (143) Statistical Data (110)
EDRS PRICE	MF01/PC12 Plus Postage.
DESCRIPTORS	*Academic Achievement; Achievement Tests; College
	Preparation; Educational Assessment; *Educational Change; Educational Facilities Improvement;
	Educational Policy; Family Characteristics; Federal Programs: High Schools: *High School Seniors:
	Institutional Characteristics: Longitudinal Studies:
	*National Surveys; School Demography; Secondary
	School Curriculum; Student Atticudes; *Student
	Characteristics; Test Results
IDENTIFIERS	*High School and Beyond (NCES); National Longitudinal Study High School Class 1972; *Test Score Decline

#### ABSTRACT

Using data from the National Longitudinal Study of the High School Seniors Class of 1972 (NLS) and from High School and Beyond (HS&B), this study documents changes in the academic achievement of high school seniors between 1972-1980. It also identifies the school and student factrs related to these changes. Its purpose is to inform policymakers about the sources of current educational problems and to identify educational practices important for educational excellence. The findings indicate declines on all three achievement tests, with the largest declines in vocabulary and reading. Numerous and significant changes were identified in the characteristics of seniors, their homes and families, their schools, and their attitudes and behavior. A "step doun" analysis of covariance was used to estimate how these changes separately affected the average score decline. Path analysis was employed to explore differences in the educational processes experienced by 1980 subpopulations with larger score declines and their 1972 counterparts. Results indicate that the changes from 1972 to 1980 at both the school and student level that contributed most to test score decline were: (1) a greater likelihood of being in a general or vocational curriculum rather than an academic one; (2) a drop in the frequency of taking traditional college preparation core courses; (3) a decrease in homework; and (4) increase in student dissatisfaction with the lack of academic emphasis in schools. Findings support the policy recommendations of recent national reports in regard to curriculum, course content, homework and programs for special populations. (BS)

Factors Associated With Decline of Test Scores of High School Seniors, 1972 to 1980

A Study of Excellence in High School Education: Educational Policies, School Quality, and Student Outcomes

Educational Testing Service Princeton, New Jersey

Donald A. Rock Ruth B. Ekstrom Margaret E. Goertz Thomas L. Hilton Judith Pollack

William Fetters Project Officer Center for Statistics

Prepared for the Center for Statistics under contract OE-300-83-0247 with the U.S. Department of Education. Contractors undertaking such projects are encouraged to express freely their professional judgment. This report, therefore, does not necessarily represent positions or policies of the Government, and no official endorsement should be inferred. This report is released as received from the contractor.

December 1985

CS 85-217



#### ACKNOWLEDGEMENTS

This report is the first of two technical studies conducted by Educational Testing Service under Contract No. 300-83-0247 with the National Center for Education Statistics. The work presented here represents the efforts of many people.

The primary authors for Chapter I were Ruth Ekstrom and Margaret Goertz. Donald Rock and Barbara Heyns also contributed material to this chapter.

The main portion of Chapter II was written by Donald Rock. Thomas Hilton and Ruth Ekstrom also contributed to this chapter.

Chapters III and VI were written by Ruth Ekstrom.

Chapter IV was written by Margaret Goertz.

Chapter V was written by Donald Rock. Ruth Ekstrom also contributed to this chapter.

Chapter VII was written primarily by Thomas Hilton. Donald Rock also contributed to this chapter.

Chapter VIII and IX were written by Donald Rock.

Margaret Goertz is the primary author for Chapter X; Rock, Ekstrom, and Hilton also made contributions.

This report benefited from the work of several members of the ETS Division of Statistics and Psychometric Research and Services. Judith Pollack directed all data analyses, assisted by Thomas Jirele. Frederic Lord and Marilyn Wingersky provided advice and assistance in the IRT analysis. ETS visiting fellows Maria Pennock-Roman and Valerie Lee also made substantial contributions to this work.

Secretarial work on this report was done by the ETS Vydec staff and by Eleanor Hibbs, and Irene Smith. Kandy Olbrick, Project Manager, coordinated report production.

This report also benefited from the assistance of in lividuals outside of ETS. The project advisory committee members -- Charles Bidwell, University of Chicago; Barbara Heyns, New York University; and Robert Linn, University of Illinois -- provided valuable input. The final version of this report owes a considerable debt to the review by Lyle Jones, University of North Carolina; and Richard Shevelson of Rand, and by other anonymous reviewers.



Finally, we wish to thank the staff of the National Center for Education Statistics who provided us with a wide range of information and assistance that facilitated our work with these data bases. Special thanks go to our Project Officer, William Fetters.



EXCELLENCE IN HIGH SCHOOL EDUCATION: CROSS-SECTIONAL STUDY, 1972-1980

#### Executive Summary

In 1983, eight major national studies reported on the status of public education in the United States. These reports sounded a common theme: The American educational system is in trouble. The major evidence cited in support of this claim was that academic achievement, as measured by performance on the College Board's Scholastic Aptitude Tests and the National Assessment of Educational Progress, had declined. This situation was attributed to demographic changes, lower standards, lower expectations for students, a less rigorous curriculum, and the poor academic preparation of new teachers. However, there is little systematic research that relates these factors to test score decline.

This study, which was carried out by Educational Testing Service (ETS) under contract to the National Center for Education Statistics (NCES), utilized NLS and HS&B data to document changes in the academic achievement of high school seniors between 1972-1980 and to identify the school and student factors related to these changes. The study findings show that there were significant changes in test scores, in high schoo's, and in student behavior. They also show that these changes were ir errelated.

There were declines on all three achievement tests between 1972 and 1980. The largest declines occurred in vocabulary and reading. The average senior in 1980 (a student at the 50th percentile in 1980 in vocabulary and reading achievement) would rank at about the 41st percentile among the 1972 seniors in both vocabulary and reading. Similarly, a 1980 senior with average mathematics achievement in 1980 would be at the 45th percentile when compared with the 1972 seniors. When these changes are measured in standard deviation units, the declines are .22 for Vocabulary, .21 for Reading, and .14 for Mathematics, indicating a greater decline in verbal than in quantitative skills.

There were also significant changes from 1972-1980 in the characteristics of high school seniors, their homes and families, the schools they attended, and their attitudes and behaviors.

- Some demographic changes occured, such as increases in percentages of minority-group students and population shifts from the Northeast to the South.
- o The proportion of students in the academic curriculum declined, as did the number of semesters of social studies, science and foreign language taken, and the amount of homework done.
- o The percentage of schools with a high dropout rate increased. The number of laboratory courses taken by students fell, the proportion of students believing there should have been more



academic emphasis increased, and students had lower opinions of their school's reputation, quality of academic instruction, and physical condition of buildings.

- The parents of the 1980 seniors were better educated and had higher educational aspirations for their children, but provided fewer study aids.
- Students' interest in correcting social and economic inequities declined, while interest in making money and in job success increased. Students became more self-confident between 1972 and 1980 but less sure of their ability to control the course of their own lives.

The impacts of the above changes on test scores were examined. It was found that:

- Changes in student behaviors and in school characteristics played the major roles in test score declines.
- The demographic shifts were a minor factor in test score decline.
- The changes in the home educational support system resisted test score declines.

Changes from 1972 to 1980 at both the school level and student level that seem to have contributed most to the decline were: (1) a greater likelihood of being in the general or vocational curriculum rather than the academic curriculum, (2) a drop in the frequency with which students report taking "traditional" college preparation core courses such as foreign languages, science and/or courses requiring laboratory work, (3) a decrease in the amount of homework done, and (4) an increasing dissatisfaction among the students with the lack of emphasis on academics in the schools.

Taken together, these findings suggest that the major factor contributing to test score decline was a decreased academic emphasis in the educational process. The impact of this shift in emphasis fell primarily on White and on upper and middle class students, however. Federal and state programs designed to strengthen basic skills in reading and mathematics appear to have prevented comparable score declines among low socioeconomic status Blacks in Vocabulary and Reading and to have contributed to the score increase among this same group in Mathematics.



## TABLE OF CONTENTS

	Page
Chapter I - Introduction	1
A. Study Rationale and Issues to Be Discussed	2
and Practice.	3
C. Major Research Questions and Hypotheses	5
<ol> <li>Descriptive Cross-Sectional Analysis.</li> <li>Relational Cross-Sectional Analysis</li> </ol>	5 8
D. Report Overview	9
Chapter II - Instrumentation and Methodology	10
A. Instrumentation	10
1. Tests	10
	10
a. 1972 lests	11
c. Common Items. $\cdots$	12
2. Questionnaires	13
B. Methodology	14
1 Descriptive Analysis	. 15
2. Relational Analysis	. 16
Chapter III - The Samples and Background Changes	. 18
	19
A. The Sample.	. 10
B. Changes in Student Characteristics.	. 19
C. Changes in Students Family Background.	
Chapter IV - The Schools, Educational Programs, and Learning Conditions	. 25
A. Student Body Characteristics	. 25
1 Pacial/Ethnia Composition	. 25
2 Student Absontanism and Dronout Rates	. 26
3 College-Bound Students.	. 26
4. High School Curriculum.	. 26
5. Students with Special Educational Needs	. 27
B. Staff Character stics	. 28
1. Student/2 ff Ratios	. 28



### TABLE OF CONTENTS (continued)

В.	Staff Characteristics (continued)	
	2. Percentage of Staff with Advanced Degrees	28
	3. Teacher Turnover.	20
	4. Racial/Ethnic Composition of Staff	20
		29
с.	Educational Programs and Teaching Methods	29
	l. Educational Programs	29
	a. Handicapped Education	29
	b. Federal Programs.	29
	c. Advanced Placement.	30
	d. Ability Grouping.	30
		30
	2. Teaching Methods	31
D.	Student Evaluations of School Facilities and	
	Educational Experiences	
		32
	1. Student Ratings of Schools	20
	2 Student Evaluation of Brazze	.32
	3 Factors Interfering with Education	33
	J. Factors interfering with Education	35
Chapt	er V - Changes in Tested Achievement and School Grades	36
Α.	Comparisons of 1972 and 1980 Senior Vocabulary	
	Test Scores	•
В.	Changes in Besting Test Sector 1070 1000	36
c.	Changes in Mathematics Tech Scores 1972-1980	37
о. П	Changes in Mathematics lest Scores 19/2-1980	38
Б.	changes in Grades 19/2-1980	39
Chapte	er VI - Students' Attitudes, Values, and Behaviors	40
Α.	Educational Agnizations	
R	Students' Blans for the Einst Very Star Will a star	40
с. С	Ability to Complete C line First fear after High School	41
U. D	Ability to complete College	46
D. Б	influences of Others on Plans for after High School	46
Ľ.	Uccupational Aspirations.	48
F.	Career Values	53
G.	Life Values	55
Н.	Self-Esteem and Locus of Control.	56
I.	Time Spent on Homework.	58
J.	Extracurricular Activities.	50
K.	Courses Taken	61
		01



9

Page

# TABLE OF CONTENTS (continued)

	Page
Chapter VII - Partitioning Mean Test Score Changes	64
A. Methodology	64
1. Moment of Group Mean	65
2. Some Comments on Mean Score Change	66
B. Variables	66
C. Results	68
l. Vocabulary	68
a. Sex	68
b. Socioeconomic Status	70
c. Race/Ethnicity	71
d. School Type	72
e. Geographic Region	72
f. Curriculum	72
	73
h. Homework per Week	73
i Education Planned	74
i Study Aide in Home	74
k Attitude towards Academic Emphasis.	74
1 Instructional Variables	74
- School Variables	74
m. School vallables.	75
n. Exposure to Subject Matter	76
0. Participation in Athletics	76
p. Mother's Educational Aspirations for Student.	70
2. Reading	76
3. Mathematics	78
	~~~
D. Summary of Mean Score Change Partitioning	80
Chapter VIII - Partitioning Test Score Changes by Blocks	
of Variables Using Analysis of Covariance	88
Chapter IX - Path Analysis	95
A. Determinants of Mothers' Educational Aspiration	•
for the Children	96
B. Determinants of Private School Attendance	96
C. Choosing the Academic Curriculum	96
D. Number of Semesters of Language Courses	97
E. Amount of Homework	97
F. Semesters of Mathematics Courses	97
G. Determinants of Tested Achievement	98



## TABLE OF CONTENTS (continued)

			Page
Chapt	ter ]	K - Summary and Policy Implications	10 <b>2</b>
۸.	Sur	mmary of Study Findings	102
	1.	Test Score Change	102
		a. Vocabulary	102
		b. Reading	102
		c. Mathematics	103
		d. Comparison with SAT Score Decline	103
	2.	Changes in the Characteristics of High School	
		Seniors and Their Schools	103
	3.	Factors Affecting Test Score Change	104
В.	Pol	icy Implications	105
	1.	Curriculum	105
	2.	Course Content	106
	3.	Homework	106
	4.	Programs for Special Populations	10 <del>6</del>
Refer	ence	S	107
Appen	dix	A - Items Common to the 1972 and 1980 Student Questionnaires	
Appen	dix	B - Definitions of Variables and Coding Procedures	
Appen	dix	C - Cross-Tabulations of Test Scores	
Appen	dix (	D - Test Score Partitioning	
Appen	dix 3	E - Analysis of Covariance Partition of Score Change	



х

Table No.	Title	<u>P</u>	age		
3-1	Number cf Cases		20		
3-2	Parental Occupation		22		
3-3	Highest Education Level Completed by Father	Between	23	δ.	24
3-4	Highest Education Level Completed by Mother	Between	23	&	24
3-5	Number of Study Aids in Home	Between	23	&	24
3-6	Amount of Schooling Mother Wants Student	Between	23	å	24
4-1	Percent of Current Students Who Are White	Between	25	&	26
4-2	Percent of Current Students Who Are Black	Between	25	δ.	26
4-3	Percent of Current Students Who Are Hispanic	Between	25	δε	26
4-4	Approximate Average Daily Percentage Attendance	Between	26	δ	27
4-5	Percent of Students Who Enter 10th Grade But		_	_	
	Drop Out Before Graduation	Retween	26	δ	27
4-6	Percent of Last Year's Graduates Who Enrolled				
	in College	Between	26	&	27
4-7	Percent Enrolled in Academic or College Prepara-				
	tory Curriculum	Between	26	&	27
4-8	Percent Enrolled in General Curriculum	Between	27	&	28
4-9	Percent Enrolled in Vocationa-Technical				
	Curriculum	Between	27	&	28
4-10	Percent of Students Classified as Handicapped	Between	27	â	28
4-11	Percent of Students Classified as Disadvantaged	Between	27	&	28
4-12	Number of Students Per Teacher	Between	28	&	29
4-13	Percent of Full-Time High School Teachers With Master's or Doctor's Degrees	Between	28	&	29
4-14	Percent of Full-Time High School Teacher; Leaving				
	Since End of Last School Year	Between	28	&	29
4-15	Percent of Current Faculty Who Are White	Between	28	&	29
4-16	Extent of Special Accommodations for Handicapped Students	Between	29	&	30
4-17	Percent of High Schools Participating in Title I,				
- 1/	Elementary and Secondary Education	Between	29	&	30
4-18	Percent of High Schools Participating in				
4 10	Bilingual Education	Between	29	&	30
4-19	Percent of High Schools Participating in				
4 17	Vocational Education	Between	29	۶£	30
4-20	Percent of High Schools Participating in				
- 20	Consumer and Homemaking Education	Between	29	8	30
4-21	Percent of High Schools Offering Advanced				
	Placement Courses	Between	29	&	30
4-22	Percent of High Schools Using Ability Groupings	Between	29	&	30
4-23	Summary of Instructional Methods Used		31		
4-24	Use of Teacher's Lecture	Between	31	&	32
4-25	Use of Writing Essays, Themes, Poetry or Stories	Between	31	&	32
4-26	Use of Student Centered Discussions	Between	31	&	32
4-27	Use of Project or Laboratory	Between	31	&	32
4-28	Use of Individual Instruction	Between	31	&	32
4-29	Use of Teaching Machines or Computer Assisted				
2	Instruction	Between	31	&	32



xi

Table No.	Title		Pag	e	
4-30	Student Rating of Schools		33		
4-31	Student Rating of School Reputation	Between	33	3	34
4-32	Student Rating of School Condition	Botween	33	2	34
4-33	Student Rating of Academic Instruction	Between	33	3	34
4-34	Student Rating of School Library	Between	33	ع ع	34
4-35	Student Rating of Teacher Interest in Students	Between	33	a i	34
4-36	Student Evaluation of Program: Summary	Detween	34		54
4-37	School Should Have Placed More Emphasis on Basic		54		
	Academic Subjects	Retween	34	3	35
4-38	School Did Not Offer Enough Practical Work	Decween	54	ŭ	55
	Experience	Rot woon	34	a	35
4-39	School Should Have Placed More Emphasis on	Desween	• •		55
	Vocational and Technical Programs	Between	34	æ	35
4-40	School Provided Counseling That Will Help	Decacen	34	~	55
	Continue Education	Between	34	£	35
4-41	High School Provided Counseling That Will Help	Detween	34	ŭ	55
	Find Employment	Between	34	£	35
4-42	Factors Interfering With Education: Summary	Detween	35	~	55
4-43	Hard to Adjust to School Routine	Between	35	æ	36
4-44	Courses Too Hard	Between	35	å	36
4-45	Poor Teaching	Between	35	ۍ چ	36
4-46	Poor Study Habits	} ;ween	3. <sup>5</sup>	&	36
5-1	IRT Vocabulary Formula Score	Botucon	26	2	27
5-2	IRT Reading Formula Score	Between	27	04 04	21
5-3	IRT Mathematics Formula Score	Between	20	20 .2	20
5-4	Grades in High School	Between	30	Ω.	23
		Detween	55	u	40
6-1	Highest Level of Education Planned	Between	40	&	41
6-2	Students' Plans for First Year After High School		42-	-44	ŀ
6-3	Ability to Complete College	Between	46	&	47
6-4	Influence of Significant Others on Students' Post-High School Plans: Summary		47		
6-5	Influence of Parents on Plans for After High		-/		
	School	Botwoon	47	£	48
6-6	Influence of Teachers on Plans for After High	Detween			40
	School	Between	47	æ	68
<b>6-</b> 7	Influence of Guidance Counselor on Plans for	Detween		u.	40
	After High School	Between	47	æ	48
6-8	Influence of Friends and Relatives on Plans for	Detreen			-0
	After High School	Retween	47	æ	48
6-9	Type of Work Student Wants to Do at Age 30	Detween	49-	-51	40
6-10	Changes in Values Related to Career Plans:		-7	51	
	Summary		54		
6-11	Work That Seems Important and Interesting	Between	54	æ	55
6-12	Meeting and Working With Sociable. Friendly	2004001		-	
	People	Between	54	æ	55
6-13	Freedom to Make My Own Decisions	Between	54	ž	55
6-14	Job Security and Permanence	Between	54	&	55



Table No.	Title	<u> </u>	age	2	
6-15	Good Income	Between	54	&	55
6-16	Previous Work Experience in the Area	Between	54	&	55
6-17	Changes in Students' Life Values		56		
6-18	Success in Work	Between	56	&	57
6-19	Strong Friendships	Between	56	&	57
6-20	Marriage and Family Life	Between	56	&	57
6-21	Steady Work	Between	56	&	57
6-22	Better Opportunities for Your Children	Between	56	&	57
6-23	Working to Correct Social and Economic				
	Inequalities	Between	56	&	57
6-24	Lots of Money	Between	56	&	57
6-25	Being a Community Leader	Between	56	δ	57
6-26	Living Close to Parents and Relatives	Between	56	&	57
6-27	Getting Away from This Area of the Country	Between	56	&	57
6-28	Changes in Self-Esteem and Locus Control: Summary		57		
6-29	I Feel I Am a Person of Worth	Between	57	&	58
6-30	I Am Able to Do Things as Well as Most People	Between	57	&	5 <b>8</b>
6-31	I Take a Positive Attitude Toward Myself	Between	57	&	58
6-32	On the Whole, I'm Satisfied With Eyself	Between	57	&	58
6-33	Good Luck is More Important Than Hard Work for				
0 55		Between	57	&	58
6-34	Planning Only Makes a Person Unhappy	Between	57	&	<b>58</b>
6-35	Every Time I Try to Get Ahead, Something or				
0 00	Somebody Stops Me	Between	57	&	58
6-36	People Who Accept Their Condition in Life are				
0.00	Happier	Between	57	&	58
6-37	Time Per Week Spent on Homework	Between	58	&	59
6-38	Percentage of Students in Extracurricular				
0.00	Activities		59		
6-39	Athletics	Between	60	&	61
6-40	Debating and/or Music	Between	60	&	61
6-41	Subject Matter Clubs	Between	60	&	61
6-42	Vocational Education Clubs	Between	60	&	61
6-43	Newspaper and/or Yearbook	Between	60	δ.	61
6-44	Student Government	Between	60	δ.	61
6-45	Hobby Clubs	Between	60	&	61
6-46	Cheerleading	Between	60	&	61
6-47	Honorary Clubs	Between	60	<u>&amp;</u>	61
6-48	Semesters (Half-Years) of Mathematics	Between	62	&	63
6-49	Semesters of English	Between	62	. &	63
6-50	Semesters of Science	Between	62	<u>&amp;</u>	63
6-51	Semesters of Social Studies	Between	62	8	63
6-52	Semesters of Foreign Languages	Between	62	<u>&amp;</u>	63
7-1	Mean IRT Vocabulary Scores in 1972 and 1980 by Sex, With Partitioning		69	)	
7-2	Mean IRT Reading Scores in 1972 and 1980 by Sex, With Partitioning		77	,	



14

-

<u>Table No.</u>	Title	Page
7-3	Mean IRT Mathematics Scores in 1972 and 1980 by Sex, With Partitioning	79
7-4	Population Shift Effect, Subgroup Mean Change Effect, and Change in Moment of Grouping	
7-5	Variables by IRT Vocabulary Score Population Shift Effect, Subgroup Mean Change	81
	Effect, and Change in Moment of Grouping Variables by IRT Reading Score	82
7-6	Population Shift Effect, Subgroup Mean Change Effect, and Change in Moment of Grouping	
	Variables by IRT Mathematics Score	83
7-7	Largest Contributors for all Three Tests	84
8-1	Differences Between 1972-80 Senior Tests Score Means	
	When Each Block is Separately Held Constant	90
9-1	1972-80 Contracts of the Direct Effects of Major	
	Explanatory Variables on Test Outcomes	99
10-1	SAT Test Score Changes, 1972-1980	103

Figure No.

<u>Title</u>

Page

8-1	Adjusted Mean Differences for 1972-80 Test Scores by Selected Blocks of Explanatory Variables		92	&	93
9-1	Hypothesized Path Model to Explain Vocabulary Achievement	Between	95	£	96
9-2	Hypothesized Path Model to Explain Reading Achievement	Between	95	۔ ھ	96
9-3	Hypothesized Path Model to Explain Mathematics Achievement	Between	95	&	96



/

#### CHAPTER I

#### INTRODUCTION

The quality and effectiveness of American education has once again become a critical national issue. The National Commission on Excellence in Education (1983), appointed by Secretary of Education T. H. Bell, decries the "rising tide of mediocrity" in public education. The prestigious Twentieth Century Fund (1983) asserts that American public schools are in trouble. The National Task Force on Education for Economic Growth (1983), consisting of forty-one eminent leaders from state government and the corporate world, concludes that declining standards in public schools undermine both this country's efforts to sustain economic recovery and our competitive economic position internationally.

Research on schools tends to echo the message of such commissions, if not the tone. Recent studies of American high schools are premised on the assumption that secondary education is the weakest link in the instructional chain and the one most in need of reform. They argue that the basic structure of the American high school has not changed in nearly a century and no longer serves its purpose well (Sizer, 1983). While our schools have adjusted to a host of new demands in the last twenty-five years, a large gap remains between school achievement and the type of education students need in order to meet the demands of a technological society (Boyer, 1983). Students engage in a relatively narrow range of classroom activities and become more interested in personal and vocational goals and less interested in the intellectual goals of school as they get older (Goodlad, 1983).

Such studies have received and will continue to receive much publicity. Regrettably, the analyses and conclusions rest on relatively small samples of schools and pay scant attention to changing conditions. Does the average high school assign less homework, require fewer course credits for graduation, or permit more off-campus or part-time study than was true ten years ago? To what extent has fiscal retrenchment or school policy altered the quality, size, salary schedules, and degree of turnover among teachers? Commentators have linked the decline in test score performance among students to changing educational standards and criteria for graduation, but little systematic evidence exists to suggest that changes in school organization or curriculum are responsible. These and a host of related questions on the causes and consequences of educational effectiveness demand scrutiny.

Without doubt, the longitudinal studies initiated by the National Center for Education Statistics are the best resource for a systematic examination of both the current state of secondary schooling and the degree of change since the early 1970s. They offer an opportunity to conduct research pertinent to policy recommendations regarding effective reform. No other national data set on administrative practices and policy, on curriculum and requirements, and on student outcomes exist for an assessment of the changing nature of secondary education in this country. These data promise to yield a rigorous and exacting portrait of



American secondary education and the American high school student during a momentous decade of change. The longitudinal frame permits an investigation of the effects of variation across schools in educational processes and an adequate data base to infer causal relationships between school and student characteristics.

#### A. STUDY RATIONALE AND ISSUES TO BE DISCUSSED

This technical report is one of two reports that will be produced by Educational Testing Service as part of the Study of Excellence in High School Education. The general, long-term goal of this project is to improve school quality and, thus, produce excellence in high school education. The specific, short-term goals are to conduct two studies: 1) a cross-sectional analysis comparing 1972 high school seniors and their schools with 1980 high school seniors and their schools, which is the basis of this report, and 2) a longitudinal analysis relating growth and development of 1980 high school sophomores to their schooling experience over the period 1980-82.

This cross-sectional study has three major objectives: 1) to document changes in achievement and other student outcomes over time both nationally and by selected subpopulations, 2) to identify the school and student variables that are related to changes in student achievement and other outcomes, and 3) to present this information to educational policymakers in a way that will illuminate and assist their decision making.

There are both substantive and analytical issues addressed in this report. By substantive issues, we mean <u>what</u> we are looking at. By analytical issues, we mean <u>how</u> we look at these topics. The basic problem concerns the identification of school and student factors that are related to student outcomes. The major focus, however, is on those variables that can be changed through educational policy rather than on predetermined school characteristics.

The substantive issues are:

- How did the American high school and its students change between 1972 and 1980?
  - o Changes in student characteristics and family background.
  - Changes in student body characteristics, staff characteristics, educational programs, teaching methods, school facilities, and students' educational experiences.
  - o Changes in tested achievement and in school grades.
  - o Changes in students' educational and occupational aspirations, attitudes and values, and school behaviors.
- What factors account for changes in high school student outcomes?



- -3-
- o Demographic characteristics of students.
- o Student behaviors and attitudes (e.g., amount of homework done, number and type of courses taken, educational vs. occupational aspirations, etc.).
- o School characteristics (e.g., characteristics of teachers, curricular offerings, instructional methods, etc.).
- o Home educational support (e.g., parental influence on students' plans, study aids in the home, etc.).

The analytical issues are:

- What kinds of methodologies are needed to identify determinants of change in cross-sectional data?
- How can the effects of student characteristics be differentiated from the effects of school characteristics on student outcomes?
- B. RELEVANCE OF STUDY FINDINGS FOR EDUCATIONAL POLICY AND PRACTICE

In the last year, eight major studies have reported on the status of American education today (National Commission on Excellence in Education, 1983; Twentieth Century Fund, 1983; Education Commission of the States, Task Force on Education for Economic Growth, 1983; College Entrance Examination Board, 1983; The Carnegie Corporation, 1983; Sizer, 1983; Boyer, 1983; and Goodlad, 1983). These studies sounded a common theme: The American educational system is in trouble. The National Commission report issued the strongest indictment of the system, stating that the <u>average graduate</u> of our schools and colleges today is not as well-educated as "the average graduate of 25 or 35 years ago, when a much smaller proportion of our population completed high school and college."

The reports presented the following evidence of the scope and seriousness of the decline in academic achievement:

- Average achievement of high school students on most standardized tests is now lower than 26 years ago when Sputnik was launched (National Commission, p. 8).
- o The College Board's Scholastic Aptitude Tests demonstrate a virtually unbroken decline from 1963 to 1980, with the number and proportion of students demonstrating superior achievement also having dramatically declined (National Commission, pp. 8-9).
- o Successive national assessments throughout the 1970s have shown a steady decline in mathematics and science achievement (National Task Force, p. 5).
- o Remedial mathematics enrollments at 4-year colleges increased 72 percent between 1975 and 1980 (National Task Force, p. 5).



 Many 17-year-olds do not possess the "higher order" intellectual skills needed to function in a technological society.

The reports identified five areas that in large part explained the student outcomes mentioned above. First, demographic changes and changes in societal values have changed the role of schools. The schools have to teach more "hard-to-educate" youngsters skills that were once possessed by only a few, while providing a range of social services, such as performing the role of a parent, nurse, nutritionist, sex counselor and policeman (Twentieth Century Fund, 1983). Second, schools now expect and require less of students. The amount of homework assigned to high school seniors has decreased, the difficulty of subject matter has been reduced, grades have become inflated, and "minimum competency" examinations have replaced more rigorous standards of performance. Third, the content of education is less rigorous. More students are taking "general track" courses; fewer students are choosing to enroll in advanced mathematics and science courses. An emphasis on "back-to-basics" has diminished the concern for science and has emphasized computational skills rather than the mastery of mathematical concepts. Fourth, American high school students spend too little time on school work in terms of the number of hours spent in school, the number of days in the school year, and the time spent in class on academic instruction. For example, within a week's time of approximately 25 instructional hours in the nation's elementary schools, only one hour is devoted to science and less than four hours are devoted to arithmetic. Finally, not enough of the more academically able students are attracted to teaching. Existing teacher preparation and in-service training programs need improvement.

This study's research questions, listed in the following section, have been designed to inform policymakers about the sources of our current educational problems and to identify educational practices that appear to be important for educational excellence. For example, commentators have attributed the declines in test scores to higher levels of truancy and negative attitudes toward school; to increased drug and alcohol abuse by students; to increased amounts of time spent watching television rather than doing homework; to an increase in labor force participation by students; and to family factors such as increased divorce, marital disruption and smaller family size. This study compares the amount and rates of cognitive change among students having some of these attributes, and determines whether the declines in test scores are equal across such groups. If such factors yield common patterns of change, or if the composition of students in such categories between 1972 and 1980 does not change, these factors are unlikely candidates to explain the structural declines in test scores and other educational outcomes. These data, since they apply to comparable cohorts, are more pertinent to assess such changes than prior analyses based on SAT scores (Austin & Garber, 1982; Jencks, 1978). By comparing the 1972 and 1980 student populations, it is possible to isolate more sources of test score decline that relate to a changing profile of student characteristics during this period.



Although prior SAT score decline studies focused on a small portion of the high school cohort, they have been useful in generating hypotheses to explain test score decline. The Advisory Panel on the Scholastic Aptitude Test Score Decline (Wirtz, 1976) concluded that the decline had occurred in two phases, with a different explanation of each. In the first, from 1960 to 1972, the explanation rested in the fact that the SAT-taking population had undergone a drastic change, from a relatively small segment of the high school population headed for elite private colleges largely in the East to a much larger segment of the high school population that was more broadly representative of the range of abilities of high school seniors in the United States. In the second phase of the score decline, from 1972 to 1980, the number of students taking the SAT remained roughly constant, and also the total number of high school seniors. But throughout the period the score decline continued, apparently because of some other factor or factors. After considering approximately 75 hypotheses, the panel members concluded that "there is no one cause of the SAT score decline, at least as far as we can discern, and we suspect no single pattern of causes."

The panel did, however, mention six possible causes:

- 1. the proliferation of elective courses,
- 2. the lowering of academic standards,
- 3. the competition of television,
- 4. the weakening of the role of the family in the educational process,
- 5. national tensions, and
- 6. diminution of students' learning motivation.

This section has provided an introduction to the policy issues that are addressed in this study. The research questions expand on these issues.

#### C. MAJOR RESEARCH QUESTIONS AND HYPOTHESES

The following questions are examples of those addreased in the analysis.

- 1. Descriptive Cross-Sectional Analysis
  - a. How much and in what direction did test scores change between 1972 and 1980? Are these changes consistent across type of student (gender, race/ethnicity, SES), type of school, region of the country and curriculum? Are these changes consistent across test content areas?



20

We hypothesize, based on similar SAT analyses, that:

- o Test scores declined between 1972 and 1980;
- The test score decline was less for minority students than for majority students;
- The size and direction of the test score changes varied across curriculum (e.g., academic, general and vocational); and
- The changes in test scores were greater on things taught directly (e.g., mathematics) than on things taught less directly (e.g., vocabulary).
- b. How much and in what direction did students' self-reported grades change between 1972 and 1980? Are these changes consistent across type of student, type of school, region of the country and curriculum?

We hypothesize, based on other studies showing grade inflation, that there was an increase in mean grades reported from 1972 to 1980. We also hypothesize that this change was relatively uniform across students, schools, regions, and curriculum.

c. How much and how did students' educational and occupational aspirations change from 1972 to 1980? Are these changes consistent across type of student curriculum?

We hypothesize that:

- More 1980 seniors than 1972 seniors planned post-secondary education;
- o More 1980 seniors than 1972 seniors planned to attend a community college;
- o The rise in educational aspirations was greatest among women and minorities;
- o More seniors in the general curriculum had aspirations for college in 1980 than in 1972;
- Students' occupational aspirations changed between 1972 and 1980; more 1980 seniors were interested in professional and technical occupations, while fewer 1980 seniors were interested in clerical occupations or in full-time homemaking; and
- o Occupational aspirations changed more for females and minorities.



d. How much and how did student behaviors, attitudes and values change between 1972 and 1980?

We hypothesize that:

- o The amount of homework done by students decreased from 1972 to 1980 and that this decline was consistent across curriculum;
- o The participation rates in extracurricular activities increased from 1972 to 1980, but the percentage involved in honorary societies and other academically oriented activities declined;
- o Students' evaluations of their school experiences were less positive in 1980 than in 1972;
- Students took fewer math, science and foreign language courses in 1980 than in 1972;
- o There were fewer students enrolled in the academic curriculum and more enrolled in the general curriculum in 1980 than in 1972;
- o There was an increase, from 1972 to 1980, in students' confidence about their ability to complete college and this increase was greatest for women and minorities; and
- More 1980 students were concerned with money and job security and fewer with social problems than 1972 students; these patterns were consistent across all groups of students;
- e. How much and how did student background and family characteristics change between 1972 and 1980?

We hypothesize that:

- o There was a higher proportion of minority students, educationally disadvantaged students, and students classified as handicapped in the schools in 1980 as compared with 1972; and
- o Parents of 1980 students had a higher mean education level than parents of 1972 students.
- f. How much and how did schools change between 1972 and 1980?

We hypothesize that:

- o There were more schools in which the majority of students were enrolled in a nonacademic curriculum in 1980 than in 1972;
- o The student teacher ratio decreased from 1972 to 1980;
- Advanced placement courses were more available in 1980 than in 1972;



- Student absenteeism and dropout rates were higher in 1980 than in 1972;
- o Teacher turnover rates were higher in 1980 than in 1972;
- A larger percentage of teachers had master's degrees or doctorates in 1980 than in 1972; and
- o There were more schools that were prelominantly minority in 1980 than in 1972.

#### 2. <u>Relational Cross-Sectional Analysis</u>

a. What was the effect of changes in demographic characteristics of high school seniors, changes in student attitudes and schoolrelated behaviors, changes in students' home environments, and changes in school characteristics on test scores?

We hypothesize that the test score decline reflects:

- A change in the racial/ethnic mixture in the sample of testtakers;
- A decline in the amount of time devoted to traditional academic subjects, such as English, foreign languages, science, mathematics, and social science;
- A reduction in the amount of writing required of students and in the number of laboratory courses taken;
- o A reduction in the amount of time students devote to doing homework;
- o An increase in the holding power of the high school;
- o An increase in parental education; and
- o An increase in parents' educational aspirations for students.
- b. Do members of different subgroups experience different educational processes which explain differences in achievement outcomes? Did these subgroups go through different educational processes in 1972, and in 1980?

There are a number of other hypotheses which have been discussed in the popular press and would therefore be tempting to investigate (such as score decline being related to an increase in student television watching, to a deterioration in discipline in the schools, or to an increase in students coming from single parent families). Unfortunately, the crosssectional data do not permit these comparisons. Many of these hypotheses, however, will be explored in the longitudinal 1980-82 study.



-8-

#### D. REPORT OVERVIEW

-9-

The remainder of this report is divided into ten chapters. The next chapter (Chapter II) describes the study instrumentation and methodology. Next are four chapters which provide a descriptive analysis of the changes between 1972 and 980. Chapter III provides a description of the two samples. It also covers changes in student background and family characteristics. Changes in schools are covered in Chapter IV; this includes changes in student body characteristics, in staff characteristics, in educational programs and teaching methods, and in student evaluations of school facilities vs. their educational experiences. Chapter V covers changes in test scores and in self-reported grades. In Chapter VI changes in students' school-related behaviors, attitudes and values are described. The next three chapters provide a relational analysis focusing on test score changes as the major outcome. Chapter VII covers the partitioning of mean score changes. In Chapter VIII partitioning is done using analysis of covariance to look at the relative impact of blocks of variables on score changes while controlling for other blocks of variables. The final relational analysis, covered in Chapter IX, uses path analysis to explore how the members of those groups which showed the greatest test score decline differ in educational processes from the members of those groups with less decline. Chapter X provides a summary and policy recommendations.



## -10-

#### CHAPTER II

#### INSTRUMENTATION AND METHODOLOGY

This chapter describes, briefly, the tests and questionnaires used in this study and the methods used to analyze them. Readers interested in more detailed information about the tests should refer to the Psychometric Analysis report for this study (Rock et al., 1984a). Further information about the analysis methodology can be found in the project Research Design Statement (Rock et al., 1984b).

#### A. INSTRUMENTATION

Two different types of instruments were used to obtain the data used in this analysis--tests and questionnaires. Both the test battery and the questionnaires underwent a number of changes between 1972 and 1980. This cross-sectional analysis is, therefore, limited to the test and questionnaire items that are common to both years.

#### 1. Tests

The cognitive<sup>1</sup> tests used in the National Longitudinal Study of the High School Seniors Class of 1972 (NLS) and in <u>High School and Beyond</u> (HS&B) have a long and complex history. In this chapter we will provide a brief description of the test batteries and their interrelationships.

a. <u>1972 Senior Tests</u>. In the spring of 1972, 18 randomly selected students in each of a sample of 1,044 randomly selected high schools took a battery of cognitive tests as part of the base-year survey of the longitudinal study which was to continue for an unspecified time. As of this writing, four follow-ups have been conducted, and a fifth follow-up is in the planning stage. The battery consisted of six tests which are listed in the left-hand column of Figure 1. These tests and a brief description of each follow:

- Vocabulary Fifteen moderately difficult items consisting of one word followed by five possible synonyms. Test-taker selects one word or phrase whose meaning is closest to that of stem. Time - 5 minutes.
- Picture-Number Test of short-term associative memory in which the testtaker first studies pairs of pictures and 2-digit numbers and then is shown the pictures only and is asked to select the number on the answer sheet that was paired with picture. Time -3 minutes to study 15 items in Part 1, and 2 minutes to answer; similarly for Part 2.

We use the adjective "cognitive" to describe a broad category of tests that includes basic intellectual skills, achievement, developed ability, and scholastic aptitude.



Reading - Relatively unspeeded measure of reading comprehension in which 5 reading passages are given and test-taker answers multiplechoice questions (with 5 options) concerning what is stated or implied in each passage. Time - 15 minutes.

- Letter Groups A test of inductive reasoning where each item consists of five groups of letters. The test-taker determines which four groups share a common characteristic and indicates the group which differs from the others. Time - 15 minutes.
- Mathematics Twenty-five items in which the test-taker indicates which of two quantities is greater, or equal, or that the data given are insufficient to make a decision. The items were selected not to require specific algebraic, geometric or trigonometric skills. Time - 15 minutes.
- Mosaic Comparisons This test was used as a highly speeded measure of perceptual speed and accuracy. The subject compares one hundred and sixteen pairs of tile-like patterns to detect the location of small differences in the designs. Time - Part 1 (56 items), 3 minutes; Part 2 (33 items), 3 minutes; Part 3 (27 items), 3 minutes.

Total testing time - 69 minutes.

The test battery was administered by a survey administrator in each school who usually was a guidance counselor or an experienced teacher. The students marked their answers in a separate mark-sensed answer sheet, not in the test booklet.

b. <u>1980 Tests</u>. In the spring of 1980, as part of High School and Beyond, 36 randomly selected seniors in each of 1,015 high schools took test batteries that roughly paralleled the 1972 test. As shown in Figure 1, the 1980 tests were quite similar to the 1972 tests. The entire Letter Groups test was dropped, as well as parts of two other tests, to make room for a test of spatial relations (Visualization in Three Dimensions) and a self-report measure of the student's reactions to the testing situation ("Questions About the Tests"). Brief descriptions of the two instruments added to the 1980 battery follow.

Visualization in Three Dimensions - This test is a measure of "the ability to visualize how a figure would look after manipulation in three-dimensional space, by folding a flat figure to make a three-dimensional figure." Each of the 16 items in the test has a drawing of a flat piece of metal in the left-hand column and drawings of five objects on the right, only one of which could be made by folding the flat piece of metal. The test-taker selects the one object that could have been made. Time - 9 minutes.



Questions About the Tests - This 6-item self-report questionnaire was designed to tap factors that may have prevented the test-takers from performing as well as they might have under optimum testing conditions. Included are questions inquiring about the importance of the testing to the students, their concern about doing well, how much they enjoyed participating, and how they felt while taking the tests. Time - 5 minutes.

The total 1980 test battery and the time allowed were as follows:

Vocabulary

Part l	5	minutes
Part 2	4	minutes
Reading	15	minutes
Mathematics		
Part l	15	minutes
Part 2	4	minutes
Picture-Number	5	minutes
Mosaic Comparisons		
Part 1	3	minutes
Part 2	3	minutes
Visualization in Three Dimensions	9	minutes
Questions About the Tests	5	minutes

Total time - 68 minutes

c. <u>Common Items</u>. In a report to NCES (Donlon et al., 1978), ETS recommended that the Letter Groups, Picture-Number, and Mosaic Comparisons tests be dropped from the test battery. Surveys of users of the 1972 public release tape and of the research literature indicated that data from these three tests had been little used. Also, it had been argued that "Measures of basic cognitive skills are not designed to assess patterns of change over time" (Haertel & Wiley, 1978). ETS concurred with these views. In the Psychometric Analysis, we found that mean scores increased dramatically between 1972 and 1980 on both the Mosaic Comparisons test and the Picture-Number test. Both of these tests require careful supervision during their administration. Without this control, examinees can refer back to the study pairs of pictures while taking the Picture-Number test and inflated scores would result. Mosaic Comparisons is highly speeded, and scores can be dramatically increased if time limits are not carefully monitored. The Psychometric Analysis found the correlations between the first and second halves of the two



parts of the Mosaic Comparisons test were low (.25 to .36), strongly suggesting that this test has low reliability. In addition, as described in the Psychometric Report, in 1972 the sample members responded on a separate answer sheet and in 1980 they responded in the test booklet--a change which is known to affect responses. Given all of these factors, it was decided to eliminate the Picture-Number and Mosaic Comparisons tests from further analysis for this study.

Of the remaining tests, Vocabulary and Reading were identical in the two batteries and 18 of the 25 mathematics items either were identical (12) or had minor editorial or format changes (6). Item response theory (IRT) was used to score and equate tests across populations. Using IRT, Mathematics, Vocabulary and Reading scores were put on the 1972 score scale. The IRT equated items are the basis of the test score comparisons in this cross-sectional study. Additional technical information is provided in the Psychometric Analysis (Rock, et al., 1984a).

#### 2. Questionnaires

The 1972 and 1980 data collections also utilized questionnaires to gather information from students and their schools. These questionnaires provide a rich source for studying the changing demographics of American high schools, changing school conditions, and changing attitudes, values and behaviors among the students.

There were four data collection forms used in 1972 in addition to the student tests. These were: the Student Questionnaire, the Student's School Record Information Form, the School Questionnaire, and the Counselor Questionnaire. In 1980, a Student Questionnaire, a School Questionnaire, and a Teacher Questionnaire were the main data collection instruments supplementing the student tests.

The 1972 Questionnaire is divided into four sections covering: 1) high school experiences, 2) attitudes and plans, 3) plans for the future (with separate subsections for those planning to work full-time during the year they leave high school, those planning to enter military service, those planning to become homemakers, those planning to take vocational or technical courses, those planning to go to a two- or four-year college or to a university, and those planning part-time work), and 4) a final section with information primarily demographic in nature. There were a total of 107 items in this questionnaire. The 1980 Senior Questionnaire covers much of the same material as the 1972 version. There are a total of 121 items. Although many of the items are the same in these two questionnaires, there was addition, deletion, and rephrasing of questions. Table 1 in Appendix A shows the comparable or similar items from the two student questionnaires used in this analysis.

The 1972 School ( estionnaire is divided into three sections covering: 1) program and student information, 2) resources, and 3) the grading system. The 1980 School Questionnaire is similar but, again, includes



-13-

various changes. Table 2 in Appendix A shows the comparable or similar items from the two school questionnaires.

Data from the other questionnaires were used primarily to confirm or elaborate on the information in the Student and School questionnaires.

#### B. METHODOLOGY

This section describes the methodology for the 1972 and 1980 comparisons. The descriptive analysis not only documents changes in student achievement, background, behavior and attitudes and in their schools, but it also provides a subset of critical input and process variables for use in the relational analysis.

The classification variables and subcategories used in this crosssectional descriptive analysis are shown below. A complete list of the outcome variables are included in Appendix A, and the classification variables are defined in Appendix B.

- 1. Sex--male and female;
- Race/Ethnicity--White, Black, Asian-American, American Indian, Mexican-American, Puerto Rican, and Other Hispanics;
- 3. Socioeconomic Level--high, middle, and low;
- 4. Type of School--public, private, and private-parochial;
- 5. Community Type--urban, suburban/small city, and rural;
- 6. Geographic Region--Northeast, North-Central, South, and West;
- 7. Curriculum Type--academic, general, and vocational; and
- 8. Administrative Population--1972 and 1980.

For each continuous outcome variable, we provide an introductory descriptive analysis table showing the mean and standard deviation for that variable, for 1972 and 1980, by each of the first seven classification variables. We also show the 1980-1972 mean difference and the effect size of this difference. For categorical outcome variables we show the percentage choosing each option and the 1980-1972 in percentages. For most outcome variables, we provide additional descriptive analyses showing 1972 and 1980 differences categorized in three-way tables which include sex by curriculum, socioeconomic status by race, socioeconomic status by school type, socioeconomic status by geographic region, socioeconomic status by curriculum, and socioeconomic status by community type.

An asterisk on a number in the column "1980-1972 difference" indicates that the difference between means is statistically significant at



the .05 level or less. The standard errors used in the statistical test of the difference between test score means uses a sample design effect correction (deft) of approximately 1.6. The test score correction factor (deft) used was based on the 1980 total Senior sample. The deft for the 1972 sample was slightly less, but in the interest of providing a conservative test, the constant value of 1.6 was used for test scores in both samples. This total sample deft was also used to correct the test score standard errors within subpopulations. This decision also leads to conservative statistical tests since in the vast majority of 1972-1980 subsample comparisons, the subsample defts were slightly smaller than the deft of the total sample. It was felt that it would be more judicious to err on the side of finding no differences (i.e., no change), especially in those instances where there were relatively few numbers of data cases. Separate defts for 1972 and 1980 were used for correcting the standard errors of percentages. The total sample defts were used in the subsamples as well as for totals.

The column labeled "effect size" is the difference between means divided by the pooled standard deviation. This measure of effect size is scaled in terms of standard deviation units, and since it is independent of sample size, it allows one to make rough comparisons of the relative magnitude of changes across populations and/or in outcome variables having different metrics.

What can one say about whether an effect size is small, moderate, or large? Cohen (1969) suggests that comparisons of treatments in the social sciences frequently yield effects sizes of .20 and below while very few ever yield effect sizes as large as .80 and above. Similarly, Smith and Glass (1977) report average effect sizes of .68 in treatmentcontrol comparisons. It should be pointed out here that these notions about what is a small, moderate, or a large effect are for the most part gathered from empirical data where the comparison is between a group receiving a formal intervention and a non-treated control group, or alternatively a group receiving what is believed to be an inferior treatment.

Since the comparison here is between two relatively similar populations, receiving similar treatments but separated in time, one should probably be more modest with respect to expectations about obtained effect sizes. That is, considering the context of these 1972-1980 comparisons the following categories of effect sizes will be used in succeeding interpretations. A statistically significant effect between 10 percent and 20 percent of a pooled standard deviation will be considered a small but practically significant effect. Effect sizes of 21 percent to 50 percent of a standard deviation will be considered to be moderate-sized effects while 51 percent of a standard deviation and larger will be considered large effects.

#### 1. Descriptive Analysis

The descriptive analysis is targeted toward answering four major questions:



a. <u>How did the students and their background characteristics change</u> between 1972 and 1980?

In Chapter III we describe the 1972 and 1980 samples. We show changes in the percentage of males and females, the percentage of students in each racial/ethnic category, the percentage of students enrolled in different curricula, the socioeconomic background of the students, and the community type and region of the country in which they reside. We also examine changes in parental education and occupation.

b. <u>How did schools, their educational programs, and other learning</u> conditions change between 1972 and 1980?

Changes in student body characteristics, including absenteeism and dropout rates and the percentage of college-bound students; in staff characteristics; in educational programs; and in students' evaluation; of their school experiences are presented by four of the major classification variables in Chapter IV.

c. <u>How much did tested achievement and school grades change between</u> 1972-1980?

In Chapter V, summary statistics for the mean test score changes are presented in IRT scaled units and in effect size scaled units. Information on self-reported grades is also included.

d. <u>How much did behaviors, attitudes, etc., change for various</u> groups of students and schools?

Changes in homework, extracurricular activities, attitudes toward school, educational and occupational aspirations, selfesteem, and life/work goals are presented in Chapter VI using the seven major classification variables. When the dependent variables are on a quantitative scale, means and standard deviations are presented. Scaled effect sizes are presented where there is a comparison of two means. When the data is nominal, tables show cell, row, and column marginal percentages and frequencies.

#### 2. Relational Analysis

One major concern of the relational analysis is to determine the extent that changes in test scores are related to changes in population composition. When interpreting the difference between distributions of the same outcome for two populations, demographers and social scientists must be very cautious to recognize structural differences in the populations that might partly or wholly explain the observed differences (Das Gupta, 1978). Researchers can begin to postulate external causes for the observed change only if there are no shifts in population characteristics or if the effects of those shifts can be accounted for.



The first question the relational analysis of test scores addresses is: What are the relatively manipulable and non-manipulable individual background and family, school, and community characteristics that are related to each of the achievement test scores within each population? These explanatory variables were selected from the list of items common to the 1972 to 1980 Student and School questionnaires. Secondly, the shift in the distribution of 2ach variable common to both 1972 and 1980 is examined since the degree to which a particular variable may contribute to a change in the mean score for the general population depends on both the change in mean score for the members of a particular category and on the extent to which the relative size of the category may have changed from 1972 to 1980.

Two techniques were used to partition the test score decline. The first technique describes the extent of the relationship between selected population and behavioral shifts to score decline. This type of analysis provides considerable detail about how classifying an individual on one or two variables at a time relates to test score changes between 1972 and 1980. In addition, this methodology attempts to partition the total score change into that part that was due to population shifts in the classification variables and that part that was due to mean changes within the classification groupings.

This partitioning procedure, however, does not lend itself to evaluating the impact of any one given classification variable or set of classification variables on score changes while controlling for the effects of numerous confounding variables. The second procedure that was used attempts to look at the relative impact of selected blocks of variables on the 1972-1980 mean score changes both before and after controlling for other confounding blocks of variables. The four blocks of variables are (1) demographics (e.g., race/ethnicity, sex); (2) student behavior and attitudes; (3) school characteristics; and (4) home educational support systems (e.g., parental influences, parental education, etc.). This partitioning procedure uses a "step down" analysis of variables that form a block while controlling for the remaining blocks. A second step in this step down ANCOVA is the identification of the variables in each block that contribute the most to that particular block's net effect on mean score change.

In a sense this method takes the multiplicity of findings from the first method and sorts them into logical sets or blocks and summarizes their net impact on mean test score change.

The above two methods are primarily exploratory and descriptive in nature. The third and final method contrasts path analysis models for the 1972 and 1980 cohorts separately in an effort to shed light on what changes in processes might have occurred to account for both the overall and, as well, differential score decline.



#### CHAPTER III

-18-

#### THE SAMPLES AND BACKGROUND CHANGES

This chapter describes the 1972 and 1980 samples and how the background characteristics of students in these samples changed between 1972 and 1980.

#### A. THE SAMPLE

Table 3-1 shows the size of the 1972 and 1980 student samples. Both the actual number of cases and the weighted estimate of the population size (N), used for generalizing to national samples, are presented. Also shown here, but not included in other tables, is the number of cases for which information on one or more classification variables is missing. Definitions of the classification codings are presented in Appendix B.

In both years, students were selected through a two-stage probability sample, with schools as the first stage units and students within schools as the second stage units. With the exception of special strata, schools were selected with probability proportional to estimated enrollment, and within each school, seniors were randomly selected.

The NLS-72 sample design called for the selection of a deeply stratified national probability sample of 1,200 public and private high schools and the selection of simple random samples of 18 seniors, where possible, and one or two counselors from each school. Schools in low income areas or with high percentages of minority-group students were over-sampled. Students from backup or substitute schools were also included in the study, resulting in a final sample of 1,318 schools.

The 1980 High School and Beyond study design called for a highly stratified national probability sample of 1,122 high schools with 36 seniors and 36 sophomores per school. (In those schools with fewer than 36 seniors or sophomores all eligible students were included in the sample.) Over 30,000 sophomores and 28,000 seniors enrolled in 1,015 public and private high schools across the nation participated in the base-year survey. Once again over-sampling was done for special strata schools including schools that were predominately Hispanic, Catholic schools that had substantial Black enrollments, alternative schools, high performance private schools, and other non-Catholic private schools.

Detailed information about the 1972 sample can be found in the NLS Data File User's Manual (Levinson, Henderson, Ricaobono, & Moore, 1978). Detailed information about the 1980 sample can be found in the High School and Beyond Sample Design Report (Frankel, 1981).

Although the NLS-72 and HS&B sample designs specified that students in all but the special strata would be selected with approximately equal probabilities, the probabilities are only approximately equal. The



sample as realized did not equal the sample as drawn, creating further deviations from a self-weighting sample. Weights were introduced for schools and for students, giving each school or each student a weight equal to the number of schools or students in the universe of schools or students which that school or student represents.

#### **B. CHANGES IN STUDENT CHARACTERISTICS**

As Table 3-1 shows, the population estimate of the number of high school seniors was virtually the same in 1972 and 1980. But the composition of the senior group changed considerably. In 1972 high school seniors were about evenly divided between males and iemales. Estimates from the 1980 HS&B sample showed males constituting 48.1 percent of all high school seniors and females 51.9 percent of all seniors. These proportions do not include about five percent of the students who did not report their gender. Further analysis enabled us to estimate that this group was approximately 60 percent male. Readjustments using this information would indicate that the 1980 seniors were 48.6 percent male and 51.4 percent female. Smaller declines in the proportion of male high school seniors have been reported elsewhere. For example, NCES data on high school graduates shows that males were 49.4 percent of 1972 graduates and 49.0 percent of 1980 graduates (The Condition of Education, 1984).

Because of the way individuals were classified into SES groups, no interpretation of shifts in SES group membership will be made.

Racial/ethnic composition also changed. White students declined from an estimated 85.8 percent of 1972 seniors to 79.9 percent of 1980 seniors. Black high school seniors increased from an estimated 8.7 percent in 1972 to 11.6 percent in 1980. Hispanics, including Mexican-Americans, Puerto Ricans, and other Hispanics, increased from an estimated 3.5 percent of 1972 seniors to 6.5 percent of 1980 seniors. There was also a slight increase in the estimated proportion of Asian-American students (from 0.9 percent to 1.3 percent). The estimated proportion of American Indian students declined from 1.1 percent to 0.7 percent. All these population estimates are, of course, subject to sampling and non-sampling errors.

There were also major changes between 1972 and 1980 in the curriculum tracks in which seniors were enrolled. The estimated proportion of seniors enrolled in the academic curriculum decreased from 45.7 percent in 1972 to 38.1 percent in 1980. Estimated enrollments in the general curriculum increased from 31.8 percent of the 1972 seniors to 37.2 percent of the 1980 seniors. There was also a slight increase in the estimated proportion of students in the vocational curriculum (from 22.5 percent in 1972 to 24.7 percent in 1980).

A slightly smaller proportion of 1980 seniors (90.0 percent) than 1972 seniors (91.5 percent) was enrolled in public schools. The estimate of the proportion enrolled in Catholic schools declined from 7.9 percent in 1972 to 6.6 percent in 1980 while the estimated proportion of all



## TABLE 3-1

# NUMBER OF CASES

		NLS 1972		H58 1980					
	SAMPLE	WEIGHTED N	×	SAMPLE	WE1GHTED N	*			
TOTAL	16683	3043598		28240	3040928				
SEX									
Male	828 1	1517010	49.9	12907	iu00722	<b>49</b> 1			
Female	8397	1525571	50.1	14086	1512617	51 0			
No Data	5	1017		1247	127789	/1.7			
SES									
Low	4827	74 16 12	24.5	8409	811768	27.4			
Middle	7927	1554775	51.3	12901	1423448	49.1			
Hiah	3863	735728	24.2	.180	723528	26 5			
No Data	66	1 148 3		850	82184	2407			
RACE/ETHNICITY									
White	12847	2527200	85.8	19852	2364647	79 9			
Black	2127	256777	8.7	3775	344397	11.6			
Asian-American	193	27740	.9	365	39373	1.3			
American Indian	189	3 1400	1.1	217	22254	.7			
Mexican-American	558	73285	2.5	1893	102170	3.5			
Puerto Rican	96	9764	.3	308	18169	.6			
Other Hispanic	122	18844	.7	976	67166	2.4			
No Data	55 1	98589		854	82753				
SCHOOL TYPE									
Fublic	14957	2701422	91.5	24678	2736069	90.0			
Private	67	16549	.6	875	104730	3.4			
Catholic	1027	235795	7.9	2687	200129	6.6			
No Data	632	89832		0	0				
GEOGRAPHIC REGION									
Northeast	3618	804775	26.4	5689	696768	22.9			
North Central	4568	917658	30.2	8102	869669	28.6			
South	5513	796009	26.2	9309	924433	30.4			
West	2984	525157	17.2	5140	550057	18.1			
NO Data	Û	0		0	0				
CURRICULUM									
General	5673	968623	31.8	10293	1112603	37.2			
Academic	6812	139 1944	45.7	10532	1138492	38.1			
Vocational	4197	682728	22.5	6959	740965	24.7			
No Data	1	303		456	48867	•••			
COMMUNITY TYPE									
Urban	4563	787529	26.5	6524	610511	20.1			
Suburban	7965	1540863	51.9	13580	1502435	49.4			
Rural	3684	639947	21.6	8136	927981	30.5			
No Data	471	75260		0	0	-			



.

seniors enrolled in private schools increased from 0.6 percent in 1972 to 3.4 percent in 1980.

National shifts in population are also evident in these data. The estimated proportion of all high school seniors from the Northeastern and North Central regions declined while the estimated proportion from the South and the West increased.

There were also shifts in the type of community from which the seniors came. In 1972, it was estimated that 26.5 percent of the seniors came from urban communities, 51.9 percent from suburban communities, and 21.6 percent from rural communities. By 1980, it was estimated that 20.1 percent of the seniors were from urban communities, 49.4 percent from suburban communities, and 30.5 percent from rural communities.

In sum, the 1980 high school seniors were more likely to be female, members of a minority group, enrolled in a nonacademic curriculum, enrolled in a non-Catholic private school, from the South or West, and from a rural area than were the 1972 seniors. These shifts in the nature of the high school population have important consequences for achievement and for attitudes and values, as will be seen in Chapters V and VI.

#### C. CHANGES IN STUDENTS' FAMILY BACKGROUND

In this section we describe changes in parental occupation, in parental education, and in educational influences in the students' homes.

There is relatively little difference between the occupation of fathers of the 1972 and the 1980 seniors (Table 3-2). The major changes are a decline of 2.3 percentage points in fathers employed in craft occupations, an increase of 1.9 percentage points in fathers who are proprietors, and an increase of 1.3 percentage points in fathers holding managerial positions.

The apparent changes in mothers' employment are, unfortunately, a confounding of the actual increase in women's participation in paid work, which took place during this period, and a change of phrasing in the parental occupation question in 1972 and in 1980. In the later year, the student was asked to indicate the parents' "most recent occupation." Therefore these figures for mother's occupation in 1980 may be either her current occupation or her occupation whenever she last held a paid job. There was a decline, from 55.2 percent in 1972 to 15.1 percent in 1980, in the proportion of seniors' mothers whose occupation was homemaker. The increases for employment of mothers were primarily in clerical occupations (up from 16.2 percent in 1972 to 26.8 percent in 1980), professional occupations (up from 9.0 percent to 18.0 percent), and service occupations (up from 5.6 percent to 11.9 percent). The decline in the percentage of students reporting homemaker as their mother's occupation was consistent across SES and racial/ethnic groups. The type of occupation pursued, however, varied across these classifications very



-21-

TABLE 3	-2	
---------	----	--

# Parental Occupation: 1972 and 1980 Total Group

	<u>Cler</u> .	<u>Craft</u>	<u>Farm</u>	Home	Labor	Mgmt.	<u>Military</u>	Oper.	Prof.	Prop.	Protec.	Sales	Service	Tech.
Fathers														
1972	2.9	18.2	5, 1	0.2	11.0	13.7	2.6	11.8	14.0	6.9	2.6	6.0	2.1	3.0
1980	2.3	15.9	4.8	0.2	9.6	14.0	2.4	11.9	15,3	8.8	2.7	5.4	2.1	4.5
<u>Mothers</u>														
1972	16.2	0.7	1.0	55.2	I.3 .	1.7	0,2	3.2	9.0	1.2	0.3	3.6	5.6	0.7
1980	26.8	2.1	0.7	15, 1	3.0	5.4	0.1	6.1	18.0	2,3	0.3	6.5	11.9	1.7

•

38

# BEST COPY AVAILABLE



-22-
predictably. For example, the largest proportion of low SES mothers was employed in service occupations, while the middle SES mothers predominated in clerical occupations, and high SES mothers in professional occupations.

Parental education was measured on a scale ranging from 1 = less than high school education to 5 = graduate or professional school. The mean for father's education (see Table 3-3) rose from 2.32 in 1972 to 2.62 in 1980. This mean education level indicates that the typical father had completed high school. This increase is significant for the total group and for all classification groups, except Puerto Ricans and Other Hispanics. Mother's education (see Table 3-4) increased from a mean of 2.19 in 1972 to 2.41 in 1980. This increase is also significant. These increases parallel national trends for increasingly higher levels of education in successive age cohorts. The slightly lower level of education for mothers than for fathers is also keeping with national data for adults.

To obtain a sense of the home support for learning, the students were asked to indicate whether or not certain study aids (a specific place to study, daily newspaper, encyclopedia/reference books, and typewriter) were available in their homes. The scale uscd in Table 3-5 ranges from 0 = have none of these, to 4 = have all of these. In 1972 the mean number of study aids in the seniors' homes was 3.21, indicating that the average senior had most of these aids. By 1980, however, the average number of study aids declined to 2.97, a significant change. This decline was similar for most of the classification groups.

As another indicator of home support for students' learning and home influence on students' educational aspirations, the students were asked to indicate the amount of schooling that their mother or female guardian wanted them to obtain. (See Table 3-6.) The scale ranges from 1 = less than high school to 5 = graduate or professional school. Using this scale, the mean level of education which the mothers wanted for the 1972 seniors was 3.63; for the 1980 seniors it was 3.73. This increase is significant. This change in parental educational aspirations for the The students differs considerably, however, for males and for females. 1972-1980 increase is significant for females but not for males. Thus, the differential parental educational aspirations for sons and daughters, evident in 1972, had all but disappeared in 1980. Mothers' educational aspirations for their children increased more for high SES students than for low SES students, thus increasing the gap in parental aspirations for high and low SES students. Mothers' educational aspirations for their children increased significantly between 1972 and 1980 for White, Black, Asian-American, and American Indian seniors but did not increase significantly for Hispanic students.

These data present a mixed picture of 1972-1980 changes in home pressure for student school achievement. Although parents provided fewer study aids for 1980 seniors than they did for 1972 seniors, more 1980 parents were, at the same time, providing their children with higher educational expectations, in terms of the amount of schooling to be



	NLS 1972					HSB 1	980				
	SAMPLE	WEIGHTED				UETCUTEN			<b>Baaaa</b>		
	N	N	MEAN	S.D.	N	N	MEAN	S.D.	POOLED S.D.	1980-1972 Difference	EFFECT
TOTAL	14054	2629005	2.32	1.2	21608	2370359	2.62	1 3	1 24	0 70 *	0.22
SEX:									1.00	0.30 *	U.24
MALE	4877	1201710	0 75	• •							
FFMALE	7177	1291310	2.35	1.2	10110	1112665	2.68	1.3	1.27	0.34 ¥	0.27
- Ender	/1/3	1330451	2.28	1.2	10778	1181524	2.57	1.3	1.25	0.29 ×	0.23
SES:											
LON	3578	574854	1 28	ΛE	5703			• •			
MIDDLE	6862	1366160	2 63	0.5	2016	350570	1.45	0.6	0.59	0.17 ¥	0.28
HIGH	3614	687991	3 75	1 1	10443	11/0251	2.36	0.9	0.84	0.34 ¥	0.40
	3024	007771	3.73	<b>T</b> .T	3447	0 39 985	9.10	0.9	0.97	0.35 ¥	0.36
RACE :											
WHITE	11517	22700A3	2 10	1 2	14780		<b>-</b>	-			
BLACK	1348	145049	1 70	1.6	10352	1950751	2.70	1.3	1.26	0.31 ¥	0.25
ASIAN-AMERICAN	166	203740	2.70	0.9	2072	190346	2.12	1.2	1.08	0.42 *	0.39
AMERICAN INDIAN	130	21943	2.43	1.3	294	32805	3.03	1.4	1.35	0.60 ¥	0.44
MEXICAN-AMERICAN	174	21043	1.91	1.1	155	16148	2.47	1.3	1.18	0.56 ¥	0.47
PUERTO PICAN	570	20337	1.51	0.8	1364	74343	1.89	1.1	1.07	0.37 ×	0.35
OTHER HISPANIC	00	14057	1.51	U.8	181	10757	1.74	1.1	1.04	0.24	0.23
	70	14057	2.20	1.4	716	48101	2.35	1.3	1.27	0.09	0.07
SCHOOL TYPE:											
PUBLIC	12556	2324125	2 31	1 2	30400						
PRIVATE	64	16701	2.JL 9 72	1.2	10005	2110602	2.55	1.3	1.25	0.24 ¥	0.19
CATHOLIC	96.8	210757	2.13	1.4	/51	88385	3.49	1.4	1.37	0.74 ×	0.54
	740	217333	C.44	1.2	2175	171371	2.95	1.3	1.25	0.51 *	0.41
GEOGRAPHIC REGION:											
NORTHEAST	3113	708085	2 77	1 0				_			
NORTH CENTRAL	3068	800039	C.JJ 9 97	1.2	4346	551147	2.72	1.3	1.26	0.39 ×	0.31
SOUTH	6688	607030	2.27	1.2	6496	710145	2.55	1.2	1.22	0.29 ¥	0.24
WEST	2485	440721	2.23	1.3	6889	689501	2.46	1.3	1.26	0.23 ×	0.18
	6403	447/21	2.50	1.3	3877	419566	2.84	1.3	1.29	0.34 ×	0.27
CURRICULUM:											
GENERAL	4624	805320	9 12	1 2	74 7 8	070/77					
ACADEMIC	6205	1282026	2 41	1.2	/030	038631	2.41	1.2	1.18	0.30 *	0.25
VOCATIONAL	3226	541357	1 44	1.3	8/0U	962327	3.07	1.3	1.29	0.44 *	0.34
•	2007	37233 <i>1</i>	4.00	1.0	4703	541292	2.14	1.1	1.04	0.28 *	0.27
COMMUNITY TYPE:											
URBAN	3794	672793	2.32	1.2	4475	4944 EF					
SUBURBAN	6957	1370669	2.48	1 1	10740	720033	<.30	1.3	1.25	0.24 ×	0.19
RURAL	3105	550544	1.91	1.0	T0/40	761740	2.60	1.3	1.29	0.33 ×	0.25
			/-	2.0	6433	143107	2.55	1.2	1.15	0.45 *	0.39

# HIGHEST EDUCATIONAL LEVEL COMPLETED BY YOUR FATHER OR MALE GUARDIAN (1=LESS THAN HIGH SCHOOL; 5=GRADUATE/PROFEDSIONAL SCHOOL)

\*SIGNIFICANT AT .05 OR LESS



BEST COPY AVAILABLE

40

## TABLE 3-3

ang da ta

### TABLE 3-4

HIGHEST EDUCATIONAL LEVEL COMPLETED TO THE MOTHER OR FEMALE GUARDIAN (1=LESS THAN HIGH SCHOOL; 5= 9 20 TR/PROFESSIONAL SCHOOL)

		ALS 1	972		HSB 1980						
	SAMPLE N	WE IGHTED N	MEAN	\$.D.	SAMPLE N	NEIGHTED N	MEAN	S.D.	POOLED S.D.	1980-1972 DIFFERENCE	EFFECT SIZE
-TOTAL +	14374	2678867	2.19	1.0	24188	2631211	2.41	1.0	1.03	0.22 *	0.21
SEX:											
MALE	7084	1314944	2.23	1.0	10951	1197447	2.46	1.9	1.03	0.23 *	0.22
FEMALE	7366	1363158	2.15	1.0	12370	1342256	2.37	1.0	1.04	0.23 <b>*</b>	0.22
SES:											
LON	3754	596172	1.37	0.5	6955	677347	1.57	0.6	0.60	0.20 *	0.33
MIDDLE	6964	1386492	2.03	0.7	11530	1287377	2.31	0.8	0.74	0.28 *	0.37
HIGH	3653	695946	3.19	1.0	5627	658547	3.45	1.0	1.01	0.26 *	0.26
BACE :											
MATE	11681	2309666	2.24	1.0	17709	<b>21091</b> 24	2.46	1.0	1.02	0.22 *	0.22
BLACK	1459	179597	1.86	1.0	2920	266891	2.24	1.1	1.05	0.38 *	0.36
ASIAN-AMERICAN	167	24099	2.25	1.1	293	32155	2.70	1.2	1.16	0.45 *	0.39
AMERICAN INDIAN	139	22906	1.77	0.9	173	17744	2.30	1.1	1.04	0.52 *	0.50
MEXICA-AMERICAN	402	54034	1.42	0.7	1502	81619	1.78	0.9	66.0	0.36 *	0.41
PLIERTO RICAN	63	6421	1.56	0.9	241	14240	1.69	0.8	0.85	0.14	0.16
OTHER HISPANIC	84	13362	2.04	1.0	805	54739	2.19	1.0	1.03	0.16	0.15
SCHOOL TYPE:											
PUBLIC	12848	2371440	2.18	1.0	21013	2359113	2.37	1.0	1.02	0.18 *	0.18
PRIVATE	63	15105	2.42	1.2	792	91288	3.07	1.2	1.19	0.65 *	0.55
CATHOLIC	953	219202	2.30	1.0	2383	180810	2.59	1.0	1.03	0.29 *	0.28
GEOGRAPHIC REGION:											
NORTHEAST	3182	722571	2.19	1.0	4891	604693	٠.45	1.1	1.04	0.26 *	0.25
NORTH CENTRAL	4036	820416	2.19	1.0	7114	77056.	40	1.0	0.99	0.21 *	0.21
SOUTH	4622	678145	2.10	1.0	7851	788613	2.29	1.1	1.05	0.19 *	0.18
NEST	2534	457736	2.30	1.0	4332	467339	2.57	1.1	1.05	0.27 *	0.25
CURRICULUM:											
GENERAL	4747	823315	2.03	1.0	8639	944018	2.27	1.0	0.97	0.24 *	0,25
ACADEMIC	6308	1299971	2.43	1.1	9566	1039352	2.74	1.1	1.08	0.30 #	0.28
VOCATIONAL	3318	555279	1.85	0.9	5696	614860	2.08	0.9	0.89	0.24 *	0.27
COMMINITY TYPE:											
LEBAN	3903	687569	2.15	1.0	5321	506744	2.34	1.0	1.02	0.19 *	0.18
SERBAN	7082	1394279	2.28	1.0	11816	1314401	2.51	1.1	1.05	0.22 *	0.21
RURAL	3182	561393	1.99	1.0	7051	810066	2.29	1.0	0.99	0.30 *	0.30

ARTCHIFICANT AT .05 OR LESS

Full Text Provided by ERIC

BEST COPY AVAILABLE 41

to for an a star and the star

#### TABLE 3-5

# NUMBER OF "STUDY ALOS" AVAILABLE IN HOME (Count of responses to: place for study; gaily newspaper; "NCYCLOPEDIA/Reference books; typewriter)

NLS 1972 HSB 1980 SAMPLE WEIGHTED SAMPLE WEIGHTED POOLED 1980-1972 EFFECT N N MEAN S.D. N N MEAN S.D. S.D. DIFFERENCE SIZE TOTAL 16412 3002567 3.21 0.9 26191 2830358 2.97 1.0 0.94 -0.23 ¥ -0.25SEX: MALE 80% 1487778 3.19 0.9 12376 1343612 3.01 1.0 0.96 -0.19 ¥ FEMALE -0.208312 1514024 3.22 0.9 13728 1478552 2.94 0.9 0.92 -0.28 × -0.30 SES: LON 4717 727933 2.54 1.1 7720 742694 2.29 1.1 1.06 -0.24 \* MIDDLE -0.237829 1539145 3.31 0.8 12170 1357352 3.10 0.8 t.80 -0.22 \* HIGH -0.27 3832 729451 3.64 0.6 5934 694330 3.47 0.7 0.62 -0.18 × -0.29 RACE: MHITE 12729 2506111 3.28 0.9 19000 2261452 3.03 0.9 0.89 -0.25 \* BLACK -0.28 2046 247777 2.75 1.1 3229 293820 2.71 1.1 1.07 -0.03 ASIAN-AMERICAN -0.03 192 27663 3.33 8.0 349 38092 3.05 1.0 AMERICAN INDIAN 0.97 -0.28 \* -0.29 185 30932 2.99 1.0 198 20286 2.66 1.1 1.07 -0.33 MEXICAN-AMERICAN -0.31 541 71223 2.60 1.2 1741 93325 2.65 1.0 1.07 0.05 PUERTO RICAN 0.05 91 9196 2.48 1.2 284 16362 2.51 1.2 1.25 0.03 OTHER HISPANIC 0.02 114 17540 2.88 1.1 889 58928 2.74 1.1 1.08 -0.14 -0.13 SCHOOL TYPE: PUBLIC 14708 2663708 3.19 0.9 22802 2537784 2.93 1.0 0.95 -0.25 × PRIVATE -0.26 66 16256 3.53 0.7 847 101094 3.28 0.8 0.82 -0.25 CATHOLIC -0.30 1022 234707 3.46 0.7 2542 191480 3.29 0.7 0.73 -0.17 × -0.23 GEOGRAPHIC REGION: NORTHEAST 3552 793263 3.38 0.8 5399 663197 3.11 0.9 0.85 NORTH CENTRAL -0.27 ¥ -0.32 4514 908789 3.19 0.9 7481 810034 3.01 0.9 0.90 -0.19 \* SCUTH 5424 -0.21 783599 3.05 1.0 8481 837239 2.84 1.0 1.02 WEST -0.22 \* -0.21 2922 516916 3.19 0.9 4830 519888 2.96 1.0 0.97 -0.23 \* -0.24 CURRICULUM: GENERAL 5564 951961 3.04 1.0 9485 1026439 2.85 1.0 0.99 -0.19 \* ACADEMIC -0.20 6735 1378439 3.39 0.8 10089 1093825 3.21 0.8 0.81 VOCATIONAL -0.18 ¥ -0.23 4112 671864 3.05 1.0 6254 670666 2.79 1.0 1.00 -0.26 \* -0.26 COMMUNITY TYPE: URBAN 4541 784249 3.22 0.9 5926 2.99 557297 1.0 0.93 -0.23 \* SUBURBAN -0.25 7932 3534602 3.30 0.8 12678 1402331 3.04 0.9 0.90 -0.25 \* -0.28 RURAL 3661 636609 2.97 1.0 7587 870731 2.84 1.0 1.01 -0.13 ¥ -0.13

\*SIGNIFICANT AT .05 OR LESS



BEST COPY AVAILABLE

42

	NLS 1972				HSE 19	780					
	SAMPLE	WEIGHTED	ME ANI	e n	SAMPLE	WEIGHTED	ME AN	e n	POOLED	1980-1972	EFFECT
	М	N	FIEAN	5.0.	n	n	HEAN	3.0.	3.0.	Varrende	
TOTAL	13294	2455320	3.63	0.8	22225	2399707	3.73	1.0	0.91	0.11 ¥	0.12
SEX:											
MALE	6425	1198604	3.75	0.8	10315	1116278	3.76	1.0	0.92	0.01	0.01
FEMALE	6866	1256192	3.51	0.8	11485	1239390	3.72	0.9	0.89	0.21 ¥	0.23
SES:											
LON	3544	547247	3.29	0.8	6236	599167	3.37	1.0	0.96	0.08 ×	0.08
MIDDLE	6 366	1261711	3.55	0.8	10334	1147887	3.67	0.9	0.87	0.11 ¥	0.13
HIGH	3371	643907	4.06	0.7	5310	619924	4.22	0.7	0.72	0.16 *	0.22
RACE :											
WHITE	10632	2097629	3.63	8.0	16002	1903285	3.70	0.9	<b>v.8</b> 9	0.07 ¥	0.08
BLACK	1484	179449	3.72	0.8	<b>28</b> 26	25897 <b>8</b>	3.96	1.0	0.93	0.25 *	0.27
ASIAN-AMERICAN	156	22408	4.02	0.8	312	33585	4.29	0.8	0.80	0.27 *	0.34
AMERICAN INDIAN	132	22150	3.20	0.9	160	16739	3.74	1.0	0.96	0.54 #	0.56
MEXICAN-AMERICAN	406	53819	3.52	0.8	1445	76815	3.58	1.0	0.97	0.06	0.06
PUERTO RICAN OTHER HISPANIC	67 85	6739 13429	3.63	0.8	240 762	13131 50679	3.66 3.70	1.1	1.05	0.19	0.03
SCHOOL TYPE:				• •	10053	01 F047F			0 01	0 00 ×	0 00
PUBLIC	11904	2174121	3.61	0.8	19253	2150435	3.70	1.0	0.71	0.00 *	0.07
PRIVATE	6 <b>0</b>	14404	3.75	0.7	739	144110	4.12	0.0	0.03	0.3/ *	0.44
CATHOLIC	8/2	201304	3.77	0.8	2233	100110	3.70	0.7	V.04	0.20 *	V. 24
GEOGRAPHIC REGION:										<b>a 1</b> / y	A 15
NORTHEAST	2911	655534	3.63	0.8	4532	557622	3.//	1.0	0.93	U.14 #	0.15
NORTH CENTRAL	3654	740465	3.57	0.8	6415	D73474	3.01	0.9	0.07	0.04	0.05
SOUTH	4395	642139	3.0/	0.8	7321	123915	3.//	1.0	0.72	0.10 *	0.11
MEST	2334	417182	3.00	0.8	375/	424070	3.02	0.7	0.0/	V.10 *	0.17
CURRICULUM:			-								
GENERAL	4356	746427	3.39	0.8	7815	847685	3.55	1.0	0.90	0.15 #	0.17
ACADEMIC	5914	1212548	4.00	0.7	8978	971095	9.17	0.8	0.75	U.1/ #	0.23
VOCATIONAL	302 <i>3</i>	496042	3.08	0.7	5153	549800	3.20	0.9	0.85	0.19 *	V.22
COMMUNITY TYPE:											ē
URBAN	3644	636392	3.69	0.8	5003	471608	3.85	1.0	0.90	0.16 *	0.18
SUBURBAN	6563	1280168	3.69	8.0	10807	1195851	3.79	1.0	0.90	0.10 *	0.11
RURAL	2906	508396	3.40	0.8	6415	732248	3.57	1.0	0.91	0.16 #	0.18

#### HOW MUCH SCHOOLING DOES YOUR MOTHER OR FEMALE GUARDIAN WANT YOU TO GET (1=LESS THAN HIGH SCHOOL; 5=GRADUATE/PROFESSIONAL SCHOOL)

TABLE 3-6

\*SIGNIFICANT AT .05 OR LESS



**4**3

BEST COPY AVAILABLE

-----

obtained. These changing parental expectations for the 1972 and 1980 seniors' educations were differential, however, affecting females much more than males, high SES students more than low SES students, and non-Hispanic minority students more than White or Hispanic students.

-24-

.



14

#### CHAPTER IV

-25-

## THE SCHOOLS, EDUCATIONAL PROGRAMS, AND LEARNING CONDITIONS

This chapter describes how schools, educational programs, and learning conditions changed between 1972 and 1980. Four areas are examined: 1) student body characteristics, 2) staff characteristics, 3) educational programs and teaching methods, and 4) students' evaluations of school facilities and their educational experiences. Data on student and staff characteristics and educational programs are drawn from the school questionnaires; those on teaching methods and student evaluation come from the student questionnaire. Schools are grouped by four classification variables: 1) average SES of their students, 2) school type, 3) geographic region, and 4) community type.

## A. STUDENT BODY CHARACTERISTICS

Students' achievement, attitudes and behaviors are influenced by the environment of the schools they attend as well as by their personal backgrounds. The 1972 and 1980 school questionnaires contain five measures of student body composition: 1) racial/ethnic composition, 2) student absenteeism and dropout rates, 2) percentage of college-bound students, 4) percentage of students in the academic track, and 5) the percentage of students classified as disadvantaged or handicapped.

#### 1. Racial/Ethnic Composition

Table 4-1 shows the percentage of schools that were predominately White, predominately non-White, and integrated in 1972 and 1980. In 1972, 52.8 percent of the schools were 95 to 100 percent White, 36.3 percent were 50 to 94 percent White, while 10.9 percent were less than 50 percent White. In 1980, the percentages were 53.5, 35.0 and 11.5, respectively. These figures vary widely when schools are grouped by student SES, school type, geographic region and community type, however. For example, in 1972, 22.7 percent of the schools that had a low SES student body were predominately non-White compared to 6.8 percent and 1.5 percent for medium and high SES schools. A larger percentage of schools in the South and in urban communities also were non-White. The percentage of predominately minority schools and predominately White schools remained unchanged, in general, between 1972 and 1980. Catholic schools provide the exception to this statement. The percentage  $c^{\pm}$ Catholic schools that were 50 to 94 percent White nearly doubled between 1972 and 1980, from 33.5 percent to 57.7 percent, while the percentage that were predominately White dropped from 59.9 percent to 35.7 percent.

Tables 4-2 and 4-3 show the percentage of schools that have varying concentrations of Black and Hispanic students. Between 1972 and 1980, the percentage of schools that were majority Black increased slightly, while the percentage with enrollments that were only 0 to 4 percent Black



## PERCENT OF CURRENT STUDENTS WHO ARE WHITE

			NLS 1972		HSB 1980					
	NUMBER OF SCHOOLS	% WITH 0-49%	% WITH 50-79%	% WITH 80-94%	% WITH 95-100%	NUMBER OF SCHOOLS	% WITH 0-49%	% WITH 50-79%	% WITH 80-94%	% WITH 95–100%
TOTAL	1237	10.9	14.5	21.8	52.8	959	11.5	15.2	19.8	53.5
AVERAGE SES OF STUDENTS:										
LOW	306	22 7	26 4	22.0	78 0					
MIDDLE	413	4.	11 1	22.0	30.9	225	38.0	11.5	13.7	36.9
HIGH	718	0.0	11.1	17.2	64.8	472	3.6	20.0	16.3	60.1
	210	1.5	0.0	33.8	58.1	239	1.2	9.2	33.1	56.5
SCHOOL TYPE:										
PUBLIC	1109	11.4	15 4	20 E	E9 E					
PRIVATE	11	0 0	2.0	20.3	52.5	841	12.6	15.9	18.4	53.1
CATHOLITC	71	4.4	2.4	40.8	50.7	37	8.7	6.3	20.4	64.6
	/1	0.0	10.9	22.7	59.9	81	6.7	24.8	32.9	35.7
GEOGRAPHIC REGION:										
NORTHEAST	252	4.4	6.0	18 9	70 6	205		14 4		
NORTH CENTRAL	321	5.8	3.1	16 4	70.0 75 E	203	2.0	10.9	24.2	59.1
SOUTH	460	20 4	21 4	20 4	73.3	270	3.3	9.1	12.1	80.5
WEST	204	8 4	72 4	27.0	20.4	290	20.1	27.1	20.4	32.4
	644	0.4	32.0	20.5	38.2	194	14.2	15.2	26.5	44.1
COMMUNITY TYPE:										
UPBAN	376	24.4	20.2	25 0	30 4	04.0	<b>a</b> / a			
SUBURBAN	617	7.5	17 7	27 1	JV.7 Eg 1	646	20.9	20.2	21.9	30.9
RURAL	237	97	17.0	14 0	JC.1	462	0.2	16.6	27.9	49.3
		7.1	13.9	10.0	ov.4	255	9.8	12.2	12.9	65.1

NOTE: PERCENTAGES ARE BASED ON WEIGHTED DATA

**4**6



## TABLE 4-2

#### PERCENT OF CURRENT STUDENTS HHO ARE BLACK

			NLS 1972	:	HSB 1980					
	NLMBER OF SCHOOLS	% WITH 0-4%	% MITH 5-19%	% WITH 20-49%	% NITH 50–100%	NUHBER OF SCHOOLS	% WITH 0-4%	% WITH 5-19%	% WITH 20-49%	% WITH 50-100%
TOTAL	1237	68.0	15.8	9.6	6.6	<b>958</b>	68.4	14.2	9.8	7.7
AVERAGE SES OF STUDENT'S:										
LON	306	47.3	21.6	14.8	16.3	225	54.8	8.6	10.4	26.2
MIDDLE	613	76.5	12.6	8.3	2.6	472	72.5	13.5	12.7	1.4
HIGH	318	81.0	14.5	4.2	0.3	238	72.9	23.2	3.4	0.5
SCHOOL TYPE:										
PUBLIC	1109	66.5	16.4	10.3	6.8	840	67.4	13.0	11.0	8.6
PRIVATE	11	92.0	8.0	0.0	0.0	37	77.1	12.5	5.2	5.2
CATHOLIC	71	78.4	12.7	6.7	2.3	81	60.7	29.6	6.4	3.3
GEOGRAPHIC REGION:										
NORTHEAST	252	78.4	13.8	4.6	3.1	205	67.2	19.8	9.8	3.2
NORTH CENTRAL	321	84.6	10.6	2.2	2.6	270	87.6	7.1	2.5	2.8
SCUTH	460	37.0	25.4	22.8	14.8	289	46.6	18.6	18.9	15.9
HEST	204	89.1	7.8	<b>2</b> .2	1.0	194	77.6	12.6	5.2	4.7
COMMINITY TYPE:										
LIRBAN	376	43.9	21.9	18.3	15.9	242	44.9	19.3	17.8	18.8
SERVERAN	617	67.0	18.7	9.9	4.4	461	65.8	20.8	9.5	3.9
	237	76.5	11.3	6.7	5.5	255	79.0	7.3	7.0	6.7

NOTE: PERCENTAGES ARE BASED ON WEIGHTED DATA

47



17.1

# BEST COPY AVAILABLE

.\*

### TABLE 4-3

## PERCENT OF CURRENT STUDENTS HHO ARE HISPANIC

			NLS 1978	2	HSB 1980					
	NUMBER OF SCHOOLS	% WITH 0-4%	% NITH 5-19%	% WITH 20-49%	% WITH 50-100%	NUMBER OF SCHOOLS	% WITH 0-4%	% WITH 5-19%	% WITH 20-49%	
TOTAL	1237	87.0	9.1	2.7	1.3	959	81.8	11.8	4.0	2.4
AVERAGE SES OF STUDENTS:										
LON	704									
MTODLE	200	00.2	5.6	3.2	3.0	<b>22</b> 7	74.8	11.6	6 4	79
NTCH	013	85.0	8.7	2.8	0.5	471	85 5	 Q 2	7.4	1.6
11201	318	81.9	16.4	1.5	0.2	239	84.4	12 5	3.0	1.4
SCHOOL TYPE:								10.5	2.0	V.5
PUBLIC	1100									
BRTVATE.	1109	87.4	8.5	2.7	1.4	841	83.6	10 2	<b>A A</b>	
	11	90.6	9.4	0.0	0.0	37	79 7	14 7	4.0	2.2
CAINOLIC	71	81.7	13.7	4.3	0.4	81	68.0	10.7	0.0	3.5
GEOGRAPHIC REGION:								20.3	-1.1	2.0
NORTHEAST										
	252	89.7	8.8	1.5	0.1	205	88.0	7 9		
RATTU	321	97.7	2.1	0.1	0.1	267	04.9	7.7	2.9	1.2
30017	460	88.6	8.2	1.4	1.8	202	77.0	3.0	1.0	0.0
ME21	204	58.6	25.4	12 2	1.4	105	03.0	9.9	4.1	2.4
					3.0	173	59.9	30.5	8.1	7.0
CUMPUNITY TYPE:										
URBAN	376	78.6	14 5							
SUBURBAN	617	84 8	10 7	4.3	2.9	241	63.8	24.9	5.6	5.7
RURAL	917	01.4	10./	3.0	1.0	460	80.9	12.8	4.6	1.6
	631	71.0	5.0	1.4	1.2	258	89.1	6.2	2.9	1.7

NOTE: PERCENTAGES ARE BASED ON NEIGHTED DATA

48



remained constant (with the exception of Catholic schools and schools in the Northeast and West). The percentage of schools that were predominately Hispanic increased during this period across every classification variable except in the North Central region. Once again, the Catholic schools appeared to have absorbed Hispanic students to a greater extent than non-Catholic schools. The percentage of Catholic schools with more than 5 percent Hispanic enrollment grew from 18.3 percent to 32.0 percent.

#### 2. Student Absenteeism and Dropout Rates

In Table 4-4, schools are grouped by approximate average daily attendance rates. Between 1972 and 1980, the percentage of schools with low absenteeism (attendance rates of 96 to 100 percent) showed a moderate decline, while those with high absenteeism (attendance rates of 0 to 84 percent) remained constant. This finding generally holds across all four classification variables. The sharpest decrease in the percentage of schools with low absenteeism rates occurred in urban areas, in Catholic schools, in the Northeast and North Central regions, and in low and middle SES schools.

Dropout rates are measured as the percent of students who entered the 10th grade but dropped out before graduation, as reported by the schools. Table 4-5 shows a general increase in dropouts. The percentage of schools with a dropout rate of 10 to 19 percent increased from 13.4 percent to 20.4 percent, and those with a rate of 20 percent or more grew from 3.6 percent to 9.6 percent. Increases in this latter category were most evident in the South and West, in the suburbs, among public schools, and in schools with a low SES student body. Middle SES and rural schools showed a large increase between 1972 and 1980 in the percentage of schools with a 5 to 19 percent dropout rate.

#### 3. College-Bound Students

Changes in the concentration of college-bound students (both 2-year and 4-year) between 1972 and 1980 are shown in Table 4-6. During this period the percentage of schools with a preponderance (70 percent or more) of college-bound students increased from 9.3 percent to 18.8 percent. Large increases in the number of schools with high percentages of collegebound students took place in the nonpublic schools and among high SES schools. Students in middle SES schools, however, showed slightly less interest in college attendance.

### 4. High School Curriculum

Data in Chapter III showed that between 1972 and 1980 substantially more high school seniors chose to enroll in general education rather than academic/college preparatory programs. Table 4-7 shows the percentage of schools with different concentrations of students in the academic program in both of these years. There was growth in both the percentage of schools falling in the lowest (0 to 29 percent) and the highest (70 to



## APPROXIMATE AVERAGE DAILY PERCENTAGE ATTENDANCE

			NLS 1978	2	HSB 1980					
	NUMBER OF SCHOOLS	% WITH 96-100%	% WITH 90-95%	% MITH 85-89%	% WITH 0-84%	NUMBER OF SCHOOLS	х WITH 96-100%	% WITH 90-95%	% NITH 85-89%	% HITH 0-84%
TOTAL	1251	22.5	57.2	14.6	5.7	958	17.1	67.5	9.6	5.8
AVERAGE SES OF STUDENTS:										
LON	307	14 4	40 4	0E E						
MIDDLE	491	20.4	47.0	25.5	0.5	227	10.2	62.7	11.1	15.9
HICH	021	24.7	63.2	9.3	2.8	471	17.9	71.0	8.7	2.4
W100	323	26.6	53.4	10.8	9.1	237	24.3	67.5	7.6	0.5
SCHOOL TYPE:										
PUBLIC	1120	18 5	40 3	14 0	4 7					
PRIVATE	12	40.9	50.5	14.9	0.3	840	14.9	69.3	9.3	6.5
CATHOLIC	75	40.0	52.0	0.0	0.0	37	20.2	59.0	15.6	5.2
CAMOLIC	73	53.4	37.4	7.7	1.5	81	32.7	65.7	1.3	0.2
GEOGRAPHIC REGION:										
NORTHEAST	262	27 A	49 A	17 9	A E					_
NORTH CENTRAL	321	11 0	F0 4	±7.7	4.5	203	14.5	68.2	9.8	7.5
SOLITH	5C1 4E0	31.7	37.0	5./	2.8	274	23.3	67.9	4.4	4.4
LERT	437	15.4	67.1	14.0	5.5	293	15.0	72.0	10.5	2.6
MEST	209	16.8	41.0	28.8	13.4	188	13.1	58.7	16.2	11.9
COMMUNITY TYPE:										
URBAN	382	30 7	74 E	74 F	10.4					
SI BI DRAN	497	30.7	34.3	40.5	10.4	242	8.0	51.0	25.9	15.2
	063	20.2	61.3	13.2	5.3	462	17.7	69.4	7.2	5.6
RURAL	239	21.5	61.7	15.0	1.8	254	20.1	72.1	5.4	2.4

NOTE: PERCENTAGES ARE BASED ON NEIGHTED DATA

,



BEST COPY AVAILABLE

50

## SEST COPY AVAILARLE

#### TABLE 4-5

## PERCENT OF STUDENTS WHO ENTER THE 10TH GRADE BUT DROP OUT BEFORE GRADUATION

			NLS 197	2	HSB 1980					
	NUMBER OF SCHOOLS	% WITH 0-4%	2 HITH 5-92	% NITH 10-19%	% WITH 20-100%	NUMBER OF SCHOOLS	% WITH 0-4%	% WITH 5-9%	% WITH 10–19%	% WITH 20-100%
TOTAL	1192	62.2	20.8	13.4	3.6	956	46.5	23.5	20.4	9.6
AVERAGE SES OF STUDENTS:										
LON	297	<b>1</b> 8 8	32 0	22 Q	4 7	00E	7/ 7			
MIDDLE	592	71 9	16 1	0.	7.1	223	30.3	15.4	30.3	19.9
HIGH	303	79 1	15.1	7.0	3.1	4//	40.8	28.5	22.3	8.6
	303	/0.3	13.7	5.0	0.1	235	72.3	19.4	6.8	1.5
SCHOOL TYPE:										
PUBLIC	1063	58.2	23.3	15.0	3.6	979	74 4	24 1	2E /	10 1
PRIVATE	12	96.1	2 4	1 5	0.0	34	30. <del>4</del>	20.1	23.4	12.1
CATHOLIC	71	97.7	0 5	1.6	0.0	37 87	01.7	10.0	1.5	0.0
		****	0.5	1.0	0.0	03	70.0	8.2	1.2	0.0
GEOGRAPHIC REGION:										
NORTHEAST	247	73.0	19.8	5.1	2.2	200	52 2	23 6	19.9	<b>A E</b>
NORTH CENTRAL	312	78.8	10.8	8.1	2.3	271	49 3	27 4	10.2	4.3
SOUTH	440	46.6	21.4	26.4	5.5	200	47.5 42 Q	10 7	24.6	4.0
NEST	193	49.6	40.9	5 7	1.9	194	42.7	17.J 94 E	24.0	13.7
				2.7	5.0	100	73.2	64.5	10.0	10.2
CONTRACTLY TYPE:										
URBAN	355	61.0	13.9	13.5	11.6	235	54 5	13 9	16 6	16.0
SUBURBAN	596	62.8	23.0	12.3	1.9	458	46 7	24 0	10.0	10.4
RURAL	234	61.8	21.2	14.4	2.6	263	43.8	24 1	12.0	10.0
					2.0	205	<b>TJ.O</b>	C4.T	64.9	1.1

NOTE: PERCENTAGES ARE BASED ON HEIGHTED DATA

LIST COPY AVAILABLE

C

Full Text Pro

11

# -25T COPY AVAILABLE

## TABLE 4-6

## PERCENT OF LAST YEAR'S GRADUATES NOW ENROLLED IN A REGULAR THO-YEAR OR FOUR-YEAR COLLEGE

			NLS 1977	2		H\$8 1980				
	NUMBER OF SCHOOLS	% WITH 0-29%	2 NITH 30-492	% WITH 50-69%	X NITH 70-100%	NUMBER OF SCHOOLS	% WITH 0-29%	% WITH 30-49%	% WITH 50-69%	X WITH 70-100%
TOTAL	1254	30.6	34.9	25.3	9.3	973	27.2	32.9	21.1	18.8
AVERAGE SES OF STUDENTS:										
LON	310	58.1	30.9	9.7	1.4	232	54.3	30.5	12.8	2.3
MIDDLE	623	19.9	43.2	30.2	6.7	488	24.7	43.5	24.7	7.2
HIGH	321	11.4	19.0	39.3	30.4	240	4.4	8.4	20.7	66.5
SCHOOL TYPE:										
PUBLIC	1122	33.1	38.2	22.8	5.9	857	32.1	38.1	22.3	75
PRIVATE	12	0.0	15.5	55.2	29.3	33	13.1	12.4	12.8	41.6
CATHOLIC	73	9.4	18.1	38.3	34.2	83	0.0	12.5	21.6	65.9
GEOGRAPHIC REGION:										
NORTHEAST	262	20.1	34.3	27.3	18.3	208	15.9	20 X	23.2	<b>1</b> 1 7
NORTH CENTRAL	326	24.8	43.3	27.0	4.9	275	19.8	44.5	23.3	12.4
SOUTH	461	39.0	29.9	21.8	9.3	298	35.6	27.3	18.0	19 1
NEST	205	37.2	29.6	26.9	6.3	192	36.1	26.3	20.9	16.7
COMMUNITY TYPE:										ļ
URBAN	381	21.8	29.5	29.3	19.4	241	20.8	26.1	14.5	<b>78 6</b>
SUBURBAN	625	17.7	36.9	31.7	13.6	469	23.0	31.1	22.1	23 8
RURAL	241	45.1	34.8	18.0	2.1	263	32.5	36.5	22.4	8.7

NOTE: PERCENTAGES ARE BASED ON NEIGHTED DATA



## PERCENT ENROLLED IN ACADEMIC OR COLLEGE PREPARATORY CURRICULUM

			NLS 1978	2	HSB 1980					
	NUMBER OF SCHOOLS	X MITH 0-29%	7 NITH 30-49%	X NITH 50-69%	% WITH 70-100%	NUMBER OF SCHOOLS	% NITH 0-29%	% NITH 30-49%	% NI TH 50-69%	X HITH 70-100X
TOTAL	11%	37.3	32.2	17.8	12.7	892	40.9	21.7	16.5	21.0
AVERAGE SES OF STUDENTS:										
LON	200	50 B	98 E						_	
MIDDLE	E 77 E 04	37.0	20.5	9.8	1.9	206	67.5	20.8	5.9	5.8
NTCH	370	30.6	39.0	18.5	12.0	441	44.8	26.7	18.5	10.0
NTON .	301	16.2	20.3	29.9	33.5	226	11.2	12.9	16.3	59.5
SCHOOL TYPE:										
PUBLIC	1070	<b>A1 1</b>	76 6	14 E	7 8	774		<b>.</b>		
PRTVATE	19		34.0	10.5	7.0	//8	49.7	26.6	15.5	8.3
CATHOLIC	46	0.0	28.7	21.1	50.2	35	16.2	4.5	25.7	53.7
CAMOLIC	/1	8.3	16.7	31.0	43.9	79	4.3	7.3	9.0	79.5
GEOGRAPHIC REGION:										
NORTHEAST	241	18.8	28 A	30 7		101	15.4			
NORTH CENTRAL	316	129 9	20.4	37.7		171	12.0	25.0	22.0	36.4
SOUTH	447	JE.E	37.1	21.7	/.1	250	45.7	25.6	15.7	13.0
LEET	443	43.1	32.9	10.4	14.1	275	49.4	19.1	8.0	23.5
NE31	196	57.7	23.0	10.1	9.2	176	39.9	16.9	28.0	15.1
COMPLINITY TYPE:										
URBAN	357	33.6	10 7	94 0	<b>61 6</b>					
SUBURBAN	401	97 E	47.3		61.0	218	29.7	15.8	16.0	38.6
PI E AI	001	67.3	34.5	25.9	12.0	428	27.1	25.3	21.1	26.5
	633	4/./	37.3	7.4	7.5	246	54.6	21.1	13.3	11.0

NOTE: PERCENTAGES ARE BASED ON HEIGHTED DATA



53

100 percent) categories. The shift away from academic programs took place primarily among low and middle SES schools, schools in the public sector, rural schools, and those in the North Central region. The largest increase in the number of schools with predominately academic programs (70 to 100 percent) occurred among high SES schools and in the Catholic sector.

Tables 4-8 and 4-9 present similar data for enrollments in general and vocational curricula. The percentage of schools with a low proportion (0 to 29 percent) of students in the general curriculum decreased sharply, from 55.3 percent of the 1972 schools to 35.6 percent of the 1980 schools. Schools with a high proportion (70 to 100 percent) of students in the general curriculum showed a corresponding increase from 14.2 percent of the 1972 schools to 31.3 percent of the 1980 schools. The increase toward higher proportions of students in the general curriculum took place primarily among low and middle SES schools, public and non-Catholic private schools, in schools outside of the Northeast, and in rural schools.

The percentage of schools with a high proportion (20 to 100 percent) of students in the vocational curriculum decreased moderately from 59 percent in 1972 to 48 percent in 1980, while schools with a low proportion (0 to 9 percent) of vocational curriculum students increased. The increase toward fewer students in the vocational curriculum was primariiy among high SES schools, Catholic schools, schools in the West, and schools in urban areas.

## 5. <u>Students with Special Educational Needs</u>

Tables 4-10 and 4-11 show the mean percent of students classified by schools as handicapped and as disadvantaged in 1972 and 1980. It is difficult to make direct comparisons because of the way in which the questions were worded in these two years. In 1972, only the schools that classified students were asked to report the number of students classified as handicapped or disadvantaged. The 1980 school questionnaire asked <u>all</u> schools to report the percent of students classified as handicapped or disadvantaged. As a result, the mean in 1972 does not include a large number of schools that may not have had any students with special educational needs.

One can examine differences across classification variables in 1530, however. In that year, an average of 5 percent of students in low SES schools were classified as handicapped compared to 2 percent in high SES schools, and 4 percent of students in public schools compared to 1 percent in Catholic schools. One finds even greater contrasts in the percent of students classified as disadvantaged. More than 30 percent of students in low SES schools were disadvantaged in 1980 compared to 3 percent in high SES schools. Large differences also existed between public and Catholic schools (17.3 percent versus 5 percent) and between the South and other regions of the country.



54

#### TABLE 4-8

## PERCENT ENROLLED IN GENERAL CURRICULUM

			NLS 1972	-	HSB 1980					
	NCMBER OF SCHOOLS	% WITH 0-29%	% WITH 30-49%	% HITH 50-69%	% WITH 70-100%	NUMBER OF SCHOOLS	% WITH 0-29%	% WITH 30-49%	% WITH 50-69%	% WITH 70-100%
TOTAL	1196	55.3	19.6	10.9	14.2	883	35.6	17.0	16.1	31.:
AVERAGE SES OF STUDENTS:										
LOW	200	AE 7	<u> </u>							
	277	45.3	21.9	10.5	22.5	208	22.1	11.1	17.5	49.4
	590	58.2	19.0	12.8	10.1	435	27.8	20.0	19.0	33.2
HIGH	301	65.1	17.3	6.9	10.7	221	67.4	13.9	5.9	12.8
SCHOOL TYPE:										
PUBLIC	· • • 1	50.7	22 1	10 4	14 7	-	-			
PRIVATE	,	72 9		10.4	10.7	/68	20.8	18.5	18.8	35.9
CATHOLIC		76.0	0.0	27.2	0.0	37	49.0	16.9	9.9	24. <b>2</b>
94110EIC	/1	81.1	8.5	9.7	0.8	78	89.6	3.3	3.3	3.8
GEOGRAPHIC REGION:										
NORTHEAST	241	88.8	8.6	2.0	0 5	195	71 7	14 7		
NORTH CENTRAL	316	44.9	22 0	17.2	15 0	263	71.7	10./	0.1	3.4
SOUTH	443	66 6	21 4	12 4	13.7	247	24.3	20.1	20.0	35.5
MEST	104		23.4	12.0	17.5	274	38.2	15.2	12.0	34.7
H201	170	99.I	20.4	5.7	20.8	177	17.4	15.9	24.1	42.7
CONNUNINY TYPE:										
URBAN	357	63.6	15.7	5 4	14 0	917	44 0		/	
SUBURBAN	601	57 6	22 3	9.6	10 7	217	44.2	18.7	11.6	25.5
	211	57.5	10 7	7.0	10./	424	48.Z	18.5	15.0	17.8
	233	<b>30.3</b>	10.1	12.9	17.1	242	23.6	15	18.4	42.9

NOTE: PERCENTAGES ARE BASED ON WEIGHTED DATA



## TA. 4-9

PERCENT ENROLLED IN VOCATIONAL-TECHNICAL CURRICULUM


		NLS 1972	2	HSB 1980					
NUMBER OF SCHOOLS	% WITH 0-9%	% Hith 10–14%	% WITH 15 19%	% MITH 2 <b>0-</b> 100%	NUMBER OF SCHOOLS	% WITH 0-9%	% WITH 10-14%	% NITH 15-19%	% WITH 20-100%
11%	31.9	3.0	6.1	59.0	900	41.6	6.1	4.4	48.0
299	28.9	2.0	3.2	65.9	211	38.7	1.9	3.7	55.7
596	29.8	2.5	5.0	62.7	444	31.6	5.9	5.1	57.4
301	42.9	6.4	14.2	36.5	226	61.7	11.1	3.7	23.6
1070	29.3	3.1	3.9	63.7	784	29.7	5.5	5.2	59.6
12	77.4	0.0	18.7	3.9	37	84.3	5.0	0.0	10.8
71	42.1	4.4	21.9	31.7	79	65.4	13.6	5.5	15.4
241	18.2	2.6	9.2	69.9	193	28.5	10.9	4.5	56.1
316	35.0	3.3	6.7	55.0	254	38.7	5.0	4.8	51.4
443	35.7	2.9	5.7	55.7	279	41.8	6.8	2.5	48.8
196	34.6	3.2	1.9	60.2	174	57.3	2.0	7.1	33.7
357	35.8	4.6	9.2	50.5	222	58.6	7.2	5.2	29.0
601	29.5	4.6	10.2	55.7	433	34.3	6.8	4.8	54.1
233	32.9	1.0	1.0	65.0	245	40.6	5.1	3.8	50.6
	NUMBER OF SCHOOLS 1196 299 596 301 1070 12 71 241 316 443 196 357 601 233	NUMBER         %           OF         HITH           SCHOOLS         0-9%           1196         31.9           299         28.9           596         29.8           301         42.9           1070         29.3           12         77.4           71         42.1           241         18.2           316         35.0           443         35.7           196         34.6           357         35.8           601         29.5           233         32.9	NLS 1972           NUMBER         2         2           OF         HITH         HITH           SCHOOLS         0-9%         10-14%           1196         31.9         3.0           299         28.9         2.0           596         29.8         2.5           301         42.9         6.4           1070         29.3         3.1           12         77.4         0.0           71         42.1         4.4           241         18.2         2.6           316         35.0         3.3           443         35.7         2.9           196         34.6         3.2           357         35.8         4.6           601         29.5         4.6           233         32.9         1.0	NUMBER $\chi$ $\chi$ $\chi$ $\chi$ OF         WITH         WITH         WITH         WITH         WITH           SCHOOLS         0-9%         10-14%         15         19%           1196         31.9         3.0         6.1           299         28.9         2.0         3.2           596         29.8         2.5         5.0           301         42.9         6.4         14.2           1070         29.3         3.1         3.9           12         77.4         0.0         18.7           71         42.1         4.4         21.5           241         18.2         2.6         9.2           316         35.0         3.3         6.7           443         35.7         2.9         5.7           196         34.6         3.2         1.9           357         35.8         4.6         9.2           601         29.5         4.6         10.2           233         32.9         1.0         1.0	NURBER $\chi$	NLS 1972         NUMBER $2$ $\chi$ <t< td=""><td>NLS 1972         NUMBER         <math>\chi</math> <t< td=""><td>NLS 1972         HSB 1980           NUMBER         <math>\chi</math> <math>\chi</math></td><td>NLS 1972         HSB 1980           NUMBER OF SCHOOLS         <math>\chi</math> <math>\chi</math></td></t<></td></t<>	NLS 1972         NUMBER $\chi$ <t< td=""><td>NLS 1972         HSB 1980           NUMBER         <math>\chi</math> <math>\chi</math></td><td>NLS 1972         HSB 1980           NUMBER OF SCHOOLS         <math>\chi</math> <math>\chi</math></td></t<>	NLS 1972         HSB 1980           NUMBER $\chi$	NLS 1972         HSB 1980           NUMBER OF SCHOOLS $\chi$

NOTE: PERCENTAGES ARE BASED ON NEIGHTED DATA



ter le

## BEST COPY AVA

## TABLE 4-10

## PERCENT OF STUDENTS CLASSIFIED AS HANDICAPPED

	NLS 1972				HSB 1980						
	SAMPLE N	NEIGHTED N	MEAN	MEAN S.D.	SAMPLE N	WEIGHTED N	MEAN	S.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	841	10573	3.86	4.8	886	19006	3.70	5.6	5.22	-0.16	-0.03
AVERAGE SES OF STUDENTS:											
LON	195	2924	6.33	7.1	202	4208	4.99	4 9	6 11	-1 74	
MIDDLE	432	5773	3.15	3.2	441	10266	1 92	4.7	5 20	-1.34	-0.22
HIGH	214	1875	2.19	1.9	226	4079	1.97	2.6	2.28	-0.22	-0.19
SCHOOL TYPE:											
PUBLIC	788	9599	4.06	4.9	76.9	14550	6 26	E .	E 74	A 1A	
PRIVATE	1	19	0.64	0.0	35	2800	7.64	5.0 E E	2.34 E 40	9.10	0.03
CATHOLIC	20	657	1.41	1.1	82	1556	1.09	3.3 1.9	1.75	-0.32	-0.18
GEOGRAPHIC REGION:											
NORTHEAST	183	2698	3.44	3.5	190	3075	3 77	0 2	7 66		
NORTH CENTRAL	225	3534	3.74	z 0	254	50/5	3.//	7.2	7.00	0.34	0.05
SOUTH	272	291.8	5.02	6 9	244	2747	3.40	3.0	3.05	-0.20	-0.07
MEST	161	1423	2 60	2 7	174	7/47	3.37	3.7	5.65	~1.63	-0.29
		1763		C.J	1/0	3643	4.24	0.0	5.02	1.95 *	0.39
CONTINITY TYPE:											
URBAN	258	1532	2.28	2.2	221	3087	3.25	9 1	<b>4 17</b>	0 04	A 15
SUBURBAN	439	4776	3.60	4.5	622	6650	3.37	4.9	4 79	<b>v.70</b> <b>0</b> 26	U.15
RURAL	139	4217	4.74	5.5	243	9268	4.09	4.4	4.86	-0.66	-0.13

\* SIGNIFICANT AT .05 OR LESS



. •



227

### TABLE 4-11

#### PERCENT OF STUDENTS CLASSIFIED AS DISADVANTAGED

		NLS ]		HSB 1980							
	SAMPLE N	WEIGHTED N	MEAN	\$.D.	SAMPLE N	WEIGHTED N	MEAN	s.D.	POOLED S.D.	1980-1972 DIFFERENCE	EFFECT SIZE
TOTAL	671	92 35	21.82	22.1	921	19177	13.90	19.8	20.82	-7.92 *.	-0.38
AVERAGE SES OF STUDENTS:											
LON	205	3363	36.90	24.7	218	4436	30.72	28.5	26.76	-6.18	-0.23
MIDDLE	329	4592	15.20	15.7	454	10151	10.76	12.8	14.11	-4.44 *	-0.31
HIGH	137	1281	5.95	7.1	229	4120	3.09	5.5	6.14	-2.84 <b>*</b>	-0
SCHOOL TYPE:											
PUBLIC	622	8465	22.21	21.6	804	14707	17.27	21.1	21.33	-4.94 *	-0.23
PRIVATE	0	0	0.0	0.0	36	3013	1.82	5.6	5.80	0.0	
CATHOLIC	16	414	9.34	13.8	81	1457	4.79	9.2	10.24	-4.54	-0.44
GEOGRAPHIC REGION:											
NORTHEAST	139	2054	16.28	20.0	192	3137	14.03	17.6	18.69	-2.25	-0.12
NORTH CENTRAL	158	2729	15.87	16.4	263	6019	8.82	13.1	14.49	-7.05 *	-0.49
SOUTH	276	348 <b>0</b>	31.59	25.2	287	6601	20.47	25.4	25.34	-11.12 *	-0.44
WEST	98	972	15.19	15.3	179	3419	10.02	14.7	14.95	-5.17	-0.35
CONTUNITY TYPE:											
URBAN	194	1077	22.27	25.0	227	3258	14.87	24.4	24.74	-7.40	~0.30
SUBURBAN	324	3605	14.31	17.7	450	6863	9.77	15.5	16.46	-4.54 *	-0.28
RURAL	150	4511	27.78	22.8	244	9056	16.68	20.3	21.35	-11.11 *	-0.52

\* SIGNIFICANT AT .05 OR LESS



÷.



#### **B. STAFF CHARACTERISTICS**

Another set of factors affecting the quality of students' educational experiences is the nature of the school's faculty. Relatively comparable data were available for both 1972 and 1980 in four areas: 1) number of students per classroom teacher, 2) percentage of staff with advanced degrees, 3) teacher turnover rates, and 4) racial/ethnic composition of the school's staff.

## 1. <u>Student/Staff Ratios</u>

Between 1972 and 1980, the average number of students per high school classroom teacher dropped from 17.6 to 14.7. (See Table 4-12.) This decline was statistically significant for schools at all three SES levels, in all three types of communities, in public schools, and in the South and West. The effect size varied across these groups, however. Middle SES schools and suburban and rural school showed a moderate effect size, while high SES, urban, and Southern and Western schools showed large effect sizes.

## 2. Percentage of Staff with Advanced Degrees

Table 4-13 shows the percentage of schools with low, moderate and high numbers of high school teachers holding master's or doctor's degrees. Nationally, the percentage of schools where the majority of the staff hold advanced degrees increased from 22.1 percent to 31.6 percent between 1972 and 1980. The largest shifts occurred in low SES schools, rural schools, and schools located in the South. In each of these three categories, the percentage of schools where more than 50 percent of the teachers held master's or doctor's degrees increased from approximately 11 percent to 27 percent over this eight-year period. Large differences among groups of schools remained in 1980, however. For example, 48 percent of high SES and only 24 percent of low SES schools had a majority of their teachers with advanced degrees. Similar contrasts are 57 percent in the Northeast versus 22.5 percent in the West, and 43 percent in urban schools versus 22 percent in rural schools.

### 3. <u>Teacher Turnover</u>

Administrators were asked to report the percentage of full-time high school teachers who left their school for reasons other than death or retirement. The results are contained in Table 4-14. The percentage of schools losing more than 20 percent of their staff in one year increased from 17.5 percent to 20.1 percent between 1972 and 1980. The problem of growing teacher turnover rates is magnified when one looks at the change in the percentage of schools with turnover rates of 10 percent or more. Nationally, the figures increased from 37.8 percent to 46.2 percent between 1972 and 1980, and in urban communities, the percentages grew from 26.9 percent to 39.5 percent. By 1980, there was little difference in teacher turnover rates among low, medium and high SES schools, but nonpublic schools, rural schools, and schools in the North Central regions had a disproportionate number of schools with high staff turnover.



#### NUMBER OF STUDENTS FER TEACHER

-----

	NLS 1972					HSB 1980					
	SAMPLE N	WEIGHTED N	MEAN	<b>S.</b> D.	SAMPLE N	MEIGHTED N	MEAN	\$.D.	POOLED S.D.	19 <b>80-1972</b> Difference	EFFEC SIZE
TOTAL	1238	18666	17.61	5.8	908	19234	14.74	6.7	6.23	-2.86 *	-0.46
AVERAGE SES OF STUDENTS:											
LOH	306	58%	17.58	6.3	207	4257	14.98	8.0	7.01	-2.61 *	-0.37
MIDDLE	616	9290	17.15	5.3	454	10397	15.04	5.8	5.52	-2.11 *	-0.38
HIGH	316	3481	18.86	6.2	229	3983	15.01	6.7	6.44	-3.84 *	-0.60
SCHOOL TYPE:											
PUBLIC	1108	15775	17.76	5.7	793	14816	15.97	6.3	5.97	-1.79 <b>*</b>	-0.30
PRIVATE	12	811	11.58	4.4	34	2898	6.86	3.8	4.07	-4.71	-1.16
CATHOLIC	72	1644	17.93	5.2	81	1519	17.83	4.3	4.78	-0.10	-0.02
GEOGRAPHIC REGION:											
NORTHEAST	258	3655	16.91	5.7	192	3112	15.22	7.1	6.34	-1.68	-0.27
NORTH CENTRAL	320	5793	16.12	5.5	257	5900	14.92	5.7	5.60	-1.20	-0.21
SOUTH	457	6234	17.68	4.9	278	6559	14.78	6.4	5.52	-2.90 *	-0.53
WEST	203	2984	21.20	6.7	181	3664	13.99	8.4	7.57	-7.21 *	-0.95
CONTUNITY TYPE:											
URBAN	372	2836	19.66	5.6	224	3111	13.81	9.2	7.18	-5.84 *	-0.81
SUBURBAN	619	7375	18.70	5.7	438	6795	17.33	5.8	5.74	-1.37 <b>*</b>	-0.24
RURAL	240	8361	15.94	5.6	246	9327	13.17	5.7	5.68	-2.77 *	-0.49

\* SIGNIFICANT AT .05 OR LESS



# PERCENT OF FULL TIME HIGH SCHOOL TEACHERS WITH MASTER'S OR DOCTOR'S DEGREES

			NLS 1978	2	HSB 1980					
	NUMBER OF SCHOOLS	% WITH 0-29%	% HITH 30-49%	% WITH 50-69%	% WITH 70-100%	NUMBER OF SCHOOLS	% WITH 0-29%	% WITH 30-49%	% WITH 50-69%	% HITH 70-100;
TOTAL	1254	47.9	30.0	15.2	6.9	<b>%9</b>	36.3	32.1	19.9	11.7
AVERAGE SES OF STUDENTS:										
LON	200	49 8	20.0							
MIDDLE	624	47.6	20.0	7.2	3.0	229	36.1	39.9	11.3	12.7
HIGH	221	43.4	54.0	15.6	6.9	483	41.4	29.2	19.0	10 4
	361	22.0	36.7	27.4	13.3	234	21.2	30.6	32 7	16 6
SCHOOL TYPE:										13.3
PUBLIC	1125	E1 0								
PRIVATE	1129	31.2	28.5	14.2	6.2	849	36.6	31.2	18.8	13 4
CATHOLTC	16	30.5	59.1	13.6	1.8	37	39.2	32 8	24.2	13.7
	12	18.8	37.8	28.1	15.3	83	27.6	39.1	22 3	3.0
GEOGRAPHIC REGION:										11.0
NORTHFAST										
NORTH CENTRAL	200	29.5	30.9	23.5	16.1	206	20.3	21.8	26 0	71 0
SOITH	324	49.2	30.1	14.5	6.3	275	40.7	31 6	14 0	31.7
LIFET	462	55.5	32.2	10.9	1.4	297	38.6	34 0	21 7	11./
HEJI	208	52.2	24.5	15.3	0.3	191	39 2	79.0	21.3	0.1
COMPRESSION TYPE						-/-	37.2	30.C	10.2	9.3
	380	31.9	37.7	20.7	9.8	238	77 C			
JUDUKDAN	626	34.6	34.9	19.8	10.8	668	33.3	23.3	26.6	16.5
RUKAL	241	65.4	23.5	8.8	23	943	30.3	30.5	23.7	15.5
						203	41.0	35.3	14.7	7.3

NOTE: PERCENTAGES ARE BASED ON WEIGHTED DATA





÷۲۰..

### TABLE 4-14

## PERCENT OF FULL TIME HIGH SCHOOL TEACHERS LEAVING SINCE END OF LAST SCHOOL YEAR

			NLS 197	2				HS8 1980	0	
	NUMBER OF SCHOOLS	2 WITH 0-4%	2 WITH 5-9%	% ИІТН 10–19%	% WITH 20-100%	NUMBER DF SCHOOLS	2 MITH 0-42	2 WITH 5-92	% WITH 10-19%	% WITH 20-100%
TOTAL	1258	46.8	15.4	20.3	<sup>.</sup> 17.5	979	38.1	15.7	26.1	20.1
AVERAGE SES OF STUDENTS:										
LOH	310	51.2	12.3	15.3	21.3	232	43.3	7.7	24.7	24.3
MIDDLE	625	42.7	17.7	23.5	16.1	487	35.4	19.1	25.4	20.2
HIGH	323	50.2	14.4	20 - 3	15.1	238	37.3	17.1	28.1	17.4
SCHOOL TYPE:										
PUBLIC	1126	47.7	16.3	19.0	17.0	860	41.4	16.1	25.4	17.0
PRIVATE	12	22.9	5.5	62.3	9.2	36	27.0	15.6	26.4	31.0
CATHOLIC	73	52.2	9.8	8.8	29.2	83	25.8	12.2	31.9	30.1
GEOGRAPHIC REGION:									_	
NORTHEAST	262	54.1	20.2	20.6	5.1	207	55.1	18.8	17.9	8.2
NORTH CENTRAL	327	40.2	14.4	20.5	24.9	276	30.9	16.0	27.0	26.1
SOUTH	461	44.3	14.7	21.7	19.3	299	33.9	13.5	32.3	20.3
WEST	208	56.0	12.6	16.5	14.9	197	42.4	16.4	20.5	20.7
COMUNITY TYPE:										
URBAN	382	57.1	16.0	14.0	12.9	241	44.6	15.9	23.3	16.2
SUBURBAN	628	47.4	17.3	21.4	13.9	472	42.1	20.2	24.3	13.5
	241	43 0	17 6	21 3	22 1	266	33.0	12.4	28.3	26. <b>2</b>

21.3

22.1

13.6

.

NOTE: PERCENTAGES ARE BASED ON WEIGHTED DATA

241

43.0



RURAL

62

266

33.0

12.4

## 4. Racial/Ethnic Composition of Staff

As shown in Table 4-15, the percentage of schools with predominantly minority staff (more than 50 percent minority faculty) was very small in both 1972 and 1980 (4 percent), while the percentage with nearly all white staff was large (70 percent). There was, however, a large increase in the percentage of low SES schools with mostly minority staff members and a moderate increase in the percentage of schools with more than 20 percent minority staff in the South. Looking across the classification variables in 1980, one finds that larger percentages of low SES schools, urban schools, and schools in the South have high concentrations of minority staff than do other school types.

-29-

## C. EDUCATIONAL PROGRAMS AND TEACHING METHODS

### 1. Educational Programs

The 1972 and 1980 school questionnaires provided few comparable variables describing school programs. Within the limits of existing data, this section examines the availability of special educational programs and the use of ability grouping in 1972 and 1980.

a. <u>Handicapped Education</u>. Table 4-16 shows the type of placement given handicapped students in 1972 and 1980 (1 = no special classes, 2 = some special classes, and 3 = all special classes). The figures imply that schools began to make slightly greater use of special classes during that period. The small change masks two opposing trends: the larger number of severely handicapped students served by schools in 1980 who require special classes, and the effort to "mainstream" mildly handicapped students into regular classrooms. Public schools appear to make greater use of special classes than do private or Catholic schools. This variation may be explained by the different mix of handicapped students served in each sector.

b. <u>Federal Programs</u>. Tables 4-17 through 4-20 report the percentage of schools participating in four federal programs: Title I (Education of Children of Economically Disadvantaged), Title VII (Bilingual Education), Vocational Education Basic Programs, and Vocational Education Consumer and Homemaking Education. The percentage of schools participating in Title I decreased generally from 67.1 percent to 55.6 percent between 19/2 and 1980. The largest declines are found among low and high SES schools, and schools in the South and West. Program participation increased among middle SES and Catholic schools.

More schools chose to participate in the federal bilingual education program between 1972 and 1980. This increase was consistent across all classifications, but reached significance only for high SES, public, urban, and suburban schools. The increase was greatest in urban schools. More surprisingly, the participation rate of high SES schools nearly tripled, from 3.8 percent to 10.7 percent.



## PERCENT OF CURRENT FACULTY WHO ARE WHITE


			NLS 1972	}	HSB 1980					
	NUMBER OF SCHOOLS	% WITH 0-49%	% MITH 50-79%	% MITH 80-94%	% WITH 95-100%	NUMBER OF SCHOOLS	% WITH 0-49%	% WITH 50-79%	% HITH 80-94%	% HITH 95–188
TOTAL	1232	4.4	9.3	15.3	71.0	919	3.9	7.6	18.3	70.2
AVERAGE SES OF STUDENTS:										
LON	299	6.9	17.3	21.3	54.6	213	14.4	19.2	20.2	40.Z
MIDDLE	616	4.3	6.9	13.3	75.5	458	0.8	4.8	21.2	73.Z
HIGH	317	0.4	2.2	10.3	87.1	226	0.0	2.5	11.1	86.4
SCHOOL TYPE:										<i></i>
PUBLIC	1104	4.2	10.2	16.1	69.5	807	5.0	8.8	20.4	65.8
PRIVATE	11	0.0	0.0	14.3	85.7	35	0.0	3.9	7.1	89.8
CATHOLIC	71	2.3	0.0	7.2	90.5	77	0.0	2.7	17.4	79.8
GEOGRAPHIC REGION:										
NORTHEAST	250	2.9	1.7	8.5	87.0	195	1.4	2.4	8.6	87.6
MODTH CENTRAL	321	3.2	3.1	10.3	83.4	260	1.2	2.6	5.6	90.6
SOUTH	456	7.5	22.9	23.9	45.6	281	8.8	15.4	30.0	45.7
HEST	205	2.0	1.7	14.8	81.5	183	1.6	6.3	25.3	66.8
COMPRINTTY TYPE:										
IDRAM	372	4.8	19.5	22.5	53.2	226	7.4	17.1	16.1	59.4
	618	4.6	8.2	15.2	72.0	448	1.3	6.0	22.7	70.1
RURAL	235	4.1	7.3	13.2	75.4	245	4.7	5.7	15.7	73.9

NOTE: PERCENTAGES ARE BASED ON WEIGHTED DATA





## TABLE 4-16

# EXTENT OF SPECIAL ACCOMODATIONS FOR HANDICAPPED STUDENTS ( 1=NO SPECIAL CLASSES, 3=ALL SPECIAL CLASSES )

	NLS 1972				HS3 1980						
	SAMPLE N	WEIGHTED N	MEAN	S.D.	SAMPLE N	WEIGHTED N	MEAN	s.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	917	11459	1.59	0.5	924	17724	1.68	0.5	0.50	0.10 ×	0.19
AVERAGE SES OF STUDENTS:											
LOM	208	3158	1 4 2	0 4				_			
MIDDLE	479	4 2 9 1	1.00	0.0	218	4368	1.83	0.4	0.50	0.21 ¥	0.42
HTGH	777	0201	1.33	0.5	465	9651	1.69	0.4	0.49	0.14 ¥	0.29
112011	230	2020	1.65	0.5	219	3221	1.53	0.5	0.49	-0.12	-0.25
SCHOOL TYPE:											
PUBLIC	857	10745	1 / 7		<b>.</b>						
PRIVATE	037	10305	1.03	0.5	845	14932	1.78	0.4	0.47	0.15 *	0.32
CATHOLIC	<u> </u>	30	1.00	0.0	22	1623	1.22	0.4	0.41	0.22	0 55
CATHOLEC	24	/ 30	1.05	0.1	57	1169	1.06	0.2	0.18	0.01	0.07
GEOGRAPHIC REGION:											
NORTHEAST	200	2870	1 60	• •							
NORTH CENTRAL	244	2010	1.50	0.0	196	3146	1.68	0.5	0.53	0.10	0.19
SOLITH	240	3017	1.54	0.5	263	5670	1.74	0.4	0.47	0.20 *	0.42
HEST	29/	3262	1.58	0.6	280	5667	1.71	0.4	0.52	0.12	0 24
HE51	174	1512	1.72	0.5	185	3241	1.54	0.5	0.47	-0.18 ¥	-0.38
COMMUNITY TYPE:											
URBAN	981	1 7 74	1 64								
SURIDRAM	201	1//4	1.54	0.5	229	2654	1.55	0.5	0.51	0.01	0.02
DIDA1	4/4	5172	1.60	0.5	439	6428	1.61	0.4	0.49	0 01	0.02
RVRAL	152	4465	1.59	0.6	256	8642	1.78	0.4	0.49	0.18 *	0.34

\* SIGNIFICANT AT .05 OR LESS

65

.



. Aborth

٩

		NLS 1972						
	SAMPLE N	WEIGHTED N	PERCENT	SAMPLE N	WEIGHTED N	PERCENT	1980-1972 Difference	
TOTAL	1169	17705	67.1	962	20384	55.6	-11.4 *	
AVERAGE SES OF STUDENTS:								
LOW	287	5736	89.0	227	4640	66.4	-22.6 *	
MIDDLE	585	3852	61.8	475	10763	65.9	4.1	
HIGH	297	3117	41.5	238	4350	24.4	-17.1 *	
SCHOOL TYPE:								
PUBLIC	1050	15126	75.1	845	157 <b>06</b>	69.7	-5.4	
PRIVATE	10	726	0.0	37	3170	1.1	1.1	
CATHOLIC	69	1455	14.8	80	1508	23.6	8.8	
GEOGRAPHIC REGION:								
NORTHEAST	246	3296	65.6	205	3395	68.4	2.8	
NORTH CENTRAL	305	5480	66.9	270	6140	68.5	1.5	
SOUTH	432	6017	71.1	295	6948	52.3	-18.8 *	
HEST	186	2912	60.8	192	3901	30.3	-30.5 <b>*</b>	
COMUNITY TYPE:								
URBAN	342	2670	33.2	238	3454	24.5	-8.7	
SUBURBAN	590	6884	62.1	463	7124	49.2	-12.8 *	
RURAL	230	8058	82.9	261	98 <b>06</b>	71.2	-11.7 *	

#### PERCENT OF HIGH SCHOOLS PARTICIPATING IN TITLE I, ELEMENTARY AND SECONDARY EDUCATION ACT (EDUCATION OF CHILDREN OF ECONOMICALLY DISADVANTAGED)

\* SIGNIFICANT AT .05 OR LESS



- -

.....

\_\_\_\_\_

## TABLE 4-18

PERCENT	OF	HIGH	SCHOOLS	PARTICIPATING	IN	TITLE	VII,	ELEMENTARY	AND	SECONDARY	EDUCATION	ACT
				(81)	LIN:	GUAL EC	UCAT	ION)				

-----

		NLS 1972				s , <b></b>	
	SAMPLE N	WEIGHTED N	PERCENT	SAMPLE N	WEIGHTED N	PERCENT	198 <b>0</b> -1972 Difference
TOTAL	1051	15583	6.9	957	20333	10.6	3.7 *
AVERAGE SES OF STUDENTS:							
LOW	241	4643	7.4	228	4627	13.1	5.6
MTDDLF	522	7926	7.7	474	10782	10.0	2.3
HIGH	288	3014	3.8	236	4327	10.7	6.9 ¥
SCHOOL TYPE:							
PUBLIC	941	13222	8.0	838	15628	13.0	5.0 ×
PRIVATE	10	726	0.0	37	3170	3.4	3.4
CATHOLIC	66	1399	0.5	82	1534	0.4	-0.1
GEOGRAPHIC REGION:							
NORTHEAST	223	2742	7.6	201	3356	12.0	4.3
NORTH CENTRAL	276	4921	4.6	270	6122	7.6	3.0
SOUTH	377	5121	6.1	290	6876	7.9	1.8
WEST	175	2799	11.4	1%	3978	18.6	7.2
COPPUNITY TYPE:							
URBAN	313	2503	8.2	238	3438	20.8	12.6 *
SUBURBAN	539	6082	7.7	456	7035	14.5	6.8 *
RURAL	192	6904	5.6	263	9859	4.2	-1.3

\* SIGNIFICANT AT .05 OR LESS

67



Ŀ.

### TABLE 4-19

# PERCENT OF HIGH SCHOOLS PARTICIPATING IN TITLE I-B, VOCATIONAL EDUCATION ACT OF 1963 (VOCATIONAL EDUCATION BASIC PROGRAMS)

	NLS 1972						
	SAMPLE N	WEIGHTED N	PERCENT	SAMPLE N	NEIGHTED N	PERCENT	1980-1972 Difference
TOTAL	1140	16862	62.5	948	19926	52.7	-9.8 ¥
AVERAGE SES OF STUDENTS:							
LOW	272	E179	76 6				
MIDDLF	676	91/0	/	222	4335	66.4	-8.0
HTCH	2/2	0373	00.2	468	10635	59.2	-1.0
11201	273	2089	49.2	237	4333	28.8	-20.4 ¥
SCHOOL TYPE:							
PUBLIC	1021	14292	70.4	829	16333	47 0	
PRIVATE	11	766	1.5	74	19333	07.0	-2.7
CATHOLIC	66	1621	77	30	5021	1.1	-0.4
		1461	1.3	83	1572	5.2	-2.2
GEOGRAPHIC REGION:							
NORTHEAST	233	3056	55.6	1 96	7761	40.0	
NORTH CENTRAL	303	5390	55 A	240	3301	47.0	-0.5
SOUTH	417	5521	66 7	207	0020	20.9	1.2
WEST	197	2805	74 5	290	0039	57.5	-9.4
	107	2075	/4.3	193	3959	41.8	-32.7 ¥
COMMUNITY TYPE:							
URBAN	348	2671	54.2	230	7704	77 4	<b></b>
SUBURBAN	573	4588	41 5	230	3374	33.4	-20.8 *
RURAL	212	7500	44 6	458	7104	54.2	-7.3
	C1C	7347	00.V	260	9428	58.6	-7.5

\* SIGNIFICANT AT .05 OR LESS



		NLS 1972					
	SAMPLE N	WEIGHTED N	PERCEN7	SAMPLE N	WEIGHTED N	PERCENT	1980-1972 Difference
TOTAL	1097	15967	50.2	%3	20347	59.3	9.1 ¥
AVERAGE SES OF STUDENTS:							
LOW	260	4590	54.5	227	4507	73.4	18.9 ¥
MIDDLE	550	8318	49.8	474	<b>10</b> 84 <b>8</b>	67.9	18.0 ¥
HIGH	287	3059	44.8	239	4352	29.8	-15.0 *
SCHOOL TYPE:							_
PUBLIC	984	13528	57.6	843	15605	76.3	18.7 ¥
PRIVATE	11	766	1.5	37	3170	1.1	-0.5
CATHOLIC	64	1398	0.8	83	1572	8.3	7.5
GEOGRAPHIC REGION:							
NORTHEAST	220	2844	38.3	198	3336	44.7	6.4
NORTH CENTRAL	288	5249	45.9	271	6051	66.3	20.4 ¥
SOUTH	406	5542	56.5	297	69 <b>80</b>	68.2	11.6 ×
WEST	183	2332	59.5	197	3981	45.6	-13.9
CONTUNITY TYPE:							
URBAN	324	2566	37.7	235	3429	33.4	-4.3
SUBURBAN	560	6380	49.5	468	7201	58.7	9.1 ¥
RURAL	206	6927	55.3	260	9717	69. <b>0</b>	13.7 ×

PERCENT O F HIGH SCHOOLS PARTICIPATING IN TITLE I-F, VOCATIONAL EDUCATION ACT OF 1963 (CONSUMER AND HOMEMAKING EDUCATION)

\* SIGNIFICANT AT .05 OR LESS

69



ST COPY AVAILABLE

\_\_\_\_

The percentage of schools involved in the federal vocational education program declined nationally from 62.5 percent to 52.7 percent. Large and significant decreases in participation occurred among high SES schools, in urban schools, and in schools located in the West. In 1980, the program was available primarily in the public schools and in low and middle SES schools.

The percentage of schools participating in consumer and homemaking education (Title I-F, Vocational Education Act of 1963) increased significantly, from 50.2 percent in 1972 to 59.3 percent in 1980. In 1972 this program was available somewhat more often in low and middle SES schools than in high SES schools. By 1980, however, this gap had widened considerably with low and middle SES schools showing significant increases in participation while high SES schools showed a significant decrease in participation. This program is available primarily in public schools, which showed a significant increase in participation between 1972 and 1980. In 1972, more schools in the South and the West than in the Northeastern and North Central regions participated. By 1980 the program was more often available in the North Central and Southern regions, and schools in both of these regions showed significant increases in participation. This program is available more frequently in rural than in urban schools. Both suburban and rural schools showed significant increases in participation between 1972 and 1980, but a decrease in urban school participation widened the difference across schools in different types of communities.

Advanced Placement. The percentage of schools offering Advanced Placement (AP) courses increased significantly, from 15 percent in 1972 to 30 percent in 1980. (See Table 4-21.) The availability of these courses varied greatly across types of schools. For example, in 1980 only 22 percent of low SES schools but 62 percent of high SES schools offered AP courses. AP courses were less available, in 1980, to public than to nonpublic school students. Nearly 60 percent of schools in the Northeastern region but only 18 percent of schools in the North Central region offered AP courses in 1980. Moreover, 42.8 percent of suburban schools but only 18.2 percent of cural schools provided AP for their The increases in the availability of AP courses between students. 1972 and 1980 reached significance for all SES groups but was greatest in high SES schools, thus increasing the gap in the availability of AP courses in high and low SES schools. Similarly, the increase was significant in all geographic regions but was greatest in the Northeast, again increasing the difference among regions. The increase in the availability of AP courses was also significant for public schools and for suburban and rural schools.

d. <u>Ability Grouping</u>. The percentage of schools using ability grouping declined significantly between 1972 and 1980. (See Table 4-22.) In 1972, 59.5 percent of the schools used grouping. By 1980 this had decreased to 51.2 percent. Grouping was used more often in high than in low SES schools in both years. The decline in the use of grouping was

70

<u>.</u>•

### TABLE 4-21

## PERCENT OF HIGH SCHOOLS OFFERING ADVANCED PLACEMENT COURSES

	NLS 1972				HSB 1980			
	SAMPLE N	WEIGHTED N	PERCENT	SAMPLE N	WEIGHTED N	PERCENT	1980-1972 Difference	
TOTAL	1129	15688	15.0	971	20184	30.0	15 0 *	
AVERAGE SES OF STUDENTS:			•					
LOW	266	4296	5.8	231	4675	22.0	16.2 <b>*</b>	
MIDDLE	563	8380	13.7	479	19848	21.8	8.1 ¥	
HIGH	300	3011	31.7	240	4200	62.1	30.3 *	
SCHOOL TYPE:								
PUBLIC	1016	13238	14.3	855	15801	28.3	14.0 *	
PRIVATE	11	799	6.8	36	2857	33.9	27.0	
CATHOLIC	62	1326	25.4	80	1526	40.2	14.8	
GEOGRAPHIC REGION:								
NORTHEAST	237	3227	28.4	204	3389	59.6	31.2 <b>*</b>	
NORTH CENTRAL	299	5158	10.0	273	6123	18.4	8.3 ¥	
SOUTH	403	5166	11.2	297	6979	25.1	13.9 <b>*</b>	
WEST	190	2137	16.0	197	3693	31.2	15.2 *	
COMPUNITY TYPE:								
URBAN	357	2630	32.0	241	3160	38.4	6.4	
SUBURBAN	563	6301	19.5	465	7095	42.8	23.3 *	
RURAL	202	6663	3.9	265	9929	18.2	14.2 <b>*</b>	

\* SIGNIFICANT AT .05 OR LESS

71



۰.

- - e

BEST COPY AVAILABLE

.

## TABLE 4-22

## PERCENT OF HIGH SCHOOLS USING ABILITY GROUPINGS

		NLS 1972					
	SAMPLE N	WEIGHTED N	PERCENT	SAMPLE N	WEIGHTED	PERCENT	1980-1972 DIFFERENCE
TOTAL	1118	16446	59.5	984	20410	51.2	-8.3 *
AVERAGE SES OF STUDENTS:							
LOW	275	5039	<b>5</b> 7 4	277	4 E 70		
MIDDLE	663	9166	50 1	233	45/9	44.2	-9.2
HTGH	200	2023	37.1	486	10801	49.1	-10.1 *
	290	3233	07.7	242	4390	61.7	-8.1
SCHOOL TYPE:							
PUBLIC	997	13665	58.7	864	15925	E0 E	- 6 0 ×
PRIVATE	12	811	41 0	37	19014	50.5	-0.2 -
CATHOLIC	67	1545	72 9	3/	3014	52.1	11.1
		1909	16.0	03	1572	50.6	-16.1
GEOGRAPHIC REGION:							
NORTIEAST	242	3454	80.7	208	3450	71 5	-9.2
NORTH CENTRAL	279	3976	46.6	278	6206	49 3	- 7.2
SOUTH	412	5220	58.3	303	7074	46.3	-4.3
WEST	185	2705	50.5	303	7030	40.3	-10.0
	205	C7 75	50.5	1 43	5/18	52.9	-5.6
COMMUNITY TYPE:							
URBAN	343	2663	56.5	264	3485	56 6	-0.1
SUBURBAN	566	6630	70.5	475	7112	46 E	-4.0
RURAL	203	7087	50 4	7/3	/110	7.3	-0.0
	243	/ 00/		203	2012	59.8	-10.7

\* SIGNIFICANT AT .05 OR LESS

122



•• .

significant only for middle SES schools. Catholic schools used grouping more than public or private schools. The extent of this difference decreased between 1972 and 1980, however, due to a decline in the use of grouping in Catholic schools and an increase in the use of grouping in private schools. Grouping was used more often in the Northeast than in other regions and more often in suburban than in rural or urban schools. It should be noted, however, that the information on grouping may not have identical meaning in 1972 and 1980 since the questions were phrased somewhat differently in the two questionnaires.

#### 2. Teaching Methods

Students were asked how often each of six instructional methods were used in their classes: listening to teachers' lectures; participating in student-centered discussions; working on a project or in a laboratory; writing essays, themes, poetry or stories; having individualized instruction; and using teaching machines or computer-assisted instruction. Answers were scaled as 1 = Never, 2 = Seldom, 3 = Fairly Often, and 4 = Frequently.

Table 4-23 summarizes the mean responses for 1972 and 1980 and the changes that occurred during that period. The most frequently used instructional method in both years was listening to lectures by the teacher with a mean rating of 3.26. Writing essays, themes, etc., and student-centered discussions had average ratings slightly below 3, which is Fairly Often. The use of these three approaches showed little change between 1972 and 1980. Effect sizes were 0.01, -0.03, and -0.08, respectively. The remaining three methods, which were less frequently used, have changes with small effect sizes, ranging from -0.12 for working on a project or in a lab to 0.17 for use of theaching machines or computer-assisted instruction. The different rates of change did not affect the rank order of the instructional approaches over the eight years.

#### Table 4-23

	Mean Re	sponse	Difference	Effect <u>Size</u>	
Instructional Method	<u>1972</u>	1980	<u>1980-1972</u>		
Listening to Teachers' Lectures	3.26	3.27	0.01	0.01	
Writing Essays. Themes, etc.	2.85	2.82	-0.03	-0.03	
Student-Centered Discussions	2.73	2.67	-0.06*	-0.08	
Work on Project or in Lab	2.48	2.37	-0.12*	-0.12	
Individualized Instruction	1.98	2.09	0.11*	0.13	
Teaching Machines or CAI	1.48	1.62	0.15*	0.17	

\*Significant at .05 or less

Some variations can be seen when the data are arrayed by the classification variables. (See Tables 4-24 through 4-29.) The use of lectures increased somewhat in academic programs (0.12), among high SES students (0.11), and in Catholic (0.18) schools. The largest significant effect size is found for high SES students attending Catholic schools (0.30).



	NLS 1972					HSB 1980					
	SAMPLE	WEIGHTED		******	84471						
	N	N	MEAN	S.D.	N	N	MEAN	S.D.	POOLED	1980-1972	EFFEC
TOTAL	16517	3018167	3.26	0.8	27909	3009033	3.27	0.8	0.0.		3120
SEX:							2.67	0.0	0.77	0.01	0.01
MALE		1									
FEMALE	6105	1501397	3.22	0.8	12789	1389867	3.23	0.8	0 77	0.02	
	0327	1515753	3.31	0.8	13972	1502299	3.32	0.8	0.76	0.02	0.02
SES:										0.01	0.02
LON	A 777	7007/7									
MIDDLE	4/33	129/03	3.17	0.8	8274	799167	3.14	0.8	0.81	-0 04	
HIGH	70/2	1547017	3.26	0.8	12719	1414929	3.27	0.8	0.76	0 01	-0.04
	2021	733544	3.36	0.7	6153	720755	3.44	0.7	0.70	0.01	0.02
RACE:									••••	V. VO ×	0.11
MHITE	19780										
BLACK	12/00	2513880	3.28	0.7	19745	2352084	3.30	0.8	0.75	0.02	
ASIAN-AMERICAN	2002	249152	3.20	0.8	3670	335200	3.16	0.8	0.41	-0.02	0.03
AMERICAN THITAN	191	27544	3.35	0.7	357	38562	3.37	0.8	0.75	-0.04	-0.05
MEXTCAN-AMERICAN	10/	31116	3.09	0.8	214	21937	3.16	0.9	0.75	0.02	0.02
PUERTO PICAN	547	71936	3.17	0.7	1864	100223	3.22	0.8	0 78	0.07	0.98
OTHER HISPANIC	93	9513	2.88	0.9	303	17864	3.12	0.9	0.75	0.04	0.05
- THE TROPANEC	119	18268	3.17	0.8	952	65199	3.15	0.8	0.84	V.24 -0.07	0.28
SCHOOL TYPE:									0.04	-0.03	-0.03
PUBLIC	14004										
PRIVATE	14804	2678145	3.26	0.8	24372	2705198	3.26	0.8	0.77	-0.00	
CATHOLIC	100	16549	3.15	0.7	868	104229	3.33	0.7	0 73	-0.00	-0.00
	1922	234467	3.32	<b>0.</b> 7	2009	199606	3.45	0.7	0.70	0.10	0.24
GEOGRAPHIC REGION:								•••	0.70	0.13 *	0.18
NORTHEAST	7547										
NORTH CENTRAL	350/	799437	3.19	0.8	5632	691948	3.17	0.8	0 81	0.00	
SOUTH	953/	911851	3.30	0.7	8030	862802	3.35	0.7	0 74	-0.02	-0.03
WEST	5434	786089	3.30	0.8	9170	910828	3.24	0.8	0 77		0.07
	2959	520791	3.24	0.7	5077	543455	3.34	0.7	0 76	-0.0/ #	-0.08
CURRICULUM:								•••	V./4	0.10 *	0.13
	-/										
ACADEMIC	5610	959914	3.20	0.8	10194	1102206	3.20	0.8	0 77		•
VOCATTONA	6779	1385411	3.38	0.7	10480	1134104	3.46	0.0	0.70	0.00	0.01
TOCALIONAL	4127	672540	3.12	0.8	6824	727874	3.10	0.8	0.70	0.08 *	0.12
COMMENTTY TYPE:								•.•	4.02	-0.02	-0.03
LIDRAN											
SIRIPRAN	9519	780716	3.26	0.8	6418	600682	3.24	0.8	0 70		
	7906	1530876	3.28	0.7	13441	1489163	3.32	0.8	U./7 0 7E	-0.02	-0.03
	3650	634714	3.22	0.8	8050	919188	3.22	0.8	V./3 0 78	0.03	0.05
					2			4.0	U./O	-0.00	-0.00

# HOW OFTEN USED IN COURSES YOU ARE TAKING THIS YEAR: LISTENING TO THE TEACHER'S LECTURE (1=NEVER; 4=FREQUENTLY)

\*SIGNIFICANT AT .05 OR LESS


#### **BEST COPY AVAILABLE**

#### TABLE 4-25

HOW OFTEN USED IN COURSES YOU ARE TAKING THIS YEAR: WRITING ESSAYS, THEMES, POETRY, OR STORIES (1=NEVER; 4=FREQUENTLY) 

		NLS 19		HSB 19	80						
	SAMPLE N	WE IGHTED N	MEAN	<b>5.</b> D.	SAMPLE N	ME IGHTED N	MEAN	<b>S.</b> D.	POCLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	16293	2980504	2.85	0.9	27061	·2924015	2.82	0.9	0.93	-0.03	-0.03
CEV:											
MATE	8075	1482943	2.76	0.9	12374	1346264	2.74	0.9	0.92	-0.02	-0.02
FEMALE	8213	1496544	2.93	0.9	13603	1465934	2.91	0.9	0.94	-0.02	-0.02
SES:											
Ini	4654	717878	2.77	1.0	7983	773487	2.66	1.0	0.97	-0.11 *	-0.11
MTNDI F	7765	1524962	2.83	0.9	12350	1375199	2.79	0.9	0.93	-0.04	-0.04
HIGH	3817	727814	2.97	0.8	6009	703739	3.08	0.9	0.86	0.12 *	0.14
PACF:											
LAITTE	12642	2487150	2.84	0.9	19255	2295569	2.81	0.9	0.92	-0.03	-0.03
BLACK	2011	242964	3.01	0.9	3511	320564	2.92	1.0	0.95	-0.09 ¥	-0.10
ASTAN_AMEDIC IN	188	27232	2.79	0.9	341	37300	2.92	0.9	0.88	0.14	0.16
AMEDICAN THE SAM	184	30890	2.74	1.0	209	21261	2.78	1.0	0.99	0.03	0.03
MEYTCAN_AMERTCAN	511	70046	2.64	1.0	1782	95184	2.70	1.0	0.98	0.06	0.07
DIEDTO DICAN	A7	8982	2.78	1.0	284	16741	2.90	1.0	0.98	0.12	0.13
OTHER HISPANIC	115	17801	2.75	1.0	914	61729	2.75	1.0	0.97	-0.00	-0.00
SCHOOL TYPE:											
URI TC	14599	2643162	2.83	0.9	23597	2627037	2.78	1.0	0.94	-0.05 *	-0.05
DETVATE	67	16549	3.23	0.7	848	101395	3.29	0.8	0.78	0.07	0.08
CATHOLIC	1015	233645	2.94	0.8	2616	195583	3.05	0.8	0.85	0.11	0.12
SECORAPHIC REGION:											
NORTHEAST	35 39	789394	2.88	0.9	5454	670748	2.95	0.9	6.88	0.07 ×	0.68
NORTH CENTRAL	4468	901841	2.77	0.9	7 <b>817</b>	841526	2.68	1.0	0.95	-0.08 ×	-0.09
SOLITH	5350	774155	2.97	0.9	8885	884055	2.87	0.9	0.92	-0.10 ¥	-0.11
WEST	2916	515114	2.74	0.9	4905	5276 <b>8</b> 6	2.78	1.0	0.96	0.04	0.04
CENIDAI	5521	945160	2.75	0.9	9863	1069980	2.70	0.9	0.93	-0.05	-0.05
	6720	1374240	3.02	0.8	10240	1109038	3.15	0.8	0.83	0.13 *	0.16
VOCATIONAL	4051	660800	2.63	1.0	6570	702094	2.49	1.0	0.98	-0.15 *	-0.15
CONTINITY TYPE:											
LIPBAN	6443	768967	2.89	0.9	6190	581498	2.87	0.9	0.93	-0.03	-0.03
SUBURBAN	7816	1515929	2.86	0.9	13065	1450137	2.85	0.9	0.93	-0.01	-0.01
DIDAI	3601	626060	2.76	0.9	7806	892380	2.74	1.0	0.95	-0.02	-0.02

\*SIGNIFICANT AT .05 OR LESS

0 ed by ERIC

-----

	NLS 1972				HSB 1	980					
	SAMPLE	HEIGHTEN	, J	*******							
	N	N	MEAN	5.0.	N	NETENIED	MEAN	S.D.	POOLED 3.D.	1980-1972	EFFECT
TOTAL	14308	2000614						<b></b>		DILLENDE	<b>317</b> £
10172	10340	544410	2.73	0.8	27571	2974334	2.67	0.8	0.83	-0.06 ×	-0.08
SEX:											
MALE	8120	1491730	2.66	0.8	12612	1371624	2 61				
FEMALE	8273	1506669	2.81	0.8	13843	1468341	2.73	V.O	U.83	-0.05 *	-0.06
							6.73	V.0	U.03	-0.0/ *	-0.09
963+ 174											
lut Minni s	9687	723384	2.67	0.8	8132	784875	2.57	0.9	0.85	~0.10 #	-8 12
MICH	7816	1534994	2.72	0.8	12590	1401819	2.67	0.8	0.82	-0.06 *	-0.07
1141	3638	731460	2.82	0.8	£102	714646	2.81	0.8	0.80	-0.01	-0.07
RACE:											
MITE	12719	2602840		• •							
BLACK	1C/17 9026	2506007	2./3	0.8	19563	2330143	2.68	0.8	0.82	-0.06 <b>#</b>	-0.07
ASIAN-AMERICAN	189	244773 97909	2.03	0.9	3580	326875	2.69	0.9	0.88	-0.14 #	-0.16
AMERICAN INDIAN	182	6/676 70584	2.0Y	U.8	354	38197	2.62	0.9	0.87	-0.07	-0.08
MEXICAN-AMERICAN	563	30300 71331	2.0J	V.8	213	21880	2.58	0.9	0.86	-0.07	-0.08
PUERTO RICAN	272	71331	2.0J 9 4 E	0.8	183z	98373	2.57	0.9	0.88	-0.06	-0.06
OTHER HISPANIC	115	7303 17730	6.0J 2 41	1.0	293	17007	2.58	0,9	0.93	-0.08	-0.08
		1//37	2.01	0.9	995	64766	2.66	0.9	0.89	0.05	0.06
SCHOOL TYPE:											
PUBLIC	14691	2659727	2.73		94848	7473681					
PRIVATE	47	16549	2.93	0.7	24000	20/3401	2.05	0.8	0.83	-0.07 *	-0.09
CATHOLIC	1023	235156	2.84	0.8	94 <b>69</b>	102979	2.94	0.8	0.83	0.01	0.01
		<b></b>	6101	v.v	2036	17/0/4	2./0	0.8	0.78	-0.05	-0.07
GEOGRAPHIC REGION:											
NORTHEAST	3559	7 <b>9</b> 4939	2.74	0.8	5564	482843	9 4 7				
NORTH CENTRAL	4509	905869	2.73	0.8	7917	451260	2.0/	0.8	0.83	-0.07 *	-0.09
SOUTH	5393	781015	2.74	0.8	9053	073607	2.00 9 4 E	0.8	0.81	-0.05	-0.06
NEST	2937	517593	2.70	0.8	5017	700335 527867	2.03	0.9	0.85	-0.09 #	-0.11
		-		•••		33/37/	2.00	V.0	0.03	-0.07	-0.03
CURRICULUM											
GENERAL	5565	951307	2.65	0.8	10080	1091701	2.50		0.07	-0.04 -	
ACADEMIC	6745	1380290	2.85	0.8	10399	1124855	2.21	0.0	V.03	-0.00 #	-0.07
VOCATIONAL	4087	667517	2.62	0.8	6706	715293	2.57	N.0	U./Y	-0.03	-0.04
						/ / _	6	<b>V</b> .7	V.00	-0.03	~0.06
COMUNITY TYPE:		<i>·</i>									
	4492	77717 <b>2</b>	2.77	8.0	6326	592471	2.67	0.9	0 45	-0 10 #	- 12
JUDUKBAN	7860	1523507	2.76	0.8	13277	1471795	2.70	0.8	0.82	-0.10 *	-0.12
RURAL	3613	629445	2.64	0.8	7968	910068	2.62	0.8	0.83	-0.00 -	-0.07
4		·· _			• '	1 -		•••			-0.03
	•	-			•	•					

HON OFTEN USED IN COURSES YOU ARE TAKING THIS YEAR: PARTICIPATING IN STUDENT-CENTERED DISCUSSIONS (1=NEVER; 4=FREQUENTLY) 

\*SIGNTEICANT AT .05 OR LESS



BEST COPY AVAILABLE 76

						NS5 1900					
		HLS 19	72			NSB 19	<b>104</b>				
	SAMPLE	NEIGHTED N	MEAN	S.D.	SAMPLE	NEIGHTED N	MEAN	\$.D.	POOLED S.D.	1980-1972 DIFFERENCE	EFFECT SIZE
TOTAL	16247	<b>2</b> 9 <b>73</b> 373	2.48	1.0	27240	2938544	<b>2.3</b> 7	1.0	1.01	-0.12 #	-0.11
efy:											
MAIF	8056	1480468	2.47	1.0	12445	1352884	2.43	1.0	1.02	-0.04	-0.04
FEMALE	8165	1492349	2.49	1.0	13691	1473332	<b>2</b> .32	1.0	1.00	-0.18 #	-0.18
SES:											
LOK	4615	713052	2.37	1.0	8039	777562	2.22	1.1	1.04	-0.15 #	-0.14
NTDOLE	7765	1524174	2.48	1.0	12448	1385026	2.37	1.0	1.01	-0.11 ¥	-0.11
HIGH	3815	726835	2.60	0.9	6034	706204	2.55	1.0	0.94	-0.06	-0.06
BACE:											
MHITE	12637	2485739	2.50	1.0	19355	2304562	2.39	1.0	1.00	-0.11 *	-0.11
BLACK	1978	239048	2.42	1.0	3533	323130	2.25	1.0	1.04	-0.18 *	-0.17
ASTAN-AMEDICAN	187	27121	2.75	0.9	350	3807%	2.64	1.0	0.94	-0.10	-0.11
AMERICAN INDIAN	179	29872	2.41	1.1	206	21 <b>368</b>	2.39	1.1	1.08	-0.03	-0.02
MEXTCAN-AMERICAN	534	70108	2.31	1.0	1803	96392	2.24	1.0	1.00	-0.07	-0.07
PLEPTO PTCAN	91	9360	2.22	1.1	287	16693	2.43	1.1	1.10	0.21	0.19
OTHER HISPANIC	117	18067	2.46	1.0	938	64041	2.30	1.1	1.05	-0.16	-0.16
SCHOOL TYPE:											
PUBLIC	14555	2636811	2.48	1.0	23760	2640948	2.36	1.0	1.01	-0.11 ¥	-0.11
PRIVATE	67	16549	2.31	1.0	847	101168	2.40	1.0	0.97	0.09	0.09
CATHOLIC	1913	232911	2.57	0.9	2633	196428	2.42	9.9	0.94	-0.15 *	-0.16
GEOGRAPHIC REGION:											
NORTHEAST	3529	788068	2.51	1.0	5517	678363	2.45	1.0	1.00	-0.06	-0.06
NORTH CENTRAL	4480	899635	2.54	1.0	7845	843714	2.45	1.0	1.01	-0.10 ¥	-0.10
SOUTH	5322	771020	2.37	1.0	8912	885 308	2.21	1.0	0.99	-0.16 ¥	-0.17
WEST	2916	514651	2.49	1.0	4966	531160	2.40	1.0	1.03	-0.09 *	-0.09
CURRICULUM:											
GENERAL	5499	940649	2.29	1.0	9949	1076397	2.18	1.0	0.99	-0.11 ¥	-0.12
ACADEMIC	6703	1371319	2.65	0.9	10313	1115347	2.59	0.9	0.93	-0.06 ¥	-0.07
VOCATIONAL	4044	661103	2.40	1.1	6590	704310	2.31	1.1	1.11	-0.09 *	-0.09
COMMUNITY TYPE:										_	
URBAN	4436	768217	2.48	1.0	6238	585131	2.33	1.0	1.02	-0.15 *	-0.15
SUBURBAN	7798	1512560	2.50	1.0	13129	1454780	2.41	1.0	1.00	-0.09 ¥	-0.09
RURAL	3588	62 38 35	2.47	1.0	7873	898634	2.32	1.0	1.02	-0.15 *	-0.15

#### NON OFTEN USED IN COURSES YOU ARE TAKING THIS YEAR: NORKING ON A PROJECT OR IN A LABORATORY (1=HEVER; 4=FREQUENTLY)

\*SIGNIFICANT AT .05 OR LESS



77

## BEST COPY AVAILABLE

### BEST COPY AVAMABLE

#### TABLE 4-28

HOW OFTEN USED IN COURSES YOU ARE TAKING THIS YEAR: HAVING INDIVIDUALIZED INSTRUCTION (1=NEVER; 4=FREQUENTLY)

		NLS 1	972			HSB 1	980				
	SAMPLE	WEIGHTED		*****	SAMDI F	UE TOUTED					
	N	N	MEAN	5.0.	N	N	MEAN	S.D.	POOLED S.D.	1980-1972 DIFFERENCE	EFFECT STZF
TOTAL +	16313	2983748	1.98	0.9	27600	2977080	2.09	0.9	Ô 87	0 11 ×	
SEX:									0.07	V.11 *	0.13
MALE	8001	1484027	1 04								
FEMALE	A217	1405967	1.74	0.8	12639	1373446	2.04	* 3	0.86	0.10 ×	0.12
	0017	1472004	2.01	0.9	13850	1490713	2.1.		₹.88	0.11 *	0.13
SES:											
LON	4450	71 71 94	2 61								
NICOLE	7796	1528158	2.01	0.9	8143	787667	2.10	0.9	0.90	0.09 ¥	0.10
HIGH	3820	778404	1.98	0.9	12613	1402637	2.07	0.9	0.87	0.08 ¥	0.10
•	3020	/20000	1.42	0.8	6113	715909	2.11	0.9	0.84	0.19 ¥	0.22
RACE:											
MHITE	12676	2401254	3.04								
BLACK	2000	2473634	1.90	0.9	19616	2335861	2.06	0.9	0.86	0.10 ¥	0.12
ASIAN-AMERICAN	190	27714	2.13	0.9	3576	326736	2.22	0.9	0.91	0.07	0.07
AMERICAN INDIAN	190	20021	2.1/	0.9	354	38608	2.10	0.9	0.89	-0.07	-0.07
MEXICAN-AMERICAN	E 12	67761	1.97	0.9	213	21880	2.19	0.9	0.94	0.22	0.24
PUERTO RICAN	950	07014	1.95	0.9	1832	97705	2.15	0.9	0.89	0.19 ¥	0.22
OTHER HISPANIC	117	17072	2.00	0.9	284	16150	2.08	0.9	0.93	0.26	0.28
	***	1/7/2	2.05	1.0	941	64261	2.11	0.9	0.92	0.06	0.07
SCHOOL TYPE:											
PUBLIC	14624	2648716	1 07								
PRIVATE	67	14540	1.67/	0.9	24087	2676511	2.08	0.9	0.87	0.11 ¥	0.13
CATHOLIC	1013	231024	2 10	0.9	861	102736	2.39	0.9	0.90	0.56 ¥	0.62
	2013	CJ1764	2.10	0.9	2652	197833	2.07	0.8	0.85	-0.02	-0.03
GEOGRAPHIC REGION:											
NORTHEAST	3530	798179	1 00			/					
NORTH CENTRAL	6692	001533	2 01	0.0	5562	682853	1.99	0.9	0.85	0.09 ¥	0.11
SOUTH	5-56	776034	1 00	0.7	7956	855008	2.10	0.9	0.87	0.09 ¥	0.10
HEST	2 6	516001	2 83	0.9	9045	899447	2.11	0.9	0.88	0.13 ¥	0.14
•	× 20	310001	2.03	0.9	5037	539773	2.16	0.9	0.89	0.12 ¥	0.14
CURRICULUH:											
GENERAL	5526	945888	1 04		1007/	1001000		_			
ACADEMIC	6735	1 377566	1 07	0.9	10076	1091282	2.10	0.9	0.86	0.05 ¥	0.06
VOCATIONAL	4051	659991	2 64	0.0	10417	1125998	2.13	0.9	0.85	0.16 ¥	0.18
	4034	V37771	6.04	0.9	6720	717250	2.16	0.9	0.93	0.12 ¥	0.13
COMUNITY TYPE:											
URBAN	4456	771620	2 01		4 701						
SUBURBAN	7820	1516714	1 05	9.7	1020	590509	2.13	0.9	0.90	0.12 ¥	0.13
RURAL	3607	627046	2 00	0.0	13362	14/550/	z.08	0.9	0.86	0.12 *	0.15
			2.00	4.7	/ 9//	ATT00 <del>0</del>	2.08	0.9	0.68	0.08 ¥	0.09

\*SIGNIFICANT AT .05 OR LESS



.

		NLS 1972				HSB 14	80				
	SAMPLE N	MEIGHTED N	MEAN	<b>S.</b> D.	SAMPLE N	HE IGHTED N	MEAN	<b>5.</b> D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	16280	2979350	1.48	0.8	27728	29 <b>90390</b>	1.62	0.9	0.87	0.15 *	0.17
SEX :											
MALE	8094	1487273	1 39	0.7	12695	1380252	1.58	0.9	0.80	0.19 *	0.24
FEMALE	8181	1491060	1.56	0.9	13912	1495369	1.66	0.9	0.93	0.09 ×	0.10
SES :											
LON	4625	714276	1.52	0.9	8203	792279	1.64	0.9	0.92	0.12 ×	0.13
MIDDLE	7775	1526616	1.59	0.8	12650	1407368	1.67	0.9	88.0	0.14 *	0.16
HIGH	3825	728836	1.40	0.7	6130	717732	1.5.	0.8	0.80	0.17 ×	0.21
RACE:											
MHITE	12664	2491624	1.46	0.8	19652	2340930	1.59	0.9	0.85	0.13 ×	0.15
BLACK	1986	240052	1.60	0.9	3617	330047	1.75	1.0	0.98	0.15 *	0.15
ASIAN-AMERICAN	191	27429	1.46	0.8	353	38222	1.55	0.8	0.78	0.09	0.12
AMERICAN INDIAN	180	29703	1.67	0.9	215	22107	1.77	1.1	0.99	0.10	0.10
MEXICAN-AMERICAN	530	69581	1.47	0.8	1853	99277	1.70	1.0	0.94	0.23 ×	0.25
PUERTO RICAN	88	9069	1.32	0.7	293	16957	1.87	1.0	0.97	0.55 ×	0.57
OTHER HISPANIC	115	17730	1.58	0.9	950	651 <b>35</b>	1.69	0.9	0.94	0.11	0.12
SCHOOL TYPE:											
PUBLIC	14599	2644201	1.47	0.8	24203	2687709	1.63	0.9	0.88	0.16 ¥	0.19
PRIVATE	67	16549	1.32	0.6	864	103684	1.40	6.7	0.70	0.08	0.12
CATHOLIC	1008	222295	1.54	0.8	2661	198997	1.58	0.8	0.83	0.05	0.06
GEOGRAPHIC REGION:											
NORTHEAST	3531	768601	1.50	0.8	5593	686895	1.64	0.9	0.89	0.14 ¥	0.16
NORTH CENTRAL	4489	902614	1.49	0.8	7976	857177	1.64	0.9	0.87	0.15 ×	0.17
SOUTH	5344	773686	1.47	0.8	9106	904839	1.61	0.9	0.88	0.15 ¥	0.16
NEST	2916	514450	1.43	0.8	5053	541480	1.58	0.9	0.85	0.15 ×	0.18
• •											
CENEDAI	5525	044044	1.41	0.8	10119	1095445	1.51	0.8	0.81	0.10 ×	0.13
ACADENTC	6715	1374682	1.40	0.7	10450	1130718	1.57	0.8	0.78	0.17 ¥	0.22
VOCATIONAL	4039	659422	1.73	1.0	6763	720727	1.86	1.1	1.06	0.13 ×	0.12
COMMINITY TYPE:											
URBAN	444N	769800	1.52	0.9	6366	595730	1.67	0.9	0.91	0.15 ¥	0.16
SUBURBAN	7817	1514582	1.46	0.8	13361	1481119	1.62	0.9	0.86	0.16 ¥	0.19
RURAL	3595	626438	1.44	0.8	8001	913542	1.58	0.9	0.86	0.14 ¥	0.16
RURAL	3595	626438	1.44	0.8	8001	913542	1.58	0.9	0.86	0.14 *	

# HOW OFTEN USED IN COURSES YOU ARE TAKING THIS YEAR: USING TEACHING MACHINES OR COMPUTER-ASSISTED INSTRUCTION (1=NEVER; 4=FREQUENTLY)

TABLE 4-29

\*SIGNIFICANT AT .05 OR LESS

.

.

79

.



The average reported frequency of writing essays, themes, poetry or stories did not change between 1972 and 1980. Variations do appear, however, across race, SES, school type and curriculum. Small negative effect sizes are found for low SES (-0.11), Black (-0.10), and vocational (-0.15) students, while small positive effect sizes are shown for high SES (0.14) and academic (0.16) students. High SES students attending Catholic schools (0.33) or enrolled in academic programs (0.26) show even greater increases.

Participation in student-centered discussions had a moderate decrease among low SES students (-0.12) and Blacks (-0.16). Change was greatest for low SES Blacks (-0.21) and low SES students in the South (-0.18).

While the decline in working on a project or in a laboratory showed a mean effect size of -0.12, it is an insignificant -0.04 for males but a significant -0.18 for females. These sex differences are consistent across curricula. A small negative effect size is found for low and middle SES students, for those enrolled in the general curriculum (-0.12), and for students living in the South (-0.14).

Students generally report having individualized instruction more frequently in 1980 than in 1972. The increase was greatest, however, for high SES students, particularly those who are Black (0.38), for students attending private schools (0.62), and for Mexican-Americans (0.22). While the use of teaching machines and computer-assisted instruction is not as widespread as other instructional methods, a small significant effect size was found. This effect differed by gender, SES, race and curriculum.

To summarize, only three of the six instructional approaches had significant changes with a small effect size. Students tended to work on projects or in a laboratory somewhat less frequently in 1980 than in 1972 and to receive individualized instruction or use teaching machines or computer-assisted instruction somewhat more frequently. The largest increases in the latter two categories were found among minority and high SES students. Women were involved less in projects and laboratory work than men in 1980. Although men gained greater access to teaching machines and computer-assisted instruction, they still used these approaches slightly less frequently than women in 1980.

### D. STUDENT EVALUATIONS OF SCHOOL FACILITIES AND EDUCATIONAL EXPERIENCES

The student questionnaire provided information about students' perception of the quality of their school and their educational program and about school-related factors that interferred with their education.

#### 1. Student Ratings of Schools

Students answered questions evaluating the condition of school facilities, quality of academic instruction, and reputation of the school



in the community. The ratings ranged from 1 = Poor to 4 = Excellent. Table 4-30 summarizes the mean ratings and change measures for five common variables in 1972 and 1980. In 1972, the ratings ranged from 2.99 (good) for reputation of the school in the community to 2.52 (between fair and good)

#### **Table 4-30**

#### Student Ratings of Schools

	Mean Re <u>1972</u>	sponse <u>1980</u>	Difference 1980-1972	Effect Size
Reputation in Community	2.99	2.86	-0.13*	-0.15
Condition of Buildings and Classrooms	2.86	2.73	-0.12*	-0.15
Quality of Academic Instruction	2.78	2.72	-0.05*	-0.07
Library Facilities Teacher Interest in Students	2.74 2.52	2.81 2.57	0.07* 0.05*	0.08 0.06

\*Significant at .05 or less

for teacher interest in students. Three of the categories--library facilities, quality of instruction, and teacher interest--show little overall change between 1972 and 1980, while two--condition of buildings and reputation of school--have a small negative effect size.

Cross tabulations by the classification variables show some variation in the effect sizes. (See Tables 4-31 through 4-35.) For example, students living in the Northeast report a moderate, negative change in school reputation in the community. The condition of school buildings and classrooms also showed a moderate decline for Other Hispanic students and schools located in the Northeast. While change in the quality of academic instruction was negligible for all students, a small significant decline is found among female students in general (-0.16) and vocational (-0.17) curricula, low SES Whites (-0.15), and low SES students living in suburban (-0.15) communities, in rural (-0.19) communities, and in the South (-0.21). A small significant increase in teacher interest in students is reported by those who are high SES, are enrolled in academic programs, or attend Catholic schools.

#### 2. Student Evaluation of Program

The second set of questions asked students whether their high schools provided them with adequate programs. Table 4-36 summarizes their responses to the following statements:



#### STUDENT RATING OF SCHOOL: REPUTATION IN CONTUNITY (1=POOR; 4=EXCELLENT)

---

	NLS 1972			HSB 1	980						
	SAMPLE	WEIGHTED			SAMPLE	LIF TOHTED				1000 10	
	N	N	MEAN	S.D.	N	N	MEAN	S.D.	S.D.	1980-1972 DIFFERENCE	EFFECT SIZE
TOTAL	15352	2811646	2.99	0.9	26251	2835188	2.86	0.9	0.90	-0.13 *	-0 15
SEX:											V.15
MALE	7596	1396787	2 07	• •	10175						
FEMALE	7751	1413842	3.01	0.9	13330	1321000	2.86	0.9	0.91	-0.11 *	-0.12
SFC :				•••		1430634	2.00	0.9	0.89	-0.15 *	-0.17
	4315	667781	2.86	0.9	7587	733707	2.68	0.9	0.90	-017 #	-0.10
MICH NICH	7365	1447688	2.99	0.9	12130	1347262	2.86	0.9	0.90	-0 13 #	-0.19
NTON	3619	687274	3.12	0.9	5969	698497	3.05	0.9	0.88	-0.06 <b>*</b>	-0.07
RACE:									_		••••
MHITE	11994	2350262	7 67	• •	1						
BLACK	1853	221044	9 70	0.9	18921	2249808	2.90	0.9	0.88	-0.13 *	-0.15
ASIAN-AMERICAN	170	24594	2.70	1.0	3281	298481	2.68	1.0	0.95	-0.02	-0.02
AMERICAN INDIAN	166	29100	C./J 9 77	1.0	332	36579	2.87	0.9	0.94	0.14	0.14
MEXICAN-AMERICAN	495	2010 <del>7</del>	2.1/	0.9	180	18650	2.63	0.9	0.93	-0.14	~0.15
PUERTO RICAN	472 A1	4102	2.03	0.9	1718	92791	2.70	0.9	0.93	-0.14	-0.15
OTHER HISPANIC	110	17161	2.40	1.0	275	15715	2.67	1.0	0.97	0.27	0.28
		1/101	3.00	0.9	898	60427	2.84	0.9	0.92	-0.16	-0.17
SCHOOL TYPE:											
PUBLIC	13736	2488779	2.96	0.9	22820	2542218					
PRIVATE	62	15198	3.34	0.7	847	100205	2.00	0.9	0.90	-0.16 *	-0.17
CATHOLIC	990	227436	3.37	0.8	2575	100275	3.21	0.8	0.82	-0.12	-0.15
				•••	2373	172035	3.41	0./	0.74	0.04	0.06
GEOGRAPHIC REGION:											
NORTHEAST	3308	740393	3.04	0.9	5319	455590	2 81				
NORTH CENTRAL	4240	853750	2.93	0.9	7610	A1 A790	2 87	0.9	0.91	-0.24 *	-0.26
SOUTH	5098	738302	3.06	0.9	8546	848092	2 80	0.7	0.09	-0.06	-0.07
WEST	2706	479201	2.91	0.9	4776	512716	2 85	0.7	0.00	-0.17 *	-0.19
						510/10	2.05	0.7	0.91	-0.06	-0.07
GENERAL	5169	883986	2.88	0.9	9460	1021316	2.74	0.9	n ag	-0.14 ×	
	6394	1307092	3.10	0.9	10154	1098377	3.05	0.9	0.07	-0.14 *	-0.16
TUCALIUNAL	3788	620265	2.91	0.9	6273	674802	2.73	0.9	0.90	-0.18 *	-V.06 -0 20
COMMUNITY TYPE:											
URBAN	<b>61 86</b>	726816	2 00								
SUBURBAN	7386	1430404	2 84	0.9	5959	561356	2.78	1.0	0.95	-0.12 *	-0.13
RURAL	3612	E0E1A7	J. VO 2 OE	0.9	12714	1407914	2.92	0.9	0.89	-0.14 *	-0.15
	3416	375107	C · 73	V.0	/5/8	865918	2.80	J.9	0.86	-0.15 *	-0.17

\*SIGNIFICANT AT .05 OR LESS



#### STUPENT RATING OF SCHOOL: CONDITION OF BUILDING (1=POOR; 4=EXCELLENT)

NLS 1972 HSB 1980 \*--\*-\*-SAMPLE WEIGHTED SAMPLE WEIGHTED POOLED 1980-197 EFFECT N N MEAN S.D. N N MEAN S.D. S.D. DIFFERENCE SIZE TOTAL 16511 3014567 2.86 0.9 27468 - 62384 2.73 0.9 0.86 -0.12 # -0.15 SEX: MALE 8188 1502015 2.89 0.9 12703 1379840 2.77 0.9 0.87 -0.11 \* -0.13 FEMALE 8318 1511535 2.83 8.0 13947 1099238 2.71 0.8 0.84 -0.13 \* -0.15 SES: LON 4741 729265 2.74 0.9 8129 784932 £.62 0.9 0.86 -0.11 \* -0.13 MICOLE 7859 1541683 2.86 0.9 12586 1399480 2.74 5.9 0.86 -0.12 \* -0.14HIGH 3848 733111 2.98 0.8 6109 715245 2.88 8.0 0.82 -0.10 \* -0.13 RACE: NHITE 12766 2511438 2.90 0.9 19574 2330799 2.78 0.8 -0.12 \* 0.84 ~0.14 BLACK 2072 249769 2.55 0.9 3576 325870 2.50 0.9 0.87 -0.06 -0.07 ASIAN-AMERICAN 190 27039 2.65 0.7 353 38467 2.76 0.3 0.78 0.10 0.13 AMERICAN INDIAN 20961 185 30621 2.82 0.9 204 2.55 0.9 0.92 -G.27 -0.30 MEXICAN-AMERICAN 550 72509 2.74 0.9 1827 97273 2.61 -0.12 0.8 0.86 -0.14 PUERTO RICAN 92 9200 2.54 0.9 295 17101 2.52 0.8 0.82 -0.03 -0.03 OTHER LISPANIC 122 18844 3.01 0.8 938 63734 2.71 0.9 0.85 -0.30 # -0.36 SCHOOL TYPE: PUBLIC 19797 2674234 2.85 0.9 23954 2662393 2.72 0.9 0.86 -0.13 \* -0.15 PRIVATE 67 16549 3.04 0.9 -0.21 865 103530 2.83 0.8 0.77 -0.27 CATHDLIC 1022 234931 3.02 0.8 2649 196460 2.87 6.8 0.79 -0.15 \* -0.19 GEOGRAPHIC REGION: NORTHEAST 3582 797143 2.91 0.8 5572 684237 2.69 0.8 -0.21 \* 9.84 -0.35 NORTH CENTRAL 4533 910429 2.85 0.9 7920 851412 2.77 0.8 0.86 -0.09 ¥ -0.10 SOUTH 5437 785716 2.83 2.75 0.9 0.9 8976 890967 0.87 -0.08 \* -0.09 HEST 2959 521289 2.84 0.8 5000 535768 2.71 0.9 C.84 -0.13 \* ~0.16 CURRICULUM: 9991 1080186 GENERAL 5603 957822 2.78 0.9 2.66 0.9 0.86 -0.12 \* -0.14 2.93 ACADEMIC 6768 1382246 0.9 -0.09 # 10402 1125119 2.84 0.8 0.84 -0.11 VOCATIONAL 4139 674197 2.82 0.9 6679 714060 2.68 -0.15 \* 0.9 0.85 -0.17 COMMUNITY TYPE: URBAN 4517 780706 2.79 6.9 6270 588373 2.63 0.8 0.85 -0.17 # -0.19 SUBURBAH 7903 1528727 2.93 0.8 13257 1466637 2.79 0.8 0.84 -0.14 \* -0.16 RURAL 3644 633765 2.79 0.9 7941 907373 2.72 -0.07 ¥ 0.9 0.88 -0.08

\*SIGNIFICANT AT .05 OR LESS

5---



83

#### STUDENT RATING OF SCHOOL: ACADEMIC INSTRUCTION (1=POOR; 4=EXCELLENT)

		NLS 1	NLS 1972			HSB 1	980				
	SAMPLE N	MEIGHTED N	MEAN	<b>S.</b> D.	SAMPLE N	WEIGHTED N	MEAN	s.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	14937	2749599	2.78	0.8	25781	2790945	2.72	0.8	0.78	-0.05 ¥	-0.07
SEX:											
MALE	7550	1394345	2.76	0.8	12122	1310515	2 7E			- 0 - 02	
FEMALE	7382	1354238	2.79	C.8	12935	1396757	2.71	0.8	0.76	-0.02 +	-0.02
SES:											
LON	4000	617536	2 69	<b>A</b> A	7970	784554		• •			
MIDDLE	7187	1418905	9 79	0.0	11085	/04334	2.37	0.8	0.79	-0.10 *	-0.13
HIGH	3697	703976	2 84	0.0 A A	E 00E	1333633	2.72	0.8	0.//	-0.06 *	-0.08
			2.04	•.•	3703	100401	C.00	0.0	0.78	0.03	0.04
RACE:											
MHITE	11766	2321659	2.80	0.8	18612	2217775	2.74	0.8	0.77	-0.05 #	-0.07
BLACK	1714	205858	2.67	0.8	3226	293223	2.64	0.8	0.82	-0.03	-0.07
ASIAN-AMERICAN	175	25073	2.79	0.8	339	36801	2.79	0.8	0.78	-0.01	-0.03
AMERICAN INDIAN	162	27388	2.67	0.7	188	19420	2.61	0.9	0.81	-0.06	-0.01
MEXICAN-AMERICAN	473	62255	2.69	0.8	1652	88998	2.63	0.8	0.80	-0.00	-0.07
PUERTO RICAN	73	7323	2.55	0.8	267	15596	2.59	0.8	0.83	-0.00	-0.06
OTHER HISPANIC	104	16251	2.80	0.9	874	58416	2.67	9.8	0.82	-0.13	-0.15
SCHOOL TYPE:											
PUBLTC	13332	2626366	2 74		22777	04 04 787		• •			
PRIVATE	65	16012	2.70 9 QE	0.0	620	2470/03	2.09	0.8	0.77	-0.07 *	-0.09
CATHOLIC	980	226737	3 01	0.0		101907	3.13	0.8	0.82	0.18	0.22
	/00		3.41	V.0	6374	196393	2.99	U.8	0.80	-0.02	-0.02
GEOGRAPHIC REGION:											
NORTHEAST	3238	731445	2.85	0.8	5246	648292	2.80	0.8	0.77	-0.05	-0.04
NORTH CENTRAL	4153	836726	2.75	0.8	7485	806660	2.70	0.8	0.77	-0.05	-0.00
SOUTH	4842	702701	2.75	0.8	8354	827940	2.68	0.8	0.80	-0.07 #	-0.00
HEST	2704	478728	2.75	0.8	46 96	508053	2.73	0.8	`.78	-0.02	-0.02
CURRICULUM:											
GENERAL	4883	834600	2.64		0200	000250	0 EE	• •	a		
ACADEMIC	6580	1345409	9 RG	0.0	7607	770237	2.35	0.3	0.77	-0.09 *	-0.12
VOCATIONAL	3473	568888	2.70	0.8	10214	444761	2.94	0.8	0.75	0.05 *	0.06
		200000	2000			940/01	6.03	v.o	V./O	-0.0/ *	-6.03
CONTUNITY TYPE:											
URBAN	4080	709043	2.80	0.8	5794	540930	2.75	0 A	0 79	-0.05	-0.07
SUBURBAN	7237	1411969	2.80	0.8	12577	1393350	2.77	0.8	0.78	-0.03	-0.0/
RURAL	3262	570747	2.70	0.8	7410	851665	2.63	0.8	0.78	-0.03	-0.04
			-					<b></b>			~v.vo

\*SIGNIFICANT AT .05 OR LESS



#### STUDENT RATING OF SCHOOL: LIBRARY (1=POOR; 4=EXCELLENT)

	NLS 1972					HSB 19	980				
	SAMPLE	MEIGHTED			SAMPLE	NEIGHTED			POOLED	1980-1972	EFFECT
	N	N	MEAN	S.D.	N	N	MEAN	S.D.	5.0.	DIFFERENCE	SIZE
TOTAL	16249	2975642	2.74	0.9	27162	293123 <b>9</b>	2.81	0.9	0.86	0.07 *	0.08
SEX:											
MALE	8070	1464003	2.75	0.9	12573	1365851	2.82	0.9	0.87	0.07 ¥	0.08
FEMALE	8174	1490623	2.72	0.8	13788	1483232	2.80	0.8	0.85	0.08 *	0.09
SES:											
LOW	4634	714744	2.79		8028	776977	2 83		0 85	0 04	A AF
MTDDI F	7731	1691111	2 71	A Q	19661	1794000	2.03	0.7	0.05	0.04	0.05
HIGH	3824	729557	2.71	0.9	6060	710499	2.80	0.9	0.66	0.00 *	0.10 0.10
51CE •											
	10417	04 04 F 7/			107/0	07040//		• •	• •-		
	12013	2404536	2 / 3	0.9	19362	2304966	2.81	0.9	0.85	0.08 *	0.10
ARTAN AMERICAN	1994	241271	2.75	0.8	3529	322085	2.81	0.9	0.87	0.06	0.07
ASTAN-AREKILAN	190	2/291	2./1	0.9	351	38334	2.76	0.8	0.87	0.05	0.06
ARERICAN INDIAN	180	29863	2.76	0.9	204	21005	2.69	0.9	0.89	-0.07	-0.08
MEXICAN-AMERICAN	539	70481	2.92	0.9	1809	96661	2.79	0.8	0.85	-0.12	-0.15
PUERTO RICAN	91	9314	Z.82	0.9	289	16845	2.84	0.8	0.85	0.02	0.03
OTHER MISPANIC	117	18156	2.97	0.8	929	63176	2.80	0.9	0.87	-0.17	-0.19
SCHOOL TYPE:											
PUBLIC	14560	2638684	2.75	0.9	23674	2633000	2.83	0.9	0.85	0.08 ¥	0 10
PRIVATE	67	16549	2.85	0.9	860	102852	2.57	1.0	0.96	-0.28	-0.29
CATHOLIC	1015	233666	2.55	0.8	2628	195388	2.63	0.9	0.86	0.08	0.09
GEOGRAPHIC REGION:											
NORTHFAST	3527	788512	2 73		5475	474168	9 78		0 94	0 04	
NOPTH CENTRAL	4455	804157	2 72	0.4	7877	863781	2 84	0.7	0.00	0.00 A 10 ×	0.07
SOLITH	5157	774 888	5.7 <u>6</u> 9 74	0.0	7033	801770	6.04	0.0	V.05	0.12 *	0.14
LEST	9357 901 A	770000 E14004	6./0 9 7E	0.7	0071	001/30	2.01	0.9	0.00	0.05	0.06
MEDI	2710	214000	2./3	0.9	4703	5315/2	2.80	0.8	0.85	0.05	0.06
CURRICULUM:											
GENERAL	5529	946807	2.76	0.9	9873	1067174	2.79	0.9	0.86	0.03	0.04
ACADEMIC	6719	1375090	2.69	0.9	10347	1119592	2.79	0.9	0 87	L 10 H	0.19
VOCATIONAL	4000	653443	2.82	0.8	6555	702222	2.68	0.8	0.84	0.06 *	6.07
COPELNITY TYPE:											
LIPRAN	6620	769061	2 79		4180	589448	2 81		6 AE	0 04	
SIRIDRAN		1508047	9 74		1 2004	306440	2.0J 9 44	U.7	V.03	U.U4	0.05
	7771	1300003	5./4 9 4 8	0.7	13070	T44 3/ 35	6.04	U.O	U.05	0.10 #	U.12
RURAL	2012	067073	6.00	U.Y	/0/5	87899 <b>9</b>	2.75	0.9	U.55	U.07 🗮	0.08

\*SIGNIFICANT AT .05 OR LESS



#### STUDENT RATING OF SCHOOL: TEACHER INTEREST IN STUDENTS (1=POOR; 4=EXCELLENT)

	NLS 1972				HSB 19	80					
	SAMPLE N	HEIGHTED N	MEAN	S.D.	SAMPLE N	WEIGHTED N	MEAN	\$.D.	<b>POOLED</b> S.D.	1980-1972 Diffe <b>rence</b>	EFFECT SIZE
TOTAL	16209	2966051	2.52	0.9	27021	2917070	2.57	0.9	0.88	0.05 ¥	0.06
SEX:											
MALE	7993	1469767	2 51	• •	12640	1 36 71 70	9 E0	• •		<b>A A A Y</b>	
FENALE	8211	1495267	2.54	0.9	13765	1479368	2.59	0.9	0.87	0.08 *	0.10
efe •								•••	••••		0.03
	46.37	716112	2.52	0.9	7959	770254	2.49	0.9	0.87	-0.03	-0.03
HIDULE	//30	1518674	2.51	0.9	12390	1377113	2.55	0.9	0.88	0.04	0.05
U TOU	3784	721141	2.55	0.9	6053	708836	2.73	0.9	0.86	0.18 *	0.20
RACE:											
MHITE	12581	2476375	2.53	0.9	19322	2300553	9 69	<b>n</b> q		0 04 ×	A 47
BLACK	2007	242323	2.47	0.9	3475	316935	2 40	0.7	0.89	0.00 *	0.07
ASIAN-AMERICAN	186	26769	2.64	0.9	353	38316	2.66	0.9	0.87	0.01	0.02
AMERICAN INDIAN	181	29840	2.31	1.0	196	20224	2.47	<b>n</b> 9	0.07	0.01	0.01
MEXICAN-AMERICAN	523	69423	2.49	0.9	1795	96541	2.54	n.9	0.75 0.96	0.16	0.17
PUERTO RICAN	92	9450	2.51	1.0	288	16640	2.66	n 9	n 02	0.00	0.07
OTHER HISPANIC	115	17719	2.61	0.8	918	61550	2.58	0.9	0.88	-0.04	-0.04
SCHOOL TYPE:											
PUBLIC	14521	2620162	2 40		07630	0410744		• •			
PRIVATE	67	14640	2.47 9 0E	0.7	23339	2019/00	2.52	0.9	0.8/	0.03	0.03
CATHOLIC	1016	233400	2.73	0.0	001	102922	3.15	0.9	0.87	0.20	0.23
	1014	233477	2.02	0.7	2021	194382	2.94	0.8	0.86	0.12 #	0.14
GEOGRAPHIC REGION:											
NORTHEAST	3521	785166	2.53	0.9	5475	674136	2.60	0.9	0.87	0.06 *	0.07
NORTH CENTRAL	444 3	893979	2.49	0.9	7798	838666	2.53	0.9	0.87	0.05	9.05
SOUTH	5348	774015	2.54	0.9	8827	876689	2.56	0.9	0.89	0.02	0.02
NEST	2897	512891	2.55	0.9	4921	527579	2.63	0.9	0.88	0.09 ¥	0.10
CURRICULUM:											
GENERAL	6497	941622	2 66		0787	1458702	0 47	• •			
ACADEMIC	6683	1346799	2 41	0.7	7/0/	1120/02	2.43	0.9	0.8/	-0.01	-0.01
VOCATIONAL	4028	658327	2.45	0.9	6536	793023	2.47	0.0	0.85	U.17 # 0.02	0.20
CANAGE ANTTY TYPE							••				
100AA	A498			• -				_	_		
	44 50 777 /	/00196	2.54	0.7	6168	579997	2.57	0.9	0.88	0.03	0.03
JUDUKDAN	//64	1504/51	2.52	0.9	13054	1445125	2.60	0.9	0.87	0.08 *	0.09
RURAL	3589	029367	2.52	0.9	7799	891947	2.53	0.9	0.88	0.01	0.01

\*SIGNIFICANT AT .05 OR LESS



-34	-
-----	---

```
Table 4-36
```

	Mean R 1972	esponse 1980	Difference 1980-1972	Effect Size
My high school should have placed more emphasis on academic subjects.	2.52	2.13	-0.39*	-0.46
My high school did not offer enough practical work experience.	2.13	2.17	0.04*	0.04
My high school should have placed more emphasis on vocational and technical programs.	2.06	1.98	-0.08*	-0.09
My high school provided me with counseling that will help me continue my education.	2.68	2.78	+0.10*	+0.10
My high school provided me with counseling that will help me find employment.	2.16	2.43	+0.27	+0.26

\*Significant at .05 or less

The responses ranged from 1 to 4 with 1 showing negative assessment of the high school (agreement with the first three questions and disagreement with the last two) and 4 showing a positive assessment of the high school (disagreement with the first three questions and agreement with the last two). In 1972, students were generally neutral about the scope of the basic academic program and of counseling for further education. They were slightly negative regarding technical education, and counseling to find employment. By 1980, students' assessments of the basic academic program had dropped substantially, but they had become slightly more satisfied with their school's counseling services, especially with regard to finding work.

Cross-tabulations show that growing dissatisfaction with schools' emphasis on basic academic subjects was shared by students of all SES and racial/ethnic groups, and by those enrolled in all three types of schools and three kinds of curriculum. High SES students and those enrolled in Catholic schools reported the largest increase in ratings for education-oriented counseling services, while maler, high SES students, and those enrolled in academic programs had the largest positive change for work-oriented counseling. In the latter case, however, the mean rating for these groups still remained substantially below ratings given by low SES and vocational education students in 1980, primarily because of the lower interest in post high school employment among academic students. (See Tables 4-37 through 4-41.)



		NLS 1	972			H5B 1	980				
	SAMPLE	NEIGHTED			SAMPLE	WETCHTER				1980-1972	FFF
	N	N	MEAN	S.D.	N	N	MEAN	S.D.	S.D.	DIFFERENCE	SIZE
TOTAL	14563	2651072	2.52	0.9	256 76	2772524	2.13	0.8	0.84	-0.39 ¥	-0.47
SEX:											
MALE	7217	1319495	2.53	0.9	11689	1273699	2.14	0.8	0.85	-0.38 ×	-0.45
FEMALE	7343	1331030	2.51	0.9	13056	1404758	2.12	0.8	0.84	-0.39 *	-0.47
SES:											
LOW	4252	651012	2.38	0.9	7655	739636	2.10	0.8	0.83	-0 28 *	-0 34
MIDDLE	6932	1357913	2.55	0 8	11793	1311942	2 14	0.0	0.05	-0 41 #	-0.49
HIGH	3328	633107	2.60	0.9	5593	659241	2.14	0.9	0.87	-0.46 *	-0.55
PACE:											
MHITE	11220	2200988	2 5A	0.8	18220	2175560	2 18	0.8	ñ 91	-0 40 ¥	-0.48
BLACK	1869	226743	2.13	0.0	1124	303419	1 01	0.0	0.85	-0.77 #	-0.40
ASTAN-AMERICAN	177	25406	2 20	0.0	349	27704	1 72	0.0	0.07	_0 E7 #	-0.20
AMERICAN INDIAN	166	27285	2.32	n 9	193	19626	2 00	0.7	0.77	-0.37 #	-0.74
MEXICAN-AMERICAN	483	63102	2.22	0.5	1737	92082	2 00	0.0	0.04	-0.31 ×	-0.37
PUERTO RICAN	84	8715	2 02	0.8	280	16412	1 02	0.0	0.01	-0.21 #	-0.20
OTHER HISPANIC	104	16107	2.30	0.8	873	60303	2.07	0.8	0.85	-0.23	-0.13
SCHOOL TYPE:											
PUBLIC	13134	2369596	2 51	n o	22613	2504742	. 2 .00	<b>A A</b>	0 97	-0 41 *	-0 50
PRIVATE	59	14728	2.83	1 9	715	2500702	2.07	1.0	0.83	-0.41 *	-0.30
CATHOLIC	811	186337	2.72	0.8	2348	176743	2.42	0.9	0.91	-0.30 *	-0.33
SEACHADHIC DESIGN:											
NODTHEAST	3051	679733	2 ( 0	<b>^</b> •	E1 72	497094	2 22		0.04	-0 47 *	0 EF
NORTH CENTRAL	6030	A10119	2 64	0.0	7102	901044	2 15	0.7	0.00	-0.47 *	-0.35
SOUTH	4816	602716	2 16	n a	9421	919074	2 07	0.0	0.03	-0.40 ×	-0.40
WEST	2657	468505	2.47	0.9	4721	504559	2.07	0.8	0.83	-0.39 *	-0.48
C100TC18194:											
CENEDAL	5021	SEEG44	2 60	• •	0414	1010267	0 10		0 02		
ACADEMIC	5021	1192008	2 51	0.7	9410 9410	1017207	2.10	0.0	0.02	-0.38 #	-0.47
VOCATIONAL	3702	602798	2 53	0.9	6307	671559	2.10	0.9	0.84	-0.42 * -0.36 *	-0.49
COMMINITY TYPE:											
LIPRAN	2045	687137	2 50	A 9	2447	E	9 64		0 04	- 0 44 ×	
SUBAN	3703 4028	1276120	2.3V 9 87	V.7 A G	19761	340330 1779981	6.04	V.O	V.04	-U.40 #	-0.55
DIDAI	07CV 1974	1337137	6.7/ 9 AA	V.7	16341 7454	13/6601	6.10	V.0	V.85	-0.41 *	-0.48
RUMAL	32/0	300531	C.99	V.0	/453	<b>Q</b> 21706	Z.14	<b>U.</b> 8	<b>V.8</b> 2	-0.31 *	-0.38

.

.

#### HIGH SCHOOL SHOULD HAVE PLACED MORE EMPHASIS ON BASIC ACADEMIC SUBJECTS (1=AGREE STRONGLY; 4=DISAGREE STRONGLY)

**TABLE 4-37** 

\*SIGNIFICANT AT .05 OR LESS



83

#### HIGH SCHOOL DID NOT OFFER ENOUGH PRACTICAL NORK EXPERIENCE (1=AGREE STRONGLY; 4=LISAGREE STRONGLY)

**TAILE 4-38** 

	NL\$ 1972			HSB 1980							
	SAMPLE N	NEIGHTED N	MEAN	S.D.	SAMPLE N	NEIGHTEO N	MEAN	S.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	14242	2589751	2.13	0.9	24999	2686542	2.17	1.0	0.95	0.04 #	0.84
SEX:					-						
MALE	6967	1275800	2.09	0.9	11303	1224892	2 14	1 0	A 94	0 04	
FEMALE	7271	1313186	2.17	1.0	12752	1366671	2.21	1.0	0.96	0.04	0.04
SES:											
LON	4280	657932	2.14	1.0	7643	777667	2 80	1.0	0.04	-0.05	
MIDDLE	6809	1330381	2.10	n. 9	11445	1271496	2 17	1.0	U.70 8 05		-0.03
HIGH	3107	593439	2.19	0.9	5269	615327	2.29	1.0	0.93	0.10 ¥	0.11
BACF:											
MAITE	10952	2149796	2.13		17575	2804288	2 10	1.0	8 04	a at x	
BLACK	1840	220657	2.12	1.0	17575	184239	2.17	1.0	1.00	U.UO *	0.07
ASIAN-AMERICAN	153	21530	2.12	<b>A A</b>	3350	38561	2 74	1.0	1.00	-0.00	-4.08
AMERICAN INDIAN	164	27659	2.21	8.9	197	10771	1 00	0.7	0.00	U.CC -0.27	V.23
MEXICAN-AMERICAN	485	63619	2 27	<b>n</b> .o	1715	98984	2 10	1.0	0.70	-0.23	-0.25
PUERTO RICAN	79	8044	2.20	6.9	276	15032	9 14	1 1	0.70	-0.09	-0.09
OTHER HISPANIC	99	15203	2.28	1.0	867	59958	2.25	1.0	1.00	-0.04	-0.04
SCHOOL TYPE:											
PUBLIC	12754	2296583	2.14	<b>A</b> Q	21812	2414381	2 10	1.0	0 OF	0.05 *	
PRIVATE	60	14413	1 01	0.7	711	6410301	2.17	1.0	U. 75		0.00
CATHOLIC	872	198996	1.94	0.9	2436	180584	1.93	0.9	0.85	-0.01	-0.01
GEOGRAPHIC REGION:											
NORTHEAST	2986	661156	2.08		4088	408404	2 87		0.04	-0 02	
NORTH CENTRAL	3975	798184	2.08	A. 9	71 92	776276	2 20	<b>N</b> Q	0.74	-0.02	-v.uz
SOUTH	6670	670085	2 18	<b>n</b> .9	\$2 <b>7</b> 7	#14397	2 16	1.0	0.73	U.12 #	0.13
HEST	2611	460328	2.21	1.0	4592	489448	2.29	1.0	0.97	0.08 ×	0.08
CLERR TOULUM:											
GENERAL	4947	843835	2 85	0.9	9269	1881324	2 07				
ACADEMIC	5444	1120364	2.15	0.9	0828	073 201	2.07	1.0	0.75	U.U3 A 12 M	
VOCATIONAL	3831	625553	2.22	1.0	6332	673693	2.17	1.0	1.00	-0.05	-0.05
CONTINITY TYPE:											
URBAN	<b>1858</b>	663451	2.20	1.0	5729	63676A	2 24	1.			
SUBURBAN	6726	1299785	2.14	0.9	11045	1320607	2 9 9 1	1.0	V.70 A GE	U.U4 A A7 #	U. U.
RURAL	3290	567371	2.04	8.9	7285	2320077	5.CL 2 87	1.4	V. 73	0.0/ #	U.UD
				<b>**7</b>	/ 203	034407	C.V/	¥.7	¥.73	V.VC	w.w.#

\*SIGNIFICANT AT .05 OR LESS

Full Text Pro by ERIC

2

		NLS 1	972			H58 19	780				
	SAMPLE	MEIGHTED			SAMPLE	NEIGHTED			POOLED	1980-1972	EFFEC1
	N	N	MEAN	5.D.	N	N	MEAN	5.D.	\$.D.	DIFFERENCE	SIZE
TOTAL	14281	2592072	2.06	0.9	25480	2740625	1.98	0.8	0.86	-0.08 *	-0.09
SEX:											
MALE	7112	1296705	2.07	0.9	11684	1268196	1.96	<b>n</b> a	0.45	-0 17 #	-0 13
FENALE	7166	1294819	2.05	0.9	12858	1378556	2.01	0.8	0.84	-0.04	-0.44
SES :											
LON	4254	652942	1.96	8.9	7718	744807	1 80				-0.10
MIDDLE	6838	1332242	2.02	0.9	11676	1297176	1 96	0.0	0.84	-0.10 *	-0.17
HIGH	3146	599238	2.25	0.9	5439	635735	2.25	0.9	0.89	-0.00	-0.00
RACE:											
MITE	11016	2154293	2.07	0.9	17966	2137957	2.03	0.9	0.86	-0 05 #	-8.64
BLACK	1829	220766	2.00	8.9	3380	307776	1 78	0.4	0.84	-0.05 #	
ASIAN-AHERICAN	160	22700	1.96	0.8	340	37101	2.00	0.6	0 78	-V.CJ ~	
AMERICAN INDIAN	163	26889	1.96	0.9	198	20241	1 75	0.0	0.88	-0 21	_8 24
MEXICAN-AMERICAN	486	63639	2.01	0.8	1747	92905	1.86	0.8	0.61	-7.21	-0.27
PUERTO RICAN	73	7271	1.94	0.9	270	15842	1 44	0.0	0.01	-0.15 -	-0.17
OTHER HISPANIC	102	15982	1.93	0.9	882	61360	1.88	0.8	0.83	-0.05	-0.06
SCHOOL TYPE:											
PUBLIC	12823	2306243	2.06	0.9	22333	2469113	1 97		0.86	-0 00 #	-0.10
PRIVATE	55	12760	2.00	0.9	736	90512	2 87	n a	0.60	0.07	-0.10
CATHOLIC	857	194573	2.07	0.9	2411	180999	2.05	0.9	0.90	-0.02	-0.02
GEOGRAPHIC REGION:											
NORTHEAST	2888	642744	2.08	0.9	5011	611781	2.04	0.9	0.89	-0.04	-0.04
NORTH CENTRAL	4010	806350	2.02	0.9	7318	788035	2.01	0.6	0.85	-0.00	-0.00
SOUTH	4755	679101	2.13	0.9	8418	834627	1.94	0.9	0.87	-0.18 #	-0.21
MEST	2628	463877	2.00	0.9	- 4733	505981	1.92	0.8	0.81	-0.09 *	-0.11
CURRICULUM:											
GENERAL	i976	851407	1.96	0.8	9430	1018126	1.89	0.8	0.81	-0.07 #	-8 69
ACADEMIC	5491	1119587	2.24	0.9	9242	997488	2.25	0.9	0.85	0.01	0.01
VOCATIONAL	3813	620776	1.87	0.9	6434	684724	1.73	0.8	0.82	-0.13 #	-0.16
CONTRINITY TYPE:											
URBAN	386 3	667492	2.09	0.9	5843	546286	1.94	0.A	0.87	-0.15 #	-0 18
SUBURBAN	6757	1298259	2.08	0.9	12234	1 349622	2.02	0.9	0.87	-0.05 #	-0.10
DIDA1	7710	348804	1 00		7401						-0.00

### HIGH SCHOOL SHOULD HAVE PLACED MORE EMPHASIS ON VOCATIONAL AND TECHNICAL PROGRAMS (1=AGREE STRONGLY; 4=DISAGREE STRONGLY)

#SIGNIFICANT AT .05 OR LESS

.



#### NLS 1972 HSB 1980 SAMPLE WEIGHTED SAMPLE WEIGHTED POOLED 1980-1972 EFFECT N N MEAN S.D. N N MEAN S.D. S.D. DIFFERENCE SIZE TOTAL 14832 2709555 2.68 1.0 26194 2816867 2.78 1.0 0.99 0.10 # 0.18 SFY: MALE 7344 1346509 2.67 1.0 11904 1290375 2.79 7344 1346509 2.67 1.0 11904 1290375 2.79 1.0 0.98 0.12 # 7484 1362280 2.69 1.0 13332 1431068 2.77 1.0 1.01 0.08 # 0.12 FEMALE 0.08 SES: 4281 654250 2.80 0.9 7778 747023 2.87 1.0 0.96 0.07 \* 0.87 7003 1379016 2.68 1.0 11915 1323347 2.75 1.0 0.99 0.06 \* 0.07 3495 666931 2.57 1.0 5861 685426 2.76 <sup>1</sup>.0 1.02 0.18 \* 0.18 LOH MIDDLE HIGH PACE: 18482 2199288 2.73 1.0 0.99 0.08 \* 3492 318064 3.04 1.0 0.98 0.08 349 37721 2.87 1.0 0.93 0.08 200 20362 2.78 1.0 1.00 0.08 1771 94549 2.88 1.0 0.97 0.01 279 16145 2.83 1.0 1.04 0.07 905 62376 2.90 1.0 0.96 0.10 11389 2248457 LEITE 2.65 1.0 1945 235376 9.08 BLACK 2.96 0.9 0.08 171 163 507 2.79 0.9 349 2.70 1.0 200 2.88 0.9 1771 2.76 1.0 279 2.80 0.9 905 ASIAN-AMERICAN 24189 0.08 AMERICAN INDIAN 27329 0.08 MEXICAN-AMERICAN 66576 0.01 PUERTO RICAN 87 8898 2.76 0.06 109 OTHER HISPANIC 16585 2.80 0.18 SCHOOL TYPE: PUBLIC 13285 2401271 1.02283725310420.9806952511.02551190574 2.77 1.0 0.99 0.09 \* 2.99 1.0 1.02 0.17 2.84 1.0 1.01 0.24 \* 2.69 0.09 PRIVATE 6**1** 15111 2.82 2.16 CATHOLIC 911 211000 **2.**60 190574 0.24 GEOGRAPHIC PEGION: NORTHEAST 3184 713869 2.62 1.052636450050.975168073551.08594849958 2.75 1.0 1.02 2.71 1.0 0.98 2.89 1.0 0.98 2.75 1.0 1.00 0.13 \* 0.13 NORTH CENTRAL 4070 817119 2.66 0.05 0.05 SG. TH 4926 710955 2.80 0.08 \* 0.09 HCST. 2652 467613 2.63 1.0 4821 514549 0.12 # 0.12 CLRR ICULLIN: GENERAL 4920 838805 2.59 0.9 9419 1015026 2.69 1.0 0.99 0.10 # 0.10 62A7 1285162 2.73 1.0 10127 1095671 2.86 1.0 1.00 0.13 # 0.13 7424 Energine 72 1.0 4277 444112 2.70 1.0 0.07 0.03 # 0.13 4920 ACADEHIC VOCATIONAL 6277 666112 2.79 1.0 0.97 0.08 # 0.08 COMMUNITY TYPE: URBAN 40787065242.701.060325636242.791.01.01710513778652.651.01264313975492.761.01.0032665629742.740.975198556942.811.00.98 0.09 <del>\*</del> 0.11 <del>\*</del> 0.06 0.09 SUBURBAN 0.11 RURAL

HIGH SCHOOL PROVIDED HE WITH COUNSELING THAT WILL HELP HE CONTINUE HY EDUCATION (1=DISAGREE STRONGLY; 4=AGREE STRONGLY) 

**\*SIGNIFICANT AT .05 OR LESS** 

91

0.07



#### HIGH SCHOOL PROVIDED ME WITH COUNSELING THAT HILL HELP ME FIND EMPLOYMENT (1=DISAGREE STRONGLY; 4=AGREE TRONGLY)

		NLS 19	72			HSB 19	980				
	SAMPLE	WEIGHTED N	MEAN	S.D.	SAMPLE N	WEIGHTED N	MEAN	S.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	13547	2466444	2.16	1.0	24229	2598298	2.43	1.0	1.03	0.27 *	0.26
SEX:									-		
MALE	67 <b>0</b> 1	1228967	2.10	1.0	10889	1178007	2.43	1.0	1.01	9.33 *	0.33
FEMALE	6842	12 <b>36</b> 711	2.21	1.0	12418	1326860	2.40	1.0	1.04	0.19 *	0.19
SES:											
1.0M	4169	639252	2.46	1.0	7468	717622	2.64	1.0	1.02	0.18 *	0.17
MIDDLE	6449	1268634	2.14	1.0	11109	1233822	2.39	1.0	1.02	0.26 *	0.25
HIGH	2882	550204	1.84	0.9	5028	585892	2.21	1.0	0.97	0.37 *	0.38
RACE											
MITE	10278	2026572	2.10	1.0	16865	2003282	2.36	1.0	1.01	0.26 #	0.26
BLACK	1868	226320	2.56	1.0	3383	309714	2.71	1.1	1.08	0.15 *	0.14
ASIAN-AMERICAN	159	22971	2.01	0.9	306	33988	2.49	1.0	0.97	0.48 *	0.50
AMERICAN INDIAN	151	25247	2.30	1.1	193	19447	2.54	1.1	1.07	0.24	0.2 <b>2</b>
MEXICAN-AMERICAN	463	60867	2.42	0.9	1695	91709	2.62	1.0	1.00	0.19 *	0.19
PUERTO RICAN	81	8163	2.31	1.1	268	15643	2.53	1.0	1.05	0.22	0.21
OTHER HISPANIC	101	15263	2.36	1.0	845	58981	2.62	1.0	0.98	0.26	0.27
SCHOOL TYPE:											
PUBLIC	12149	2192937	2.15	1.0	21243	2346032	2.45	1.0	1.02	0.30 #	0.29
PRIVATE	49	11383	1.33	0.8	675	82425	2.19	1.0	1.00	0.35	0.35
CATHOLIC	806	184845	2.04	1.0	2311	169841	2.20	1.0	1.00	0.17 *	0.17
GEOGRAPHIC REGION:											
NORTHEAST	2834	628598	2.06	1.0	4831	585642	2.33	1.1	1.04	0.27 *	0.26
NORTH CENTRAL	3746	755770	2.12	1.0	6944	746746	2.38	1.0	1.00	0.25 *	0.25
SOUTH	4537	652489	2.35	1.0	7988	790473	2.57	1.0	1.04	0.23 *	0.22
HEST	2430	42 9586	2.06	1.0	4466	475438	2.38	1.0	1.00	0.32 *	Q. 32
CURRICULUM:											
GENERAL	4690	804946	2.13	1.0	8923	961542	2.38	1.0	1.01	0.26 *	0.25
ACADEMIC	5213	1067245	1.95	0.9	8756	940574	2.26	1.0	0.98	0.32 *	0.32
VOCATIONAL	3643	593950	2.57	1.0	6180	656 360	2.71	1.0	1.04	9.14 *	0.14
COPPLNITY TYPE:											
URBAN	3727	644511	2.17	1.0	5658	527550	2.51	1.1	1.05	0.33 *	0.32
SUBURBAN	6382	1232775	2.09	1.0	11545	127 <b>368</b> 1	2.35	1.0	1.02	0.26 #	0.26
RURAL	3087	532065	2.28	1.0	7026	<b>7970</b> 67	2.49	1.0	1.01	0.21 #	0.21

**#SIGNIFICANT AT .05 OR LESS** 



#### 3. Factors Interfering with Education

The final set of questions asked students how much of each four factors had interferred with their education. Using a rating scale of 1 = A Great Deal to 3 = N ot At All, they reported the following:

#### Table 4-42

Response 1980	Difference 1980-1972	Effect <u>Size</u>
2.64	-0.06*	-0.10
2.48	-0.08*	-0.15
2.2 <b>9</b>	-0.11*	-0.17
2.12	-0.18*	-0.28
	Response 1980 2.64 2.48 2.29 2.12	Response      Difference        1980      1980-1972        2.64      -0.06*        2.48      -0.08*        2.29      -0.11*        2.12      -0.18*

\*Significant at .05 or less

In 1972, the responses tended to fall between Somewhat and Not At All, with poor study habits causing the most interference. All four factors became more of a problem between 1972 and 1980, particularly poor study habits. High SES Blacks (-.41) and students attending Catholic schools (-0.34) showed moderate negative effect sizes for difficult courses. Academic (-0.09) and Black students (-0.07) reported the smallest increase in poor teaching as a cause of educational problems. Poor study habits interfered with students' education more in 1980 than in 1972, but particularly so for women (-0.35) and high SES Blacks (-.43) and students attending Catholic schools. (See Tables 4-43 through 4-46.)

In summary, student evaluations of school facilities and of their educational experiences generally became more negative between 1972 and 1980. The variables that showed a small negative change were the condition of school buildings and classrooms, the reputation of the school in the community, and teaching and course difficulty as factors interfering with students' education. Study habits and adequate emphasis on basic academic subjects had moderate negative effect sizes. The factors which showed improvement were adequacy of high school counseling for (1) future educational plans and (2) for future occupational plans.



	SAMPLE N	NEIGHTED N	MEAN	\$.D.	SAMPLE N	WEIGHTED N	HEAN	\$.0.	POOLED \$.D.	1980-1972 Difference	EFFECT SIZE
OTAL	16395	2999461	2.70	0.6	27429	2958855	2.64	0.6	0.58	-0.06 *	-0.10
SEX:											
MALE	8131	1493920	2.65	0.6	12683	1379363	2.59	0.6	0.61	-0.06 *	-0.10
FEMALE	8260	1504776	2.75	0.5	13915	1495280	2.70	0.6	0.55	-0.05 *	-0.09
SES:											
LOW	6691	724090	2.69	0.6	8150	786518	2.61	0.6	0.59	-0.08 ×	-0.13
MIDDLE	7810	1534093	2.71	0.6	12562	1397555	2.66	0.6	0.57	-0.05 *	-0.08
HIGH	3838	731438	2.70	0.6	6083	712718	2.67	0.6	0.58	-0.04	-0.06
ACF:											
LATTE	12718	2502673	2.71	0.6	19545	2327420	2.65	0.6	0.57	-0.06 *	-0.11
BLACK	2020	244782	2.62	0.6	3566	324911	2.64	0.6	0.60	0.02	0.04
ASTAN-AMERTCAN	192	27656	2.73	0.5	356	38404	2.68	0.5	0.54	-0.05	-0.09
AMEDICAN THITAN	193	30799	2.67	0.6	206	20962	2.46	0.7	0.63	-0.22	-0.34
MEXICAN-AMERICAN	535	70078	2.75	0.5	1829	98079	2.63	0.6	0.56	-0.12 *	-0.21
PUEDTO PTCAN	222	9422	2.67	0.5	293	17111	2.61	2.6	0.57	-0.06	-0.11
OTHER HISPANIC	117	17912	2.67	0.6	932	63380	2.57	0.6	0.62	-0.10	-0.16
SCHOOL TYPE:											
PUBLIC	14697	2661722	2.70	0.6	23929	2659016	2.64	0.6	0.58	-0.06 ¥	-0.09
PRIVATE	67	16549	2.74	0.6	860	103441	2.62	0.6	0.61	-0.12	~0.20
CATHOLIC	1019	234059	2.72	0.5	2640	196398	2.68	0.5	0.54	-0.04	-0.08
EOGRAPHIC REGION:											
NORTHEAST	3565	796392	2.68	0.6	5552	682668	2.65	0.6	0.59	-0.03	-0.04
NORTH CENTRAL	4512	206 379	2.69	0.6	7893	848459	2.64	0.6	0.59	-0.05 *	-0.09
SOUTH	5377	778053	2.72	0.6	8981	892084	2.66	0.6	0.56	-0.06 *	-0.11
WEST	2941	518637	2.71	0.6	5003	535644	2.61	0.6	0.59	-0.10 *	-0.17
URPICULUM:											
GENERAL	5550	950522	2.64	0.6	9972	1078816	2.59	0.6	0.62	-0.35 ¥	-0.09
ACADEMIC	6769	1380198	2.75	0.5	10368	1122030	2.72	0.5	0.53	-0.03	-0.05
VOCATIONAL	4095	668439	2.68	0.6	6690	714627	2.61	0.6	0.59	-0.07 *	-0.12
CONTUNITY TYPE:											
URBAN	4485	776295	2.71	0.6	6257	586688	2.65	0.6	0.57	-0.06 ¥	-0.10
SUBURBAN	7844	1520768	2.69	0.6	13239	1466726	2.63	0.6	0.59	-0.06 ×	-0.11
RURAL	3634	633262	2.73	0.5	7933	905442	2.67	0.6	0.55	-0.06 ×	-0.12

# HOW MUCH HAS THE FOLLOWING INTERFERED WITH YOUR EDUCATION: FIND IT HARD TO ADJUST TO SCHOOL ROUTINE (1=A GREAT DEAL; 3=NOT AT ALL)

\*SIGNIFICANT AT .05 OR LESS



91

#### TABLE 4-43

		NLS 1	972			HSB 1	980				
	SAMPLE	NEIGHTED			SAMPLE	WEIGHTED	* -		POOLED	1980-1972	FEFFOT
	N	N	MEAN	\$.D.	N	N	MEAN	S.D.	S.D.	DIFFERENCE	SIZE
TOTAL	16502	3015571	2.56	0.5	27472	2963821	2.48	0.6	0.55	-0 08 *	-0.15
SEX:									•	-0.00 *	-0.13
MALE	010/	1 5000 70									
FEMALE	0104	1902239	2.54	0.5	12697	1380744	2.47	0.6	0.56	-0.07 <b>*</b>	-0.13
	0314	1512566	2.57	0.5	13945	1498719	2.48	0.5	0.54	-0.09 #	-0.16
SES:											
LON	6762	720402	9 69								
MIDDLE	7859	1649463	C.JC 0 F4	0.5	8172	789459	2.43	0.6	0.50	-0.09 *	-0.16
HIGH	7050	1346403	2.54	0.5	12582	1399375	2.48	0.6	0.55	-0.06 #	-0.10
	3047	/33231	2.03	0.5	6091	713572	2.52	0.5	0.53	-0.11 *	-0.20
RACE :											
MHITE	12775	251 3404	2 E7	0 E	10540						
BLACK	2052	247842	2.3/	0.5	19569	2330352	2.48	0.6	0.55	-0.08 *	-0.15
ASIAN-AMERICAN	192	27454	C.36 0 EE	0.0	3578	326182	2.48	0.6	0.56	-0.04	-0.07
AMERICAN INDIAN	186	21034	2.33	0.5	355	38265	2.39	0.5	0.54	-0.16	-0.29
MEXICAN-AMERICAN	EAE	31066	6.4/	0.0	204	20723	2.47	0.6	0.56	-0.01	-0.02
PUERTO PTCAN	045	/15/0	2.50	0.5	1837	98766	2.39	0.6	0.56	-0.12 *	-0.21
OTHER SPANTC	. 110	75/1	2.52	0.5	291	17117	2.41	0.6	0.60	-0.11	-0.18
	117	10345	2.45	0.6	938	63862	2.44	0.5	0.54	-0.01	-0.02
SCHOOL TYPE:											
PUBLIC	14791	247557 <b>8</b>	9 EE	<b>A F</b>							
PRIVATE	67	14540	2.33	0.5	23968	2663773	2.48	0.6	0.55	-0.07 <b>*</b>	-0.13
CATHOLIC	1025	235324	C./U 9 E0	0.5	002	103193	2.51	0.6	0.55	-0.19	-0.34
	1063	233324	6.39	0.5	2642	196854	2.40	0.6	0.55	-0.19 *	-0.34
BEOGRAPHIC REGION:											
NORTHEAST	3586	700570	9 E 7	0 E	5550						
NORTH CENTRAL	4530	000850	6.33 9 EE	0.5	2227	683774	2.47	0.6	9.55	-0.07 ¥	-0.13
SOUTH	5420	707030 784715	C.JJ 9 E4	0.5	7913	851502	2.46	0.6	0.55	-0.08 *	-0.15
WEST	2957	F21407	6.30	0.5	8995	892835	2.48	0.6	0.56	-0.78 #	-0.14
	2757	361407	2.00	0.5	5005	535710	2.49	0.5	0.54	-0.10 *	-0.19
CURRICULUM:											
GENERAL	5595	957814	9 E1		0007						
ACADEMIC	6773	1384429	9 40	0.0	9997	1001/9/	2.43	0.6	0.56	-0.08 *	-0.14
VOCATIONAL	4133	477496	6.0V 9 E9	0.5	103/6	1122726	2.54	0.5	0.53	-0.06 *	-9.12
	4233	073025	6.96	0.0	00 40	715529	2.44	0.6	0.56	-0.08 *	-0.14
COMPLINITY TYPE:											
URBAN	4508	778865	2 64	0 E	4047						
SUBURBAN	7902	1520001	0 EA	V.J A E	0207	567987	2.50	0.6	0.55	-0.06 *	-0 11
RURAL	2455	4150771	2.30 9 E4	V.J	13255	1968065	2.46	0.6	0.55	-0.09 *	-0.17
	2022	033747	c. 30	0.5	7950	907768	z.48	0.6	0.55	-0.08 ¥	-0.14

### HOW MUCH HAS THE FOLLOWING INTERFERED NITH YOUR EDUCATION: COURSES ARE TOO HARD (1=A GREAT DEAL; 3=NOT AT ALL)

\*SIGNIFICANT AT .05 OR LESS



95

#### TABLE 4-44

#### HOW MUCH HAS THE FOLLOWING INTERFERED WITH YOUR EDUCATION: POOR TEACHING (1=A GREAT DEAL; 3=NOT AT ALL)

H58 1980 NLS 1972 POOLED 1980-1972 SAMPLE WEIGHTED SAMPLE WEIGHTED EFFECT N N MEAN S.D. N N MEAN S.D. S.D. DIFFERENCE SIZE 16354 2991378 2.40 0.7 27380 2953544 2.29 0.7 0.66 -0.11 # -9.17 TOTAL SEX: 8103 1489272 2.39 0.7 12656 1375883 2.28 0.7 0.67 -0.11 \* -0.16 8247 1501341 2.40 0.7 13906 1495052 2.29 0.6 0.64 -0.11 \* -0.18 MALE FEMALE SES: 4682 722237 2.47 0.7 8118 784108 2.36 0.7 0.65 -0.11 \* -0.16 7788 1529764 2.40 0.7 12549 1395527 2.27 0.7 0.66 -0.13 \* -0.20 3826 729303 2.33 0.7 6082 712341 2.25 0.6 0.65 -0.08 \* -0.12 LOH MIDDLE HIGH RACE : 

 12694
 2496925
 2.40
 0.7
 19534
 2326021
 2.27
 0.6
 0.65
 -0.13
 \*

 2014
 244339
 2.45
 0.7
 3551
 323462
 2.40
 0.7
 0.66
 -0.05

 MERICAN
 188
 27022
 2.32
 0.7
 356
 38462
 2.21
 0.7
 0.66
 -0.10

 N INDIAN
 184
 30776
 2.35
 0.7
 201
 20545
 2.29
 0.7
 0.71
 -0.06

 -AMERICAN
 535
 70230
 2.46
 0.7
 1827
 97915
 2.38
 0.6
 0.63
 -0.08

 RICAN
 93
 9503
 2.37
 0.7
 287
 16483
 2.36
 0.7
 0.66
 -0.09

 ISPANIC
 1.8
 18111
 2.43
 0.7
 935
 63543
 2.35
 0.7
 0.66
 -0.09

-0.19 MHITE -0.07 BLACK -0.15 ASIAN-AMERICAN -0.09 AMERICAN INDIAN MEXICAN-AMERICAN PUERTO RICAN -0.12 -0.01 -0.13 OTHER HISPANIC SCHOOL TYPE: 14652 2653569 2.39 0.7 23881 2653821 2.28 0.7 0.66 -0.11 \* 67 16549 2.54 0.6 861 103222 2.42 0.6 0.62 -0.12 1016 233260 2.41 0.7 2638 196501 2.33 0.6 0.65 -0.08 -0.17 Rəlic -0.19 PRIVATE 67 10349 2.41 0.7 1016 233260 2.41 0.7 CATHOLIC -0.12 GEOGRAPHIC REGION: 

 NORTHEAST
 3555
 794004
 2.38
 0.6
 5539
 680914
 2.24
 0.6
 0.64
 -0.14 \*

 NORTHEAST
 3555
 794004
 2.38
 0.6
 5539
 680914
 2.24
 0.6
 0.64
 -0.14 \*

 NORTH CENTRAL
 4498
 903913
 2.40
 0.7
 7893
 849714
 2.27
 0.6
 0.65
 -0.13 \*

 SOUTH
 5372
 776911
 2.41
 0.7
 8960
 889146
 2.33
 0.7
 0.67
 -0.08 \*

 NEST
 2929
 516551
 2.39
 0.7
 4988
 533770
 2.30
 0.7
 0.66
 -0.09 \*

-0.21 -0.20 -0.08 # -0.12 -0.09 \* -0.14 CURRICULUM: 5528 946465 2.41 6740 1378058 2.36 4685 666552 2.45 
 9964
 1078429
 2.26
 0.7
 0.67
 -0.15 \*
 -0.22

 10360
 1119537
 2.30
 0.6
 0.64
 -0.06 \*
 -0.09

 6661
 712216
 2.31
 0.7
 0.67
 -0.15 \*
 -0.22
 0.7 GENERAL 0.7 ACADEMIC 0.7 VOCATIONAL CONTINITY TYPE: 4476 774040 2.39 0.7 6236 584847 2.31 0.6 0.65 -0.08 \* 7828 1517219 2.38 0.7 13227 1465326 2.27 0.7 0.66 -0.12 \* 3627 631456 2.44 0.7 7917 903370 2.31 0.7 0.65 -0.14 \* -9.12 (IRBAN -0.18 SUBURBAN -0.21 RURAL

**#SIGNIFICANT AT .05 OR LESS** 



		NLS 1972			HSB 1980						
	SAMPLE	WEIGHTED			SAMDI F					1014 1070	
	N	N	MEAN	S.D.	N	N	MEAN	S.D.	S.D.	DIFFERENCE	EFFECI
TOTAL	16398	2997534	2.30	0.7	27417	2956796	2.12	0.7	0.67	-0.18 *	-0 28
SEX:											-V.LU
MALE	<b>\$1</b> 21	1601171		• -							
FEMALE	9272	14711/1	2.19	0.7	12666	1376636	2.04	0.7	0.67	-0.15 ¥	-0.22
	02/3	194997/	2.42	0.6	13926	1496954	2.19	0.7	0.65	-0.23 *	-0.35
SES:											
LON	A705	725472	2 22	0 7	<b>A</b> 1 70						
MIDDLE	7802	1621646	2 20	0.7	0134	/8481/	2.16	0.7	0.66	-0.16 *	-0.24
HIGH	1010	770002	2.30	0.7	12503	1397610	2.11	0.7	0.67	-0.19 *	-0.29
	3035	/30772	2.29	U.7	6083	712694	2.10	0.7	0.68	-0.20 *	-0.29
RACE :											
WHITE	12710	2400942	2 20		10541						
BLACK	2031	245482	2 74	0.7	17341	232/143	2.11	0.7	0.67	-0.20 *	-0.29
ASIAN-AMERICAN	189	97110	2 20	0.0	3504	324188	2.20	0.7	0.66	-0.16 *	-0.25
AMERICAN INDIAN	182	20492	9 90	0.7	355	38304	2.15	0.6	0.68	-0.06	-0.08
MEXICAN-AMERICAN	53A	70840	2.27	0.6	204	20722	2.14	0.7	0.70	-0.15	-0.21
PUERTO RICAN	950	70007	2.27	0.0	1828	97694	2.16	0.6	0.64	-0.13 *	-0.20
OTHER HTSPANTC	119	70/0	2.29	0.7	290	16647	2.27	0.7	0.67	-0.02	-0.03
	110	10224	2.35	0.7	935	63713	2.19	0.7	0.67	-0.15	-0.22
SCHOOL TYPE:											
PUBLIC	14700	2659887	2.30	07	2701 8	94E74E1			<b>•</b> • -		
PRIVATE	67	16549	2 32	0.4	23710	203/031	2.12	0.7	0.67	-0.18 ¥	-0.27
CATHOLIC	1018	233636	2 22	0.0	050	102554	2.16	0.7	0.66	-0.16	-0.24
	2010	233030	C.JJ	V./	2041	140241	2.10	0.7	9.67	-0.23 *	-0.34
GEOGRAPHIC REGION:											
NORTHEAST	3561	795129	2.31	07	5550	441667	2 14				
NORTH CENTRAL	4504	904508	2.29	0.7	79.05	001333	2 10	0.7	U.67	-0.17 *	-0.25
SOUTH	5397	780558	2 33	0 7	8071	0-7037	2.10	0.7	0.67	-0.20 *	-0.29
WEST	2936	517339	2 28	0.7	5001	070177	2.15	0.7	0.67	-0.18 *	-0.27
		527557		V./	2001	222201	2.08	0.7	0.67	-0.20 <del>*</del>	-0.29
CURRICULUM:											
SENERAL	5554	950808	2.22	0.7	9979	1070644					
ACADEMIC	6745	1378124	2.35	0.7	10749	1120407	2.04	6.7	0.68	-0.18 *	-0.26
VOCATIONAL	4098	668300	2.34	0.7	10302	712704	2.17	0.7	0.66	-0.18 *	-0.26
					0070	110100	c.10	V./	0.00	-0.18 *	-0.27
CONTUNITY TYPE:											
URBAN	4485	774994	2.31	0.7	6264	587084	2 14	07			
SUBURBAN	7847	1520250	2.29	0.7	1 2 2 2 4	1444545	2 00	U./	U.00	-0.15 *	-0.23
RURAL	36 38	632990	2.32	0 7	701 4	1400343	C.V7 2 17	0.7	0.67	-0.20 *	-0.30
				w.,	/714	403103	c.13	0./	ጣ.67	-0.19 ¥	-0.28

### HOW MUCH HAS THE FDLLOWING INTERFERED WI H YOUR EDUCATION: POOR STUDY HABITS (1=A GREAT DEAL; 3=NOT AT ALL)

TABLE 4-46

\*SIGNIFICANT AT .05 OR LESS



#### -36-

#### CHAPTER V

#### CHANGES IN TESTED ACHIEVEMENT AND SCHOOL GRADES

This chapter deals with changes in measured achievement of high school seniors from 1972 to 1980. The two achievement areas that will be discussed here are tested achievement and school achievement as reflected in high school grades. These two measures of achievement are particularly important since in theory they should provide corroborative evidence for either stability or change in educational outcomes over this eight-year period. If changes in test scores parallel changes in grades, both in direction and size, then one can feel relatively assured that there is at least a consistency between an objective external standard (test scores) and a more subjective internal standard (grades). Conversely, if the test scores and grades change in opposing directions, one suspects that there has been a change in school standards. This interpretation, of course, assumes that the test scores are either identical or properly equated across administrations.

The tables presented in this chapter show comparisons of 1972 and 1980 means by total population and by the standard subpopulation classification variables. Differences between the 1980 and 1972 test score means are shown in the formula-corrected number-right true score metric and by effect size. The formula-corrected number-right true scores are on the same scale as the formula-corrected raw scores and thus can be interpreted in the same way.

A. COMPARISONS OF 1972 AND 1980 SENIOR VOCABULARY TEST SCORES

Inspection of Table 5-1 indicates there was a decline in Vocabulary scores from 1972 to 1980 of .85 of a test score point or 22 percent of a standard deviation. The typical senior in 1980 (a student at the 50th percentile) would rank at about the 41st percentile among the 1972 seniors in Vocabulary. A closer look at the subpopulations indicates the groups that contributed disproportionally to the observed decline. Females declined more than males. Whites declined more than Blacks. The greater decline for women was of sufficient magnitude to reverse what was, in 1972, a measured superiority for females compared to males on Vocabulary to a slight inferiority in 1980. The considerably greater decline of Whites when compared to Blacks reduced the disparity between Whites and Blacks in IRT scale score units, but the two groups are still almost a standard deviation apart.

The decline is relatively consistent across SES levels with a slight increase in decline as one goes up the SES ladder. Part of this increasing decline with higher levels of SES may be due to the possibility of floor effects at the lower SES levels.

The decline is consistent across geographic areas, but there appears to be some differential rate of decline when comparisons are made by



#### Table 5-1

#### IRT VOCABULARY FORMULA SCOPE (SCALED TO NLS VOCABULARY TEST)

	NLS 1972				HSB 1980						
	SAMPLE	WEIGHTED N	MEAN	s.D.	SAMPLE N	WEIGHTED N	MEAN	S.D.	POOLED S.D.	1980-1972 Difference	EFFEC SIZE
TOTAL	15696	2860438	6.55	4.0	24936	2666481	5.70	3.7	3.79	-0.85 *	-0.22
SEX:											
MALE	7804	1425843	6.44	4.0	11374	1218450	5.90	3.7	3.82	-0.54 ¥	-0.14
FEMALE	7887	1433577	6.67	4.0	12657	1355969	5.69	3.6	3.76	-0.98 *	-0.26
SES:											
LOW	4491	693845	4.59	3.6	7369	707354	4.05	3.1	3 32	-0 53 *	-0 14
MIDDLE	7486	1460802	6.52	3.8	11474	1269007	5 77	3 4	3.50	-0.75 *	-0.10
HIGH	3663	695762	8.63	3.8	5457	627386	7.71	3.6	3.70	-0.93 *	-0.25
RACE:											
WHITE	12174	2383015	7 68	3.0	1 786 2	2110778	6 24	3.6	3 71	_0 84 ¥	_0 23
BLACK	1927	234726	<b>1</b> 2A	3.0	3173	285008	1 20	2.0	2 00	-0.04 -	~0.23
ASTAN-AMERICAN	182	25667	6 72	4.2	31/3	33794	5.20	10	2.70 4 0E	-0.08	-0.03
AMERICAN INDIAN	181	20787	6 M6	1 6	200	20116	5.07 6 19	3.7	4.05	-0.05	-0.21
MEXTCAN-AMERTCAN	E14	49774	1 47	2.2	1420	07074	7.10	3.1	3.33	0.14	0.04
PLIEDTO DICAN	94	8174	3.4/	2.7	244	14210	3.30	3.0	2.75	0.03	0.01
OTHER HISDANIC	112	17322	3.00	3.0	654	14610	3.36	3.0	3.03	-0.20	-0.09
OTTER HISPANIC	116	1/366	4.04	3.4	030	20430	3.70	2.9	5.00	-0.87	-0.29
SCHOOL TYPE:											
PUBLIC	14097	2540625	6.44	4.0	21677	2399504	5.52	3.6	3.76	-0.92 ¥	-0.24
PRIVATE	66	16235	7.88	3.8	736	84461	7.80	4.0	4.03	-0.08	-0.02
CATHOLIC	990	224332	8.24	3.7	2523	182516	7.06	3.5	3.56	-1.17 *	-0.33
GEOGRAPHIC REGION:											
NOPTHEAST	3493	777868	7.33	4.0	5013	609829	6.48	3.8	3 88	-0.86 ¥	-0 22
NORTH CENTRAL	4089	831402	6.58	3.9	7301	785432	5.85	3.5	3.66	-0.74 *	-0.20
SOUTH	5334	773440	5.63	3.9	8208	806215	4.84	3.5	3.68	-0.80 *	-0.22
WEST	2780	477727	6.71	4.0	4414	465005	5.94	3.6	3.79	-0.77 *	-0.20
CURRICULUM:											
GENERAL	5305	900620	5 32	36	9070	077554	6 A1	12	3 35	_0 60 ×	-0.15
ACADEMIC	6681	1322785	A 20	1.0	946.8	1000106	7 4 2	3.2	3.33	-0.47 *	-0.15
VOCATIONAL	3909	636730	4.70	3.4	6042	642443	4.15	3.0	3.14	-0.55 *	-0.18
COMMENTEL TYPE:											
LIPBAN	6002	702623	6 28	2 0	6624	E13004	E 20	37	7 70	-1 07 *	
SUBIRBAN	7642	1480061	7 11	4.0	11014	1101500	4 14	3.1	3./7	-1.0/ *	-0.28
DIDAI	2524	411217	5 75	1.0	7400	1303307	C.17	3./	3.01	-0.9/ #	-0.20
	3330	OTTET/	3.13	3.0	/478	047000	3.36	3.3	3.02	-0.42 #	-0.12

\*SIGNIFICANT AT .05 OR LESS



community type. The students living in rural areas were characterized by a somewhat smaller decline than those students classified as either urban or suburban. It may be that rural areas were more stable both in respect to population shifts and schooling practices during the 1972-1980 period.

The pervasiveness of the decline is emphasized by the fact that it is consistent across the three curriculum areas. There are declines for students from both public and Catholic schools. The magnitude of the decline is somewhat greater for students in Catholic schools. There is not sufficient data in 1972 on the private, non-Catholic schools to draw any conclusion about the stability of Vocabulary scores for students in these schools.

When sex is cross-classified with curriculum (see Tables 5-1.1 to 5-1.6 in Appendix C for the cross-classification tables), the female Vocabulary score decline is relatively consistent across the acad mic and vocational curricula with only a slightly lower rate of decline for the general curriculum student. When comparing Black and White declines while controlling for SES, one notes that there is a larger White decline in standard deviation units at all SES levels. The smaller decline for Blacks may be partly artifactual in the sense that the majority of Blacks are in the low SES category (especially in 1972), and there may be a test "floor effect" operating in their favor. That is, the low SES Black mean in 1972 is only 2.76 items correct, leaving little room for decline while the low SES White students had a mean of 5.52 (in 1972), indicating that they may have somewhat less of a floor effect. Only when one gets to the high SES Blacks are the scores sufficiently high to negate the possibility of a floor effect, and here the sample size is too small to draw any reliable conclusions.

When students are cross-classified by SES and school type, the sample sizes are too small to make any comparison that includes private schools. What is clear, however, is that middle SES students in both public and Catholic schools are showing declines in Vocabulary scores.

In summary, there is a decline in Vocabulary test scores between 1972 and 1980. Females tend to show greater declines than males. Whites show greater declines than Blacks, but this comparison is partially confounded with test floor effects. The decline is relatively pervasive and cuts across SES levels, geographical regions, and curriculum type. Two possible interactions were found between score decline and subpopulation classifications. There was a somewhat smaller decline for students in rural areas and for students from public schools as compared to their counterparts attending Catholic schools.

B. CHANGES IN READING TEST SCORES 1972-1980

Table 5-2 indicates an overall decline in Reading test scores of 1.05 test score points or 21 percent of a pooled standard deviation. Again, the typical senior in 1980 would rank at about the 41st percentile among the 1972 seniors in Reading. Although the overall decline (i.e.,



#### Table 5-2

#### IRT READING FORMULA SCORE (SCALED TO NLS READING TEST)

-----

	NL\$ 1972			HSB 1980							
	SAMPLE N	WEIGHTED N	MEAN	\$.D.	SAMPLE N	WEIGHTED N	MEAN	\$.D.	POOLED S.D.	1983–1972 DIFFERENCE	EFFEC SIZE
TOTAL	15713	286 3482	9.89	5.0	24864	26589 <b>58</b>	8.84	5.1	5.09	-1.05 *	-0.21
SEX:											
MALE	7811	1427414	9.83	5.0	11352	1215335	8.95	5.3	5.16	-0.88 *	-0.17
FEMALE	7897	1435051	9.95	5.0	12614	1352068	8.96	5.0	5.00	-0.99 *	-0.20
SES:											
LON	4503	695440	7.65	5.0	7350	705500	6.73	4.7	4.83	-0.92 *	-0.19
HIDDLE	7489	1461882	9.92	4.8	11438	1264873	9.05	4.9	4.87	-0.86 *	-0.18
HIGH	3664	695829	12.13	4.5	5448	626850	11.20	4.8	4.71	-0.94 *	-0.20
RACE											
NHITE	12180	2384253	10.56	4.8	17818	2105217	9.60	5.0	4 89	-0.96 *	-0 20
BLACK	1935	235572	5.94	4.5	3157	283823	5.56	4.5	4.48	-0.38	-0.09
ASIAN-AMERICAN	182	25667	10.14	5.2	320	33695	9.11	5.0	5.12	-1.03	-0.20
AMERICAN INDIAN	181	29787	6.51	4.9	199	20103	7.11	4.8	4.86	0.60	0.12
MEXICAN-AMERICAN	517	68498	6.28	4.6	1620	83914	5.60	4.4	4.43	-0.69	-0.15
PUERTO RICAN	84	8376	6.11	4.7	261	14047	5.68	4.3	4.42	-0.43	-0.10
OTHER HISPANIC	112	17498	6.68	4.8	853	57027	5.73	4.7	4.70	-0.95	-0.20
SCHOOL TYPE:											
PUBLIC	14114	2543636	9, 78	5.0	21618	2391769	8.66	5.1	5.09	-1.12 <b>*</b>	-0.22
PRIVATE	66	16235	11.41	5.1	734	84636	11.31	5.2	5.16	-0.09	-0.02
CATHOLIC	989	224161	11.61	4.4	2512	182553	10.06	4.7	4.62	-1.55 *	-0.33
GEOGRAPHIC REGION:											
NORTHEAST	3493	777802	10.55	4.9	4974	605302	9.57	5.0	4.97	-0.98 *	-0.20
NORTH CENTRAL	4100	833505	9.97	4.9	7292	783970	9.21	5.0	4.96	-0.76 *	-0.15
SOUTH	5338	773939	9.14	5.2	8199	806228	7.83	5.2	5.20	-1.31 *	-0.25
WEST	2782	478236	9.88	5.1	4399	463458	9.01	5.1	5.07	-0.87 *	-0.17
CURRICULUM:											
GENERAL	5310	901464	8.48	4.8	9041	974753	7.71	4.8	4.79	-0.78 ¥	-0.16
ACADEMIC	6486	1323908	11.99	4.4	9439	1006476	11.33	4.7	4.62	-0.66 *	-0.14
VOCATIONAL	3916	637808	7.51	4.7	6035	641041	6.81	4.7	4.67	-U.70 *	-0.15
COMMUNITY TYPE:											
URBAN	4099	703504	9.46	5.0	5511	511844	8.21	5.2	5.10	-1.25 ¥	-0 24
SUBURBAN	7649	1481264	10.49	4.9	11863	1298837	9.29	5.1	5.00	-1.20 #	-0.24
RURAL	3541	612151	9.27	5.1	7490	848277	8.52	5.2	5.14	-0.74 *	-0.14
						= = = / •					

\*SIGNIFICANT AT .05 OR LESS



for the total population) is of the same magnitude as that on the Vocabulary test in standard deviation terms, it is even more pervasive or consistent across subpopulations. The male and female declines are reasonably equivalent as are the declines by SES level, geographic areas, and curriculum types. As in the case of Vocabulary, the magnitude of the decline is somewhat smaller for students in rural areas. Blacks also show less of a decline than do the White students. Mexican-Americans show declines only slightly less than the Whites. The sample sizes are too small to draw any conclusions about the magnitudes of decline for the remaining ethnic groups.

It is interesting to note that there is a somewhat greater decline in Reading test scores for students in Catholic schools than for students in public schools. This decline was also present for Vocabulary test scores.

In summary, Reading test scores showed a decline of the same magnitude as the Vocabulary scores. The decline in Reading scores tended to be somewhat more consistent across subpopulations than the Vocabulary scores. That is, declines were consistent across SES, curriculum type, sex groups, and geographic regions. Catholic school students tended to show greater declines than public school students. Students in fural areas showed slightly less declines in tested Reading than did students from urban and suburban areas. Blacks showed smaller declines than Whites. These three interactions, which were also present in the Vocabulary test analysis, were not tested for significance.

C. CHANGES IN MATHEMATICS TEST SCORES 1972-1980

Although Mathematics test scores showed a significant decline (see Table 5-3) for the total population (14 percent of the pooled standard deviation), the decline was less than that found for the Vocabulary and Reading tests. A senior with average Mathematics achievement in 1980 would be at the 45th percentile when compared with the 1972 seniors. The Mathematics test score decline is consistent across sex groups, SES groups, public and Catholic school students, curriculum types, and community types. What is particularly interesting is the finding of a considerably greater decline in Mathematics scores for students in the South. As in the case of Vocabulary and Reading, White students show greater declines than do Blacks and Mexican-Americans. Once again the estimated greater decline for Whites may be somewhat exaggerated due to the greater likelihood of test score floor effects working in favor of the Blacks.

In summary, there is a small but significant decline in Mathematics scores. There is, however, a proportionately greater decline for students attending school in the South. Otherwise the declines tend to generalize across the remaining subpopulations with the exception of Blacks and Mexican-Americans where there are small but not significant increases.



#### Table 5-3

#### IRT MATHEMATICS FORMULA SCORE (SCALED TO NLS MATHEMATICS TEST)

SAMPLE MEIGNTED      SAMPLE MEIGNTED      POOLED      1980-1972      EFFECT        TOTAL      15705      2662252      12.94      7.3      11321      121364      1.03      7.0      -0.14        FENALE      7007      146603      12.90      7.10      1.101      900120      8.44      6.6      6.9      -0.14        HITE		NLS 1972			HSB 1980							
TOTAL      15705      2662252      12.94      7.3      24758      2650446      11.90      7.7      7.24      -1.03      -0.14        SEX:      MALE      7807      1426314      13.79      7.3      11321      1213609      12.83      7.4      7.34      -0.96      *      -0.13        SES:		SAMPLE N	WEIGHTED N	MEAN	S.D.	SAMPLE N	WE IGHTED N	MEAN	S.D.	POOLED S.D.	198 <b>0-1972</b> Dífference	EFFECT SIZE
SEX:    7807    1426314    13.79    7.3    11321    1213609    12.83    7.4    7.34    -0.96    *    -0.10      SES:    LOM    4493    694282    9.39    7.1    7303    701703    8.44    6.6    6.75    -0.95    *    -0.10      SES:    LOM    4493    694282    9.39    7.1    7303    701703    8.44    6.6    6.75    -0.95    *    -0.14      HIDDLE    7490    1641083    12.00    7.0    11410    1263636    12.16    6.8    6.90    -0.74    *    -0.12      RACE:    MHITE    12179    2304219    13.95    6.9    17756    2099496    12.98    6.9    -0.74    *    -0.12      RACE:    MHITE    12179    230421    13.95    6.9    17756    2099496    12.98    6.9    6.90    -0.98    *    -0.12      RACE:    MHITE    12179    230421    15.96    6.7    317    33116    15.97    7.3    15.97    6.36    6.19	TOTAL	15705	2862252	12.94	7.3	24758	2650446	11.90	7.2	7.24	-1.03 <b>*</b>	-0.14
MALE FEMALE      7807      1426314      13.79      7.3      11321      1213009      12.83      7.4      7.34      -0.96 #      -0.13        SES:      LOM      4493      694282      9.39      7.1      7303      701703      8.44      6.6      6.75      -0.95 #      -0.14        MIDDLE      7490      1461663      12.90      7.0      11410      126303      12.16      6.8      6.90      -0.70 #      -0.11        RACE:	SEX:											
FEHALE      7893      1434921      12.09      7.2      12549      1346152      11.39      6.9      7.00      -0.70      *      -0.10        SES:      LOM      4493      694282      9.39      7.1      7303      701703      8.44      6.6      6.75      -0.95      *      -0.11        HITE      7490      146162      12.90      7.0      11410      1263636      12.16      6.8      6.90      -0.74      *      -0.11        HITE      12179      2384219      13.95      6.9      17756      2099R66      12.98      6.9      6.90      -0.98      *      -0.12        RACE:      MHITE      12179      2384219      13.95      6.9      17756      2099R66      12.98      6.9      6.90      -0.98      *      -0.14        BLCK      1931      235045      6.50      6.2      3133      33104      15.50      7.2      7.03      -0.67      -0.07        ATTERICAN      101      29767      7.74      6.4      198      220	MALE	7807	1426314	13.79	7.3	11321	1213609	12.83	7.4	7.34	-0.96 ¥	-0.13
SES:    LOM    4493    694282    9.39    7.1    7303    701703    8.44    6.6    6.75    -0.95    *    -0.14      MIDDLE    7490    1461863    12.90    7.0    11410    1263636    12.16    6.89    -0.74    *    -0.11      HIGH    3666    696135    16.62    6.3    5428    624635    15.83    6.4    6.37    -0.79    *    -0.12      RACE:	FEMALZ	7893	1434921	12.09	7.2	12549	1346152	11.39	6.9	7.00	-0.70 <b>*</b>	-0.10
LOM      4493      694282      9.39      7.1      7303      701703      8.44      6.6      6.75      -0.95      *      -0.14        MIDDLE      7490      1461863      12.90      7.0      11410      1263636      12.16      6.8      6.69      -0.79      *      -0.11        MIGH      3666      690135      16.62      6.3      5426      624635      15.83      6.46      6.37      -0.79      *      -0.12        RACE:      MITE      12179      2364219      13.95      6.9      17756      2099866      12.98      6.9      6.30      -0.90      *      -0.14        MATCRICAN      102      25667      15.96      6.7      317      33116      15.50      7.2      7.03      -0.47      -0.07        AMERICAN ARCHICAN      181      29787      7.74      6.4      196      20013      8.28      6.5      6.48      0.54      0.08        MEXICAN-ARERICAN      84      8376      6.33      6.2      256      13398      7.19	SES:											
HIDDLE    7495    1441863    12.70    7.0    11410    125353    12.16    6.8    6.63    -0.74    -0.11      HIGH    3666    590135    16.62    6.3    5428    624635    15.83    6.4    6.37    -0.79    #    -0.11      RACE:	LOH	4493	694282	9.39	7.1	7303	701703	8.44	6.6	6.75	-0.95 ¥	-0.14
NTGH    3666    696135    16.62    6.3    5428    624335    15.83    6.4    6.37    -0.79    *    -0.12      RACE:    IMITE    12179    2384219    13.95    6.9    17756    2099866    12.98    6.9    -0.19    0.03      ASTAN-AHTERICAN    182    25667    15.96    6.7    3115    3284281    6.69    6.3    6.22    0.19    0.03      ASTAN-AHTERICAN    182    25667    15.96    6.7    3117    33116    15.50    7.2    7.03    -0.47    -0.07      ASTAN-AHTERICAN    182    25667    15.96    6.7    3117    33116    15.50    7.2    7.03    -0.47    -0.07      ASTAN-AHTERICAN    164    68165    8.02    6.8    1610    82650    7.54    6.8    6.79    -0.48    -0.07      PUERTO RICAN    64    8376    6.33    6.2    256    13898    7.19    7.5    7.24    0.85    0.12      PUBLIC    14105    2542234    12.79    7.3    21503	MIDDLE	7490	1461863	12.90	7.0	11410	126 36 36	12.16	6.8	6.89	-0.74 *	-0.11
RACE:        MITE      12179      2364219      13.95      6.9      17756      2099A66      12.98      6.90      -0.98      -0.14        BLACK      1931      235045      6.50      6.2      3153      204201      6.69      6.3      6.22      0.19      0.03        ASIAN-AMERICAN      102      25667      15.96      6.7      317      33116      15.50      7.2      7.03      -0.47      -0.07        AMERICAN AMERICAN      101      27787      7.74      6.4      198      20013      8.28      6.5      6.48      0.54      0.08        MEXICAN-AMERICAN      84      8376      6.33      6.2      256      13898      7.19      7.5      7.24      0.65      0.12        OTHER MISPANIC      112      17461      8.04      5.9      839      55699      8.08      6.6      6.69      0.04      0.01        SCHOOL TYPE:      PUBLIC      14105      2542234      12.79      7.3      21503      2383107      11.59      7.2      7.	HIGH	3666	696135	16.62	6.3	5428	624635	15.83	6.4	6.37	-0.79 *	-0.12
MHITE      12179      2384219      13.95      6.9      17756      2099866      12.98      6.90      -0.98      *      -0.14        BLACK      1931      235045      6.50      6.2      3153      284281      6.69      6.3      6.22      0.19      0.03        ASIAN-AHERICAN      182      25667      15.96      6.7      317      33116      15.50      7.2      7.03      -0.47      -0.07        AMERICAN IMDIAN      181      29787      7.74      6.4      198      20013      8.28      6.5      6.48      0.54      0.08        PUERTO RICAN      514      68376      6.33      6.2      256      13896      7.19      7.5      7.24      0.85      0.12        OTHER HISPANIC      112      17461      8.04      5.9      839      55699      8.08      6.8      6.69      0.04      0.01        SCHOOL TYPE:      -      -      -      7.3      21503      2383107      11.59      7.2      7.25      -1.20      *      0.17	RACE :											
BLACK      1931      235045      6.50      6.2      3153      204201      6.69      6.3      6.22      0.19      0.03        ASIAN-AHERICAN      182      25667      15.96      6.7      317      33116      15.50      7.2      7.03      -0.47      -0.07        AMERICAN INDIAN      181      29707      7.74      6.4      198      20013      8.28      6.5      6.48      0.54      0.08        MEXICAN-AHERICAN      514      66165      8.02      6.8      1610      82650      7.54      6.8      6.79      -0.48      -0.07        PUERTO RICAN      84      8376      6.33      6.2      256      13896      7.19      7.5      7.24      0.85      0.12        OTHER MISPANIC      112      17461      8.04      5.9      839      55699      8.08      6.8      6.69      0.00      0.01        SCHOOL      17PE:      PUBLIC      14105      2542234      12.79      7.3      21503      2383107      11.59      7.2      7.25 <td< td=""><td>MHITE</td><td>12179</td><td>2384219</td><td>13,95</td><td>6.9</td><td>17756</td><td>2099886</td><td>12.98</td><td>6.9</td><td>6.90</td><td>-0.98 ¥</td><td>-0.14</td></td<>	MHITE	12179	2384219	13,95	6.9	17756	2099886	12.98	6.9	6.90	-0.98 ¥	-0.14
ASTAN-AHERICAN    182    25667    15.96    6.7    317    33126    15.50    7.2    7.03    -0.47    -0.07      ANTRRICAN    181    29787    7.74    6.4    198    20013    8.28    6.5    6.48    0.54    0.08      MEXICAN-AHERICAN    514    66165    8.02    6.8    1610    82650    7.54    6.8    6.79    -0.47    -0.67      PUERTO RICAN    84    8376    6.33    6.2    256    13898    7.19    7.5    7.24    0.85    0.12      OTHER MISPANIC    112    17461    8.04    5.9    839    55699    8.08    6.8    6.69    0.04    0.01      SCHOOL TYPE:    PUELIC    14105    2542234    12.79    7.3    21503    2383107    11.59    7.2    7.25    -1.20    *    -0.17      PUELIC    14105    2542234    12.79    7.3    21503    2383107    11.59    7.2    7.25    -1.20    *    -0.17      PUELIC    14099    833316    13.29	BLACK	1931	235045	6.50	6.2	3153	284281	6.69	6.3	6.22	0.19	0.03
AMERICAN INDIAN    101    29787    7.74    6.4    198    20013    8.28    6.5    6.48    0.54    0.08      MERICAN-AMERICAN    514    66165    8.02    6.8    1610    82650    7.54    6.8    6.79    -0.48    -0.07      PUERTO RICAN    64    8376    6.33    6.2    256    13898    7.19    7.5    7.24    0.85    0.12      OTHER HISPANIC    112    17461    8.04    5.9    839    55699    8.08    6.69    0.04    0.01      SCHOOL TYPE:    PUBLIC    14105    2542234    12.79    7.3    21503    2383107    11.59    7.2    7.25    -1.20    *    -0.17      PRIVATE    66    16235    15.50    6.0    735    84922    15.48    6.9    6.80    -0.02    -0.00      CATHOLIC    990    224332    15.36    6.4    2520    182516    14.35    6.2    7.1    7.0    7.03    -0.07    *    -0.16      GEOGRAPHIC REGION:    NORTH CENTRAL <td< td=""><td>ASIAN-AMERICAN</td><td>182</td><td>25667</td><td>15.96</td><td>6.7</td><td>317</td><td>33116</td><td>15.50</td><td>7.2</td><td>7.03</td><td>-0.47</td><td>-0.07</td></td<>	ASIAN-AMERICAN	182	25667	15.96	6.7	317	33116	15.50	7.2	7.03	-0.47	-0.07
MEXICAN-AMERICAN    514    68165    8.02    6.8    1610    82650    7.54    6.8    6.79    -0.48    -0.07      PUERTO RICAN    84    8376    6.33    6.2    256    13898    7.19    7.5    7.24    0.85    0.12      OTHER HISPANIC    112    17461    8.04    5.9    839    55699    8.08    6.8    6.69    0.04    0.01      SCHOOL TYPE:    PUBLIC    14105    2542234    12.79    7.3    21503    2383107    11.59    7.2    7.25    -1.20    *    -0.17      PRIVATE    66    16235    15.50    6.0    735    84822    15.48    6.9    6.80    -0.02    -0.00      CATHOLIC    990    224332    15.36    6.4    2520    182516    14.35    6.2    6.27    -1.02    *    -0.16      GEOGRAPHIC REGION:	AMERICAN INDIAN	181	29787	7.74	6.4	198	20013	8.28	6.5	6.48	0.54	0.08
PUERTO RICAN    64    8376    6.33    6.2    256    13898    7.19    7.5    7.24    0.85    0.12      OTHER HISPANIC    112    17461    8.04    5.9    839    55699    8.08    6.8    6.69    0.04    0.01      SCHOOL TYPE:    PUBLIC    14105    2542234    12.79    7.3    21503    2303107    11.59    7.2    7.25    -1.20    *    -0.17      PRUAIC    190    224322    15.50    6.0    735    64822    15.48    6.9    6.80    -0.02    -0.00      CATHOLIC    990    224332    15.36    6.4    2520    182516    14.35    6.2    6.27    -1.02    *    -0.16      GEOGRAPHIC REGION:                                   <	MEXICAN-AMERICAN	514	68165	8.02	6.8	1610	82650	7.54	6.8	6.79	-0.48	-0.07
OTHER HISPANIC      112      17461      8.04      5.9      839      55699      8.08      6.8      6.69      0.04      0.01        SCHOOL TYPE:      PUBLIC      14105      2542234      12.79      7.3      21503      2383107      11.59      7.2      7.25      -1.20      *      -0.17        PRIVATE      66      16235      15.50      6.0      735      64822      15.48      6.9      6.80      -0.02      -0.00        CATHOLIC      990      224332      15.36      6.4      2520      182516      14.35      6.2      6.27      -1.02      *      -0.16        GEOGRAPHIC REGION:      NORTHEAST      3494      778067      13.90      7.1      4955      604620      13.36      7.2      7.15      -0.53      -0.07        NORTH CENTRAL      4099      833316      13.29      7.1      7277      780062      12.59      7.0      7.03      -0.70      *      -0.10        SOUTH      5303      900844      10.41      6.8      8988      969228 </td <td>PUERTO RICAN</td> <td>84</td> <td>8376</td> <td>6.33</td> <td>6.2</td> <td>256</td> <td>13898</td> <td>7.19</td> <td>7.5</td> <td>7.24</td> <td>0.85</td> <td>0.12</td>	PUERTO RICAN	84	8376	6.33	6.2	256	13898	7.19	7.5	7.24	0.85	0.12
SCHOOL TYPE:    PUBLIC    14105    2542234    12.79    7.3    21503    2383107    11.59    7.2    7.25    -1.20 *    -0.17      PRIVATE    66    16235    15.50    6.0    735    84822    15.48    6.9    6.80    -0.02    -0.00      CATHOLIC    990    224332    15.36    6.4    2520    182516    14.35    6.2    6.27    -1.02 *    -0.16      GEOGRAPHIC REGION:	OTHER HISPANIC	112	17461	8.04	5.9	839	55699	8.08	6.8	6.69	0.04	0.01
PUBLIC    14105    2542234    12.79    7.3    21503    2383107    11.59    7.2    7.25    -1.20 *    -0.17      PRIVATE    66    16235    15.50    6.0    735    64822    15.48    6.9    6.80    -0.02    -0.00      CATHOLIC    990    224332    15.36    6.4    2520    182516    14.35    6.2    6.27    -1.02 *    -0.16      GEOGRAPHIC REGION:	SCHOOL TYPE:											
PRIVATE    66    16235    15.50    6.0    735    84822    15.48    6.9    6.80    -0.02    -0.00      CATHOLIC    990    224332    15.36    6.4    2520    182516    14.35    6.2    6.27    -1.02    *    -0.16      GEOGRAPHIC REGION:	PUBLIC	14105	2542234	12.79	7.3	21503	2383107	11.59	7.2	7.25	-1.20 ¥	-0.17
CATHOLIC    990    224332    15.36    6.4    2520    182516    14.35    6.2    6.27    -1.02 *    -0.16      GEOGRAPHIC REGION:    NORTHEAST    3494    778067    13.90    7.1    4955    604620    13.36    7.2    7.15    -0.53    -0.07      NORTHEAST    3494    778067    13.90    7.1    4955    604620    13.36    7.2    7.15    -0.53    -0.07      NORTHEAST    4099    833316    13.29    7.1    7277    783062    12.59    7.0    7.03    -0.70 *    -0.10      SOUTH    5322    773173    11.95    7.4    8171    805015    10.07    7.1    7.25    -0.34    -0.26      WEST    2780    477695    12.37    7.4    4355    457749    12.04    7.1    7.25    -0.34    -0.05      CURRICULM:    GENERAL    5303    900844    10.41    6.8    8988    969228    9.89    6.5    6.62    -0.52 *    -0.08      ACADEMIC    GENERAL    5303    900844	PRIVATE	66	16235	15.50	6.0	735	84822	15.48	6.9	6.80	-0.02	-0.00
GEOGRAPHIC REGION:      NORTHEAST    3494    778067    13.90    7.1    4955    604620    13.36    7.2    7.15    -0.53    -0.07      NORTH CENTRAL    4099    833316    13.29    7.1    7277    783062    12.59    7.0    7.03    -0.70    #    -0.10      SOUTH    5332    773173    11.95    7.4    8171    805015    10.07    7.1    7.23    -1.88    #    -0.26      WEST    2780    477695    12.37    7.4    4355    457749    12.04    7.1    7.25    -0.34    -0.05      CURRICULUM:      GENERAL    5303    900844    10.41    6.8    8986    969228    9.89    6.5    6.62    -0.52    #    -0.06      ACADEMIC    6497    1323927    16.66    6.2    9433    1007295    16.17    6.2    6.24    -0.49    #    -0.08      VOCATIONAL    3914    637177    8.78    6.2    5990    637325    8.48    6.2    6.23    -0.	CATHOLIC	990	224332	15.36	6.4	2520	182516	14.35	6.2	6.27	-1.02 *	-0.16
NORTHEAST    3494    778067    13.90    7.1    4955    604620    13.36    7.2    7.15    -0.53    -0.07      NORTH CENTRAL    4099    833316    13.29    7.1    7277    783062    12.59    7.0    7.03    -0.70    *    -0.10      SOUTH    5332    773173    11.95    7.4    8171    805015    10.07    7.1    7.23    -1.88    *    -0.26      WEST    2780    477695    12.37    7.4    4355    457749    12.04    7.1    7.25    -0.34    -0.05      CURRICULUM:    5303    900844    10.41    6.8    8988    969228    9.89    6.5    6.62    -0.52    *    -0.08      ACADEMIC    6497    1323927    16.66    6.2    9433    1007295    16.17    6.2    6.24    -0.49    *    -0.08      VOCATIONAL    3914    637177    8.78    6.2    5990    637325    8.48    6.2    6.23    -0.31    -0.05      URBAN    4097    703168 <t< td=""><td>GEOGRAPHIC REGION:</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></t<>	GEOGRAPHIC REGION:											
NORTH CENTRAL    4099    833316    13.29    7.1    7277    783062    12.59    7.0    7.03    -0.70 *    -0.10      SOUTH    5332    773173    11.95    7.4    8171    805015    10.07    7.1    7.23    -1.88 *    -0.26      WEST    2780    477695    12.37    7.4    4355    457749    12.04    7.1    7.25    -0.34    -0.05      CURRICULUM:    5303    900844    10.41    6.8    8988    969228    9.89    6.5    6.62    -0.52 *    -0.08      ACADEMIC    6497    1323927    16.66    6.2    9433    1007295    16.17    6.2    6.24    -0.49 *    -0.08      VOCATIONAL    3914    637177    8.78    6.2    5990    637325    8.48    6.2    6.23    -0.31    -0.05      COMMUNITY TYPE:    URBAN    4097    703168    12.16    7.3    5466    509127    10.98    7.4    7.34    -1.18 *    -0.16      SUBURBAN    7648    1481125    13.81    7.2<	NORTHEAST	3494	778067	13.90	7.1	4955	604620	13.36	7.2	7.15	-0.53	-0.07
SOUTH WEST    5332    773173    11.95    7.4    8171    805015    10.07    7.1    7.23    -1.88 *    -0.26      WEST    2780    477695    12.37    7.4    4355    457749    12.04    7.1    7.25    -0.34    -0.05      CURRICULUM: GENERAL    5303    900844    10.41    6.8    8988    969228    9.89    6.5    6.62    -0.52 *    -0.08      ACADEMIC    6497    1323927    16.66    6.2    9433    1007295    16.17    6.2    6.24    -0.49 *    -0.08      VOCATIONAL    3914    637177    8.78    6.2    5990    637325    8.48    6.2    6.23    -0.31    -0.05      COMMUNITY TYPE:    URBAN    4097    703168    12.16    7.3    5466    509127    10.98    7.4    7.34    -1.18 *    -0.16      SUBURBAN    7648    1481125    13.81    7.2    11820    1295212    12.70    7.1    7.14    -1.12 *    -0.16      RURAL    3538    611858    12.15    <	NORTH CENTRAL	4099	833316	13.29	7.1	7277	783062	12.59	7.0	7.03	-0.70 *	-0.10
WEST    2780    477695    12.37    7.4    4355    457749    12.04    7.1    7.25    -0.34    -0.05      CURRICULUM: GENERAL    5303    900844    10.41    6.8    8986    969228    9.89    6.5    6.62    -0.52 #    -0.08      ACADEMIC    6457    1323927    16.66    6.2    9433    1007295    16.17    6.2    6.24    -0.49 #    -0.08      VOCATIONAL    3914    637177    8.78    6.2    5990    637325    8.48    6.2    6.23    -0.31    -0.05      COMPUNITY TYPE:    URBAN    4097    703168    12.16    7.3    5466    509127    10.98    7.4    7.34    -1.18 #    -0.16      SUBURBAN    7648    1481125    13.81    7.2    11820    1295212    12.70    7.1    7.14    -1.12 #    -0.16      RURAL    3538    611858    12.15    7.3    7472    846107    11.24    7.1    7.14    -0.91 #    -0.13	SOUTH	5332	773173	11.95	7.4	8171	805015	10.07	7.1	7.23	-1.88 <b>*</b>	-0.26
CURRICULUM:    5303    900844    10.41    6.8    8988    969228    9.89    6.5    6.62    -0.52 *    -0.08      ACADEMIC    6487    1323927    16.66    6.2    9433    1007295    16.17    6.2    6.24    -0.49 *    -0.08      VOCATIONAL    3914    637177    8.78    6.2    5990    637325    8.48    6.2    6.23    -0.31    -0.05      COMMUNITY TYPE:    URBAN    4097    703168    12.16    7.3    5466    509127    10.98    7.4    7.34    -1.18 *    -0.16      SUBURBAN    7648    1481125    13.81    7.2    11820    1295212    12.70    7.1    7.14    -1.12 *    -0.16      RURAL    3538    611858    12.15    7.3    7472    846107    11.24    7.1    7.14    -0.91 *    -0.13	WEST	27 <b>8</b> 0	477695	12.37	7.4	4355	457749	12.04	7.1	7.25	-0.34	-0.05
GENERAL    5303    900844    10.41    6.8    8988    969228    9.89    6.5    6.62    -0.52 *    -0.08      ACADEMIC    6457    1323927    16.66    6.2    9433    1007295    16.17    6.2    6.24    -0.49 *    -0.08      VOCATIONAL    3914    637177    8.78    6.2    5990    637325    8.48    6.2    6.23    -0.31    -0.05      COMMUNITY TYPE:    URBAN    4097    703168    12.16    7.3    5466    509127    10.98    7.4    7.34    -1.18 *    -0.16      SUBURBAN    7648    1481125    13.81    7.2    11820    1295212    12.70    7.1    7.14    -1.12 *    -0.16      RURAL    3538    611858    12.15    7.3    7472    846107    11.24    7.1    7.14    -0.91 *    -0.13	CURRICULUM:											
ACADEMIC    6457    1323927    16.66    6.2    9433    1007295    16.17    6.2    6.24    -0.49 *    -0.08      VOCATIONAL    3914    637177    8.78    6.2    5990    637325    8.48    6.2    6.23    -0.31    -0.05      COMPLINITY TYPE:    URBAN    4097    703168    12.16    7.3    5466    509127    10.98    7.4    7.34    -1.18 *    -0.16      SUBURBAN    7648    1481125    13.81    7.2    11820    1295212    12.70    7.1    7.14    -1.12 *    -0.16      RURAL    3538    611858    12.15    7.3    7472    846107    11.24    7.1    7.14    -0.91 *    -0.13	GENERAL	5303	900844	10.41	6.8	8988	969228	9.89	6.5	6.62	-0.52 *	-0.08
VOCATIONAL      3914      637177      8.78      6.2      5990      637325      8.48      6.2      6.23      -0.31      -0.05        COMPLINITY      TYPE:      URBAN      4097      703168      12.16      7.3      5466      509127      10.98      7.4      7.34      -1.18      -0.16        SUBURBAN      7648      1481125      13.81      7.2      11820      1295212      12.70      7.1      7.14      -1.12      -0.16        RURAL      3538      611858      12.15      7.3      7472      846107      11.24      7.1      7.14      -0.91      -0.13	ACADEMIC	6457	1323927	16.66	6.2	9433	1007295	16.17	6.2	6.24	-0.49 *	-0.08
CONFLINITY TYPE:      URBAN      4097    703168    12.16    7.3    5466    509127    10.98    7.4    7.34    -1.18    *    -0.16      SUBURBAN    7648    1481125    13.81    7.2    11820    1295212    12.70    7.1    7.14    -1.12    *    -0.16      RURAL    3538    611858    12.15    7.3    7472    846107    11.24    7.1    7.14    -0.91    *    -0.13	VOCATIONAL	3914	637177	8.78	6.2	5990	637325	8.48	6.2	6.23	-0.31	-0.05
URBAN      4097      703168      12.16      7.3      5466      509127      10.98      7.4      7.34      -1.18      +      -0.16        SUBURBAN      7648      1481125      13.81      7.2      11820      1295212      12.70      7.1      7.14      -1.12      *      -0.16        RURAL      3538      611858      12.15      7.3      7472      846107      11.24      7.1      7.14      -0.91      *      -0.13	COMPUNITY TYPE:											
SUBURBAN      7648      1481125      13.81      7.2      11820      1295212      12.70      7.1      7.14      -1.12      *      -0.16        RURAL      3538      611858      12.15      7.3      7472      846107      11.24      7.1      7.14      -0.91      *      -0.13	URBAN	4097	703168	12.16	7.3	5466	509127	10.98	7.4	7.34	-1.18 ¥	-0.16
RURAL 3538 611858 12.15 7.3 7472 846107 11.24 7.1 7.14 -0.91 * -0.13	SUBURBAN	7648	1481125	13.81	7.2	11820	1295212	12.70	7.1	7.14	-1.12 *	-0.16
	RURAL	3538	611858	12.15	7.3	7472	846107	11.24	7.1	7.14	-0.91 *	-0.13

\*SIGNIFICANT AT .05 OR LESS



#### D. CHANGES IN GRADES 1972-1980

On the student questionnaires, the 1972 and 1980 seniors were asked to describe their high school grades. The scale ranged from 1 = Mostly below D (below 60) to 8 = Mostly A (a numerical average of 90-100). As can be seen from Table 5-4, the mean grade reported increased slightly from 5.55 in 1972 to 5.70 in 1980, or from the middle to slightly above the middle of the mostly C (70 to 74) category. Thus, while test scores have shown small to moderate declines, grades have gone up during the eight-year period. For both the total population and most subpopulations there is a small positive effect indicating an increase in self-reported grades. The increase in self-reported grades is consistent for Blacks and Whites and for all SES areas. The increase in grades is disproportionately greater for students in the academic curriculum and for students in the Northeast and North Central Regions.

As indicated earlier, an increase in grades while achievement test scores declined suggests a slippage in school standards as reflected in the school grades. The fact that both the test score declines and the grade increases were consistent across both curriculum and SES groups suggests that the slippage in standards is pervasive. These results taken in combination with the self-reports of spending less time doing homework (see Chapter VI) suggests there is not only a lowering of grading standards but a lowering of other educational demands and requirements. While no one social indicator can be interpreted in isolation from other relevant and critical indicators, when related social indicators are considered as constellations, we can make causal inferences, albeit tentatively.

If there has been a decrease in the involuntary learning (school required work), then gains in achievement must rely more heavily on voluntary learning. Unfortunately, there are many competing demands on the time for high school students today, not the least of which are extracurricular activities, TV watching, paid work, and non-school related hobbies and social activities. Some of these will be discussed in Chapter VI. The majority, however, were not covered in this cross-sectional stud 1d must await analysis in the 1980-82 longitudinal study.

In sum, these data on grades tend to confirm the much discussed "grade inflation," but they also show that the size of this trend toward higher grades is not substantial for most groups of students.



#### Table 5-4

#### GRADES IN HIGH SCHOOL (1=BELOW D; 8=MOSTLY A)

\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*

SAMPLE WEIGNTED      SAMPLE WEIGNTED      POOLED      1980-1972      EFFECT        TOTAL      16576      SAMPLE WEIGNTED      N      N      N      N      N      N      N      N      N      N      N      N      N      N      N      N      N      N      N      N      N      N      DOLED      1980-1972      EFFECT        TOTAL      16576      S.26      1.4      1.4      1.4      1.4      1.4      1.4      1.4      1.4      1.4      1.4      1.4      1.4      1.4      1.4      1.4      1.4      1.4      1.4      1.4      1.4      1.4      1.4      1.4      1.4		NLS 1972			HSB 1980							
TOTAL    16576    3024388    5.55    1.4    28013    3017166    5.70    1.4    1.43    0.14 *    0.10      SEX:    MALE    8320    1506570    5.26    1.4    12815    1389400    5.46    1.4    1.43    0.20 *    0.14      FFMALE    8350    154601    5.64    1.4    14000    5.97    1.4    1.33    0.20 *    0.14      MIDDLE    7864    155670    5.26    1.4    12815    1389400    5.46    1.4    1.43    0.20 *    0.08      MIDDLE    7864    15567    5.26    1.4    127712    5.77    1.4    1.43    0.20 *    0.08      MALE    27093    5.99    1.4    6150    719407    6.14    1.4    1.39    0.16 *    0.11      RACE:		SAMPLE N	WEIGHTED N	MEAN	\$.D.	SAMPLE N	WEIGHTED N	MEAN	S.D.	POOLED S.D.	1980-1972 DIFFERENCE	EFFECT SIZE
SEX: MALE 6221 1506570 5.26 1.4 12815 1389400 5.46 1.4 1.43 0.20 * 0.14 FEMALE 8350 1516001 5.84 1.4 12815 1389400 5.46 1.4 1.43 0.12 * 0.08 SES: LOM 4789 736837 5.24 1.4 8335 804180 5.36 1.4 1.43 0.12 * 0.08 HIDDLE 7864 1545878 5.50 1.4 1272 1417012 5.70 1.4 1.43 0.12 * 0.08 HIDDLE 7864 1545878 5.50 1.4 1272 1417012 5.70 1.4 1.43 0.17 * 0.12 MHITE 12779 2513137 5.64 1.4 19756 2352390 5.81 1.4 1.43 0.17 * 0.12 MHITE 12779 2513137 5.64 1.4 19756 2352390 5.81 1.4 1.43 0.17 * 0.12 MHITE 12779 2513137 5.64 1.4 19756 2352390 5.81 1.4 1.43 0.17 * 0.12 MHITE 12779 2513137 5.64 1.4 19756 2352390 5.81 1.4 1.43 0.17 * 0.12 MHITE 12779 2513137 5.64 1.4 19756 2352390 5.81 1.4 1.43 0.17 * 0.13 ASIAM-AMERICAN 192 27493 5.99 1.4 362 39134 6.18 1.3 1.32 0.19 0.15 AMERICAN-AMERICAN 551 72309 5.09 1.4 380 101554 5.16 1.4 1.37 0.07 0.05 PUERTO RICAN 94 9541 5.29 1.5 302 17756 5.34 1.4 1.40 0.31 0.22 MERICAN-MERICAN 94 9541 5.29 1.5 302 17756 5.34 1.4 1.43 0.06 0.04 OTHER HISPANIC 120 18502 5.38 1.3 969 66541 5.26 1.4 1.37 -0.12 -0.09 SCHOOL TYPE: PRIVATE 16459 2683774 5.52 1.4 24473 2715176 5.64 1.5 1.44 0.13 * 0.66 * 0.43 CATHOLIC 10.3 234807 5.96 1.3 2675 198612 5.99 1.3 1.30 0.63 * 0.03 EECOGMAPHIC RESION: NORTHEAST 2501 796146 5.64 1.3 5625 687480 5.66 1.4 1.35 0.03 * 0.03 EECOGMAPHIC RESION: NORTHEAST 2066 520805 5.56 1.4 1.3 1.40 0.13 * 0.05 CLARICULUM: GENERAL 6536 961916 5.09 1.4 10216 1104363 5.26 1.4 1.35 CLARICULUM: GENERAL 6536 961916 5.09 1.4 10216 1104363 5.26 1.4 1.35 CLARICULUM: GENERAL 6536 961916 5.09 1.4 10216 1104363 5.26 1.4 1.35 CLARICULUM: GENERAL 6536 961916 5.09 1.4 10216 1104363 5.26 1.4 1.35 CLARICULUM: GENERAL 6536 961916 5.09 1.4 10216 1104363 5.26 1.4 1.35 CLARICULUM: GENERAL 6536 961916 5.09 1.4 10216 1104363 5.26 1.4 1.35 CLARICULUM: GENERAL 6526 781669 5.48 1.4 6460 605593 5.54 1.4 1.40 CLARICULUM: GENERAL 6526 781669 5.48 1.4 6460 109595 5.50 1.4 1.4 1.43 O.16 * 0.12 COMMINITY TYPE: URBAN 7722 13832176 5.60 1.4 1.4 8040 1490956 5.70 1.4 1.43 O.10 * 0.07	TOTAL	16576	3024388	5.55	1.4	28013	3017186	5.70	1.4	1.43	0.14 *	0.10
PALE      6221      1506570      5.26      1.4      12815      1339400      5.46      1.4      1.43      0.20      0.14        SES:      LDM      4789      736837      5.24      1.4      0335      00180      5.36      1.4      1.43      0.12      0.08        MIDDLE      7804      1545078      5.50      1.4      12742      1317012      5.70      1.4      1.40      0.20      0.14        HIGH      7804      1545078      5.50      1.4      12742      1317012      5.70      1.4      1.40      0.20      0.14        HIGH      3641      73090      5.90      1.4      13730      339699      5.27      1.3      1.33      0.17      0.13        BLACK      2107      2513137      5.64      1.4      19756      2352390      5.81      1.4      1.43      0.17      0.13        MATRICAN      192      27493      5.99      1.4      362      39134      6.16      1.3      1.32      0.17      0.13	SEX:											
FEMALE    8350    1516001    5.84    1.4    14000    1504276    5.97    1.4    1.38    0.13    *    0.09      SES:    LOM    4789    736837    5.24    1.4    12742    1617012    5.70    1.4    1.40    0.20    *    0.14      HIDLE    7864    1545078    5.50    1.4    12742    1617012    5.70    1.4    1.40    0.20    *    0.14      HIGH    3841    730980    5.96    1.4    6150    719407    6.14    1.4    1.39    0.16    *    0.11      RACE:    MHITE    12779    2513137    5.64    1.4    19756    2352390    5.81    1.4    1.43    0.17    0.12      MARSANAMERICAN    122    24735    5.99    1.4    362    37134    3161    1.41    1.30    0.17    0.12      MITE    1270    254734    5.10    1.3    302    17735    5.23    1.4    1.43    0.00    0.31    0.22      MITEAM-AMERICAN    104	MALE	8221	1506570	5.26	1.4	12815	1389400	5.46	1.4	1.43	0.20 ×	0.14
SES:    LOM    4789    736837    5.24    1.4    8335    804180    5.36    1.4    1.43    0.12 *    0.08      HIDDLE    78844    1545878    5.50    1.4    12742    1417012    5.70    1.4    1.40    0.20 *    0.14      HITCH    3641    730980    5.96    1.4    6150    719407    6.14    1.4    1.39    0.16 *    0.11      RACE:	FEMALE	8350	15168 <b>0</b> 1	5.84	1.4	14000	15 <b>0</b> 4276	5.97	1.4	1.38	0.13 *	0.09
LON 4789 736037 5.24 1.4 0335 004180 5.36 1.4 1.43 0.12 * 0.08 HIDDLE 7864 1545878 5.50 1.4 12742 1417012 5.70 1.4 1.40 0.20 * 0.14 HIGH 3064 1545878 5.50 1.4 12742 1417012 5.70 1.4 1.40 0.20 * 0.14 HIGH 3064 1545878 5.50 1.4 12742 1417012 5.70 1.4 1.40 0.20 * 0.14 HIGH 3064 1730980 5.99 1.4 6.150 719407 6.14 1.4 1.33 0.15 * 0.13 ALACK 2107 254734 5.10 1.3 3730 33909 5.27 1.3 1.33 0.17 * 0.12 ALACK 2107 254734 5.10 1.3 3730 33909 5.27 1.3 1.33 0.17 * 0.13 ATENICAN INDIAN 166 30926 4.92 1.4 212 21795 5.23 1.4 1.40 0.31 0.22 HEXICAN-AMERICAN 551 72389 5.09 1.4 1800 101554 5.16 1.4 1.37 0.07 0.05 PUERTO RICAN 99 9541 5.29 1.5 302 17736 5.34 1.4 1.43 0.06 0.04 OTHER HISPANIC 120 18582 5.38 1.3 969 66541 5.26 1.4 1.37 -0.12 -0.09 SCHOOL TYPE: PRIVATE 67 16549 5.58 1.3 865 103196 6.14 1.3 1.30 0.56 * 0.43 CATHOLIC 10.3 23407 5.96 1.3 2675 198812 5.99 1.5 1.47 0.13 * 0.09 PRIVATE 67 16549 5.58 1.3 865 103196 6.14 1.3 1.30 0.56 * 0.43 CATHOLIC 10.3 23407 5.96 1.3 2675 198812 5.97 1.5 1.47 0.19 * 0.13 SOUTH 1570 722013 5.64 1.4 9222 916771 5.69 1.5 1.44 0.13 * 0.09 PRIVATE 772 13531 91414 5.40 1.4 8063 665043 5.59 1.5 1.44 0.19 * 0.13 SOUTH 5476 792013 5.64 1.4 9222 916771 5.69 1.5 1.44 0.010 * 0.13 SOUTH 5476 792013 5.64 1.4 9222 916771 5.69 1.5 1.44 0.010 * 0.13 SOUTH 5476 792013 5.64 1.4 9222 916771 5.69 1.5 1.44 0.010 * 0.13 SOUTH 5476 792013 5.64 1.4 9222 916771 5.69 1.5 1.44 0.010 * 0.13 SOUTH 5476 792013 5.64 1.4 9222 916771 5.69 1.5 1.45 0.05 0.04 WEST 2966 520005 5.56 1.3 10494 1133662 6.36 1.4 1.40 0.10 * 0.13 SOUTH 4167 6772 1333415 6.05 1.3 10494 1133662 6.36 1.4 1.30 0.31 0.31 0.24 COMPLICIVE: COMPLICINAL 4167 6776 1363415 6.05 1.3 10494 1133662 6.36 1.4 1.40 0.10 * 0.13 ACADENIC 6772 1353415 6.05 1.3 10494 1133662 6.36 1.4 1.40 0.10 * 0.13 ACADENIC 4167 6772 1363415 6.05 1.3 10499 1133662 6.36 1.4 1.44 0.30 0.31 * 0.24 YOCATIONAL 4167 67764 5.60 1.4 13400 1899056 5.70 1.4 1.43 0.10 * 0.13 BURAN 4524 781669 5.48 1.4 6440 605593 5.54 1.4 1.44 0.06 0.05 SUBURGAN 7922 1552178 5.6	SES:											
HIDDLE    7884    1545878    5.50    1.4    12742    1417012    5.70    1.4    1.40    0.20 *    0.14      HIGH    3041    730980    5.98    1.4    6150    719407    6.14    1.4    1.39    0.16 *    0.11      RACE:	LON	4789	736837	5.24	1.4	8335	804180	5.36	1.4	1.43	0.12 ×	8.08
HIGH      3841      730980      5.96      1.4      6150      719407      6.14      1.4      1.39      0.16 *      0.11        RACE:      MHITE      12779      2513137      5.64      1.4      19756      2352390      5.81      1.4      1.43      0.17 *      0.12        BLACK      2107      254734      5.10      1.3      3730      339899      5.27      1.3      1.33      0.17 *      0.13        ASTAN-AMERICAN      192      27493      5.99      1.4      362      39134      6.18      1.3      1.32      0.17 *      0.13        AMERICAN INDIAN      166      30026      4.92      1.4      212      21775      5.23      1.4      1.43      0.06      0.15        MERICAN-AMERICAN      551      72369      5.09      1.4      1800      101554      5.16      1.4      1.37      0.07      0.05        PUENTO      RICAN      94      9541      5.26      1.4      1.38      0.66      1.4      1.33      0.66      1.4	MIDDLE	7884	1545878	5.50	1.4	12742	1417012	5.70	1.4	1.40	0.20 *	0.14
RACE:      MITE      12779      2513137      5.64      1.4      19756      2352390      5.81      1.4      1.43      0.17      *      0.13        ASIAN-AHERICAN      192      27493      5.99      1.4      362      39134      6.18      1.3      1.33      0.17      *      0.13        ASIAN-AHERICAN      192      27493      5.99      1.4      212      21795      5.23      1.4      1.40      0.31      0.22        AMERICAN AMERICAN      194      9541      5.29      1.4      101554      5.16      1.4      1.37      0.07      0.05        PUERTO RICAN      94      9541      5.29      1.5      302      17736      5.34      1.4      1.43      0.06      0.04        OTHER HISPANIC      120      18582      5.36      1.3      969      66541      5.26      1.4      1.37      -0.12      -0.09        SCHOOL TYPE:      PUBLIC      14859      2683974      5.52      1.4      24473      2715178      5.66      1.5      1.	HIGH	3841	730980	5.98	1.4	6150	719407	6.14	1.4	1.39	0.16 *	0.11
MHITE      12779      2513137      5.64      1.4      19756      2352390      5.81      1.4      1.43      0.17      *      0.12        BLACK      2107      254734      5.10      1.3      3730      339899      5.27      1.3      1.33      0.17      *      0.13        ASIAN-AMERICAN      192      27493      5.99      1.4      362      39136      6.18      1.3      1.32      0.19      0.15        AHERICAN INDIAN      186      30926      4.92      1.4      212      21795      5.23      1.4      1.40      0.31      0.22        MEXICAN-AMERICAN      951      72389      5.09      1.4      1880      101554      5.16      1.4      1.43      0.06      0.04        OTHER MISPANIC      120      18582      5.38      1.3      969      66541      5.26      1.4      1.37      -0.12      -0.09        SCHOOL TYPE:      PUBLIC      14659      5.52      1.4      24473      2715178      5.66      1.4      1.31      0.30<	RACE:											
BLACK      2107      254734      5.10      1.3      3730      339899      5.27      1.3      1.33      0.17      *      0.13        ASIAN-AMERICAN      192      27493      5.99      1.4      362      39134      6.18      1.3      1.32      0.19      0.15        AMERICAN INDIAN      186      30926      4.92      1.4      212      21795      5.23      1.4      1.40      0.31      0.22        MEXICAN-AMERICAN      551      72389      5.09      1.4      1880      101554      5.16      1.4      1.37      0.07      0.05        PLENTO RICAN      96      9541      5.29      1.5      302      17736      5.34      1.4      1.43      0.06      0.04        OTHER HISPANIC      120      18582      5.52      1.4      24473      2715178      5.66      1.5      1.44      0.13      0.09        PRUATE      67      16549      5.52      1.4      24473      2715178      5.66      1.4      1.30      0.05      0.43	WHITE	12779	2513137	5.64	1.4	19756	2352390	5.81	1.4	1.43	0.17 ×	0.12
ASIAN-AMERICAN    192    27493    5.99    1.4    362    39134    6.18    1.3    1.32    0.19    0.15      AMERICAN INDIAN    186    30926    4.92    1.4    212    21795    5.23    1.4    1.40    0.31    0.22      PUERTO PICAN    94    9541    5.29    1.5    302    17736    5.34    1.4    1.43    0.06    0.04      OTHER HISPANIC    120    18582    5.36    1.3    969    66541    5.26    1.4    1.37    -0.12    -0.09      SCHOOL TYPE:    PUERTO PICAN    94    9541    5.52    1.4    24473    2715178    5.66    1.5    1.44    0.13 *    0.09      SCHOOL TYPE:    PBLIC    14859    2683974    5.52    1.4    24473    2715178    5.66    1.5    1.44    0.13 *    0.09      SCHOOL TYPE:    PBLIC    14859    2683974    5.52    1.4    24473    2715178    5.66    1.4    1.3    1.30    0.56 *    0.43      CATHOLIC    10.3	BLACK	2107	254734	5.10	1.3	3730	339899	5.27	1.3	1.33	0.17 ×	0.13
AMERICAN INDIAN    186    30926    4.92    1.4    212    21795    5.23    1.4    1.40    0.31    0.22      MEXICAN-AMERICAN    551    72389    5.09    1.4    1880    101554    5.16    1.4    1.37    0.07    0.05      PUERTO RICAN    9541    5.29    1.5    302    17736    5.34    1.4    1.37    0.07    0.05      SCHOOL TYPE:    120    18582    5.38    1.3    969    66541    5.26    1.4    1.37    -0.12    -0.09      SCHOOL TYPE:    PUELIC    14859    2683974    5.52    1.4    24473    2715178    5.66    1.5    1.44    0.13    *    0.09      PRIVATE    67    16549    5.58    1.3    2675    198812    5.99    1.3    1.30    0.03    0.02      GEOGRAPHIC REGION:	ASIAN-AMERICAN	192	27493	5.99	1.4	362	39134	6.18	1.3	1.32	0.19	0.15
MEXICAN-AHERICAN    551    72369    5.09    1.4    1880    101554    5.16    1.4    1.37    0.07    0.05      PUERTO RICAN    94    9541    5.29    1.5    302    17736    5.34    1.4    1.43    0.06    0.04      OTHER HISPANIC    120    1852    5.38    1.3    969    66541    5.26    1.4    1.47    -0.12    -0.09      SCHOOL TYPE:    PUBIC    14859    2683974    5.52    1.4    24473    2715178    5.66    1.5    1.44    0.13    0.09      SCHOOL TYPE:    PRUVATE    67    16549    5.58    1.3    865    103196    6.14    1.3    1.30    0.056    *    0.43      CATHOLIC    10.3    234907    5.96    1.3    2675    198812    5.99    1.3    1.30    0.03    0.02      GEOGRAPHIC REGION:    NORTH CENTRAL    3561    796146    5.64    1.4    3625    68748    5.66    1.4    1.36    0.22 *    0.16      NORTH CESION:    NORTH CENTRAL<	AMERICAN INDIAN	186	30926	4.92	1.4	212	21795	5.23	1.4	1.40	0.31	0.22
PUERTO RICAN      94      9541      5.29      1.5      302      17736      5.34      1.4      1.43      0.06      0.04        OTHER HISPANIC      120      18582      5.38      1.3      969      66541      5.26      1.4      1.37      -0.12      -0.09        SCHOOL TYPE:      PUBLIC      14859      2683974      5.52      1.4      24473      2715178      5.66      1.5      1.44      0.13      *      0.09        PRUATE      67      16549      5.58      1.3      865      103196      6.14      1.3      1.30      0.65      *      0.43        CATHOLIC      10.3      234807      5.96      1.3      2675      19812      5.99      1.3      1.30      0.03      0.02        GEOGRAPHIC REGION:      NORTHEAST      3581      796146      5.64      1.4      8063      865643      5.59      1.5      1.47      0.19      0.13        SOUTH      5478      792013      5.64      1.4      922      916771      5.69      1.4	MEXICAN-AMERICAN	551	72389	5.09	1.4	1880	101554	5.16	1.4	1.37	0.07	0.05
OTHER HISPANIC      120      18582      5.36      1.3      969      66541      5.26      1.4      1.37      -0.12      -0.09        SCHOOL TYPE: PUBLIC      14859      2683974      5.52      1.4      24473      2715178      5.66      1.5      1.44      0.13 *      0.09        PRIVATE      67      16549      5.58      1.3      865      103196      6.14      1.3      1.30      0.56 *      0.43        CATHOLIC      10.3      234807      5.96      1.3      2675      198812      5.99      1.3      1.30      0.03      0.02        GEOGRAPHIC REGION:	PUERTO RICAN	94	9541	5.29	1.5	302	17736	5.34	1.4	1.43	0.06	0.04
SCHOOL TYPE:      PUBLIC      14859      2683974      5.52      1.4      24473      2715178      5.66      1.5      1.44      0.13      *      0.09        PRIVATE      67      16549      5.56      1.3      865      103196      6.14      1.3      1.30      0.56      *      0.43        CATHOLIC      10.3      234807      5.96      1.3      2675      198812      5.99      1.3      1.30      0.03      0.02        GEOGRAPHIC REGION:	OTHER HISPANIC	120	18582	5.38	1.3	969	66541	5.26	1.4	1.37	-0.12	-0.09
PUBLIC    14859    2683974    5.52    1.4    24473    2715178    5.66    1.5    1.44    0.13 *    0.09      PRIVATE    67    16549    5.58    1.3    865    103196    6.14    1.3    1.30    0.56 *    0.43      CATHOLIC    10.3    234807    5.96    1.3    2675    198812    5.99    1.3    1.30    0.03    0.02      GEOGRAPHIC REGION:    NORTHEAST    3581    796146    5.64    1.3    5625    686748    5.86    1.4    1.36    0.22 *    0.16      NORTHEAST    3581    796146    5.64    1.3    5625    686748    5.86    1.4    1.36    0.22 *    0.16      NORTHEAST    3581    796146    5.64    1.4    9222    916771    5.69    1.5    1.47    0.19 *    0.13      SOUTH    5478    792013    5.64    1.4    9222    916771    5.69    1.5    1.45    0.05    0.04      WEST    2966    522085    5.56    1.4    10216    <	SCHOOL TYPE:											
PRIVATE    67    16549    5.58    1.3    865    103196    6.14    1.3    1.30    0.56 *    0.43      CATHOLIC    10.3    234807    5.96    1.3    2675    198812    5.99    1.3    1.30    0.03    0.02      GEOGRAPHIC REGION:              0.03    0.02      GEOGRAPHIC REGION:             0.13    0.03    0.02      GEOGRAPHIC REGION:              0.03    0.02       NORTH CENTRAL    4551    914144    5.40    1.4    8063    865643    5.59    1.5    1.47    0.19    0.13      SOUTH    5478    792013    5.64    1.4    9222    916771    5.69    1.5    1.45    0.05    0.04      WEST    2966    522085    5.56    1.4 </td <td>PUBLIC</td> <td>14859</td> <td>2683974</td> <td>5.52</td> <td>1.4</td> <td>24473</td> <td>2715178</td> <td>5.66</td> <td>1.5</td> <td>1.44</td> <td>0.13 ×</td> <td>0.09</td>	PUBLIC	14859	2683974	5.52	1.4	24473	2715178	5.66	1.5	1.44	0.13 ×	0.09
CATHOLIC    10.3    234807    5.96    1.3    2675    198812    5.99    1.3    1.30    0.03    0.02      GEOGRAPHIC REGION: NORTHEAST    3581    796146    5.64    1.3    5625    68748    5.86    1.4    1.36    0.22 *    0.16      NORTHEAST    3581    796146    5.64    1.3    5625    68748    5.86    1.4    1.36    0.22 *    0.16      NORTHEAST    4551    914144    5.40    1.4    8065    865643    5.59    1.4    1.47    0.19 *    0.13      SOUTH    5478    792013    5.64    1.4    9222    916771    5.69    1.5    1.45    0.05    0.04      WEST    2966    522085    5.56    1.4    5103    546024    5.67    1.4    1.42    0.11    0.08      CURRICULUM:    GENERAL    5636    961916    5.09    1.4    10216    1104363    5.26    1.4    1.40    0.18 *    0.13      ACADEMIC    6772    1383415    6.05    1.3    10494	PRIVATE	67	16549	5.58	1.3	865	103196	6.14	1.3	1.30	0.56 ×	0.43
GEOGRAPHIC REGION:        NORTHEAST      3581      796146      5.64      1.3      5625      688748      5.86      1.4      1.36      0.22 *      0.16        NORTH CENTRAL      4551      914144      5.40      1.4      8063      865643      5.59      1.5      1.47      0.19 *      0.13        SOUTH      5478      792013      5.64      1.4      9222      916771      5.69      1.5      1.45      0.05      0.04        MEST      2966      522085      5.56      1.4      5103      546024      5.67      1.4      1.42      0.11      0.08        CURRICULUM:        GENERAL      5636      961916      5.09      1.4      10216      1104363      5.26      1.4      1.42      0.11      0.08        CURRICULM:        GENERAL      5636      961916      5.09      1.4      10216      1104363      5.26      1.4      1.40      0.10 *      0.13        ACADEMIC      6772      1383415      6.05      1	CATHOLIC	10.3	234807	5.96	1.3	2675	198812	5.99	1.3	1.30	0.03	0.02
NORTHEAST    3581    796146    5.64    1.3    5625    68748    5.86    1.4    1.36    0.22 *    0.16      NORTH CENTRAL    4551    914144    5.40    1.4    8063    865643    5.59    1.5    1.47    0.19 *    0.13      SOUTH    5478    792013    5.64    1.4    9222    916771    5.69    1.5    1.45    0.05    0.04      WEST    2966    522085    5.56    1.4    5103    546024    5.67    1.4    1.42    0.11    0.08      CURRICULUM:	GEOGRAPHIC REGION:											
NORTH CENTRAL      4551      914144      5.40      1.4      8063      865643      5.59      1.5      1.47      0.19 *      0.13        SOUTH      5478      792013      5.64      1.4      9222      916771      5.69      1.5      1.45      0.05      0.04        WEST      2966      522085      5.56      1.4      5103      546024      5.67      1.4      1.42      0.11      0.08        CURRICULUM:	NORTHEAST	3581	796146	5.64	1.3	5625	688748	5.86	1.4	1.36	0.22 *	0.16
SOUTH WEST    5478    792013    5.64    1.4    9222    916771    5.69    1.5    1.45    0.05    0.04      WEST    2966    522085    5.56    1.4    5103    546024    5.67    1.4    1.42    0.11    0.08      CURRICULUM: GENERAL    5636    961916    5.09    1.4    10216    1104363    5.26    1.4    1.40    0.18    0.13      ACADEMIC    6772    1383415    6.05    1.3    10494    1133862    6.36    1.3    1.30    0.31    0.24      VOCATIONAL    4167    678754    5.20    1.3    6897    735252    5.36    1.4    1.35    0.16    0.12      CONTUNITY TYPE:    URBAN    4524    781669    5.48    1.4    6460    605593    5.54    1.4    1.43    0.10    0.07      SUBURBAN    7922    1532178    5.60    1.4    13480    1490958    5.70    1.4    1.43    0.10    0.07      RURAL    3672    637824    5.60    1.4    8073	NORTH CENTRAL	4551	914144	5.40	1.4	8063	865643	5.59	1.5	1.47	0.19 *	0.13
WEST    2966    522085    5.56    1.4    5103    546024    5.67    1.4    1.42    0.11    0.08      CURRICULUM: GENERAL ACADEMIC    5636    961916    5.09    1.4    10216    1104363    5.26    1.4    1.40    0.18    0.13      ACADEMIC    6772    1383415    6.05    1.3    10494    1133862    6.36    1.3    1.30    0.31    0.24      VOCATIONAL    4167    678754    5.20    1.3    6897    735252    5.36    1.4    1.45    0.16    0.12      CONTUNITY TYPE:    URBAN    4524    781669    5.48    1.4    6460    605593    5.54    1.4    1.44    0.06    0.05      SUBURBAN    7922    1532178    5.60    1.4    13480    1490958    5.70    1.4    1.43    0.10    0.07      RURAL    3672    637824    5.60    1.4    8073    920635    5.80    1.4    1.43    0.19    0.13	SOUTH	5478	792013	5.64	1.4	9222	916771	5.69	1.5	1.45	0.05	0.04
CURRICULUM:      GENERAL    5636    961916    5.09    1.4    10216    1104363    5.26    1.4    1.40    0.18    0.13      ACADEMIC    6772    1383415    6.05    1.3    10494    1133862    6.36    1.3    1.30    0.31    0.24      VOCATIONAL    4167    678754    5.20    1.3    6897    735252    5.36    1.4    1.35    0.16    0.12      COMPLINITY TYPE:    URBAN    4524    781669    5.48    1.4    6460    605593    5.54    1.4    1.43    0.06    0.05      SUBURBAN    7922    1532178    5.60    1.4    13480    1490958    5.70    1.4    1.43    0.10    0.07      RURAL    3672    637824    5.60    1.4    8073    920635    5.80    1.4    1.43    0.19    0.13	WEST	2966	522085	5.56	1.4	5103	546024	5.67	1.4	1.42	0.11	0.08
GENERAL    5636    961916    5.09    1.4    10216    1104363    5.26    1.4    1.40    0.18 *    0.13      ACADEMIC    6772    1383415    6.05    1.3    10494    1133862    6.36    1.3    1.30    0.31 *    0.24      VOCATIONAL    4167    678754    5.20    1.3    6897    735252    5.36    1.4    1.35    0.16 *    0.12      CONTRUNTY TYPE:    URBAN    4524    781669    5.48    1.4    6460    605593    5.54    1.4    1.44    0.06    0.05      SUBURBAN    7922    1532178    5.60    1.4    13480    1490958    5.70    1.4    1.43    0.10 *    0.07      RURAL    3672    637824    5.60    1.4    8073    920635    5.80    1.4    1.43    0.19 *    0.13	CURRICULUM:											
ACADEMIC    6772    1383415    6.05    1.3    10494    1133862    6.36    1.3    1.30    0.31 *    0.24      VOCATIONAL    4167    678754    5.20    1.3    6897    735252    5.36    1.4    1.35    0.16 *    0.12      COMPLINITY TYPE:    URBAN    4524    781669    5.48    1.4    6460    605593    5.54    1.4    1.44    0.06    0.05      SUBURBAN    7922    1532178    5.60    1.4    13480    1490958    5.70    1.4    1.43    0.10 *    0.07      RURAL    3672    637824    5.60    1.4    8073    920635    5.80    1.4    1.43    0.19 *    0.13	GENERAL	56 36	961916	5.09	1.4	10216	1104363	5.26	1.4	1.40	0.18 ×	0.13
VOCATIONAL      4167      678754      5.20      1.3      6897      735252      5.36      1.4      1.35      0.16      *      0.12        COMPLINITY      TYPE:      URBAN      4524      781669      5.48      1.4      6460      605593      5.54      1.4      1.44      0.06      0.05      SUBURBAN      7922      1532178      5.60      1.4      13480      1490958      5.70      1.4      1.43      0.10      0.07        RURAL      3672      637824      5.60      1.4      8073      920635      5.80      1.4      1.43      0.19      0.13	ACADEMIC	6772	1383415	6.05	1.3	10494	1133862	6.36	1.3	1.30	0.31 *	0.24
CONTUNITY TYPE: URBAN 4524 781669 5.48 1.4 6460 605593 5.54 1.4 1.44 0.06 0.05 SUBURBAN 7922 1532178 5.60 1.4 13480 1490958 5.70 1.4 1.43 0.10 * 0.07 RURAL 3672 637824 5.60 1.4 8073 920635 5.80 1.4 1.43 0.19 * 0.13	VOCATIONAL	4167	678754	5.20	1.3	6897	735252	5.36	1.4	1.35	0.16 *	0.12
URBAN 4524 781669 5.48 1.4 6460 605593 5.54 1.4 1.44 0.06 0.05 SUBURBAN 7922 1532178 5.60 1.4 13480 1490958 5.70 1.4 1.43 0.10 * 0.07 RURAL 3672 637824 5.60 1.4 8073 920635 5.80 1.4 1.43 0.19 * 0.13	CONTRNITY TYPE:											
SUBURBAN      7922      1532178      5.60      1.4      13480      1490958      5.70      1.4      1.43      0.10      *      0.07        RURAL      3672      637824      5.60      1.4      8073      920635      5.80      1.4      1.43      0.19      *      0.13	URBAN	4524	781669	5.48	1.4	6460	605593	5.54	1.4	1.44	0.06	0.05
RURAL 3672 637824 5.60 1.4 8073 920635 5.80 1.4 1.43 0.19 * 0.13	SUBURBAN	7922	1532178	5.60	1.4	13480	1490958	5.70	1.4	1.43	0.10 ¥	0.07
	RURAL	3672	6 <b>3782</b> 4	5.60	1.4	8073	920635	5.80	1.4	1.43	0.19 *	0.13

\*SIGNIFICANT AT .05 OR LESS



#### CHAPTER VI

#### STUDENTS' ATTITUDES, VALUES, AND BEHAVIORS

In this chapter we describe changes in students' aspirations, attitudes, values, and behaviors. It begins with a discussion of changes in students' educational aspirations and post-high-school plans, their beliefs about their ability to complete college, and the influence which parents, teachers, guidance counselors and friends had on the post-highschool plans. Next we discuss changes in the students' occupational aspirations. This is followed by an analysis of changes in attitudes and values related to careers and to other life goals. Next we explore changes in students' self-esteem and locus of control. The final section deals with changes in behavior, specifically in time spent on homework, participation in extracurricular activities, and course-taking behavior.

#### A. EDUCATIONAL ASPIRATIONS

The seniors were asked, in 1972 and 1980, the highest level of education they planned to attain. The scale ranged from 1 = less than high school to 5 = graduate or professional school. The mean level of education planned by these students was 3.42 in 1972 and 3.45 for 1980, as shown in Table 6-1. This represents a mean level of aspiration for education midway between junior college/post-secondary vocational school and a four-year college (or some post-secondary education but not a college degree). The relative stability of educational aspirations is somewhat illusionary since it represents a very small decrease in male aspirations and a small increase in female aspirations. The sex difference in educational aspirations, evident in 1972, had disappeared by 1980. There was a moderate increase in the educational aspirations of Asian-Americans (the racial/ethnic group with the highest level of educational aspirations). There was a moderate increase in the educational aspirations of students in Catholic schools and in the academic curriculum and smaller increases in the aspirations of students in the general and vocational curricula and in urban and rural schools.

Examination of the interaction of sex and curriculum showed a small statistically significant increase in educational aspirations for males in the academic curriculum. Females in all curricula show greater increases in educational aspirations than do males. There was a moderate decrease in educational aspirations for low SES Hispanics and a small increase in aspirations for high SES Whites. There were moderate increases in educational aspirations for Catholic school students at middle and high SES levels and a small increase for high SES public school students. High SES students in the Northeast, in the West, and in urban and rural communities also showed moderate increases in educational aspirations. There were small to moderate increases in the aspirations of middle and high SES students in 411 curricula.



#### Table 6-1

#### HIGHEST LEVEL OF EDUCATION YOU PLAN TO ATTAIN (1=LESS THAN HIGH SCHOOL; 5=GRADUATE/PROFESSIONAL \$CHOOL)

	NLS 1972				HSB 1980						
	SAMPLE N	NEIGHTED N	MEAN	\$.D.	SAMPLE N	WEIGHTED N	MEAN	\$.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	12285	2307353	3.42	1.0	27429	2956578	3,45	1.0	1.02	0.04	0.04
SEX:											
MALE	5963	113 <b>0</b> 978	3.54	1.0	12563	1362411	3.47	1.1	1.04	-0.07 *	-0.07
FEMALE	6319	1175851	3.30	1.0	13863	1489894	3.46	1.0	0.99	0.16 ×	0.16
SES:											
LON	3059	489999	2.98	1.0	8193	790498	3.01	1.0	0.97	0.02	0.02
MIDDLE	5897	1183646	3.32	0.9	12548	1395723	3.40	1.0	0.97	0.08 ×	0.08
HIGH	3305	629448	3.94	0.8	6066	70987£	4.10	0.9	0.86	0.16 ×	0.19
RACE:											
MITE	10144	2011874	3.43	1.0	19502	2322042	3.45	1.0	1.01	0.02	0.02
BLACK	1110	136813	3.46	1.0	3631	330771	3.53	1.0	1.03	0.07	0.07
ASIAN-AMERICAN	148	21012	3.74	0.9	354	38196	4.18	0.8	0.87	0.44 ×	0.51
AMERICAN INDIAN	109	186 <b>0</b> 4	2.90	1.0	206	21504	3.21	1.0	1.00	0.31	0.31
MEXICAN-AMERICAN	324	43038	3.31	0.9	1852	<b>%9618</b>	3.23	1.0	0.99	-0.03	-0.08
PUERTO RICAN	58	5974	3.47	1.0	302	17761	3.25	1.1	1.06	-0.22	-0.21
OTHER HISPANIC	۲6 ر	12552	3.28	1.0	958	65919	3.33	1.1	1.07	0.05	0.05
SCHOOL TYPE:											
PUBLIC	10964	2035236	3.41	1.0	23938	2658694	3.40	1.0	1.01	-0.00	-0.00
PRIVATE	60	14819	3.82	0.8	845	101332	3.99	1.0	0.97	0.17	0.17
CATHOLIC	860	199125	3.55	1.0	2646	196552	3.84	0.9	0.94	0.29 *	0.31
GEOGRAPHIC REGION:											
NORTHEAST	2746	629475	3.43	1.0	5512	675375	3.55	1.1	1.05	0.11 *	0.11
NORTH CENTRAL	3517	721267	3.35	1.0	7924	852563	3.39	1.0	1.00	0.04	0.04
SOUTH	3899	573524	3.45	1.0	9039	898681	3.40	1.0	1.02	-0.05	-0.05
NEST	2123	383087	3.47	0.9	4954	529959	3.53	1.0	0.97	0.06	0.07
CURRICULUM:											
GENERAL	3896	6777 <b>01</b>	3.05	0.9	10003	1081101	3.16	1.0	0.95	0.11 ×	0.12
ACADEMIC	5710	1179639	3.89	0.8	10358	111964 <b>0</b>	4.11	0.9	0.84	0.22 ×	0.26
VOCATIONAL	2679	450013	2.72	0.8	6685	714287	2.89	0.8	0.82	0.17 ×	0.21
CONTUNITY TYPE:											
URBAN	3240	578819	3.46	1.0	6290	590400	3.53	1.0	1.01	0.08 *	0.08
SUBURBAN	6054	1197451	3.52	1.0	13204	1459473	3.53	1.0	1.01	0.02	0.02
RURA!.	2738	485943	3.16	1.0	7935	906705	3.27	1.0	1.00	0.12 ×	0.12

\*SIGNIFICANT AT .05 OR LESS



In summary, there was little change in the mean level of education planned by the average student in 1972 and in 1980. However, there were moderate increases in the educational aspirations of females, Asian-Americans, Catholic school students, and students in the academic curriculum.

Educational aspirations are a function of many different attitudes, values, and influences. These include students' beliefs about their own intellectual ability and also the influence of parents, teachers, and other key figures. In the next three sections we will examine students' plans for the first year after high school, their beliefs about their ability to do college-level work, and the extent to which parents, teachers, and friends have influenced the students' plans.

B. STUDENTS' PLANS FOR THE FIRST YEAR AFTER HIGH SCHOOL

There is often a gap between high school students' dreams and aspirations and the reality of their immediate plans. Consequently, the students were asked what they planned to do in the first year after they completed high school. The options included several types of post-secondary education, full- and part-time work, military service, and homemaking.

As can be seen in Table 6-2, four-year college was the most frequently mentioned plan, involving 33.6 percent of the 1972 group and 37.8 percent of the 1980 group. Full-time work ranked second in both years, being named as the post-high-school plan for 25.9 percent of the 1972 group and 29.5 percent of the 1980 group. There were increases between 1972 and 1980 both in plans for attending a four-year college (4.2 percentage points) and in plans for full-time work (3.6 percentage points). Fifty-nine percent of the 1972 group planned some form of post-secondary education (45 percent planned academic studies in a four-year college or in a junior college; 14 percent planned vocational studies in a junior college or in a voc-tech institute). Fifty-eight percent of the 1980 group planned some type of post-secondary education (46 percent planned academic studies in a four-year college or in a junior college; 12 percent planned vocational studies in a junior college or in a voc-tech institute). Students planning academic work in a junior college decreased by 2.7 percentage points between 1972 and 1980 as did students planning to attend a voc-tech institute. There was a very small increase in students planning to enter a vocational program in junior college.

Sex differences in students' plans for post-secondary education immediately after high school increased between 1972 and 1980, due primarily to the increased number of females planning to enter a fouryear college. In 1972, nearly equal percentages of males and females (58.4 percent of the males and 59.5 percent of the females) planned to enter some type of post-secondary education immediately after high school. By 1980, the percentage of males with such plans had declined



#### Table 6-2

#### STUDENTS' PLANS FOR FIRST YEAR AFTER HIGH SCHOOL

		College	Jr. College Academic	Jr. College Vocational	<u>Voc-lech</u>	Work <u>Full-Time</u>	Work Fart-lime	Apprentice	Militery	Homemaker	<u>Other</u>
<u>total</u>											
	1972	33.6	11.3	5.4	8.8	25.9	2. 1	2.7	3.4	2.8	4.2
	1980	37.8	8.6	5.8	6.1	29.5	1.9	2.4	3.5	1.2	3.2
<u>SE X</u>											
	Males										
	1972	34.6	11.4	5.2	7.2	24.8	1.9	3.8	6.0	Ú, I	5.0
	1980	37.2	7.1	5, 1	5.4	31.5	1.5	3.4	5.4	0.2	3, 1
	Females				_						
	1972	32.5	11.2	5.5	10.3	27.0	2.2	1.5	0.8	7.7	2.7
	1980	39.2	10.2	6.4	6.7	26.8	2.2	1.6	1.6	2.1	3.3
<u>SES</u>											
	Low										
	1972	18.7	6.9	5.0	11.1	39.1	2.6	2.9	4.5	4.6	4.6
	1980	21.7	6.2	5.8	8.2	41.6	2.7	3.2	5.0	2.2	3.6
	<u>Middle</u>										
	1972	29.2	12.3	6.1	9.8	27.0	2.2	3.1	3.3	2.7	4. 5
	1980	34.3	9.7	6.6	6.9	30.4	1.8	2.6	3.5	1.0	3, 1
	High										
	1972	58.3	13.6	4.2	4.4	10.1	1.3	1.3	2.4	1.0	3, 7
	1980	60.9	9.4	4.7	2.9	14.1	1.1	1.6	2. 1	0.5	2.8
RACE											
	White							_			
	1972	34.8	12.0	5,3	8.6	24.6	2.0	2.7	3.3	2.8	3.9
	1980	38.3	8.8	5.7	5.8	29.8	1.6	2.4	2.9	1.3	3.3
	Black										
	1972	32.1	5.4	4.9	11.6	30.6	2.5	3, 1	3.7	2.0	4.2
	1980	39.0	6.1	5-4	7.5	27.4	3.0	1.8	6.6	0.6	2.5
	<u>Hispanic</u>			10.5		<b></b>		~ ~		1 7	7.0
	1972	23.6	11.6	10.5	7.9	30.3	2.2	2.8	7.7	1.7	2.9
	1980	28.	9.5	6.9	6.7	33.4	3, 1	3.1	4.8	1.3	3.0

\$



BEST COPY AVAILABLE

#### -43-

#### Table 6-2 (continued)

#### STUDENTS' PLANS FUR FIRST YEAR AFTER HIGH SCHOOL

	College	Jr. College <u>Academic</u>	Jr. College <u>Vocational</u>	Voc-Tech	Work F <u>ill-Time</u>	Work <u>Part-</u> lime	Apprentice	<u>Military</u>	Homemaker	<u>Other</u>
SCHOOL TYPE										
Public										
1972	32.6	11.5	5.3	8.8	26.1	2.2	2,8	3.5	2.9	4.4
1980	35.3	8.8	6.0	6.3	31.0	1.9	2.5	3.7	1.3	3.2
Private										
1972	50.2	10.9	11.0	10.8	10.7	3.2	0.0	0.0	2.5	1.2
1980	61.6	5.0	4.4	4.2	13.4	2.0	0.8	1.7	0.8	6.1
<u>Cetholic</u>										
1972	46.7	9.2	5.2	9.4	21.7	1.1	1.6	1.5	1.5	2.2
1980	58.5	8.4	4.1	4.7	17.7	1.4	2.0	1.1	0.5	1.7
CURRICULUM										
Academic										
1972	57.7	14.9	4.8	5.2	9.6	1.1	1.2	2.3	0.9	2.1
1980	66.7	9.8	4.1	2.8	10.2	1.1	1.1	2.0	0.3	1.8
General										
1972	17.6	10.9	5.8	9.7	33.5	3.0	3.9	4.5	4.3	6.8
1980	25.0	9.0	6.2	6.8	37.5	2.3	2.6	4.5	1.8	4.5
Vocationa	1									
1972	7.1	4.4	5.9	14.7	48.2	2.7	3.8	3.9	4.4	5.0
1980	12.5	6.3	7.7	10.2	47.5	2.4	4.1	4.3	1.5	3.5
REGION										
Northeast										
1972	37.1	11.7	4.7	7.8	26.0	1.6	2.6	2.9	1.7	3.8
1980	44.7	7.0	4.4	5.5	25.9	1.9	2.5	4.2	0.8	3.2
North Cent	ral									
1972	33.5	7.8	5.0	10.3	27.5	2.2	3.0	3.4	3.1	4.3
1980	37.4	6.8	5.5	7.2	31.7	1.8	2.6	2.6	1.2	3.1
South										
1972	34.0	10.2	4.2	9.9	27.0	1.9	2.5	3.6	3.0	3.5
1980	35.8	8.2	5.5	6.1	32.0	1.7	2.2	4.0	1.4	2.9

HELT COPY AVAILABLE 110

BEST COPY AVAILABLE
### -44-

# Table 6-2 (continued)

# STUDENTS' PLANS FOR FIRST YEAR AFTER HIGH SCHOOL

	College	Jr. College <u>Acedemic</u>	Jr. College Vocational	<u>Voc-Tech</u>	Work <u>Full-Time</u>	York <u>Part-lime</u>	Apprentice	Military	Homemaker	Uther
<u>REGION</u> (cont.)										
West										
1972	27.4	18.2	8.9	6.0	21.3	2.7	2.4	3.8	3,4	5.9
1980	33.1	14.1	8.6	5,3	25.9	2.2	2.5	3, 1	1.3	3.9
COMMUNITY										
Urban										
1972	35, 5	11.5	5.0	7.7	26.7	1.6	2.7	2.8	2.4	4.0
1980	40.2	?.6	5.9	6.0	27.4	2.4	2.8	3.6	0.9	3.2
Suburb										
1972	37.3	12.6	5.6	7.8	22.7	2.1	2.4	3, 5	2.4	3.9
1980	40.3	9.5	6.0	5.4	27.7	1.6	2.6	2.8	1.0	3.2
Rural										
1972	24.1	7.8	5,3	12.6	31.6	2.5	3.4	4.1	4.1	4.4
1980	32.3	7.8	5.4	7.3	33.7	2.0	2.0	4.5	1.8	3.1



111

.



to 54.8 percent while the percentage of females increased to 62.5 percent. The percentage of students planning to enter an academic program (either in a four-year college or a junior college) showed a similar trend. In 1972, 46.0 percent of the males planned to enter this type of postsecondary education, but by 1980 the percentage had declined to 44.3 percent. For females, however, the percentage planning immediate posthigh-school entrance into acalemic post-secondary education rose from 44.3 percent in 1972 to 49.4 percent in 1980. The percentage of students planning to enter vocational post-secondary education in a junior college or a voc-tech institute immediately after high school declined between 1972 and 1980 for both males and females.

There was an increase of 3.5 percentage points between 1972 and 1980 in students planning to begin full-time work immediately after high school. This increase was due to a sharp rise in the percentage of males with such plans (24.8 percent in 1972 and 31.5 percent in 1980). The percentage of females with immediate plans for full-time work decreased very slightly between 1972 and 1980.

Increases in the percentage of students planning to enter a four-year college or to enter a vocational program in a junior college occurred across all SES levels. Declines in the percentage of students planning to enter an academic program in a junior college or to enter a voc-tech institute are also found across all SES levels. Increases in the percentage of students planning to enter a four-year college also occurred in all racial/ethnic groups. Blacks also showed increases in the percentage of students planning to enter a junior college, but there is a decline in the percentage of Hispanics planning this type of post-secondary education.

There was also an increase, at all SES levels, in the percentage of students planning to work full-time immediately after high school. This increase was also evident for White and Hispanic students; there was a decrease, however, between 1972 and 1980, in Black students planning to enter full-time employment immediately after high school.

It is not surprising that four-year college was the most frequently mentioned post-high-school plan in both 1972 and 1980 for students in the academic curriculum. Plans for this type of post-secondary education increased by 9 percentage points among academic students between 1972 and 1980. However, plans for all other types of post-secondary education showed a decrease among academic students. This netted a very small increase (from 82.6 percent to 83.4 percent) in the percentage of academic curriculum students planning to enter any type of post-secondary education immediately after high school.

Full-time work was the most frequently mentioned post-high-school plan in 1972 and in 1980 for students in the general and the vocational curricula. Plans for work increased among general curriculum students but decreased among vocational students. However, plans for some type

# BEST COPY AVAILABLE



of post-secondary education increased among both general and vocational students. In 1972, 44.0 percent of the general and 32.1 percent of the vocational students had immediate plans for post-secondary education; by 1980 this had increased to 47.0 percent and 36.7 percent. Perhaps most surprising is the fact that there was a decrease, from 20.6 percent in 1972 to 17.9 percent in 1980, in the percentage of vocational curriculum students planning to enter post-secondary vocational education in a junior college or in a voc-tech institute.

This picture of changes in immediate post-high-school plans is fairly consistent with the educational aspirations changes discussed earlier. The increase in females' educational aspirations is matched by their increased plans for participation in post-secondary education. Although there was an increase between 1972 and 1980 in the percentage of students planning to enter four-year colleges, declines in plans for most other forms of post-secondary education resulted in a very small net loss (0.8 percentage points) for post-secondary education participation immediately after high school. The shift to four-year colleges from other types of post-secondary education may be due to changes in the competitiveness of entrance into four-year colleges between 1972 and 1980.

#### C. ABILITY TO COMPLETE COLLEGE

The students were asked, regardless of their educational plans, if they thought that they had the ability to complete college. The scale ranged from 1 = definitely not to 5 = yes, definitely. As shown in Table 6-3, the mean in 1972 was 4.05; in 1980 it was 4.20. This indicates that the average high school students in these groups believed that they had the intellectual ability to complete college. The increase from 1972 to 1980 is significant but has only a small effect size. This increase appears inconsistent with the declining proportion of students in the academic curriculum and with the falling test scores. It is, however, consistent with the rise in high school grades.

The increase in students' belief that they have the ability to complete college was fairly consistent across sex, SES groups, and curricula. White and Black students showed small increases, and there was a large increase among Asian-Americans. Catholic school students showed a moderate increase, while students in public schools showed a small increase. Students from rural communities showed a moderate increase, while the students in urban and suburban communities showed small increases.

#### D. INFLUENCES OF OTHERS ON PLANS FOR AFTER HIGH SCHOOL

The students were asked to indicate the extent to which four groups of significant others influenced the students' post-high-school plans (whether for further education, paid work, or another activity). The scale ranges from 1 = no influence at all to 3 = a great deal. As can be



-46-

#### DO YOU THINK YOU HAVE THE ABILITY TO COMPLETE COLLEGE (1=DEFINITELY NOT; 5=YE\$, DEFINITELY)

NLS 1972 HSB 1980 --------SAMPLE WEIGHTED SAMPLE WEIGHTED POOLED 1980-1972 EFFECT N N MEAN S.D. N N MEAN S.D. S.D. DIFFERENCE SIZE TOTAL 16433 3002090 4.05 1.0 27165 2930326 4.20 1.0 0.99 0.15 × 0.15 SEX: 1.0 1.00 0.9 0.97 MALE 8142 1494648 4.06 1.0 12750 1383934 4.20 0.14 × 0.14 FEMALE 8286 1506424 4.05 1.0 13962 1499831 4.21 0.16 \* 0.17 SES: LOH 4724 727019 1.1 8059 776373 3.87 1.0 12459 1386673 4.20 1.1 1.10 0.9 0.96 0.7 0.73 3.71 0.15 × 0.14 MIDDLE 7813 1533525 4.03 0.17 × 0.18 732597 4.45 6066 HIGH 3844 0.8 711332 4.59 0.15 \* 0.20 RACE: 
 1.0
 0.98

 1.0
 0.96

 0.7
 0.80
12696 2499082 4.08 WHITE 1.0 19457 2316310 4.21 0.13 \* 0.13 2069 249702 4.04 1.0 3487 317414 4.23 BLACK 0.18 × 0.19 356 27654 4.02 38640 4.52 ASIAN-AMERICAN 192 1.0 0.50 × 0.62 204 1.0 1818 1.1 297 1.1 182 AMERICAN INDIAN 30102 3.57 20893 3.95 1.1 1.14 0.38 0.34 548 71714 3.91 1.0 0.99 1.1 1.11 1.0 1.01 MEXICAN-AMERICAN 97356 3.93 0.02 0.02 95 17292 4.07 PUERTO RICAN 9691 3.79 1.11 1.01 0.28 0.09 0.28 0.25 120 18550 3.95 OTHER HISPANIC 62469 4.04 0.09 SCHOOL TYPE: 14736 2665150 4.04 23681 2631565 
 J.0
 1.00

 0.8
 0.81
PUBLIC 1.0 0.13 × 4.17 0.13 67 16549 4.35 861 PRIVATE 0.8 103288 4.49 0.15 0.18 CATHOLIC 1013 232771 4.19 1.0 2623 195472 4.42 0.8 0.85 0.23 \* 0.27 GEOGRAPHIC REGION: 1.0 5530 1.1 7786 1.0 8882 1.0 4967 0.9 0.97 1.0 1.03 1.0 1.00 0.9 0.92 NORTHEAST 3539 789613 4.08 678565 4.27 0.19 × 0.19 NORTH CENTRAL 906230 3.98 4510 839332 4.14 0.16 ¥ 0.16 SOUTH 879003 4.16 5432 785323 4.05 0.11 × 0.11 WEST 2952 520925 4.13 0.13 × 533425 4.26 0.15 CURRICULUM: GENERAL 5569 951276 3.76 1.1 9863 1067170 3.99 1.0 1.05 0.23 × 0.22 ACADEMIC 6776 1385199 4.46 0.7 10338 1118542 4.61 0.6 0.69 0.15 × 0.22 VOCATIONAL 4087 665312 3.63 3.89 0.26 \* 1.1 6579 703008 1.1 1.08 0.24 COMMUNITY TYPE: URBAN 4495 777661 1.0 0.97 0.9 0.95 1.0 6207 4.08 582480 4.22 0.13 × 0.14 SUBURBAN 7877 1523846 4.13 1.0 13108 1450840 4.25 0.12 × 0.12 RURAL 3644 633262 3.87 1.1 7850 897006 4.11 1.0 1.04 0.23 × 0.22

\*SIGNIFICANT AT .05 OR LESS



seen in summary Table 6-4, parents had the greatest influence on students' post-high-school plans in both 1972 and in 1980. In 1972, teachers were the group which had the least influence. By 1980, however, teachers' influence had increased, and guidance counselors were the group with the least influence. Three of the four groups of significant others, teachers, parents and guidance counselors, showed an increase in influence on students' post-high-school plans between 1972 and 1980. The influence of friends and relatives, however, decreased.

Ta	Ь	le	6-	4
Ta	υ.	16	· · ·	-

Influence of S	Significant Ot	hers on Stud	ents' Post-Hi	gh-School Plans
Influences	1972 <u>Me an</u>	1980 <u>Mean</u>	Differ- ence	<u>Effect Size</u>
Parents	2.33	3 2.42	0.09*	0.15
Friends and Relat	ives 2.15	<b>5</b> 2.02	-0.14*	-0.20
Guidance Counseld	or 1.55	5 1.61	0.06*	0.08
Teachers	1.49	1.73	0.23*	0.34

\*Significant at .05 or less

Parents' influence on students' post-high-school plans showed a small increase from 1972 to 1980. (See Table 6-5.) This increase was larger for high and middle SES students than for low SES students. Students in Catholic schools showed a greater increase than public school students. Students from the West showed a greater increase than students from other regions.

Teachers' influence on students' post-high-school plans showed a moderate increase from 1972 to 1980. There was little variation in this increase across sex, racial/ethnic, SES, curriculum, geographic region, school type, or community type. (See Table 6-6.) Cross-tabulations of teacher influence show somewhat larger increases for middle SES Hispanics and for high SES Blacks. High SES students in the academic and vocational curricula consistently showed slightly larger increases in teacher influence.

Although guidance counselors' influence on students post-high-school plans increased very slightly between 1972 and 1980, it reached a small but significant effect size with students in Catholic schools. (See Table 6-7.) Cross-tabulations show small increases for influence of the counselor for students in Catholic schools regardless of SES.

Friends and relatives were the only group of significant others to show any decline in their influence on students post-high-school plans between 1972 and 1980. This decrease is 20 percent of a standard deviation. The decrease was greater for low SES than high SES students. (See Table 6-8.) It was also somewhat greater for Black and other Hispanic



#### INFLUENCE OF PARENTS ON PLANS FOR AFTER HIGH SCHOOL (1=NOT AT ALL; 3=A GREAT DEAL)

SAMPLE    MELENTED    SAMPLE    MELENTED    SAMPLE    MELAN    S.D.    N    N    MELAN    S.D.    SAMPLE    MELAN    S.D.    SAMPLE    MELAN    S.D.    DOLED    1980-1972    EFFECT      TOTAL    16599    3030694    2.33    0.6    27547    2971982    2.42    0.6    0.64    0.09 #    0.15      SEX:    HALE    8352    1509407    2.31    0.7    12690    1379228    2.41    0.6    0.65    0.10 #    0.16      SES:    LON    4764    735239    2.30    0.7    8177    790295    2.35    0.7    0.68    0.06 #    0.08      NITTE    12810    2520880    2.31    0.6    12616    1402109    2.40    0.6    0.64    0.19 #    0.14      BLACK    2090    2.5352    2.53    0.6    3610    32957    2.57    0.6    0.64    0.09 #    0.14      BLACK    2090    2.43		NLS 1972				HSB 19	980					
TOTAL    16599    3030694    2.33    0.6    27547    2971982    2.42    0.6    0.64    0.09 *    0.15      SEX: MALE    B3232    1509407    2.31    0.7    12690    1379228    2.41    0.6    0.65    0.10 *    0.14      SES:           0.16    0.65    0.10 *    0.14      SES:            0.66    0.66    0.66    0.06 *    0.08      MIDDLE              MITE                  MITE		SAMPLE N	NEIGHTED N	MEAN	S.D.	SAMPLE N	MEIGHTED N	MEAN	S.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
SEX: MALE 522 1509407 2.31 0.7 1379228 2.41 0.6 0.65 0.10 * 0.16 FEMALE 8362 1350270 2.35 0.7 12690 2.44 0.6 0.65 0.06 * 0.10 * 0.16      SES: LOM 4764 735239 2.30 0.7 8177 790295 2.35 0.7 0.68 0.06 * 0.08 * 0.08 MIDDLE 7694 1354370 2.30 0.6 12616 1402209 2.40 0.6 0.64 0.10 * 0.16      MITE 12810 2520080 2.31 0.6 12616 1402209 2.40 0.6 0.64 0.09 * 0.13 * 0.22      MITE 12810 2520080 2.31 0.6 19581 2332409 2.40 0.6 0.64 0.09 * 0.13 * 0.22      MITE 12810 2520080 2.43 0.6 3001 329657 2.57 0.6 0.63 0.04 0.07      MITE 12810 2520080 2.43 0.7 355 30717 2.49 0.6 0.64 0.23 * 0.35      MITE 12810 2520800 2.33 0.6 3010 329657 2.57 0.6 0.63 0.04 0.07      MITE 12810 2520800 2.33 0.6 3010 329657 2.57 0.6 0.64 0.23 * 0.35      MITE 12810 2520800 2.33 0.7 355 2102 2.52 0.6 0.64 0.23 * 0.35      MITE 14070 2589030 2.33 0.7 295 17066 2.36 0.7 0.70 0.04 0.05      MITE 122 138044 2.43 0.6 949 64763 2.46 0.6 0.65 0.09 * 0.14      MITE 14070 2689030 2.33 0.6 24028 2471355 2.42 0.6 0.65 0.09 * 0.35      MITE 14070 2689030 2.33 0.6 24028 2471357 2.49 0.6 0.61 0.13 * 0.21      MITE 14070 2689030 2.33 0.6 24028 2471357 2.49 0.6 0.61 0.13 * 0.21      MITE 1599 2608030 2.33 0.6 24028 2471357 2.49 0.6 0.65 0.08 * 0.12      MITE 1280 229579 2.32 0.7 5569 664337 2.39 0.6 0.65 0.08 * 0.12      MITE 12970 2689332 2.39 0.6 0.65 0.08	TOTAL	16599	3030694	2.33	0.6	27547	29 <b>719</b> 82	2.42	0.6	0.64	0.09 *	0.15
MALE    6232    1509407    2.31    0.7    12690    1379228    2.41    0.6    0.65    0.10 *    0.16      FEMALE    8362    1520270    2.35    0.6    13957    1500182    2.44    0.6    0.64    0.09 *    0.14      SES:    ION    4784    735239    2.30    0.7    8177    790295    2.35    0.7    0.68    0.06 *    0.08      MIDLE    7694    1549370    2.30    0.6    12616    106299    2.40    0.6    0.57    0.13 *    0.12      RACE:    #    MITE    12810    2520080    2.31    0.6    19581    2332409    2.40    0.6    0.64    0.09 *    0.14      MACK    2090    2.43    0.7    355    36717    2.49    0.6    0.64    0.09 *    0.14      MACK    2090    2.43    0.7    2.55    1.70    0.6    0.64    0.65    0.09 *    0.15	SEX:											
FENALE    8362    1520270    2.35    0.6    13957    1500182    2.44    0.6    0.64    0.09 *    0.14      SES:             0.64    0.09 *    0.14      SES:         0.6    12616    1402905    2.35    0.7    0.66    0.06 *    0.08      MITE        0.6    12616    1402905    2.35    0.6    0.59    0.13 *    0.22      MITE         0.6    0.64    0.09 *    0.14      BLCK           0.6    0.64    0.09 *    0.14      BLCK             0.14       MITE	MALE	8232	1509407	2.31	0.7	12690	1379228	2.41	0.6	0.65	0.10 ¥	0.16
SES:    LOM    4784    735239    2.30    0.7    8177    790295    2.35    0.7    0.66    0.06 *    0.08      HIDDLE    78949    154377    2.30    0.6    12616    1402909    2.40    0.6    0.64    0.10 *    0.10 *    0.10 *    0.10 *    0.10 *    0.10 *    0.10 *    0.10 *    0.22      RACE:	FEMALE	8362	1520270	2.35	0.6	13957	1500182	2.44	0.6	0.64	0.09 *	0.14
LOM    4764    735239    2.30    0.7    0177    790295    2.35    0.7    0.68    0.06    #    0.08      MIDDLE    7894    1549370    2.30    0.6    12616    1402909    2.40    0.6    0.64    0.10    #    0.16      MISH    3857    734919    2.42    0.6    6097    714494    2.55    0.6    0.59    0.13    0.22      RACE:    HITE    12810    2520080    2.31    0.6    19581    2332409    2.40    0.6    0.64    0.09    #    0.14      MARTRICAN    191    27609    2.43    0.7    356    38717    2.49    0.6    0.64    0.66    0.09    M    MERICAN    135    27640    2.40    0.7    1030    97952    2.50    0.6    0.64    0.63    0.09    0.15      MERICAN-AMERICAN    136    2130    0.7    2.52    0.6    0.65    0.09    0.16	SES:											
MIDDLE HIGH    7894    1549370    2.30    0.6    12616    1402900    2.40    0.6    0.64    0.10    *    0.16      MIRH    3857    734919    2.42    0.6    6097    714494    2.55    0.6    0.59    0.13    0.22      RACE:        0.22      MHITE    12810    2520080    2.31    0.6    19581    2332409    2.40    0.6    0.64    0.09    0.14      BLACK    2098    253552    2.53    0.6    3610    329657    2.57    0.6    0.64    0.09    0.14      BLACK    2098    2.43    0.7    356    36717    2.49    0.6    0.64    0.03    0.018      MERICIAN INDIAN    106    31059    2.43    0.7    1330    97995    2.50    0.6    0.64    0.03    0.015      PUERTO RICAN    96    9764    2.33    0.6    24028    2.67135	LON	4784	735239	2.30	0.7	8177	790295	2.35	0.7	0.68	0.06 ×	0.08
HIGH    3857    734919    2.42    0.6    6007    714494    2.55    0.6    0.59    0.13    #    0.22      RACE:    Imite    12810    252080    2.31    0.6    19581    2332409    2.40    0.6    0.64    0.09    #    0.14      BLACK    2090    253552    2.53    0.6    3610    3232457    2.57    0.6    0.64    0.09    #    0.14      BLACK    2090    253552    2.53    0.6    3610    3232457    2.57    0.6    0.64    0.09    #    0.14      MIERICALININDIAN    166    31059    2.29    0.7    205    21102    2.52    0.6    0.64    0.06    0.64    0.23    #    0.35      PUERTO RICAN    95    77480    2.33    0.7    295    17046    2.36    0.7    0.70    0.04    0.05    0.05      PUERTO RICAN    9    9764    2.33    0.6	MIDDLE	7894	1549370	2.30	0.6	12616	1402909	2.40	0.6	0.64	0.10 ¥	0.16
RACE:    NITE    12810    252080    2.31    0.6    19581    2332409    2.40    0.6    0.64    0.09    *    0.14      BLACK    2098    253552    2.53    0.6    3610    329657    2.57    0.6    0.63    0.04    0.07      ASIAN-AMERICAN    191    27609    2.43    0.7    356    36717    2.49    0.6    0.64    0.64    0.06    0.09      AMERICAN-AMERICAN    186    31059    2.29    0.7    205    21102    2.52    0.6    0.64    0.23    0.35      PUERTO RICAN    96    9764    2.33    0.7    295    17046    2.36    0.7    0.70    0.04    0.03    0.05      SCHOOL TYPE:	HI <del>ch</del>	3857	734919	2.42	0.6	6097	714494	2.55	0.6	0.59	0.13 *	0.22
MHITE    12810    2520080    2.31    0.6    19581    2332409    2.40    0.6    0.64    0.09 *    0.14      BLACK    2098    253552    2.53    0.6    3610    329657    2.57    0.6    0.64    0.09 *    0.14      ASIAN-AHERICAN    191    27609    2.43    0.7    356    36717    2.49    0.6    0.64    0.23 *    0.35      MERICAN INDIAN    186    31059    2.29    0.7    205    21102    2.52    0.6    0.64    0.23 *    0.35      PUERIO RICAN    952    72480    2.40    0.7    1295    17066    2.36    0.7    0.70    0.06    0.65    0.09    0.15      PUERIO RICAN    96    9764    2.33    0.7    295    103070    2.46    0.6    0.61    0.13    0.25      SCHOOL TYPE:    PUERIO RICAN    1026    235732    2.32    0.6    2651355    2.42    0.6    0.61	RACE:											
BLACK    2098    253552    2.53    0.6    3610    329657    2.57    0.6    0.63    0.04    0.07      ASIAN-AMERICAN    191    27609    2.43    0.7    356    36717    2.49    0.6    0.64    0.66    0.09      AMERICAN INDIAN    18059    2.29    0.7    205    21102    2.52    0.6    0.64    0.64    0.09    0.15      PUERTO RICAN    552    72480    2.40    0.7    1830    97985    2.50    0.6    0.64    0.64    0.09    0.15      PUERTO RICAN    96    9764    2.33    0.7    295    17046    2.36    0.7    0.70    0.04    0.05      OTHER HISPANIC    122    18844    2.43    0.6    24028    2671355    2.42    0.6    0.65    0.09    +    0.14      PUBLIC    14878    2689030    2.33    0.6    2658    197557    2.47    0.6    0.61    0.13	MHITE	12810	2520080	2.31	0.6	19581	2332409	2.40	0.6	0.64	0.09 *	0.14
ASIAN-AHERICAN  191  27609  2.43  0.7  356  30717  2.49  0.6  0.64  0.06  0.09    AMTRICAN  INDIAN  186  31059  2.29  0.7  205  21102  2.52  0.6  0.64  0.23  #  0.35    PUERTO RICAN  96  9764  2.33  0.7  295  17046  2.36  0.7  0.70  0.04  0.05    OTHER HISPANIC  122  18844  2.43  0.6  949  64763  2.46  0.6  0.64  0.03  0.05    SCHOOL TYPE:  PUBLIC  14878  2689030  2.33  0.6  24028  2671355  2.42  0.6  0.61  0.13  0.21    CATHOLIC  1026  235732  2.32  0.6  2658  197557  2.47  0.6  0.61  0.15  0.21    MORTH CENTRAL  4552  91877  2.31  0.6  7569  684337  2 39  0.6  0.64  0.07  0.11    NORTH CENTRAL  4552  91877  2.31 <td< td=""><td>BLACK</td><td>2098</td><td>253552</td><td>2.53</td><td>0.6</td><td>3610</td><td>329657</td><td>2.57</td><td>0.6</td><td>0.63</td><td>0.04</td><td>0.07</td></td<>	BLACK	2098	253552	2.53	0.6	3610	329657	2.57	0.6	0.63	0.04	0.07
AMERICAN INDIAN  186  31059  2.29  0.7  205  21102  2.52  0.6  0.64  0.23 *  0.35    MEXICAN-AMERICAN  552  72480  2.40  0.7  1830  97985  2.50  0.6  0.65  0.09  0.15    PUERTO RICAN  96  972  1830  97985  2.50  0.6  0.65  0.09  0.15    OTHER HISPANIC  122  18844  2.43  0.6  949  64763  2.46  0.6  0.64  0.03  0.05    SCHOOL TYPE:  PUBLIC  14878  2689030  2.33  0.6  24028  2671355  2.42  0.6  0.61  0.13  0.21    CATHOLIC  1026  235732  2.32  0.6  2658  197557  2.47  0.6  0.61  0.15  0.25    GEOGRAPHIC REGION:  NORTHEAST  3599  801829  2.32  0.7  5569  684337  2.39  0.6  0.65  0.08 *  0.12    NORTHEAST  3599  801829  2.32  0.7  5513363	ASIAN-AMERICAN	191	27609	2.43	0.7	356	38717	2.49	0.6	0.64	0.06	0.09
MEXICAN-AMERICAN PUERTO RICAN    552    72480    2.40    0.7    1830    97985    2.50    0.6    0.65    0.09    0.15      PUERTO RICAN    96    9764    2.33    0.7    295    17046    2.36    0.7    0.70    0.04    0.05      OTHER HISPANIC    122    18844    2.43    0.6    949    64763    2.46    0.6    0.64    0.03    0.05      SCHOOL TYPE:    PUBLIC    14478    2689930    2.33    0.6    24028    2671355    2.42    0.6    0.65    0.09    0.14      PRIVATE    67    16549    2.35    0.6    861    103070    2.48    0.6    0.61    0.13    0.21      CATHOLIC    1026    235732    2.32    0.7    5569    684337    2.39    0.6    0.65    0.08    0.12      NORTH CENTRAL    4552    913877    2.31    0.6    7925    853363    2.38    0.6    0.64    0.10	AMERICAN INDIAN	186	31059	2.29	0.7	205	21102	2.52	0.6	0.64	0.23 *	0.35
PUERTO RICAN OTHER HISPANIC    96    9764 122    2.33 18844    0.7 2.43    0.7 0.6    0.7 949    0.70 64763    0.70 2.46    0.70 0.64    0.04 0.03    0.05 0.05      SCHOOL TYPE: PUBLIC    14878    2689030    2.33 2.32    0.6    24028    2671355    2.42    0.6    0.65    0.09 *    0.14      PRIVATE    67    16549    2.35    0.6    661    103070    2.48    0.6    0.61    0.13    0.21      CATMOLIC    1026    235732    2.32    0.6    2658    197557    2.47    0.6    0.61    0.13    0.21      NORTHEAST    3599    801829    2.32    0.7    5569    684337    2.39    0.6    0.65    0.08 *    0.12      NORTHEAST    3599    801829    2.32    0.7    5569    684337    2.39    0.6    0.64    0.07 *    0.11      SOUTH    5476    791492    2.40    0.6    9042    898078    2.49    0.6    0.64	MEXICAN-AMERICAN	552	72480	2.40	0.7	1830	97985	2.50	0.6	0.65	0.09	0.15
OTHER HISPANIC    122    18844    2.43    0.6    949    64763    2.46    0.6    0.64    0.03    0.05      SCHOOL TYPE: PUBLIC    14878    2689030    2.33    0.6    24028    2671355    2.42    0.6    0.65    0.09 *    0.14      PRIVATE    67    16549    2.35    0.6    661    103070    2.48    0.6    0.61    0.13    0.21      CATHOLIC    1026    235732    2.32    0.6    2658    197557    2.47    0.6    0.61    0.13    0.21      CATHOLIC    1026    235732    2.32    0.6    2658    197557    2.47    0.6    0.61    0.15    0.25      SECOGRAPHIC REGION:	PUERTO RICAN	96	9764	2.33	0.7	295	17046	2.36	0.7	0.70	0.04	0.05
SCHOOL TYPE:  PUBLIC  14878  2689030  2.33  0.6  24028  2671355  2.42  0.6  0.65  0.09 *  0.14    PRIVATE  67  16549  2.35  0.6  861  103070  2.48  0.6  0.61  0.13  0.21    CATHOLIC  1026  235732  2.32  0.6  2658  197557  2.47  0.6  0.61  0.15 *  0.25    SECOGRAPHIC REGION:	OTHER HISPANIC	122	18844	2.43	0.6	<del>9</del> 49	64763	2.46	0.6	0.64	0.03	0.05
PUBLIC  14878  2689030  2.33  0.6  24028  2671355  2.42  0.6  0.65  0.09 *  0.14    PRIVATE  67  16549  2.35  0.6  861  103070  2.46  0.6  0.61  0.13  0.21    CATHOLIC  1026  235732  2.32  0.6  2658  197557  2.47  0.6  0.61  0.15 *  0.25    GEOGRAPHIC REGION:	SCHOOL TYPE:											
PRIVATE  67  16549  2.35  0.6  861  103070  2.48  0.6  0.61  0.13  0.21    CATHOLIC  1026  235732  2.32  0.6  2658  197557  2.47  0.6  0.61  0.13  0.21    GEOGRAPHIC REGION:	PUBLIC '	14878	2689030	2.33	0.6	24028	2671355	2.42	0.6	0.65	0.09 ¥	0.14
CATHOLIC  1026  235732  2.32  0.6  2658  197557  2.47  0.6  0.61  0.15 **  0.25    GEOGRAPHIC REGION: NORTHEAST  3599  801829  2.32  0.7  5569  684337  2.39  0.6  0.65  0.08 **  0.12    NORTHEAST  3599  801829  2.32  0.7  5569  684337  2.39  0.6  0.64  0.07 **  0.11    NORTH CENTRAL  4552  913877  2.31  0.6  7925  853383  2.38  0.6  0.64  0.07 **  0.11    SOUTH  5476  791492  2.40  0.6  9042  898078  2.49  0.6  0.64  0.10 **  0.15    NEST  2972  523495  2.28  0.7  5011  536184  2.42  0.6  0.65  0.14 **  0.22    CURRICULUM:  GENERAL  5639  963148  2.26  0.7  10019  1084871  2.37  0.7  0.66  0.11 **  0.17    ACADEMIC  6797  1389247  2.40  0.6 </td <td>PRIVATE</td> <td>67</td> <td>16549</td> <td>2.35</td> <td>0.6</td> <td>861</td> <td>103070</td> <td>2.48</td> <td>0.6</td> <td>0.61</td> <td>0.13</td> <td>0.21</td>	PRIVATE	67	16549	2.35	0.6	861	103070	2.48	0.6	0.61	0.13	0.21
GEOGRAPHIC REGION:      NORTHEAST    3599    801829    2.32    0.7    5569    684337    2 39    0.6    0.65    0.08 *    0.12      NORTHEAST    4552    913877    2.31    0.6    7925    853383    2.38    0.6    0.64    0.07 *    0.11      SOUTH    5476    791492    2.40    0.6    9042    898078    2.49    0.6    0.64    0.10 *    0.15      NEST    2972    523495    2.28    0.7    5011    536184    2.42    0.6    0.65    0.14 *    0.22      CURRICULUM:    GENERAL    5639    963148    2.26    0.7    10019    1084871    2.37    0.7    0.66    0.11 *    0.17      ACADEMIC    6797    1389247    2.40    0.6    10410    1125843    2.51    0.6    0.61    0.11 *    0.18      VOCATIONAL    4162    677997    2.28    0.7    6719    717871    2.37    <	CATHOLIC	1026	235732	2.32	0.6	2658	197557	2.47	0.6	0.61	0.15 ¥	0.25
NORTHEAST  3599  801829  2.32  0.7  5569  684337  2 39  0.6  0.65  0.08 *  0.12    NORTH CENTRAL  4552  913877  2.31  0.6  7925  853383  2.38  0.6  0.64  0.07 *  0.11    SOUTH  5476  791492  2.40  0.6  9042  898078  2.49  0.6  0.64  0.10 *  0.15    NEST  2972  523495  2.28  0.7  5011  536184  2.42  0.6  0.65  0.14 *  0.22    CURRICULUM:	GEOGRAPHIC REGION:											
NORTH CENTRAL  4552  913877  2.31  0.6  7925  853383  2.38  0.6  0.64  0.07 *  0.11    SOUTH  5476  791492  2.40  0.6  9042  898078  2.49  0.6  0.64  0.10 *  0.15    NEST  2972  523495  2.28  0.7  5011  536184  2.42  0.6  0.65  0.14 *  0.22    CURRICULUM:  6ENERAL  5639  963148  2.26  0.7  10019  1084871  2.37  0.7  0.66  0.11 *  0.17    ACADEMIC  6797  1389247  2.40  0.6  10410  1125843  2.51  0.6  0.61  0.11 *  0.18    VOCATIONAL  4162  677997  2.28  0.7  6719  717871  2.37  0.7  0.66  0.09 *  0.13    COMMUNITY  TYPE:  URBAN  4538  783616  2.34  0.6  6302  591323  2.46  0.6  0.64  0.12 *  0.19    SUBURBAN  7935  1536531  2.33<	NORTHEAST	3599	801829	2.32	0.7	5569	684337	2 39	0.6	0.65	0.08 ×	0.12
SOUTH NEST  5476  791492  2.40  0.6  9042  898078  2.49  0.6  0.64  0.10 *  0.15    NEST  2972  523495  2.28  0.7  5011  536184  2.42  0.6  0.65  0.14 *  0.22    CURRICULUM: GENERAL  5639  963148  2.26  0.7  10019  1084871  2.37  0.7  0.66  0.11 *  0.17    ACADEMIC  6797  1389247  2.40  0.6  10410  1125843  2.51  0.6  0.61  0.11 *  0.18    VOCATIONAL  4162  677997  2.28  0.7  6719  717871  2.37  0.7  0.66  0.09 *  0.13    COMMUNITY TYPE:  URBAN  4538  783616  2.34  0.6  6302  591323  2.46  0.6  0.64  0.12 *  0.19    SUBURBAN  7935  1536531  2.33  0.6  13286  1471342  2.42  0.6  0.64  0.09 *  0.14    RURAL  3670  637501  2.31  0.6  <	NORTH CENTRAL	4552	913877	2.31	0.6	7925	853383	2.38	0.6	0.64	0.07 ×	0.11
NEST  2972  523495  2.28  0.7  5011  536184  2.42  0.6  0.65  0.14 *  0.22    CURRICULUM: GENERAL  5639  963148  2.26  0.7  10019  1084871  2.37  0.7  0.66  0.11 *  0.17    ACADEMIC  6797  1389247  2.40  0.6  10410  1125843  2.51  0.6  0.61  0.11 *  0.18    VOCATIONAL  4162  677997  2.28  0.7  6719  717871  2.37  0.7  0.66  0.09 *  0.13    COMMUNITY TYPE:  URBAN  4538  783616  2.34  0.6  6302  591323  2.46  0.6  0.64  0.12 *  0.19    SUBURBAN  7935  1536531  2.33  0.6  13286  1471342  2.42  0.6  0.64  0.09 *  0.14    RURAL  3670  637501  2.31  0.6  7959  909317  2.41  0.6  0.65  0.09 *  0.14	SOUTH	5476	791492	2.40	0.6	9042	898078	2.49	0.6	0.64	0.10 *	0.15
CURRICULUM:    GENERAL  5639  963148  2.26  0.7  10019  1084871  2.37  0.7  0.66  0.11 *  0.17    ACADEMIC  6797  1389247  2.40  0.6  10410  1125843  2.51  0.6  0.61  0.11 *  0.18    VOCATIONAL  4162  677997  2.28  0.7  6719  717871  2.37  0.7  0.66  0.09 *  0.13    COMMUNITY TYPE:  urban  4538  783616  2.34  0.6  6302  591323  2.46  0.6  0.64  0.12 *  0.19    SUBURBAN  7935  1536531  2.33  0.6  13286  1471342  2.42  0.6  0.64  0.09 *  0.14    RURAL  3670  637501  2.31  0.6  7959  909317  2.41  0.6  0.65  0.09 *  0.14	NEST	2972	523495	2.28	0.7	5011	536184	2.42	0.6	0.65	0.14 <del>*</del>	0.22
GENERAL  5639  963148  2.26  0.7  10019  1084871  2.37  0.7  0.66  0.11 *  0.17    ACADEMIC  6797  1389247  2.40  0.6  10410  1125843  2.51  0.6  0.61  0.11 *  0.18    VOCATIONAL  4162  677997  2.28  0.7  6719  717871  2.37  0.7  0.66  0.09 *  0.13    COMPUNITY TYPE:  URBAN  4538  783616  2.34  0.6  6302  591323  2.46  0.6  0.64  0.12 *  0.19    SUBURBAN  7935  1536531  2.33  0.6  13266  1471342  2.42  0.6  0.64  0.09 *  0.14    RURAL  3670  637501  2.31  0.6  7959  909317  2.41  0.6  0.65  0.09 *  0.14	CURRICULUM:											
ACADEMIC  6797  1389247  2.40  0.6  10410  1125843  2.51  0.6  0.61  0.11  *  0.18    VOCATIONAL  4162  677997  2.28  0.7  6719  717871  2.37  0.7  0.66  0.09  *  0.13    COMMUNITY TYPE:  URBAN  4538  783616  2.34  0.6  6302  591323  2.46  0.6  0.64  0.12  *  0.19    SUBURBAN  7935  1536531  2.33  0.6  13286  1471342  2.42  0.6  0.64  0.09  *  0.14    RURAL  3670  637501  2.31  0.6  7959  909317  2.41  0.6  0.65  0.09  *  0.14	GENERAL	5639	963148	2.26	0.7	10019	1084871	2.37	0.7	0.66	0.11 *	0.17
VOCATIONAL    4162    677997    2.28    0.7    6719    717871    2.37    0.7    0.66    0.09 *    0.13      COMMUNITY TYPE:    URBAN    4538    783616    2.34    0.6    6302    591323    2.46    0.6    0.64    0.12 *    0.19      SUBURBAN    7935    1536531    2.33    0.6    13286    1471342    2.42    0.6    0.64    0.09 *    0.14      RURAL    3670    637501    2.31    0.6    7959    909317    2.41    0.6    0.65    0.09 *    0.14	ACADEMIC	6797	1389247	2.40	0.6	10410	1125843	2.51	0.6	0.61	0.11 ×	0.18
CONMUNITY TYPE: URBAN 4538 783616 2.34 0.6 6302 591323 2.46 0.6 0.64 0.12 * 0.19 SUBURBAN 7935 1536531 2.33 0.6 13286 1471342 2.42 0.6 0.64 0.09 * 0.14 RURAL 3670 637501 2.31 0.6 7959 909317 2.41 0.6 0.65 0.09 * 0.14	VOCATIONAL	4162	677997	2.28	0.7	6719	717871	2.37	0.7	0.66	0.09 ¥	0.13
URBAN 4538 783616 2.34 0.6 6302 591323 2.46 0.6 0.64 0.12 * 0.19 SUBURBAN 7935 1536531 2.33 0.6 13286 1471342 2.42 0.6 0.64 0.09 * 0.14 RURAL 3670 637501 2.31 0.6 7959 909317 2.41 0.6 0.65 0.09 * 0.14	CONNUNITY TYPE:											
SUBURBAN    7935    1536531    2.33    0.6    13286    1471342    2.42    0.6    0.64    0.09    #    0.14      RURAL    3670    637501    2.31    0.6    7959    909317    2.41    0.6    0.65    0.09    #    0.14	URBAN	4538	783616	2.34	0.6	6302	591323	2.46	0.6	0.64	0.12 ¥	0.19
RURAL 3670 637501 2.31 0.6 7959 909317 2.41 0.6 0.65 0.09 * 0.14	SUBURBAN	7935	1536531	2.33	0.6	13286	1471342	2.42	0.6	0.64	0.09 *	0.14
	RURAL	3670	637501	2.31	0.6	7959	909317	2.41	0.6	0.65	0.09 *	0.14

\*SIGNIFICANT AT .05 OR LESS



#### INFLUENCE OF TEACHERS ON PLANS FOR AFTER HIGH SCHOOL (1=NOT AT ALL; 3=A GREAT DEAL)

	NL3 1972				HSB 1980						
	SAMPLE N	NE 1 GHTED N	MEAN	5.D.	SAMPLE N	WEIGHTED N	MEAN	\$.D.	POOLED S.D.	1980-1972 DIFFERENCL	EFFECI SIZE
TOTAL	16444	3005323	1.49	0.7	26910	2907565	1.73	0.7	0.69	0.23 *	0.34
SEX:											
MALE	8150	1495673	1.47	0.7	12389	1349949	1.69	0.7	0.68	0.23 *	0.33
FEMALE	8289	1508634	1.52	0.7	13665	1469508	1.75	0.7	0.70	0.24 *	0.34
SES:											
LOH	4704	724912	1.54	0.7	7935	767333	1.77	0.7	0.71	0.23 ×	0.33
MIDDLE	7838	1538073	1.49	0.7	12375	1378183	1.71	0.7	0.69	0.22 *	0.32
HIGH	3844	732170	1.46	0.7	5993	701701	1.72	0.7	0.67	0.25 *	0.38
RACE:											
MHITE	12740	2505498	1.46	0.7	19258	2293844	1.68	0.7	0.67	0.22 *	0.32
BLACK	2041	246892	1.75	0.7	3451	314543	1.96	0.7	0.74	0.21 *	0.29
ASIAN-AMERICAN	191	27609	1.59	0.7	349	38009	1.78	0.7	0.71	0.19	0.27
AMERICAN INDIAN	182	30600	1.41	0.6	201	20727	1.85	0.7	0.64	0.45 ×	0.69
MEXICAN-AMERICAN	542	71046	1.58	0.7	1760	93796	1.84	0.7	0.73	0.26 *	0.36
PUERTO RICAN	93	9415	1.68	0.7	282	16597	1.92	0.7	0.70	0.24	0.34
OTHER HISPANIC	118	18323	1.47	0.6	913	62255	1.78	0.7	0.72	0.31 ×	0.44
SCHOOL TYPE:											
PUBLIC	14739	2666 327	1.49	0.7	23446	2612212	1.73	0.7	0.69	0.24 ×	0.34
PRIVATE	67	16549	1.53	0.7	842	100132	1.79	0.7	0.70	0.26	0.36
CATHOLIC	1022	234852	1.43	0.6	2622	195221	1.63	0.6	0.64	0.20 *	0.31
GEOGRAPHIC REGION:											
NORTHEAST	3561	794710	1.45	0.7	5455	671104	1.70	0.7	0.68	0.25 *	0.37
NORTH CENTRAL	4518	907612	1.48	0.7	7796	840568	1.68	0.7	0.68	0.20 ×	0.29
SOUTH	5414	783096	1.55	0.7	8781	873505	1.79	0.7	0.71	0.24 *	0.34
NEST	2951	519906	1.48	0.7	4878	522388	1.72	0.7	0.69	0.24 *	0.35
CURRICULUM:											
GENERAL	5575	953828	1.40	0.6	9764	1056616	1.61	0.7	0.66	0.20 ¥	0.31
ACADEMIC	6759	1381081	1.54	0.7	10260	1110539	1.81	0.7	0.70	0.27 ×	0.39
VOCATIONAL	4109	670112	1.53	0.7	6512	699108	1.78	0.7	0.71	0.25 *	0.35
CONTUNITY TYPE:											
URBAN	4488	776133	1.52	0.7	6106	574267	1.80	0.7	0.71	0.27 ¥	0.38
SUBURBAN	7876	1526107	1.47	0.7	12987	1439263	1.70	0.7	0.68	0.22 *	0.33
RURAL	3639	632408	1.51	0.7	7817	894035	1.73	0.7	0.17	0.22 *	0.32

**#SIGNIFICANT /T .05 OR LESS** 

.....

. . .

\_\_\_\_



#### INFLUENCE OF GUIDANCE COUNSELOR ON PLANS FOR AFTER HIGH SCHOOL (1=NOT AT ALL; 3=A GREAT DEAL)

	NLS 1972			HSB 1980							
	SAMPLE N	NEIGHTED N	MEAN	\$.D.	SAMPLE N	WEIGHTED N	MEAN	\$.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	16456	3008556	1.55	0.6	26859	2901445	1.61	0.7	0.67	0.06 ¥	0.08
SEX:											
MALE	8156	1497530	1.55	0.6	19357	1345910	1 69	07	0 44	0 04 ×	A 47
FEMALE	8295	1510009	1.55	0.6	13648	1467970	1.62	0.7	0.67	0.04 ×	0.07
SES:											
LON	4706	795997	1 4 7		701 2	7///04				• •-	
MIDDLE	7844	1510705	1.03	0.7	1716	104420	1.00	0.7	0.70	0.05 *	0.08
HIGH	3846	732895	1.50	0.6	5988	702410	1.59	0.7	0.66 0.63	0.05 * 0.05 *	0.08
DACF:											
MITE	12750	2509227	1 51	<b>n</b> 4	10274	2201505		• •	• • •	• • • •	
BLACK	2040	246853	1.51	0.0	17634	2291303	1.55	0.6	0.64	0.04 ¥	0.06
ASIAN-AMERICAN	191	27584	1.00	0.7	3430	312504	1.41	0.7	0.74	0.02	0.03
AMERICAN INDIAN	184	30745	1.02	0.0	347	30274	1.00	0.7	0.69	0.03	0.05
MEXICAN-AMERICAN	543	71102	1.02	0.7	170	20530	1.72	0.7	0.70	0.09	0.13
PUERTO RICAN	272	0473	1.73	0.7	1/50	73024	1.70	0.7	0.72	0.05	0.07
OTHER HISPANIC	118	18323	1.60	0.7	905	61051	1.67	0.7	0.72	0.05	0.08
SCHOOL TYPE:											••••
PUBLIC	34744	344 9744	3 64	• •	07/0/	0/ 0700/					
PRIVATE	47	2000/04	1.20	0.0	23406	2607886	1.60	0.7	0.67	0.05 ×	0.07
CATHOLIC	1024	235319	1.40	0.5	833 2620	98385 195174	1.61	0.7	0.65	0.20	0.32
								•••		0.27 ~	V.L3
DECORAPHIC REGION.											
NORTH CENTRAL	3571	797108	1.58	0.7	5461	671454	1.64	0.7	0.67	0.06 ¥	0.09
PORTI CENTRAL	4523	908008	1.54	0.6	7777	838501	1.58	0.7	0.65	0.04 ×	0.07
JUTT	5410	782451	1.60	0.7	8756	870479	1.64	0.7	0.69	0.04 ×	0.06
MEJI	2952	520989	1.46	<b>U.6</b>	4865	521010	1.55	0.7	0.64	0.09 ¥	0.14
CURRICULUM:											
GENERAL	5571	953597	1.48	0.6	9758	1056400	1 55	07	0 65	0 09 ×	0 12
ACADEMIC	6765	1383087	1.60	0.7	10254	1109712	1 66	0.7	0.65	0.00 ×	0.12
VOCATIONAL	4119	671569	1.55	0.6	6474	694700	1.60	0.7	0.67	0.05 ×	0.08
CONTUNITY TYPE:											
URBAN	4491	776443	1.56	0.7	6087	571443	1 65	07	0 49	0 00 *	
SUBURBAN	7880	1527350	1.53	0.6	12981	3/1441	1.05	0.7	V.07 0 45	U.UY #	U.14
RURAL	3639	633091	1.59	0.6	7701	A00011	1 62	0.7	0.05	0.05 *	0.07
	/					W7W744	4.UC	v./	V.0/	0.05	U.U5

\*SIGNIFICANT AT .05 OR LESS

118

\_\_\_\_



# INFLUENCE OF FRIENDS AND RELATIVES ON PLANS FOR AFTER HIGH SCHOOL (1=NOT AT ALL; 3=A GREAT DEAL)

	NLS 1972					HSB 19	980				
	SAMPLE N	WEIGHTED N	MEAN	<b>S</b> .D.	SAMPLE N	WEIGHTED N	MEAN	\$.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	16539	3021095	2.15	0.7	27211	2937981	2.02	0.7	0.69	-0.14 *	-0.20
SEX:											
MALE	8202	1504182	2.11	0.7	12505	1361642	1.95	0.7	0.69	-0.16 ¥	-0.23
FEMALE	8332	1515896	2.20	0.7	13830	1486238	2.08	0.7	0.69	-0.12 *	-0.18
SES:											
LON	4750	730637	2.22	0.7	8064	780219	2 64	07	0 69	-0 18 ¥	-0 27
MIDDLE	7871	1544839	2.15	0.7	12486	1389331	2 01	0 7	0.69	-0 14 *	-0.20
HIGH	3857	734924	2.10	0.7	6037	707263	2.01	0.7	0.69	-0.09 *	-0.13
RACE :											
MHITE	12790	2515773	2 13	07	19422	2312520	1 00	07	0 4 9	-0 17 *	-0.20
BLACK	2071	250157	2 36	0 7	15762	320266	2 13	0.7	0.07	-0.13 ×	-0.20
ASIAN-AMERICAN	191	27609	2.20	07	354	38737	2 07	0.7	0.07	-0.22 *	-0.32
AMERICAN INDIAN	184	30888	2.23	0.7	201	20230	2 15	0.6	0.70	-0.09	-0.17
MEXICAN-AMERICAN	545	71464	2.26	0.6	1795	95992	2 12	0.0	0.67	-0.08	-0.12
PUERTO RICAN	95	9676	2.28	0.7	286	16579	2 05	07	0.00	-0.24	-0 11
OTHER HISPANIC	121	18565	2.28	0.7	918	62192	1.99	0.7	0.69	-0.29 *	-0.42
SCHOOL TYPE:											
PUBLIC	14824	2680571	2 15	07	23722	2639559	2 01	07	0 4 9	_0 14 ×	-0.20
PRIVATE	67	16549	2 08	0.6	252	101676	2 05	0.7	0.07	-0.14 ×	-0.20
CATHOLIC	1025	235344	2.15	0.7	2637	196746	2.05	0.7	0.69	-0.10 *	-0.05
GEOGRAPHIC REGION:											
NORTHEAST	3585	799069	2 09	07	5493	675161	1 00	07	0 4 9	-0.11 *	. 0 14
NORTH CENTRAL	4533	910451	2.16	0 7	7865	847785	1 99	0.7	0.00	-0.11 ×	-0.10
SOUTH	5455	788591	2.24	0.7	8906	885681	2 07	0.7	0.07	-0.17 ×	-0.25
WEST	2966	522984	2.12	0.7	4947	529353	2.03	0.7	0.70	-0.09 *	-0.13
CURRICULUM:											
GENERAL	5612	959102	2.16	0.7	0804	1071226	2 01	07	0 69	_0 15 *	-0.22
ACADEMIC	6791	1387828	2 14	0.7	10327	1117890	2 03	0.7	0.07	-0.13 *	-0.22
VOCATIONAL	4135	673862	2.18	0.7	6603	706846	2.01	0.7	0.70	-0.17 *	-0.18
CONMUNITY TYPE:											
URBAN	4516	780089	2.17	0.7	6188	581332	2.06	0.7	0 70	-0 11 +	-0 14
SUBURBAN	7917	1533553	2.14	0.7	13125	1453997	2 00	0 7	0.70	-0.11 *	-0.10
RURAL	3656	635233	2.17	0.7	7899	902652	2.01	0.7	0.69	-0.14 *	-0.20
· · · · · · · · · · · ·						/		•. /	<b>V</b> · V 7	-4.10 %	-0.63

\*SIGNIFICANT AT .05 OR LESS

119



# **BEST COPY AVAILABLE**

students than for those from other racial/ethnic groups. The decrease was greater in the North Central and Southern regions than in the Northeast or West and greater in rural than in urban or suburban communities. Cross-tabulations show that the decrease is largest for low SES students regardless of race, school type, curriculum, or community type.

-48-

In summary, parents, teachers, and guidance counselors all showed an increase between 1972 and 1980 in their influence on students' posthigh-school plans. Friends and relatives showed a decrease in influence, especially among low SES students and among students in the North Central and Southern regions. Despite these changes, parents, friends and relatives continue to be more influential in students' post-high-school plans than teachers or guidance counselors. Teachers showed a larger increase from 1972 to 1980 in influence than did parents or guidance counselors, and this increase shows relatively little variation across groups, regions, school and community type, or curriculum. One possible interpretation of these data is that, between 1972 and 1980, students began to rely less on their peers and more on adults for advice about their post-high-school plans. The data also suggest that teachers may be assuming a larger role in counseling students, whether formally or informally.

#### E. OCCUPATIONAL ASPIRATIONS

The students were asked in 1972 and in 1980 to indicate the type of job they expected or planned to have at age 30. In both groups, more than 40 percent of the students indicated that they planned to enter professional jobs "such as accountant, artist, clergyman, dentist, physician, registered nurse, engineer, lawyer, scientist, librarian, writer, social worker, actor/actress, athlete, politician, or school teacher." (See Table 6-9.) No other occupational category attracted such a large proportion of students. Clerical work "such as bank teller, bookkeeper, secretary, typist, mail carrier, or ticket agent" was the second most popular career goal, attracting 14 percent of the students in 1972 and 10 percent of the students in 1980. Each of the other occupational categories (craftsman, farmer, homemaker, laborer, military, operative, proprietor, protective service, sales service, and technical work) attracted fewer than 10 percent of the students. Craft occupations were the third most popular career choice in both 1972 and in 1980.

Between 1972 and 1980, aspirations for managerial or administrative occupations rose most rapidly, by 4.0 percentage points, from being selected by 3.1 percent of the 1972 students to being selected by 7.1 percent of the 1982 students. Aspirations for careers as business proprietors or managers rose by 2.2 percentage points, from 1.8 percent of the 1972 students to 4.0 percent of the 1980 students. Aspirations for clerical careers declined by 4.4 percentage points.

Professional occupations were the most popular choice for both females and males in both 1972 and in 1980. The second most frequently selected occupational category for males was craftsman, "such as baker,

-49-
------

#### TYPE OF WORK STUDENT WANTS TO DO AT AGE 30

		Professional	<u>Clerical</u>	<u>Crafts</u>	<u>Technical</u>	Managerial	<u>Proprietor</u>	Service	Sales	<u>Other</u>
TOTA	L GROUP									
	1972	45.4	14.2	7.5	6.6	3.1	1.8	4.2	3.0	14 2
	198D	43.4	9.8	8.3	8.2	7.1	4.0	3.5	2.1	13.6
					-					17.0
<u>SEX</u>										
	Males	41.0			• •		_			
	1972	41.8	1.9	15.1	8.8	5.1	3.2	1.6	2.7	19.8
	1980	28.8	1.3	15.7	10.5	7.9	5.6	0.7	2.0	17.5
	Females	AQ Q	96 G	0 5					_	
	1774	40.0	27.7	U. 7	4.6	1.3	0	5.7	3.3	8.8
	1960	40.7	1/./	1.1	6.0	6.5	2.5	6.1	2.2	9.2
SES										
	Low									
	1972	30.7	21.4	9.3	6.8	2.4	1.3	6.7	3.3	18. 1
	1 <b>98</b> 0	32.5	15.4	10.8	7.9	5.3	3.0	5.1	2.0	18.D
	<u>Middle</u>									
	1972	41.8	15.4	8.3	7.3	2.9	1.6	4.4	3.2	15.1
	1980	41.5	9.9	9.0	9.3	6.7	4.1	3.6	2.0	13.9
	High									
	1972	63.8	6.2	4.6	5.3	4.1	2.6	2.1	2.5	8.8
	1980	58.4	4.5	4.7	6.8	9.5	4.4	1.7	2.1	7.9
RACE										
	White									
	1972	46.1	13.1	7.7	6.5	3.1	1.9	4.3	3, 1	16 2
	1980	43.9	9.3	8.6	7.9	7.0	4.2	3.5	2 1	14.4
									2.1	17.7
	Black									
	1972	40.0	26.5	3.6	8.7	3.3	1.1	2.8	1.9	12.1
	1980	42.6	12.5	6.0	10.2	8.3	3.1	3.8	2.5	11.0
	<u>Hispan</u> ic									
	1972	38.2	23.4	7.6	7.8	3.0	1.6	4.5	2.0	11.9
	1980	38.9	12.5	<b>9.</b> 0	9.0	5.2	3.6	3.8	2.1	15.9

BEST COPY AVAILABLE



BEST COPY AVAILABLE

-50-
------

### Table 6-9 (continued)

TYPE OF WORK STUDENT WANTS TO DO AT AGE 30

	Professional	<u>Clerical</u>	Crafts	Technical	Managerial	Proprietor	Service	Sales	<u>Other</u>
CURRICULUM									
Academi	c								
1972	65.0	5.5	3.5	6.8	3.4	1.3	2.4	2.4	9.7
1980	64.1	4.0	2.8	8.4	7.9	2.8	1.5	1.4	7.1
General									
1972	34.0	12.0	11.2	6.4	2.7	2.8	6.6	3.7	20.6
1980	36.4	8.8	10.1	7.6	7.0	4.9	4.8	2.7	17.7
Vocatio	nel_								
1972	12.9	38.9	12.3	6.6	2.9	1.7	5.5	3.6	15.6
1980	22.3	20.3	14.3	8.8	6.0	4.5	4.5	2.4	16.9
SCHOOL TYPE									
Public									
1972	45.0	13.8	7.8	6.6	3.1	1.8	4.4	3.1	14.4
1980	42.0	10.1	8.8	8.4	7.0	3.9	3.6	2.1	14.1
<u>Private</u>	2								
1972	58.4	11.5	6.9	3.4	1.7	0.0	3.1	0.0	15.0
1980	58.1	6.0	4.7	5.2	7.5	7.2	2.2	1.1	8.0
<u>Catholi</u>	<u>.c</u>								
1972	52.7	15.4	5.0	6.6	3.1	1.7	3.0	2.7	9.8
1980	56.3	8.3	4.2	7.2	8.2	3.3	2.9	2.1	7.5
REGION									
Northea	ist								
1972	49.4	14.2	6.7	6.1	3.2	1.7	3.9	3.0	11.8
1980	46.9	9.5	7.6	7.7	8.1	4.2	3.0	1.6	11.4
<u>North C</u>	Central								
1972	41.5	14.5	8.7	6.9	2.9	1.6	5.5	3.7	14.7
1980	42.5	8.9	9.4	7.8	7.1	3.5	4.1	2.3	14.4
South									
1972	44.0	16.4	6.5	7.0	3.4	2. 1	3,1	3.0	14.6
1980	41.4	11.9	7.8	8.8	6.6	3.9	3.5	2.3	13.8

BEST COPY AVAILABLE



# -51-

## Table 6-9 (continued)

TYPE OF WORK STUDENT WANTS TO DO AT AGE 30

	<b>Professional</b>	<u>Clerical</u>	<u>Crafts</u>	Technical	<u>Managerial</u>	<u>Proprietor</u>	Service	Sales	Other
<u>REGION</u> (cont.)									
West									
1972	48.3	10.1	8.3	6-6	2.9	1.8	4.2	2.0	15.8
1980	44.5	8.4	8.3	8.6	6.6	4.4	3.5	2.1	13.6
COMMUNITY									
<u>Urban</u>									
1972	45.4	16.4	6.7	7.1	2.8	1.9	4.0	3.3	12.4
1980	46.9	11.4	7.6	8.8	7.1	3.7	3.2	1.8	9.5
Suburban	<u>1</u>								
1972	50.0	12.3	6.7	6.3	3.4	1.9	3.7	3.1	12.6
1980	45.7	8.7	7.8	8.5	7.6	4.2	3.2	2.3	12.0
Rural									
1972	35.2	15.4	10.5	6.6	2.8	1.4	5.5	2.4	20.2
1980	38.0	10.6	9.6	7.3	6.2	3.8	4.2	2.0	18.3





.

automobile mechanic, machinist, painter, plumber, telephone installer, or carpenter," attracting about 15 percent of the male respondents in each year. For females, clerical work was the second most frequently selected occupational category, attracting 25 percent of the 1972 females and 18 percent of the 1980 respondents. Males showed the largest 1972 to 1980 decline in aspirations for a professional career, going from 41.8 percent in 1972 to 38.8 percent in 1980. The occupational category with the largest increase for males was proprietor, att:acting 3.2 percent of the 1972 group and 5.6 percent of the 1980 group. For females, the largest decline was in aspirations for clerical work, going from 25.5 percent of the 1972 female respondents .0 17.7 percent of the 1980 females. The largest increase in occupational aspirations among females was for managerial careers, rising from 1.3 percent of the 1972 females to 6.5 percent of the 1980 females.

Professional occupations were also the most frequently mentioned occupational aspiration by students at all socioeconomic levels. However, only about 30 percent of the low SES students aspired toward a professional occupation, as contrasted with about 60 percent of the high SES students. Clerical occupations were the second most popular choice for low and middle SES students in both 1972 and 1980 and for high SES students in 1972. By 1980, however, managerial occupations had become the second most popular choice for high SES students, being chosen by 9.5 percent of the group. Aspirations for professional occupations declined by 5.4 percentage points among high SES students between 1972 and 1980, while plans for managerial occupations rose by 4.4 percentage points. Among middle SES students, aspirations for clerical occupations decreased by 5.5 percentage points, and plans for managerial occupations increased by 3.8 percentage points. Aspirations for jobs in clerical occupations declined 6.0 percentage points among low SES students while aspirations for jobs in managerial occupations rose to 2.9 percent in this group.

Results for White, Black and Hispanic students showed similar patterns. Professional occupations were the first choice of 38 percent to 46 percent of the students in each of these three racial/ethnic groups, and clerical occupations were the area with the second highest level of interest. White students showed a slight decline between 1972 and 1980 in their plans for professional occupations while Black and Hispanic students showed a slight increase. Plans for jobs in clerical occupations decreased in all three groups; the decreases were larger for the Black and Hispanic students than for the White students. Crafts occupations were the third most popular choice for White students in both 1972 and 1980. Black and Hispanic students, however, placed technical occupations as their third choice.

It is no surprise to find that almost two-thirds of the 1972 and 1980 students in the academic curriculum aspired to professional occupations. It is somewhat surprising, however, to find that slightly more than a third of the students in the general curriculum had similar aspirations. Clerical occupations were the first choice of vocational students in 1972 but, by 1980, more than 20 percent of these students

BEST CORY AVAILABLE



planned or expected to have a job in a professional occupation by the time they reached age 30. Technical occupations were the second most frequently mentioned area by academic students in both 1972 and in 1980. In 1972, clerical occupations were the second ranked choice among general curriculum students. By 1980, however, craft occupations had gained second place, primarily because of a 3.2 percentage point decline in general curriculum students' aspirations for clerical occupations.

Aspirations for jobs in professional occupations were higher among private and Catholic school students than among public school students, but the professions remained the first choice of students in all three types of schools. Clerical occupations were the second choice of students in all three types of schools in 1972 and for public and Catholic school students in 1980. In 1980, however, private school students placed managerial occupations as their second choice.

The analysis of occupational aspirations by region and by community type follows the pattern already evident. In each group, professional occupations are the dominant choice, although the professionals are chosen less often by rural students than by those in urban or suburban communities. Clerical occupations were the second choice for all regions and for all community types in 1972 and for most regions by community uppes in 1980. By 1980, however, some variations emerged with students in the North Central region placing craft occupations in second position while those in the West ranked technical occupations as their second highest choice.

In summary, high school students in 1972 and in 1980 were most likely to aspire to a job in a professional occupation. Clerical occupations were the second most popular occupational choice in 1972, but by 1980 there was evidence of declining interest in this type of work and rising interest in managerial and technical work. The orestionnaire grouping of a large number of professional occupations makes it impossible to tease apart the likely variations among this type of work, such as the declining interest in teaching and growing interest in law and medicine which has been reported both in the popular press and in other studies of students' occupational aspirations.

#### F. CAREER VALUES

The student questionnaire asked, in both 1972 and 1980, about the importance of attitudes and values which affect career plans. The scale ranged from 1 = Not important to 3 = Very important. The results are summarized in Table 6-10. As can be seen, although 411 of these career values increased in importance between 1972 and 1980, their rank ordering did not change. The two highest ranked career values showed increases of small effect size (probably because of ceiling effect). The remaining career values showed moderate-sized increases.



-53-

The value placed on having work that is important and interesting increased at about the same rate across sex groups and across curriculum, region, and community type. (See Tables 6-11 to 6-16.) The increase in the importance placed on working with sociable, friendly people was greater for males than for females, and for students of high rather than low SES. The importance placed on the freedom to make one's own decisions increased more for females than for males; it also increased more for

nonpublic than for public school students. The increase was greater for Whites than for Blacks in all SES groups. The increase in the value placed on freedom to make one's own decisions had the largest effect size, across all groups, of all the career-related values.

# Table 6-10

# Changes in Values Related to Career Plans - 1972 and 1980

Value	Mean <u>1972</u>	Mean 1980	Differ- ence	Effect Size
Work that seems important and interesting	2.77	2.88	0.07*	0.16
Meeting and working with sociable, friendly people	2.49	2.61	0.12*	0.20
Freedom to make own decisions	2.32	2.58	0.26*	0.43
Job security and permanence	2.24	2.50	0.26*	0.38
Good income to start or in a few years	2.12	2.35	0.23*	0.34
Previous work experience in the area	1.73	2.01	0.28*	0.36

\*Significant at .05 or less

The importance placed on job security also increased more for females than for males. By 1980, both sexes placed equal value on job security. The increase was greater for high SES than low SES students. In 1972, the importance placed on this value decreased from low to high SES groups; by 1980, the importance was similar in all three SES groups. Students in the academic curriculum and those from suburban communities showed larger increases than those in other curricula and in other types of communities. The general effect of the changes in the students' view of the importance of job security was to make all groups more similar in 1980 than they had been in 1972. The increase was somewhat less for students in the vocational curriculum, who placed the highest importance on this value in 1972, than for students in other curricular areas.

The value placed on having a career with a good income increased more among females than among males and more among high SES students than low SES students. Cross-tabulations by sex and curriculum show the

		NLS 1972			HSB 1980						
	SAMPLE N	WE IGHTED N	MEAN	\$.D.	SAMPLE N	WEIGHTED N	MEAN	\$.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	16259	2975979	2.77	0.5	26688	2881077	2.84	0.4	0.43	0.07 ¥	0.16
SEX:											
MALE	8037	1477860	2.72	0.5	12442	1769403	3 70				
FEHALE	8217	1497102	2.81	0.4	13741	1332001	2.77	0.5	0.47	0.07 *	0.15
				•••		14//137	2.00	0.4	0.30	0.0/ #	0.19
SES:											
LON	4647	717850	2.72	0.5	7889	760768	2 80	0 5		A AA	
MIDDLE	7762	1524311	2.77	0.5	12293	1348840	2 84	0.5	0.40	0.00 #	0.17
HIGH	3300	724827	2.82	0.4	5945	698973	2 88	0.4	0.45	0.0/ #	0.18
				••••	3703	0,0073	2.00	V.4	0.39	0.00 #	0.16
RACE:											
HHITE	12633	2485861	2.77	0.5	19188	2284638	2 85	0 4	0 42	A A7 ¥	
BLACK	1993	241946	2.75	0.5	3384	307604	2 81	0.4	0.42		0.18
ASIAN-AHERICAN	189	27370	2.79	0.5	349	307074	2 41	0.5	0.47	0.05 *	0.11
AMERICAN INDIAN	180	30453	2.71	0.5	204	20972	2.01	0.4	0.45	0.03	0.06
MEXICAN-AMERICAN	531	69215	2.76	0 5	1740	20032	2./2	0.5	0.53	0.01	0.03
PUERTO RICAN	94	9540	2 7n	0.5	1/07	74374	2.01	0.4	0.45	0.05	0.10
OTHER HISPANIC	113	17238	2 72	0.5	291	10400	2.82	0.4	0.47	0.12	0.26
		27230		0.5	901	00.000	2.76	0.5	0.50	0.04	0.09
SCHOOL TYPE:											
PUBLIC	14573	2641506	9 77	0 5	27254	0503343		• •			
PRIVATE	66	16285	2 82	0.5	63634	250//41	2.84	0.4	0.44	0.07 ¥	0.16
CATHOLIC	1010	231101	2 77	0.7	030	100035	2.84	0.4	0.42	0.02	0.05
			L.//	V.2	23 70	142205	2.87	0.4	0.40	0.10 *	0.24
GEOGRAPHIC REGION:											
NORTHEAST	3532	789515	2 7A	0 6	E447		0.05	• •			
NORTH CENTRAL	4485	901714	2 77	0.5	344/	009200	2.85	0.4	0.43	0.07 ¥	0.16
SOUTH	5342	773103	5 7L	0.5	/000	825832	2.84	0.4	0.42	0.08 ×	0.18
WEST	2900	511667	2 74	0.5	60 70	861628	2.82	0.4	0.45	0.06 ¥	0.13
	2700	31104/	L./0	0.9	47.7	524330	2.85	0.4	0.43	0.08 <del>×</del>	0.19
CURRICULUM:											
GENERAL	5482	OTADIO	9 79	0 E	0/ 70	34/5335					
ACADEMIC	6705	1171476	2.00	0.5	9670	1045//5	2.81	0.4	0.47	0.09 ¥	0.19
VOCATIONAL	4071	13/10/3	2.02	0.4	10192	1103813	2.89	0.3	0.37	0.07 *	0.19
		004105		V+ 2	0441	689790	2.80	0.4	0.47	0.08 ¥	0.17
CONTUNITY TYPE:											
URBAN	6462	76 051 7	9 70	A E							
SUBURBAN	7817	107913	2.70 2.78	V. 5	6048	569179	2.84	0.4	0.43	0.06 *	0.14
HURAL	701/	13143/1	2./0	0.5	12904	1427652	2.85	0.4	0.42	0.07 ¥	0.16
	2014	030693	2.75	0.5	7736	884246	2.82	0.4	0.45	0.09 *	0.21

# IMFORTANCE TO CAREER PLANS: WORK THAT SEEMS IMPORTANT AND INTERESTING TO ME (1=NOT IMPORTANT; 3=VERY IMPORTANT)

Table 6-11

\*SIGNIFICANT AT .05 OR LESS



IMPORTANCE	TÜ	CAREER	PLANS:	MEETING	AND	HORKING	WITH	SOCIABLE,	FRIENDLY	PEOPLE
			(1=NOT	IMPORTAL	NT; 🗄	3=VERY I	MPORT/	ANT)		

\_\_\_\_\_

	NL\$ 1972					HSB 1	980				
	SAMPLE	WEIGHTED			SAMPLE	NEIGHTED			POOLED	1980-1972	EFFECT
	N	N	MEAN	5.9.	N	N	MEAN	S.D.	S.D.	DIFFERENCE	SIZE
TOTAL	16269	2976408	2.49	0.6	26710	2884009	2.61	0.6	0.60	0.12 *	0.20
SEX:											
MALE	8044	1479396	2.36	0.7	12444	1352436	2.52	0.6	0.63	0.16 *	0.25
FEMALE	8220	1495996	2.62	0.6	13758	1480273	2.70	0.5	0.54	0.07 *	0.14
SES:											
LOW	4662	719522	2.54	0.6	7907	762895	2.62	0.6	0.59	0.08 ¥	0.14
MIDDLE	7768	1525735	2.49	0.6	12296	1369352	2.62	0.6	0.60	0.12 *	0.21
HIGH	3791	722488	2.44	0.7	5963	699028	2.60	0.6	0.61	0.15 *	0.25
JACE:											
WHITE	12630	2485475	2.48	0.6	19188	2285533	2.61	0.6	0.60	0.13 *	0.22
BLACK	2002	242631	2.62	0.6	3399	309237	2.64	0.6	0.57	0.01	0.03
ASTAN-AMERICAN	190	27456	2.43	0.7	351	37980	2.63	0.6	0.61	0.20 ¥	0.33
AMERICAN INDIAN	180	30297	2.48	0.7	203	20803	2.57	0.6	0.63	0.09	0.15
MEXICAN-AMERICAN	535	69598	2.63	0.5	1771	94996	2.64	0.6	0.56	0.00	0.01
PUEDTO DICAN	92	92 92	2.63	0.5	290	16576	2.65	0.6	0.58	0.02	0.04
OTHER HISPANIC	111	17039	2.46	0.8	903	60682	2.66	0.6	0.59	0.20	0.33
SCHOOL TYPE:											
PUBLIC	14584	2642360	2.49	0.6	23277	2590496	2.62	0.6	0.60	0.12 *	0.21
PRIVATE	66	16285	2.50	0.6	835	99957	2.53	0.6	0.62	0.03	0.05
CATHOLIC	1006	230374	2.49	0.6	2598	193555	2.63	0.6	0.58	0.14 *	0.23
GEOGRAPHIC REGION:											
NOPTHEAST	3528	7882 52	2.49	0.6	5455	670086	2.61	0.6	0.60	0.12 *	0.20
NORTH CENTRAL	4485	901719	2.48	0.6	7678	827211	2.61	0.6	0.60	0.13 *	0.22
SOUTH	5352	774130	2.54	0.6	8706	862879	2.62	0.6	0.59	0.08 ¥	0.13
WEST	2904	512308	2.44	0.7	4871	523833	2.61	0.6	0.61	0.17 *	0.28
CLRPICULUM:											
GENERAL	5498	941754	2.49	0.6	9681	1046815	2.63	0.6	0.60	0.13 ¥	0.22
ACABEMIC	6693	1369575	2.46	0.6	10187	1104335	2.59	0.6	0.61	0.12 *	0.20
VOCATIONAL	4077	664777	2 56	0.6	6463	691307	2.64	0.6	0.58	0.09 ×	0.15
CONTRINITY TYPE:											
URBAN	4444	769792	2.53	0.6	6054	569265	2.63	0.6	0.59	0.10 *	0.17
SUBURBAN	7822	1514351	2.48	0.7	12912	1429243	2.61	0.6	0.60	0.13 ×	0.22
RURAL	3624	631505	2.49	0.6	7744	885501	2.61	0.6	0.59	0.12 *	0.20

\*SIGNIFICANT AT .05 OR LESS

\_

-----

-----

ERIC Full first Provided by ERIC

#### IMPORTANCE TO CAREER PLANS: FREEDOM TO MAKE MY OWN DECISIONS (1=NOT IMPORTANT; 3=VERY IMPORTANT)

	NL\$ 1972				HSB 1980						
	SAMPLE N	WEIGHTED N	MEAN	\$.D.	SAMPLE	WEIGHTED N	MEAN	s.D.	POOLED S.D.	1980-1972 DIFFERENCE	EFFECT SIZE
TOTAL	16180	2%1363	2.32	0.7	26682	2880465	2.58	0.6	0.60	0.26 *	0.43
SEX:											
MALE	8018	1473568	2 17	0 4	19477			• •			
FEMALE	A157	1484770	2.37	0.0	12437	1351/3/	2.5/	0.6	0.60	0.20 *	0.33
	<b>Q</b> 237	1400777	/	0.7	12/34	14///45	2.60	0.6	0.60	0.32 #	0.54
SES:											
LOW	4617	712879	2.30	0.7	7901	761969	2 E4	0 4	0 43		
MIDDLE	7728	1517691	2.30	0.7	12284	1368004	2 57	0.6	0.01	0.20 *	0.45
HIGH	3785	721715	2.38	0.6	5954	498085	2.37	0.6	0.01	0.26 *	0.44
					<i></i>	070003	2.03	V.5	0.90	0.25 #	U.92
RACE:											
MHITE	12581	2475411	2.31	0.7	19179	2283721	2.58		0 40	A 47 ×	
BLACK	1969	238713	2.45	0.6	3397	308920	2 41	0.0	0.00	0.2/ ₩	0.45
ASIAN-AMERICAN	189	27370	2.29	0.7	350	17841	2 28	0.6	0.57	0.10 #	0.2/
AMERICAN INDIAN	182	30614	2.41	0.7	202	206.02	2 41	0.0	0.37	0.29 #	0.50
MEXICAN-AMERICAN	528	68485	2.37	0.6	1745	04700	2.01	0.5	0.00	0.20	0.34
PUERTO RICAN	92	9325	2.44	0.6	288	34423	2 4 2	0.6	0.00	0.19 #	0.33
OTHER HISPANIC	112	17139	2.22	0.7	901	10461	2.06	0.5	0.57	0.18	0.32
				•••	701	04354	2.3/	V.0	0.90	0.36 🕈	0.61
SCHOOL TYPE:											•
PUBLIC	14506	2628926	2.33	0.7	23253	258701E	3 EA	• •			
PRIVATE	64	15785	2.74	0.7	874	230/013	2.30	0.0	U.60	0.26 #	0.43
CATHOLIC	1009	230819	2.26	0.7	2505	100157	2.30 9 En	0.0	0.60	0.44 *	0.73
				•••	6373	173673	2.39	0.5	0.59	0.33 #	0.56
GEOGRAPHIC REGION:											
NORTHEAST	3502	782198	2.29	0.7	5448	448748	2 50				
NORTH CENTRAL	4465	898179	2.32	0.6	7648	824366	2.37 9 EL	0.0	0.61	0.30 +	0.49
SOUTH	5318	770352	2.34	0.7	8697	842420	2.30 9 Ee	0.0	0.00	0.24 *	0.41
NEST	2895	510634	2 34	0 7	6860	002427 233344	2.30	0.6	0.61	0.24 #	0.39
		320034		•	400 7	253144	2.60	0.5	0.59	0.26 #	0.44
CURRICULUM:											
GENERAL	5456	935250	2 11	07	04.03	304/550		<b>.</b> .	• • • •		-
ACADEMIC	6665	1363700	2.33	0.7	7001	1040334	C.30	0.6	0.60	0.25 #	6.41
VOCATIONAL	405A	662110	2.10	0.7	TOT 10	1102073	2.37	U.6	0.60	0.26 #	0.43
				•••	6443	907//T	c.50	0.6	0.61	Q.28 ¥	0.47
CONTUNITY TYPE:											
LIRBAN	6623	766417	9 34	. 7	4055	E4 0776		• •			
SUBURBAN	777A	1506126	5 IS	0.7	2000	36495	2.37	0.6	0.61	0.25 *	0.42
RURAL	3608	428749	2.JE 2 10	0./	12009	1420004	2.58	0.6	0.60	0.26 *	0.44
	30.00	060302	C. JU	V./	//59	885049	2.57	0.6	0.60	0.27 #	0.45

٠

\*SIGNIFICANT AT .05 OR LESS



#### IMPORTANCE TO CAREER PLANS: JOB SECURITY AND PERMANENCE (1=NOT IMPORTANT; 3=VERY IMPORTANT)

	NLS 1972					HSB 19	980				
	SAMPLE N	NEIGHTED N	MEAN	<b>S.D.</b>	SAMPLE N	NEIGHTED N	MEAN	\$.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	16128	2953536	2.24	0.7	266 35	2877261	2.50	0.6	0.67	0.26 *	0.38
SEX:											
MALE	7993	1471388	2.29	0.7	12431	1351111	2.50	0.6	0.67	0.21 ×	C.31
FEMALE	81 30	1481131	2.20	0.7	13696	1474258	2.50	0.6	0.67	0.30 *	0.45
SES:											
LON	4588	710127	2.31	0.7	7877	760304	2.51	0.6	0.66	0.20 *	0.30
MICD! .	7721	1515868	2.26	0.7	12269	1366564	2.52	0.6	0.66	0.26 *	0.39
HIGH	3773	719353	2.14	0.7	5953	697954	2.46	0.7	0.70	0.31 *	0.45
RACE:											
WHITE	12571	2472891	2.23	0.7	19162	2281485	2.49	0.6	0.67	0.26 *	0.39
BLACK	1946	235966	2.41	0.7	3371	307278	2.57	0.6	0.63	0.17 ¥	0.26
ASIAN-AMERICAN	190	27456	2.25	0.7	349	37802	2.57	0.6	0.64	0.33 *	0.51
AMERICAN INDIAN	179	30228	2.32	0.7	202	20681	2.49	0.6	0.65	0.17	0.25
MEXICAN-AMERICAN	522	68035	2.36	0.7	1770	94978	2.51	0.6	0.64	0.15 ¥	0.24
PUERTO RICAN	90	9232	2.30	0.8	287	16397	2.53	0.7	0.68	0.23	0.33
OTHER HISPANIC	109	16417	2.21	0.7	897	60520	2.52	0.6	0.61	0.30 *	0.50
SCHOOL TYPE:											
PUBLIC	14462	2622897	2.25	0.7	23201	2583725	2.51	0.6	0.66	0.26 *	0.39
PRIVATE	66	16285	2.00	0.8	836	100017	2.28	0.7	0.75	0.29	0.38
CATHOLIC	1003	228929	2.17	0.7	2598	193519	2.49	0.6	0.66	0.32 *	0.48
GEOGRAPHIC REGION:											• • •
NORTHEAST	3489	779676	2.20	0.7	5434	667331	2.48	0.7	0.69	0.28 *	0.41
NORTH CENTRAL	4460	897499	2.25	0.7	7652	825543	2.51	0.6	0.66	0.26 *	0.39
SOUTH	5298	767416	2.29	0.7	8679	860439	2.51	0.6	0.66	0.22 *	0.34
NEST	2881	508945	2.23	0.7	4870	523948	2.49	0.6	0.68	0.26 *	0.38
CURRICULUM:											•
GENERAL	5437	932230	2.26	0.7	9646	1044017	2.49	0.6	0.67	0.24 *	0.35
ACADEMIC	6663	1363402	2.18	0.7	10182	1103302	2.48	0.7	0.68	0.30 *	0.44
VOCATIONAL	4027	657600	2.37	0.7	6433	689080	2.55	0.6	0.63	0.18 *	0.29
COMMINITY TYPE:							_				/
URBAN	4395	762492	2.28	0.7	6034	567943	2.52	0.6	0.67	0.24 *	0.36
SUBURBAN	7760	1503429	2.22	0.7	12883	1426755	2.50	0.6	0.67	0.28 *	0.42
RURAL	3604	628301	2.26	0.7	7718	882563	2.48	0.6	0.66	0.22 🕈	0.54
						130					

\*SIGNIFICANT AT .05 OR LESS

;



a 👻

-----

÷ •

#### IMPORTANCE TO CAREER PLANS: GOOD INCOME TO START OR NITHIN A FEW YEARS (1=NOT IMPORTANT; 3=VERY IMPORTANT)

		NLS 1972				HSB 19	980				
	SAMPLE	WEIGHTED			SAMPLE	NEIGHTED			POOLED	1980-1972	EFFECT
	N	N	MEAN	S.D.	N	N	MEAN	S.D.	S.D.	DIFFERENCE	SIZE
TOTAL	16203	2968041	2.12	0.7	26733	2886897	2.35	0.7	0.68	0.23 *	0.34
SEX:											
MALE	8029	1477757	2.20	0.7	12461	1354326	2.39	0.7	0.67	0.18 ×	0.28
FEMALE	8169	1489268	2.04	0.7	13756	1479680	2.32	0.7	0.68	0.27 *	0.40
SES:											
LOW	4613	71 3256	2.23	0.7	7914	763746	2.41	0.6	0.66	0.19 ¥	0.28
MIDDLE	7750	1522400	2.12	0.7	10313	1371143	2.35	0.7	0.67	0.23 *	0.34
HIGH	3791	723420	2.02	0.7	5966	699463	2.28	0.7	0.70	0.25 *	0.36
RACE:											
MHITE	12612	2481592	2.09	0.7	19213	2287649	2.32	0.7	0.68	0.23 *	0.34
BLACK	1965	239290	2.40	0.7	3395	309531	2.56	0.6	0.62	0.16 *	0.26
ASIAN-AMERICAN	190	27456	2.08	0.7	353	38113	2.33	0.7	0.68	0.25 *	0.37
AMERICAN INDIAN	182	30614	2.25	0.7	203	20805	2.37	0.7	0.69	0.11	0.16
MEXICAN-AMERICAN	527	68874	2.29	0.7	1775	94993	2.41	0.6	0.64	0.12 *	0.19
PUEDTO PICAN	91	9322	2.26	0.8	289	16417	2.29	0.7	3.75	0.03	0.03
OTHER HISPANIC	m	16968	2.03	0.8	900	60474	2.37	0.6	0.65	0.34 *	0.52
SCHOOL TYPE:											
PUBLIC	14528	2634987	2.13	0.7	23297	2593264	2.36	0.7	0.67	0.23 *	0.35
PRIVATE	66	16285	1.93	0.7	836	100017	2.15	0.8	0.75	0.22	0.30
CATHOLIC	1008	230835	2.02	0.7	2600	193615	2.30	0.7	0.67	0.28 ×	0.42
GEOGRAPHIC REGION:											
NOPTHEAST	3513	786063	2.06	0.7	5454	669801	2.33	0.7	0.69	0.26 *	0.38
NORTH CENTRAL	4472	898950	2.12	0.7	7687	829507	2.33	0.7	0.67	0.21 *	0.31
SOUTH	5322	771598	2.20	0.7	8709	862826	2.41	0.6	0.67	0.20 *	0.31
NEST	2896	511430	2.10	0.7	4883	524763	2.33	0.7	0.68	0.23 *	0.34
CURRICULUM:											
GENEDAL	5475	938486	2.15	0.7	96.98	1048946	2.38	0.7	0.67	0.23 *	0.34
ACADEMIC	6692	1369776	2.03	0.7	.0190	1104212	2.27	0.7	0.69	0.24 #	0.34
VOCATIONAL	4035	659477	2.28	0.7	6467	692376	2.44	0.6	0.64	0.17 *	0.26
COMENTTY TYPE:											
LIRBAN	4417	766496	2.15	0.7	6071	571 <b>38</b> 0	2.30	0.7	84.0	0.24 #	0.36
SLIDBAN	7799	1511371	2.09	0.7	12909	1429283	2.34	0.7	0.68	0.25 ¥	0.37
PUPAL	3612	629330	2.16	0.7	7763	AA6225	2.34	0.7	0.67	0.18 #	0.27
		~~/~~~		<b></b>		000113		<b></b>	<b>~.</b>	<b>~+4~</b> ~	

\*SIGNIFICANT AT .. 05 OR LESS

131



	NLS 1972				HSB 19	980					
	SAMPLE	MEIGHTED N	MEAN	\$.D.	SAMPLE N	WEIGHTED N	MEAN	\$.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	16246	2975798	1.73	0.8	26734	2886084	2.01	0.8	0.77	0.28 *	0.36
SEX:											
MALE	8045	1481437	1 73		12440	1764144	1 00		0 70		
FEMALE	81%	1493345	1.72	0.8	13745	1478885	2.03	0.8	0.78	0.31 *	0.32
SES:											
LON	4632	716196	1 78	0 A	701 3	74 3994	2 10	• •	0 77		
MIDDLE	7767	1525640	1.74	0.0	12314	103000	2 00	0.0	0.77	U.32 *	0.42
HIGH	3797	724811	1.65	0.7	5962	699272	1.91	0.8	0.78	0.26 *	0.34
RACE:											
HHITE	12632	2487469	1 71		10208	2287157	3 09		0 77		
BLACK	1985	240266	1 81	0.0	17200	108002	1.70 9 1E	0.0	0.77	U.2/ #	0.34
ASIAN-AMERICAN	188	27004	1 50	0.0	3372	300702	2.13	0.0	0.77	0.34 *	0.44
AMERICAN INDIAN	182	10161	1 81	0.7	303	20707	2.00	0.0	0.77	U.49 <del>*</del>	0.64
MEXICAN-AMERICAN	528	40080	1 01	0.0	1701	20703	2.09	0.0	0.77	U.26	0.34
PUERTO RICAN	90	07007	3 44	0.0	1/01	740/0	2.13	0.0	0.77	0.24 *	0.31
OTHER HISPANIC	111	17218	1.69	0.8	903	60780	2.11	0.8	0.78 0.75	0.47 * 0.38 *	0.60 0.51
SCHOOL TYPE:											
PUBLIC	14544	2443 734	3 77		27200	2502575		• •	<b>•</b>		
PRIVATE	47	14540	1./3	0.0	23277	2372333	2.02	0.8	0.77	0.29 #	0.38
	1007	220001	1.05	0.0	2507	100274	1.00	0.8	0.79	0.22	0.28
	1007	230771	1.0/	<b>V.</b> O	2597	193274	1.89	0.8	0.78	0.21 *	0.27
GEOGRAPHIC REGION:											
NORTHEAST	3518	787118	1.70	0.8	5442	667545	1.98	0.8	0.78	0.29 #	0 37
NORTH CENTRAL	4485	902441	1.75	0.8	7688	829313	2.01	0.8	0.77	0.25 #	0.37
SOUTH	5337	773000	1.72	0.8	8719	864265	2.00	0.8	0.77	0.28 #	0.33
NEST	29 <b>0</b> 6	513240	1.73	0.8	4885	524961	2.05	0.8	0.77	0.32 *	0.41
CURRICULUM:											
GENERAL	5485	940645	1.75	0.8	96.96	1048759	2.03	0.8	8 76	0 28 ×	0 17
ACADEMIC	6694	1370706	1.64	0.8	10166	1103210	1 87	0.0	0.70	0.20 *	0.3/
VOCATIONAL	4066	664145	1.88	0.8	6477	693199	2.19	0.7	0.76	0.31 *	0.41
COMMUNITY TYPE:											
URBAN	4440	769644	1.74	0.8	404 E	E70417	2 87		0 70	A 77 ×	
SUBURBAN	7819	1515671	1.70	0.8	12007	3/003/	1.00	0.0	0.70	U.33 ×	U.42
FURAL	3411	620627	1 77	0.0	1C7V/ 7744	1400130	1.77	U.O	0.70	U.27 *	0.37
	2013	067067	4.77	V.0	//02	00/31/	2.00	V.0	U./6	U.23 🗮	0.30

# IMPORTANCE TO CAREER PLANS: PREVIOUS WORK EXPERIENCE IN THE AREA (1=NOT IMPORTANT; 3=VERY IMPORTANT)

Table 6-16

3

#SIGNIFICANT AT .05 OR LESS



increase to be somewhat greater for both males and females in the general and academic curricula than for those in the vocational curriculum.

The value placed on having previous work experience in a career area also increased more for females than for males and more for low than for middle or high SES students. Cross-tabulations showed larger increases among general and vocational curriculum females than the other sex by curriculum groups.

In summary, although students' career-related values showed a small to moderate increase between 1972 and 1980, the rank ordering of these values did not change. The main trend here appears to be toward an increased emphasis on the extrinsic aspects and rewards of the job (security and pay) rather than on the intrinsic and social aspects of work. A secondary trend is toward careers providing greater autonomy or freedom of choice. This more self-centered concern is consistent with diminishing altruism.

#### G. LIFE VALUES

The students were asked in 1972 and in 1980 how important they considered each of several life goals or values. The scale ranged from 1 = Not important to 3 = Very important. The results are summarized in Table 6-17. Cross-tabulations by the major classification variables are shown in Tables 6-18 to 6-27. As can be seen, success in work was the most important life goal in both years. Other consistently high-ranking life goals and values were strong friendships, marriage and family, steady work, and better opportunities for one's children. Most of the life goals and values covered in this questionnaire showed little change between 1972 and 1980. However, two showed a moderate increase, two a small increase, and one a moderate decrease.

The life value showing the greatest increase between 1972 and 1980 was making lots of money. It moved from 1.95, slightly below 2.0 midpoint for this scale, to 2.21, or moderately important. Students in 1980 also placed more importance on living close to parents and relatives than did students in 1972. Although this item shows a moderate increase, as indicated by its effect size, it is still one of the lower ranked values. Having steady work and success in work, which were already highly rated values in 1972, became even more highly rated by 1980, although the increase shows only a small effect size (probably because of ceiling effects). The other major change in these life goals and values is the decrease in the importance of working to correct social and economic inequalities. This life value shows a moderate decrease from 2.06, or slightly important, to 1.74, or moderately unimportant.

These data show what other studies of high school students' attitudes and values have shown. Namely, the social issues concern of the late 1960s and early 1970s had diminished considerably by 1980 while more self-centered economic concerns and interest in job success increased.



-56-	•
------	---

# Changes in Students' Life Values - 1972 and 1980

Value	<u>Mean</u> 1972	Me an 1980	Differ- 	Effect <u>Size</u>
Success in Work	2.83	2.87	0.04*	0.11
Strong Friendships	2.77	2.79	0.02*	0.05
Marriage and Family Life	2.77	2.76	-0.01	-0.02
Steady Work	2.75	2.82	0.07*	0.16
Better Opportunities for Your Children	2.60	2.61	0.01	0.02
Working to Correct Inequalities	2.06	1.74	-0.32*	-0.47
Making Lots of Money	1.95	2.21	0.25*	0.40
Being a Community Leader	1.66	1.61	-0.05*	-0.07
Living Close to Parents and Relatives	1.57	1.82	0.25*	0.38
Getting Away from This Area	1.57	1.59	0.02	0.03
40' ' 41				

\*Significant at .05 or less

#### H. SELF-ESTEEM AND LOCUS OF CONTROL

The student questionnaire included four agree-disagree statements to assess self-esteem. The scale ranged from l = Disagree strongly to 4 =Agree strongly. As shown in Table 6-28, the mean 1972 response for all four items was 3.11, on the positive side of the scale's 2.5 midpoint. By 1980 the mean for these four items had risen to 3.19, indicating even higher self-esteem.

When the means are examined for the classification groups, there is a consistent tendency for Blacks to report higher self-esteem than Whites. (See Tables 6-29 to 6-32.) Changes toward more positive selfesteem were greater for males than for females, for high SES than low SES students, and for academic curriculum students in comparison to the general and vocational students.

The questionnaire also included four questions to assess whether the student had an external or internal locus of control, that is whether the students felt they had the power to control their own lives or if life events were beyond their control. The scale used ranged from 1 = I agree strongly (an index of external control) to 4 = I disagree strongly (an index of internal control). The mean 1972 score on each of the four items was on the internal control side of the scale's 2.5 midpoint. By

#### IMPORTANCE IN YOUR LIFE OF SUCCESS IN WORK (1=NOT IMPORTANT; 3=VERY IMFORTANT)

	NLS 1972					HSB 19	980				
	SAMPLE N	WEIGHTED N	MEAN	s.D.	SAMPLE N	WEIGHTED N	MEAN	S.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	16564	3024733	2.83	0.4	27532	2967920	2.87	0.4	0.38	0.04 *	0.11
SEX:											
MALE	8211	85ن1 <b>505</b>	2.85	0.4	12759	1385410	2.88	0.4	0.38	0.03 *	0.08
FEMALE	8348	1518032	2.82	0.4	13984	1502215	2.87	0.4	0.38	0.05 ×	0.14
SES:											
LOW	4778	735575	2.84	0.4	8193	790025	2.85	0.4	0.39	0.01	0.02
MIDDLE	7879	1545821	2.84	0.4	12595	1400431	2.88	0.4	0.37	0.04 ¥	0.12
HIGH	3846	732869	2.82	0.4	6095	713781	2.89	0.3	0.37	0.07 ×	0.20
RACE:											
MHITE	12789	2516140	2.83	0.4	19607	2334727	2.87	0.4	0.38	0.05 ×	0.12
BLACK	2088	252548	2.92	0.3	3588	325612	2.91	0.3	0.32	-0.01	-0.04
ASIAN-AMERICAN	191	27563	2.78	0.4	356	38118	2.87	0.4	0.39	0.09	0.24
AMERICAN INDIAN	184	30573	2.87	0.3	206	21049	2.76	0.5	0.42	-0.11	-0.27
MEXICAN-AMERICAN	550	72126	2.87	0.4	1844	98757	2.85	0.4	0.40	-0.02	-0.05
PUERTO RICAN	95	9676	2.79	0.4	300	17543	2.80	0.5	0.49	0.01	0.01
OTHER HISPANIC	122	18844	2.86	0.4	931	63453	2.85	0.4	0.40	-0.01	-0.04
SCHOOL TYPE:											
PUBLIC	14846	2683897	2.83	0.4	24019	2667797	2.87	0.4	0.38	0.04 ¥	0.10
PRIVATE	66	16256	2.76	0.4	865	103422	2.85	0.4	0.38	0.09	0.25
CATHOLIC	1024	235113	2.81	0.4	2648	196701	2.88	0.3	0.37	0.07 *	0.19
GEOGRAPHIC REGION:											
NORTHEAST	3592	799800	2.82	0.4	5587	686806	2.88	0.4	0.38	0.06 ¥	0.16
NORTH CENTRAL	4542	912316	2.82	0.4	7911	849794	2.87	0.4	0.38	0.05 ×	0.13
SOUTH	5467	790422	2.87	0.4	9016	894121	2.88	0.4	0.36	0.00	0.01
NEST	2963	522195	2.81	0.4	5018	537198	2.86	0.4	0.40	0.05 ¥	0.13
CURRICULUM:											
GENERAL	5627	962199	2.81	0.4	10024	1083512	2.85	0.4	0.41	0.05 ×	0.12
ACADEMIC	6780	1385720	2.84	0.4	10392	1123937	2.90	0.3	0.35	0.06 ¥	0.16
VOCATIONAL	4156	676512	2.85	0.4	6712	716558	2.86	0.4	0.39	0.01	0.04
CONTUNITY TYPE:											
URBAN	4533	783372	2.84	0.4	6293	589655	2.89	0.3	0.37	0.05 *	0.14
SUBURBAN	7918	1532387	2.82	0.4	13270	1467863	2.87	0.4	0.38	0.05 *	0.13
RURAL	3667	637121	2.86	0.4	7969	910402	2.86	0.4	0.38	0.01	0.02

\*SIGNIFICANT AT .05 OR LESS



#### IMPORTANCE 11 YOUR LIFE OF STRONG FRIENDSHIPS (1=NOT IMPORTANT; 3=VERY IMPORTANT)

	NLS 1972					HSB 1	980				
	SAMPLE N	WEIGHTED N	MEAN	S.D.	SAMPLE N	WEIGHTED	MEAN	S.D.	P00'_ED 5.D.	1980-1972 DIFFERENCE	EFFECT SIZE
TOTAL	16520	3018899	2.77	0.5	27444	2958134	2.79	0.4	0.45	0.02 ×	0.05
SEX:											
MALE	1018	1503351	2 70	A E	10710			• •	• · -		
FEMALE	9194	1503331	2.17	0.5	12/19	1380830	2.79	0.4	0.45	0.01	0.02
	0364	1314331	2.70	0.5	13940	1498225	2.80	0.4	0.46	0.04 <del>*</del>	0.09
SES:											
LOW	4752	732432	2 70	0 E	÷ . 60	305003			<b>•</b>		
MIDDLE	7865	1543730	2 70	0.5	5149	785893	2.72	0.5	0.51	0.01	0.03
NIGH	1945	719457	2 42	V.5	125/1	139/333	2.81	0.4	0.44	0.03 *	0.06
	3043	/3203/	2.02	V.4	0085	/122/4	2.87	0.4	0.39	0.05 ×	0.13
RACE:											
MHITE	12781	2514445	2 80	6 4	10540	0700704	9 04	• •	<b>•</b> · · ·		
BLACK	2059	249514	2 64	0.4	17500	2329/20	2.84	0.4	0.41	0.04 *	0.09
ASIAN-AMERICAN	190	27495	2 75	0.0	3337	323054	6.54	0.6	0.61	0.00	0.00
AMERICAN INDIAN	185	30651	2 72	0.5 A E	357	30257	2.84	0.4	0.43	0.09	0.21
MEXICAN-AMERICAN	546	71520	2 69	V.5	203	20801	2.70	0.5	0.51	-0.02	-0.04
PUERTO RICAN	545	9469	2 52	0.5	1043	965/1	2.69	0.5	0.52	0.00	0.01
OTHER HISPANIC	120	18668	2 60	0.0	294	10940	2.02	0.6	0.62	0.10	0.16
		10400	2.00	v.0	965	02080	2.73	0.5	0.50	0.13	0.26
SCHOOL TYPE:											
PUBLIC	14810	2679247	2 77	A E	27072	9480901			• • • •		
PRIVATE	66	16256	2 92	0.5	23732	1030201	2./9	0.5	0.46	0.02 *	0.04
CATHOLIC	1022	234794	2 84	0.4	004 9440	103204	2.05	0.4	0.39	0.03	0.07
		634774	2.04	V.4	2040	140200	2.00	0.4	0.39	0.01	0.04
GEOGRAPHIC REGION:											
NORTHEAST	3585	798477	2.78	0 5	5540	494730	2 01	• 4	• • •	<b>A</b>	
NORTH CENTRAL	4534	911280	2.77	0.5	2007	004730	2 01	0.4	0.44	0.03	0.06
SOUTH	5446	787960	2.76	0.5	2024	04/7/C	2.01	0.4	0.44	0.03 *	0.08
NEST	2955	521182	2 77	0.5	5704	070333 534034	2.70	0.5	0.48	0.00	0.00
	-/	361106	••••	0.5	3000	534070	6.06	0.4	0.44	0.04 *	0.10
CURRICULUM:											
GENERAL	5607	959812	2 75	0 5	0004	1000717	9 70				
ACADEMIC	6774	1384495	2.80	0.5	10169	1101725	2.10	0.5	0.47	0.03 *	0.06
VUCATIONAL	4138	674290	2 71	0 5	10300	1161/65	2.04	0.4	0.41	0.04 *	0.09
	1100		/3	0.5	00/9	/12/15	2.74	0.5	0.49	0.01	0.02
COMPRESITY TYPE:											
URBAN	451 9	781754	2.75	0 5	4957	505400	9 75			• • •	
SUBURBAN	7904	1530043	2 70	0.5 A E	11940	303000 1446867	C./J 2 01	0.5	0.49	0.00	0.01
RURAL	3660	636318	2 77	0.5 0 E	13649	140303/	2.01	0.4	0.44	0.02 *	0.05
	3000	030310	•••//	v.5	1429	AA1 299	C.80	0.4	0.45	0.03	0.06

\*SIGNIFICANT AT .05 OR LESS

\_\_\_\_\_\_



#### IMPORTANCE IN YOUR LIFE OF MARRIAGE AND FAMILY LIFE (1=NOT IMPORTANT; 3=VERY IMPORTANT)

		NLS 1972					5 ( <b>4 (</b>	30				
		SAMPLE N	WEIGHTED N	MEAN	\$.D.	SAMPLE N	WEIGHTED N	MEAN	S.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL		16494	3013069	2.77	0.5	27486	2961403	2.76	0.5	0.53	-0.01	-0.02
SEX:												
MALE		8167	1499247	2.72	0.6	12730	1380889	2.72	0.6	0.57	-0.00	-0.01
FEMALE		8322	1512804	2.81	0.5	13969	1500598	2.80	0.5	0.48	-0.01	-0.03
SES :												
LON		4757	733139	2.79	0.5	8177	787823	2.75	0.5	0.53	-0.04 *	-0.08
MIDDLE		7852	1540666	2.77	0.5	12569	1397334	2.77	0.5	0.52	-0.00	-0.01
HICH		3829	729360	2.74	0.6	6087	712566	2.76	0.5	0.53	0.03	0.05
RACE												
INTE		12750	2507850	2.77	0.5	19571	2329654	2.77	0.5	0.51	-0.00	-0.00
BLACK		2065	250251	2.75	0.6	3584	325720	2.69	0.6	0.59	-0.06 *	-0.10
ASIAN-AMER	ICAN	191	27592	2.61	0.6	357	38296	2.76	0.5	0.57	0.16	0.27
AMERICAN J	NDIAN	186	30838	2.77	0.6	201	20505	2.60	0.7	0.62	-0.17	-0.27
MEXICAN-AP	TERICAN	547	71796	2.79	0.5	1845	98702	2.74	0.6	0.55	-0.05	-0.10
PUERTO RIC	AN	9.5	9666	2.87	0.4	297	17181	2.73	0.6	0.55	-0.14	-0.26
OTHER HISP	PANIC	120	18446	2.74	0.6	933	63426	2.79	0.5	0.51	0.05	0.10
SCHOOL TYPE:												
PUBLIC		14784	2673380	2.77	0.5	23975	2661860	2.76	0.5	0.53	-0.01	-0.02
PRIVATE		66	16256	2.65	0.7	862	102850	2.74	0.5	0.54	0.09	0.17
CATHOLIC		1023	234792	2.76	0.5	2649	196694	2.80	0.5	0.50	0.04	0.08
GEOGRAPHIC RE	GION:											
NORTHEAST		3579	796283	2.76	0.5	5575	684464	2.73	0.5	0.54	-0.02	-0.04
NORTH CENT	RAL	4528	910051	2.76	0.5	7898	848580	2.77	0.5	0.52	0.00	0.01
SOUTH		5435	786183	2.81	0.5	9008	892826	2.77	٦.5	0.51	-0.05 ¥	-0.06
NEST		2952	520551	2.74	0.6	5005	535534	2.76	5	0.55	0.02	0.04
CURRICULUM:												
GENERAL		5595	958147	2.77	0.5	9998	1080582	2.75	0.5	0.54	-0.01	-0.03
ACADEMIC		6755	131. 231	2.76	0.5	10384	1122705	2.77	0.5	0.51	0.02	0.03
VOCATIONAI	-	4143	674389	2.79	0.5	6699	714491	2.75	0.5	0.53	-0.04 *	-0.08
COMMUNITY THE	7E :											
URBAN		4512	781230	2.77	0.5	6286	588784	2.74	0.5	0.54	-0.03	-0.05
SUBURBAN		7893	1526548	2.76	0.5	13246	1463946	2.76	0.5	0.53	-0.00	-0.00
RURAL		3656	635215	2.80	0.5	7954	908673	2.77	0.5	0.51	-0.03	-0.06

\*SIGNIFICANT AT .05 OR LESS

137

¥



## IMPORTANCE IN YOUR LIFE OF STEADY WORK (1=NOT IMFORTANT; 3=VERY IMPORT ')

	NLS 1972				HSB 1	.980					
	SAMPLE	WEICHTED	ME A LI	e n	SAMPLE	WEIGHTED		*-	POOLED	1980-1972	EFFECT
			FIEAN	5.0.	N	N	MEAN	S.D.	S.D.	DIFFERENCE	SIZE
TOTAL	16483	3010711	2.75	0.5	27331	2945918	2.82	0.4	0.45	0.07 ¥	0.17
SEX:											••••
MALE	8168	1498935	2 70	0 5	30// 7						
FEMALE	8310	1510759	2 71	V.5 A E	12007	1375468	2.84	0.4	0.44	0.05 <b>*</b>	0.10
			C./2	V.5	13643	1492723	2.81	0.4	0.46	0.11 ¥	0.23
SES.											
LOW	4742	730595	2.81	0.4	A11A	702354	2 0 2	• •			
MIDDLE	7843	1538703	2.76	0.5	12528	1702050	2.82	0.4	0.44	0.01	0.02
HIGH	3838	731015	2.67	0.5	6053	719754	2.83	0.4	0.44	0.08 *	0.17
				••••	0033	198190	2.02	0.4	0.48	0.15 ¥	0.31
RACE:											
MHITE	12747	2506649	2 74	0.5	19476	2319415	2 81		0 45		
BLACK	2065	250182	2.85	0.4	3552	322572	2 94	0.4	0.45	0.09 *	0.19
ASIAN-AMERICAN	190	27478	2.76	0.5	357	38343	9 77	0.4	0.42	-0.01	-0.03
AMERICAN INDIAN	183	30327	2.74	0.5	204	20809	2 7A	0.5	0.40	0.00	0.01
MEXICAN-AMERICAN	544	71290	2.85	0.4	1832	97886	2 82	0.5	0.51	0.04	0.07
PUERTO RICAN	93	9479	2.78	0.4	296	16998	2 81	J. E	0.43	-0.03	-0.07
OTHER HISPANIC	119	18383	2.75	0.5	924	62386	2.78	0.5	0.45	0.03	0.06
						00000		0.5	V.4/	0.04	0.08
SCHOOL TYPE:											
PUBLIC	14780	2672264	2.75	0.5	23832	2647616	2.83	0.4	8 45	0 09 #	
PRIVATE	66	16256	2.48	0.7	861	102571	2.76	0.5	0 50	0.00 *	0.1/
CATHOLIC	1017	233819	2.72	0.5	2638	195730	2.83	0.4	0.44	0 11 #	0.50
CEOCOARUTO AFATOUL									••••	V.11 *	V.24
NORTHEAST											
NORTH CONTRAL	3583	796959	2.72	0.5	5547	681323	2.83	0.4	0.46	0.11 #	0 94
SOUTH LENIKAL	4523	909467	2.75	0.5	7850	843562	2.83	0.4	0.44	0.09 #	0.24
JUTH	5432	785300	2.79	0.5	8950	887293	2.83	0.4	0.44	0.03 #	0.17
MEST	2945	518985	2.73	0.5	4984	533740	2.80	0.5	0.48	0.07 *	0.15
CLER TOULUM:											•••23
GENERAL	5504	055770									
ACADEMIC	2204	7553/U	2.76	0.5	9944	1075055	2.82	0.4	0.45	0.06 ¥	0.13
VOCATIONAL	6739	4 77 700	2.71	0.5	10336	1117735	2.82	0.4	0.46	0.11 ×	0.24
·····	4140	6/3/00	Z.81	0.4	6653	710347	2.83	0.4	0.43	0.02	0.05
COMPLINITY TYPE:											
URBAN	4506	779145	2.76	0 5	4979	507150		• •	• • •		
SUBURBAN	7887	1526524	2.73	0.5	17101	303152	2.83	0.4	0.44	0.07 ¥	0.16
RURAL	3652	634792	2.79	0.5	13171 700e	1437664	2.82	0.4	0.46	0.09 *	0.21
				0.9	7740	703542	2.82	0.4	0.44	0.03 ¥	0.08

\*SIGNIFICANT AT .05 OR LESS



		NLS 1			HSB 19	80					
	SAMPLE N	WE IGHTED	MEAN	\$.D.	SAMPLE N	WEIGHTED N	MEAN	\$.D.	POOLED S.D.	1980–1972 Difference	EFFECT SIZE
TOTAL	16430	3002472	2.60	0.6	27311	2942560	2.61	0.6	0.60	0.01	0.02
SEX:											
MALE	8141	1495145	2.60	0.6	12648	1372073	2.62	0.6	0.60	0.02	0.03
FEMALE	8284	1506310	2.61	0.6	13896	1492444	2.61	0.6	0.60	0.00	0.00
<b>9</b> 5:											
	4746	733733	2.75	05	8116	782243	2.72	0.5	0 52	-0.03 *	-0.06
	7816	1534122	2.62	0.6	12516	1390706	2.62	0.6	0.59	0.00	0.01
HIGH	3810	726533	2.42	0.7	6048	708019	2.47	0.6	0.66	0.05 *	0.08
PACE:											
LINTE	126.96	2498114	2.57	0.6	19474	2318083	2 57	0.6	0.61	0.00	0.01
BLACK	2059	249783	2 86	0.4	3540	321514	2.82	0.5	0.45	-0.04 *	-0.10
ASTAN_AMEDICAN	189	27365	2 65	0.6	354	37839	2.71	0.5	0.55	0.06	0.10
AMERICAN INDIAN	184	30573	2 74	0.6	202	20619	2.60	0.6	0.60	-0.14	-0.23
MENTCANLAMEDICAN	544	71587	2 82	0.4	1936	97893	2.78	0 5	0.47	-0.04	-0.08
DIESTO RICAN	94	9578	2 84	0 4	296	16996	2.83	0 5	0.44	-0.01	-0.02
OTHER HISPANIC	118	18251	2.78	0.5	922	62875	2.69	0.6	0.54	-0.09	-0.16
STHORE TYPE:											
	14730	2664714	2 60	0.6	23819	2644896	2 62	0.6	0 59	0.02	0.03
DDIVATE	65	15941	2.34	0.8	854	102248	2.41	0.7	0.71	0.07	0.10
CATHOLIC	1019	233862	2.55	0.6	2638	1%217	2.59	0.6	0.60	0.04	0.07
GEDGRAPHIC REGIUN:	35//	7094 57	0 5/	• /	FF 74	4 30549		• •	0 4 3	0 02	0.07
NORTHEAST	3500	/9205/	2.50	0.0	2224	0/7342	2.50	0.0	0.02		-0.03
NORTH CENTRAL	4506	906784	2.59	0.0	7852	044235	2.5/	0.0	0.00	-0.02	-0.03
SOUTH	5429	785024	2.70	0.0	8904	5363495	2.70	0.5	U.55	0.90	0.00
WEST	2934	219000	6.24	U.0	4701	220224	2.5/	9.0	0.03	0.03	9.05
CURRICULUM:											
GENERAL	5576	955 <b>050</b>	2.64	0.6	9942	1073769	2.63	0.6	0.59	-0.01	-0.02
ACADENIIC	6726	1374550	2.53	0.6	10311	1115222	2.56	0.6	0.62	0.04 ¥	0.06
VOCATIONAL	4127	672569	2.70	0.5	6659	710421	2.67	0.6	0.55	-0.04	-0.07
CONNUNITY TYPE:											
URBAN	4496	777778	2.65	0.6	6233	582812	2.68	0.6	0.57	0.04	0.06
SUBURBAN	7857	1520988	2.55	0.6	13158	1454743	2.58	0.6	0.62	0.03 *	0.05
RURAL	3644	633785	2.66	0.6	792 <b>0</b>	905005	2.61	0.6	0.57	-0.05 <b>*</b>	-0.08

#### INPORTANCE IN YOUR LIFE OF BETTER OPPORTUNITIES FOR YOUR CHILDREN (1=NOT IMPORTANT; 3=VERY IMPORTANT)

\*SIGNIFICANT AT .05 OR LESS

\_

.



IMPORTANCE	IN	YOUR	LIFE	OF	HORKING	то	CORRECT	SOCIAL	AND	ECONOMIC	INEQUALITIES
			(1:	= <b>NO</b> 1	T IMPORT	ANT	; 3=VERY	IMPORT	ANT)		

	NLS 1972				_	HSB 1	980				
	SAMPLE N	MEIGHTED N	MEAN	S.D.	SAMPLE N	MEIGHTED N	MEAN	\$.D.	POOLED S.D.	1980-1972 Difference	EFFECT STZF
TOTAL	16466	3008658	2.06	0.7	27270	2938610	1.74	0.7	0.68	-0.32 #	-0 47
SEX:											-0.4/
MALE	8154	1496944	3 04		10/74		• • • •				
FEHALE	8308	1510948	2 15	0.7	12034	13/1064	1.69	0.7	0.69	-0.27 *	-0.40
		2320/10	C.13	U./	13002	1491213	1.77	0.7	0.67	-0.38 ¥	-0.57
SES:											
LON	4739	730286	2.11	0.7	9117	782280	1 70				
MIDDLE	7839	1538397	2.03	0.7	12687	702209	1.79	0.7	0.68	-0.32 *	-0.48
HICH	3830	729969	2 07	0.7	1240/	1300011	1./1	0.7	0.68	-0.32 *	-0.47
			2.0/	0.7	0021	/0/886	1.72	0.7	0.69	-0.35 ¥	-0.51
RACE:											
WHITE	12735	2505190	2 02	0 7	30444				_		
BLACK	2063	249972	2.00	0.7	T3404	2316288	1.67	0.7	0.67	-0.35 *	-0.52
ASIAN-AMERICAN	190	27281	2 05	0.7	3523	319926	2.03	0.7	0.68	-0.34 <b>*</b>	-0.50
AMERICAN INDIAN	180	20011	2.1	0.7	355	38074	1.87	0.7	0.68	-0.15	-0.23
MEXICAN-AMERICAN	542	71175	C+1. 2 91	0./	202	20657	1.86	0.6	0.66	-0.25 *	-0.39
PUERTO RICAN	95	/1135	2.21	U.6	1833	97754	1.90	0.7	0.67	-0.31 *	-0.47
OTHER HISPANIC	110	90/0	2.20	0.6	294	17100	1.91	0.7	0.69	-0.27 ×	-0.42
	110	10543	2.05	0.7	912	61724	1.85	0.7	0.70	-0.19	-0.27
SCHOOL TYPE:											
PUBLIC	34765	9448249									
PRIVATE	14/33	2000502	2.05	0.7	23775	2640827	1.74	0.7	0.68	-0.31 *	-0.46
CATHOLIC	100	07200	2.08	0./	855	101733	1.69	0.7	0.70	-0.39 *	-0.55
GAINGELE	1055	235021	2.08	0.7	2640	196051	1.69	0.7	0.66	-0.39 *	-0.59
GEOGRAPHIC REGION:											
NORTHEAST	3574	795879	2 83	07	EEA]	( 3003 0					
NORTH CENTRAL	4515	908032	2 81	0.7	2022	679910	1.72	0.7	0.69	-0.31 *	-0.45
SOUTH	5432	795368	2 16	0.7	/831	841818	1.68	0.7	0.67	-0.36 ×	-0.53
NEST	2945	E10179	2 01	0.7	0730	885771	1.80	0.7	0.68	-0.34 *	-0.49
	6743	517370	C.VI	u./	4962	531111	1.73	0.7	0.68	-0.28 *	-0.42
CURRICULUM:											
GENERAL	5584	055044	2 02								
ACADEMIC	4763	733004	2.02	0.7	9915	1071030	1.70	0.7	0.67	-0.32 *	-0.47
VOCATIONAL	6123	1300070	2.10	0.7	10341	1118285	1.78	0.7	0.69	-0.32 ×	-0.47
	4150	0/2410	2.02	0.7	6618	706341	1.71	0.7	0.67	-0.31 *	-0.46
CONTRINITY TYPE:											
URBAN	4507	770282		<b>•</b> -							
SUBURBAN	7373	17406	C.IU	0.7	6214	581513	1.80	0.7	0.69	-0.29 *	-0.42
PURAL	70/9	1364/03	2.04	0.7	13161	1454818	1.71	0.7	0.68	-0.34 *	-0.49
	3036	0 34 / 34	2.04	0./	7895	902278	1.73	0.7	0.67	-0.31 *	-0.46

\*SIGNIFICANT AT .05 OR LESS

140

# Table 6-23

#### IMFORTANCE IN YOUR LIFE OF LOTS OF MONEY (1=NOT IMPORTANT; 3=VERY IMPORTANT)

	NLS 1972					HSB 19	980				
	SAMPLE N	WEIGHTED N	MEAN	S.D.	SAMPLE N	WEIGHTED N	MEAN	S.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	16498	3013312	1.95	0.6	27440	2958572	2.21	0.6	0.63	0.25 *	0.40
SEX:											
MALE	8183	1501168	2.10	0.6	12715	1381571	2.33	0.6	0.62	0.22 *	0.36
FEMALE	8310	1511127	1.81	0.6	13945	1498039	2.09	0.6	0.60	0.28 ¥	0.47
SES:											
LON	4748	731283	1.99	0.6	8154	786223	2.20	0.6	0.63	0.21 *	0.33
MIDDLE	7855	1540808	1.94	0.6	12561	1396763	2.20	0.6	0.62	0.26 ¥	0.42
HIGH	3837	731141	1.94	0.6	6084	712588	2.21	0.6	0.63	Ç.27 ₩	0.43
RACE:											
MHITE	12760	2509566	1.93	0.6	19544	2327886	2.18	0.6	0.62	0.25 ×	0.40
BLACK	2063	249684	2.16	0.6	3571	324358	2.37	0.6	0.63	0.21 ¥	0.34
ASIAN-AMERICAN	191	27581	1.98	0.6	356	38224	2.33	0.6	0.61	0.35 ×	0.57
AMERICAN INDIAN	184	30573	2.09	0.7	203	20981	2.23	0.6	0.65	0.14	0.21
MEXICAN-AMERICAN	547	71924	2.03	0.6	1844	98769	2.22	0.6	0.63	0.19 ¥	0.30
PUERTO RICAN	92	9394	1.97	0.6	298	17285	2.30	0.6	0.60	0.33 ¥	0.55
OTHER HISPANIC	118	17966	1.28	0.6	927	62616	2.23	0.6	0.64	0.35 ¥	0.55
SCHOOL TYPE:											
PUBLIC	14788	2673845	1.96	0.6	23936	2659224	2.21	0.6	0.62	0.25 ¥	0.41
PRIVATE	65	15%3	1.71	0.6	864	103264	2.07	0.7	0.68	0.36 *	0.54
CATHOLIC	1022	234609	1.88	0.6	2640	196084	2.16	0.6	0.60	0.29 *	0.47
GEOGRAPHIC REGION:											
NORTHEAST	3579	796500	1.92	0.6	5572	6846 93	2.22	0.6	0.63	0.30 ¥	0.48
NORTH CENTRAL	4524	909243	1.95	0.6	7885	847445	2.19	0.6	0.61	0.24 *	0.39
SOUTH	5440	786561	2.00	0.6	8985	890764	2.21	0.6	0.63	0.22 *	0.35
HEST	2955	521007	1.94	0.6	4998	535670	2.20	0.6	0.64	0.25 *	0.39
CURRICULUM:											
GENERAL	5599	957403	1.97	0.6	9989	1079281	2.22	0.6	0.63	0.25 ¥	0.40
ACADEMIC	6765	1382369	1.92	0.6	10375	1123051	2.17	0.6	0.62	0.24 *	0.39
VOCATIONAL	4133	673237	1.99	0.6	6675	712867	2.24	0.6	0.63	0.24 *	0.39
COMMUNITY TYPE:											
URBAN	4507	778654	1.98	0.6	6270	587248	2.25	0.6	0.63	0.27 *	0.43
SUBURBAN	7897	1528528	1.94	0.6	13228	1463723	2.21	0.6	0.62	0.27 *	0.44
RURAL	3655	635390	1.95	0.6	7942	907601	2.17	0.6	0.62	0.22 ¥	0.35

**\*SIGNIFICANT AT .05 OR LESS** 



### IMPORTANCE IN YOUR LIFE OF BEING A COMMUNITY LEADER (1=NOT IMPORTANT; 3=VERY IMPORTANT)

	NLS 1972					HSB 1	980				
	SAMPLE N	WEIGHTED N	MEAN	\$.D.	SAMPLE N	WE IGH TED N	MEAN	 S.D.	POOLED S.D.	1980-1972 DIFFERENCE	EFFECT SIZE
TOTAL	16474	3009664	1.66	0.7	27179	2932713	1.61	0.7	0.66	-0.05 *	~0.07
SEX:											••••
MALE	8160	1497586	1.74	07	12505	1740720			• • •		
FEMALE	8309	1511061	1.57	0.6	12375	1307367	1.07	0.7	0.68	-0.05 *	-0.07
				••••	23027	1403040	1.55	V.0	0.65	~0.04 #	-0.07
SES:											
LON	4737	730644	1.72	0.7	8058	777775	1 66	0 4	<b>A</b> 44		
MIDDLE	7845	1539054	1.63	0.7	12440	1384422	1 50	0.0	0.00	-0.1/ #	-0.26
HIGH	3833	729716	1.64	0.7	6053	708520	1 70	0.7	V.00	-0.04 #	-0.06
				•••	0033	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	1./0	V./	V.6/	0.05 #	0.08
RACE:											
MHITE	12748	2507016	1.63	0.7	19407	2312300	1.58	0.6	0 45	-0.05 ×	
BLACK	2060	249667	1.92	0.7	3502	317074	1.73	0.0	0.05	-0.10 *	-0.08
ASIAN-AMERICAN	190	27517	1.57	0.6	356	38294	1.75	0.7	0.71	-0.19 *	-0.2-
AMERICAN INDIAN	181	29947	1.78	0.7	200	20518	1 70	07	0.00	0.10	0.c
MEXICAN-AMERICAN	543	71140	1.81	07	1820	97603	1.70	07	0.72	-0.08	-0.11
PUERTO RICAN	94	957 <b>8</b>	1.84	0.7	293	16956	1 72	0.7	0.70	-0.11	-0.15
OTHER HISPANIC	121	18565	1.63	0.7	914	62292	1.71	0.7	0.73	-0.11	-0.16
								•••	0.70	0.00	0.11
SCHOOL TYPE:											
PUBLIC	14772	2671674	1.65	0.7	23705	2636079	1.60	07	0.66	-0 05 #	
PRIVATE	66	16256	1.60	0.7	853	101355	1.67	0.7	0.60	-0.05 *	-0.08
CATHOLIC	1016	233204	1.66	0.7	2621	195278	1.62	0.6	0.65	-0.04	-0 05
GEOGRAPHIC DECION.											-0.03
MODTHEAST	7570	705003		• •							
NORTH CENTRAL	35/9	/ 42481	1.55	0.6	5513	677800	1.54	0.6	0.63	-0.01	-0.02
SOUTH	4519	908749	1.65	0.7	7830	841853	1.58	0.6	0.65	-0.07 ¥	-0.11
HEST	5433	785808	1.83	0.7	8886	881441	1.71	0.7	0.70	-0.12 *	·0.17
MEST	2943	519125	1.56	0.6	4950	531618	1.57	0.6	0.65	0.00	0.01
CURRICULUM:											
GENERAL	5597	055943	1 49								
ACADEMIC	4754	733043	1.02	0.7	9885	1069597	1.56	0.6	0.65	-0.06 *	-0.09
VOCATIONAL	4132	473946	1./0	0.7	10295	1114345	1.70	0.7	0.68	0.01	0.01
	7136	0/3004	1.03	V./	66 <b>01</b>	705827	1.53	0.6	0.65	-0.10 <b>*</b>	-0.15
CONNUNITY TYPE:											
URBAN	4510	779212	1.66	07	6197	570750	• • •		• • -		
SUBURBAN	7877	1525284	1.63	0 7	11171	3/7/37	1.04	U.7	0.67	-0.02	-0.03
RURAL	3651	634837	1.72	0.7	13141 7055	1435403	1.5/	0.6	0.66	-0.05 *	-0.08
	34	034037	/6	<b>v</b> ./	(055	841440	1.64	0.7	0.67	-0.09 *	-0.13

\*SIGNIFICANT AT .05 OR LESS



#### IMPORTANCE IN YOUR LIFE OF LIVING CLOSE TO PARENTS AND RELATIVES (1=NOT IMPORTANT; 3=VERY IMPORTANT)


	NLS 1972					HSB 19	980				
	SAMPLE	MEIGHTED N	MEAN	\$.D.	SAMPLE N	NE IGHTED	MEAN	S.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	16492	3014393	1.57	0.6	27322	2946669	1.82	0.7	0.65	0.25 *	0.38
SEX:											
MALE	8170	1499806	1.55	0.6	12646	1373987	1.79	0.7	0.64	0.25 ×	C.38
FEMALE	8337	1513570	1.60	0.6	13911	1495159	1.84	0.7	0.65	0.24 *	0.37
SES:											
LON	4740	730888	1.62	0.6	8108	782036	1.84	0.7	0.66	0.22 *	0.33
MIDDLE	7861	1543064	1.59	0.6	12516	1392478	1.83	0.6	0.64	0.24 *	0.37
HIGH	3832	730191	1.48	0.6	6064	710148	1.76	0.7	0.63	0.28 *	0.44
PACF:											
MITE	12756	2510248	1.56	0.6	19482	2320368	1.80	0.6	0.64	0.25 *	0.39
BLACK	2059	249487	1.61	0.7	3546	322733	1.79	0.7	0.67	0.18 *	0.28
ASTAN-AMERICAN	191	27609	1.67	0.7	357	38441	2.03	0.7	0.69	0.37 *	0.54
ANERTCAN TNOTAN	184	30573	1.64	0.7	200	20410	1.83	0.7	0.70	0.19	0.27
MEXTCAN_AMEDICAN	546	71646	1.73	0.7	1836	98079	1.99	0.7	0.68	0.27 *	0.40
	94	9578	1.78	0.7	292	16668	1.97	0.7	0.70	0.19	0.27
OTHER HISPANIC	120	18394	1.84	0.7	924	62593	1.93	0.7	0.68	0.09	0.14
SCHOOL TYPE:											
PIBLIC	14780	2674483	1.57	0.6	23820	26478.44	1.81	0.7	0.65	0.25 *	0.38
POTVATE		16256	1.57	0.7	858	102670	1.79	0.7	0.67	0.23	0.34
CATHOLIC	1023	234951	1.62	0.6	2644	196155	1.88	0.6	0.64	0.26 *	0.41
GEOGRAPHIC REGION:											
NORTHEAST	3581	797307	1.58	0.6	5552	682282	1.82	0.7	0.65	0.24 *	0.37
NODTH CENTRAL	4524	909950	1.54	0.6	7852	844302	1.80	0.6	0.63	0.25 ¥	0.40
SOUTH	5430	785795	1.62	0.7	8943	887059	1.84	0.7	0.66	0.22 ¥	0.33
NEST	2957	521341	1.54	0.6	4975	533026	1.81	0.7	0.64	0.27 *	0.42
CLERTICUIUM:											
GENERAL	5592	957727	1.59	0.6	9943	1074891	1.81	0.7	0.65	0.22 *	0.34
ACADENTC	6761	1382429	1.53	0.6	10343	1119428	1.80	0.6	0.63	0.27 ¥	0.42
VOCATIONAL	4138	673934	1.63	0.7	6642	709595	1.85	0.7	0.66	0.22 *	0.34
COMMUNITY TYPE:											
URBAN	4514	781255	1.61	0.6	6229	583721	1.85	0.7	0.66	0.24 *	0.37
SUBLIRBAN	7888	1526911	1.55	0.6	13192	1459889	1.82	0.7	0.64	0.27 *	0.42
RURAL	3656	635974	1.59	0.6	7901	903060	1.80	0.7	0.65	0.21 *	0.32

\*SIGNIFICANT AT .05 OR LESS



IMPORTANCE	IN YOUR	LIFE	OF	GETTING	AHAY	FROM	THIS	AREA	OF	THE	COUNTRY
	0	L=NOT	IMP	ORTANT:	3=VE	NT YS	NITAN	1T )		••••	00011111

ATANIS 3-ACKI	F TUDOK I ANT 1	

	NL\$ 1972					HSB 14	980				
	SAMPLE N	WEIGHTED N	MEAN	S.D.	SAMPLE N	WEIGHTED N	MEAN	S.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	16474	3011330	1.57	0.7	27366	2949328	1.59	0.7	0.73	0.02	0.03
SEX:											
MALE	8156	1497036	1 54	07	10/70				_		
FEMALE	8313	1513277	1.50	0.7	120/9	13/5593	1.60	0.7	0.73	0.04 *	0.05
				•••	23764	1470100	1.30	U./	0.73	-0.01	-0.01
9E5:											
LOH	4731	729432	1.61	0.7	A120	797747	3 44				
MIDDLE	7850	1541081	1.56	0 7	12517	103307	1.04	0.7	0.74	0.03	0.04
HIGH	3835	730682	1.55	0.7	4849	73 06 03	1.50	0.7	0.73	0.02	0.02
				•	0407	110491	1.55	0./	0.72	-0.00	-0.00
RACE:											
HHITE	12755	2509355	1.56	0.7	19521	9191995	3 64				_
BLACK	2049	248403	1.69	0.4	1517	721187	1.30	0.7	0.72	0.00	0.00
ASIAN-AMERICAN	189	27173	1 54	0.7	3537	361107	1.78	0.8	0.76	0.08 ×	0.11
AMERICAN INDIAN	185	30651	1 45	0.7	337	30257	1.42	0.7	9.66	-0.13	-0.19
MEXICAN-AMERICAN	544	71397	1 55	0.7	204	20810	1.79	0.8	0.74	0.14	0.19
PUERTO RICAN	στ	0409	1,33	0.7	1835	97749	1.61	0.7	0.71	0.07	0.09
OTHER HISPANIC	110	7400	1.72	0.7	297	17181	1.71	0.7	0.73	-0.01	-0.01
	/	10414	1.47	U./	924	62567	1.66	0.7	0.74	0.16	0.22
SCHOOL TYPE:											
PUBLIC	14769	2672242	1 50	0 7							
PRIVATE	4/07	14954	1.30	0.7	23861	2649980	1.61	0.7	0.73	0.02	0.03
CATHOLIC	1022	236814	1.44	<b>U.O</b>	860	102920	1.44	0.7	0.65	-0.00	-0.00
	1466	C 34010	1.40	U./	2645	196428	1.48	0.7	0.68	0.00	0.00
SEOGRAPHIC REGION:											
NORTHEAST	3578	796968	1 40	0 7							
NORTH CENTRAL	4521	010274	1.00	0.7	5550	681697	1.66	0.8	0.75	0.06 #	0.08
SOUTH	5499	707274	1.37	0.7	7869	845980	1.61	0.7	0.73	0.02	0.02
NEST	2051	704703 520302	1.53	U./	8958	887846	1.57	0.7	0.72	0.04	0.05
	L 733	360376	1.5/	U./	4989	533804	1.52	0.7	0.71	-0.05	-0.06
CURRICULUM:											
GENERAL	5544	054007	1 49				_				
ACADEMIC	475 <b>8</b>	1280011	1.02	0.7	9966	1076354	1.63	0.7	0.74	0.01	0.02
VOCATIONAL	6127	477211	1.24	0.7	10356	1119967	1.53	0.7	0.71	-0.00	-0.00
	4167	0/3011	1.34	U./	6645	709764	1.62	0.7	0.73	0.03	0.04
CONTINITY TYPE:											
URBAN	4504	779534	1 57	0.7	4949	E66/ ->					
SUBURBAN	7884	1526197	1.54	0.7	17104	984651 7456746	1.59	0.7	0.73	0.02	0.02
RURAL	3450	634045	1 64	0.7	73740	1458/40	1.57	0.7	0.72	-0.01	-0.02
			4.30	<b>v</b> ./	1960	405937	1.64	0.7	0.74	0.08 *	0.11

#SIGNIFICANT AT .05 OR LESS

14.1



# Changes in Self-Esteem and Locus Control

	Me an 1972	Me an 1980	Differ- ence	Effect Size
Self-Esteem (Scale: 1 = Disagree to 4 = Agree)				
I feel I am a person of worth.	3.24	3.29	0.05*	0.08
I am able to do things as well as most other people.	3.18	3.20	0.11*	0.19
I take a positive attitude toward myself.	3.13	3.25	0.12*	0.18
On the whole, I am satisfied with myself.	2.90	3.04	0.14*	0.20
Self-Esteem Total	12.45	12.78	0.42	
Means =	3.11	3.19		
Locus of Control (Scale: 1 = Agree to 4 = Disagree)				
Good luck is more important than hard work.	3 <b>. 3</b> 0	3.19	-0.11*	-0.16
Planning only makes a person unhappy.	3.04	3.06	0.02	0.02
Every time I try to get ahead, somethin stops me.	ng 2.92	2.86	-0.06*	-0.08
People who accept their condition in life are happier than those who try	2 80	2 66	-0 1/*	-0 15
Lo change things.	12.00	11 77	-0.29	0.13
	2 01	2 0/	V. 27	
me ans =	3.01	2.94		

\*Significant at .05 or less

1980, however, scores on three of the four items had decreased, moving the students in the direction of external control but still above the scale midpoint.

Changes toward greater external control were greater for females than for males, for low and middle SES than high SES students, and for students in the general and vocational curricula than for students in the academic curriculum. (See Tables 6-33 to 6-36.)

In summary, students increased in self-esteem between 1972 and 1980. However, locus of control, while still on the internal end of the scale, moved in the direction of greater external control in 1980 than in 1972. In short, between 1972 and 1980 students became more self-confident but less sure of their ability to control the course of their own lives.

# I FEEL I AM A PERSON OF WORTH ON AN EQUAL PLANE WITH OTHERS (1=DISAGREE STRONGLY; 4=AGREE STRONGLY)

	NLS 1972				HSB 1980						
	SAMPLE	WEIGHTED			SAMOLE	WETCHTED					
	N	N	MEAN	S.D.	N	N	MEAN	S.D.	S.D.	1980-1972 DIFFERENCE	EFFECT SIZE
TOTAL	15316	2800452	3.24	0.6	25845	2791949	3.29	0.6	ñ 59	0 05 *	0.00
SEX:								••••	•	0.05 -	0.00
MALE	76.04	3 7 0 0 4 5 4	7 95	- /							
FEMALE	7000	1370030	3.25	0.6	12011	1306819	3.32	0.6	0.59	0.06 *	0.10
	//00	1401030	3.23	0.0	13185	1418416	3.27	0.6	0.59	0.04 *	0.06
SES:											
LOH	4747	471749	<b>T</b> 0 <b>T</b>	• •							
MIDDLE	7286	3470442	3.23	0.6	7530	726974	3.24	0.6	0.61	0.00	0.00
HIGH	14.21	4010442	3.23	0.6	11904	1324691	3.28	0.6	0.59	0.05 ×	0.09
	3021	071042	3.29	0.6	5868	686669	3.38	0.6	0.58	0.09 ×	0.15
RACE :											
MHITE	11088	9749779									
BLACK	1000	2341331	3.23	0.6	18546	2208345	3.28	0.6	0.58	0.05 ×	0.08
ASTAN-AMERICAN	1450	232091	3.38	0.7	3338	303771	3.41	0.6	0.63	0.03	0.05
AMEDICAN THOTAM	10.4	24562	3.11	0.6	324	35469	3.29	0.6	0.62	0.18	0.05
MEYTCAN-AMERICAN	103	26967	3.22	0.6	189	19482	3.19	0.7	0.65	-0.03	-0.05
	493	65003	3.24	0.6	1698	91058	3.23	0.6	0.60	-0.01	-0.03
ATHER HIGHNIC	85	8917	3.29	0.7	260	15450	3.18	0.7	0.73	-0 11	-0.02
UTTER HISPANIC	106	16620	3.13	0.7	867	58083	3.27	0.6	0.57	0 14	-0.10
SCHOOL TYPE .									0.37	<b>V</b> · <b>4</b> ·	0.24
PODLIC DDIVATE	13724	2481635	3.24	0.6	22493	2504108	3.29	0.6	0.59	0 05 4	
	59	14119	3.24	0.6	826	98396	3.34	0.6	0 61	0.05 ~	0.00
CATHULIC	964	222883	3.26	0.6	2526	189445	3.31	0.6	0.60	0.05	0.10
										0.05	0.07
NORTHEACT											
NORTH ADDITE	3300	736230	3.20	0.6	5249	647538	3.28	0.6	0 50	0 00 ×	
ROUTH LENIKAL	4203	848269	3.22	0.6	7433	798468	3.27	0.6	0 60	0.00 ×	0.14
30014	5103	737030	3.29	0.6	8459	838529	3.32	0.6	0.57	0.04 *	0.07
WEST	2710	478923	3.27	0.6	4704	507415	3.30	0.6	0.60	0.03	0.04
								0.0	0.37	0.03	0.04
GENERAL	5112	871191	3.21	0.6	9325	1008373	3 24	n 4	0 50		
ACADEMIC	6382	1306066	3.29	0.6	9992	1083464	1 10	0.6	0.57	0.03	0.04
VOCATIONAL	3821	622892	3.19	0.6	6176	661426	3.37	0.0	0.59	0.09 ¥	0.16
							3.66	0.0	0.39	0.03	0.05
UNTUNITY TYPE:	•										
URBAN	4197	729743	3.27	0.6	5901	556133	1 12	<b>n</b> 4			
SUBURBAN	7363	1423578	3.24	0.6	12494	1 383009	3.30	0.0	0.61	0.05 *	0.09
RURAL	3390	589128	3.21	0.6	7450	852808	3.30	0.0	0.59	0.05 ¥	0.09
					7- <b>1</b> -7- <b>1</b> -	036000	3.20	U.0	0.59	0.04 ¥	0.07

\*SIGNIFICANT AT .05 OR LESS


-----

#### I AM ABLE TO DO THINGS AS WELL AS MOST OTHER PEOPLE (1=DISACREE STRONGLY; 4=AGREE STRONGLY)

SAMPLE     HEIGHTED N     SAMPLE     HIGH     S.D.     DIFFERENCE     SIZE       TALE     7740     1422698     3.24     0.6     12180     1326004     3.35     0.6     0.57     0.12 *     0.20       SES:     LOM     4406     67855     3.16     0.6     12069     1344589     3.27     0.6     0.57     0.11 *     0.20       RACE:     MITTE     12087     238174     3.17     0.6     13804     3.38     0.6     0.57     0.11 *     0.20       MITTE     12087     238173		NLS 1972				HSB 1980				_		
TUTAL     1555     2848164     3.18     0.6     26175     2829460     3.29     0.6     0.58     0.11 *     0.19       SEX: MALE     7740     1422698     3.24     0.6     1326004     3.35     0.6     0.57     0.12 *     0.20       SES:		SAMPLE N	WEIGHTED N	MEAN	S.D.	SAMPLE N	WEIGHTED	MEAN	S.D.	POOLED S.O.	1980-1972 DIFFERENCE	EFFEC SIZE
SEX:     TALE     7740     1422698     3.24     0.6     12180     1326004     3.35     0.6     0.57     0.12 *     0.20       SES:     LOH     4406     678554     3.16     0.6     1526004     3.23     0.6     0.57     0.12 *     0.20       SES:     LOH     4406     678554     3.16     0.6     7656     739074     3.23     0.6     0.57     0.13 *     0.21       HIGH     3670     699936     3.25     0.6     5884     690313     3.38     0.6     0.57     0.11 *     0.20       MHTE     12087     238174     3.17     0.6     18000     2239281     3.28     0.6     0.57     0.11 *     0.20       MARE:     MHTE     12087     238174     3.17     0.6     18000     2239281     3.28     0.6     0.57     0.11 *     0.20       MARE:     1320     238173     3.10     0.6     0.53     0.08 *     0.12       MARE:     1320	TOTAL	15556	2848164	3.18	0.6	26175	2829460	3.29	0.6	0.58	0.11 *	0.19
MALE   7740   1422698   3.24   0.6   12100   1326004   3.35   0.6   0.57   0.12 *   0.20     SES:   LOM   4406   678554   3.16   0.6   7562   73074   3.23   0.6   0.59   0.01 *   0.20     SES:   LOM   4406   678554   3.16   0.6   7656   739074   3.23   0.6   0.57   0.12 *   0.21     HIGM   3670   699305   3.25   0.6   589   690313   3.36   0.6   0.57   0.11 *   0.24     RACE:   MITT   12067   2381764   3.17   0.6   18600   2239281   3.26   0.6   0.57   0.11 *   0.24     BLACK   19207   2331764   3.17   0.6   18600   2239281   3.26   0.6   0.57   0.11 *   0.22     BLACK   19207   233104   3.30   0.6   0.58   0.22 *   0.30     MITERICAN   168   23732   3.20   0.6   1733   93221   3.24   0.6	SEX:											
FEHALE     7812     1424701     3.12     0.6     13326     1435012     3.24     0.6     0.58     0.11     *     0.20       SES:           0.6     0.58     0.11     *     0.20       MIDULE     7429     1461128     3.16     0.6     12069     1344589     3.27     0.6     0.57     0.12     0.21       MITE     12067     2381784     3.17     0.6     13336     304411     3.38     0.6     0.57     0.11     0.20       MITE     12067     2381784     3.17     0.6     13336     304411     3.38     0.6     0.57     0.11     0.20       MARE     12087     2381784     3.17     0.6     13800     2239201     3.28     0.6     0.57     0.11     0.20       BLCK     1929     233204     3.30     0.6     1335     334     0.6     0.58     0.11     0.20       MITE     1920	MALE	7740	1422698	3.24	0.6	12180	1326004	3.35	0.6	0.57	0.12 ×	0.20
SES:   LOH   4406   67854   3.16   0.6   7656   739074   3.23   0.6   0.59   0.07 *   0.13     HIDDLE   7429   1461128   3.16   0.6   12069   1344589   3.27   0.6   0.57   0.12 *   0.21     HIGH   3670   699936   3.25   0.6   5884   690313   3.38   0.6   0.57   0.13 *   0.24     MITE   12087   231744   3.17   0.6   128000   2239281   3.28   0.6   0.57   0.11 *   0.20     ASIAN-AMERICAN   106   23732   3.12   0.6   329   3514   3.34   0.6   0.63   0.08 *   0.12     ASIAN-AMERICAN   106   23732   3.12   0.6   1733   93221   3.24   0.6   0.60   0.09   0.15     GUTHE MIDIAN   60   8048   3.20   0.6   1733   93221   3.24   0.6   0.60   0.09   0.15     SCHOOL TYPE:   PIRILC   131   17113   3.20   0.6   2251 <td>FEMALE</td> <td>7812</td> <td>1424701</td> <td>3.12</td> <td>0.6</td> <td>13326</td> <td>1435012</td> <td>3.24</td> <td>0.6</td> <td>0.58</td> <td>0.11 *</td> <td>0.20</td>	FEMALE	7812	1424701	3.12	0.6	13326	1435012	3.24	0.6	0.58	0.11 *	0.20
LOH     4406     676554     3.16     0.6     7566     739074     3.23     0.6     0.59     0.07     *     0.13       HIDDLE     7429     1461128     3.16     0.6     12069     1344569     3.27     0.6     0.57     0.12     *     0.21       HIGH     3070     699936     3.25     0.6     5884     690313     3.8     0.6     0.57     0.11     *     0.22       RACE:     HITE     12087     2381784     3.17     0.6     138000     2239281     3.28     0.6     0.57     0.11     *     0.20       BLACK     1929     233204     3.30     0.6     3326     39441     3.34     0.6     0.58     0.22 *     0.39       AIERICAN     168     23732     3.12     0.6     1733     9321     3.24     0.6     0.60     0.58     0.22 *     0.39       AIERICAN-ARERICAN     36     8846     3.28     0.7     275     15798     3.19	SES:											
HIDOLE   7429   1461120   3.16   0.6   12069   134589   3.27   0.6   0.57   0.12 *   0.21     HIGH   3670   699936   3.25   0.6   5884   690313   3.38   0.6   0.57   0.13 *   0.24     RACE:	LOW	4406	678554	3.16	0.6	7656	739074	3.23	0.6	0.59	0.07 ×	0.13
HIGH   3670   699936   3.25   0.6   5884   690313   5.38   0.6   0.57   0.13   *   0.24     RACE:   MITTE   12087   233204   3.17   0.6   18800   2239281   3.28   0.6   0.57   0.11   *   0.24     BLACK   1929   233204   3.30   0.6   3336   304411   3.38   0.6   0.57   0.11   *   0.20     ALREL   168   23722   3.12   0.6   3236   304411   3.38   0.6   0.57   0.11   *   0.20     AHERICAN   169   2330   3.16   0.7   188   19154   3.20   0.6   0.62   0.04   0.07     MEXICAN-AHERICAN   515   67732   3.20   0.6   1733   93221   3.24   0.6   0.60   0.05   0.08     PUBLIC   111   17113   3.20   0.6   22814   2541760   3.29   0.6   0.58   0.11   *   0.19     PUBLIC   13947   2528110   3.	MIDDLE	7429	1461128	3.16	0.6	12069	1344589	3.27	0.6	0.57	0.12 ×	0.21
RACE:     HITE     12087     2381784     3.17     0.6     18000     2239281     3.28     0.6     0.57     0.11     *     0.20       BLACK     1929     233204     3.30     0.6     3336     304411     3.38     0.6     0.63     0.08     *     0.12       ASIAN-AHERICAN     168     23732     3.12     0.6     329     35914     3.34     0.6     0.63     0.08     *     0.12       AHERICAN INDIAN     169     28130     3.16     0.7     188     19154     3.20     0.6     0.62     0.04     0.07       MEXICAN-AHERICAN     86     8848     3.28     0.7     275     15798     3.19     0.7     0.73     -0.09     -0.12       OTHER HISPANIC     111     17113     3.20     0.6     823     98946     3.33     0.6     0.61     0.61     0.59     0.11     *     0.19       PUBLIC     13947     2528110     3.18     0.6     2538     1	HIGH	3670	699936	3.25	0.6	5884	690313	3.38	0.6	0.57	0.13 *	0.24
MHITE     12087     2381784     3.17     0.6     18800     2239281     3.28     0.6     0.57     0.11     #     0.20       BLACK     1929     233204     3.30     0.6     3336     304411     3.38     0.6     0.63     0.08     0.12       ASIAN-AMERICAN     168     23732     3.12     0.6     329     35914     3.34     0.6     0.63     0.08     0.12       ANIERICAN INDIAN     169     28130     3.16     0.7     188     19154     3.20     0.6     0.62     0.04     0.07       MEXICAN-AMERICAN     515     67732     3.20     0.6     1733     93221     3.24     0.6     0.60     0.60     0.05     0.08       PUERTO RICAN     86.48     3.28     0.7     273     15796     3.19     0.7     0.73     -0.09     0.15       SCHOOL TYPE:	RACE:											
BLACK     1929     233204     3.30     0.6     3336     304411     5.38     0.6     0.63     0.08     #     0.12       ASIAN-AMERICAN     160     23752     3.12     0.6     329     35914     3.34     0.6     0.63     0.08     #     0.12       ATIERICAN     160     23732     3.12     0.6     329     35914     3.34     0.6     0.62     0.04     0.07       MEXICAN-AMERICAN     515     67732     3.20     0.6     1733     93221     3.24     0.6     0.60     0.05     0.08       PUBLIC     13947     2528110     3.18     0.6     22814     2541780     3.29     0.6     0.58     0.11     *     0.19       PUBLIC     13947     2528110     3.18     0.6     22814     2541780     3.29     0.6     0.58     0.11     *     0.19       PUBLIC     13947     2528110     3.18     0.6     2538     188753     3.29     0.6     0.59	MHITE	12087	2381784	3.17	0.6	18800	2239281	3.28	0.6	0.57	0.11 ×	0.20
ASIAN-AMERICAN   168   23732   3.12   0.6   329   35914   3.34   0.6   0.56   0.22 *   0.30     AHIERICAN   169   28130   3.16   0.7   188   19154   3.20   0.6   0.62   0.04   0.07     MEXICAN-AMERICAN   86   8648   3.28   0.7   275   15798   3.19   0.7   0.73   -0.09   -0.12     OTHER MISPANIC   111   17113   3.20   0.6   880   59325   3.30   0.6   0.61   0.09   0.15     School. Type:   PUEIC   13947   2520110   3.18   0.6   22014   2541780   3.29   0.6   0.58   0.11 *   0.19     PRIVATE   63   14993   3.12   0.6   823   98946   3.33   0.6   0.60   0.21   0.35     CATHOLIC   966   222262   3.19   0.6   2538   188735   3.29   0.6   0.59   0.11 *   0.19     SCORAPHIC REGION:	BLACK	1929	233204	3.30	0.6	3336	304411	3.38	0.6	0.63	0.08 ×	0.12
ATIERICAN INDIAN   169   28130   3.16   0.7   188   19154   3.20   0.6   0.62   0.04   0.07     MEXICAN-AMERICAN   515   67732   3.20   0.6   1733   93221   3.24   0.6   0.60   0.05   0.09     PUERTO RICAN   8848   3.28   0.7   275   15798   3.19   0.7   0.73   -0.09   -0.12     OTHER HISPANIC   111   17113   3.20   0.6   880   59325   3.30   0.6   0.61   0.09   0.15     SCHOOL TYPE:   PUBLIC   13947   2528110   3.18   0.6   22814   2541780   3.29   0.6   0.58   0.11 *   0.19     PRIVATE   63   14993   3.12   0.6   823   98946   3.33   0.6   0.60   0.21   0.35     CATHOLIC   966   22262   3.19   0.6   2538   188735   3.29   0.6   0.59   0.10 *   0.16     SEOGRAPHIC REGION:	ASIAN-AMERICAN	168	23732	3.12	0.6	329	35914	3.34	0.6	0.58	0.22 ×	0.39
MEXICAN-AHERICAN PUERTO RICAN     515     67732     3.20     0.6     1733     93221     3.24     0.6     0.60     0.05     0.08       PUERTO RICAN OTHER HISPANIC     86     8848     3.28     0.7     275     15798     3.19     0.7     0.73     -0.09     -0.12       OTHER HISPANIC     111     17113     3.20     0.6     880     59325     3.30     0.6     0.61     0.09     0.15       SCHOOL TYPE: PUBLIC     13947     2528110     3.18     0.6     22814     2541780     3.29     0.6     0.58     0.11     *     0.19       PRIVATE     63     14993     3.12     0.6     823     98946     3.33     0.6     0.60     0.21     0.35       CATHOLIC     966     222262     3.19     0.6     2538     188735     3.29     0.6     0.59     0.10     *     0.16       SECOGRAPHIC REGION:     0     0     6     7526     810254     3.27     0.6     0.57     0.1	AMERICAN INDIAN	169	28130	3.16	0.7	188	19154	3.20	0.6	0.62	0.04	0.07
PUERTO RICAN   86   8848   3.28   0.7   275   15798   3.19   0.7   0.73   -0.09   -0.12     OTHER HISPANIC   111   17113   3.20   0.6   880   59325   3.30   0.6   0.61   0.09   0.15     SCHOOL TYPE:   PUBLIC   13947   2528110   3.18   0.6   22814   2541780   3.29   0.6   0.58   0.11 *   0.19     PRIVATE   63   14993   3.12   0.6   823   98946   3.33   0.6   0.60   0.21   0.35     GATHOLIC   966   22262   3.19   0.6   2538   188755   3.29   0.6   0.59   0.10 *   0.16     GEOGRAPHIC REGION:   NORTHEAST   3357   751776   3.14   0.6   7526   810254   3.27   0.6   0.57   0.11 *   0.19     SOUTH   S137   742617   3.22   0.6   8533   847062   3.30   0.6   0.57   0.11 *   0.12     NORTHEAST   2792   494448   3.20   0.6	MEXICAN-AMERICAN	515	67732	3.20	0.6	1733	93221	3.24	0.6	0.60	0.05	0.08
OTHER HISPANIC     111     17113     3.20     0.6     880     59325     3.30     0.6     0.61     0.09     0.15       SCHOOL TYPE:     PUBLIC     13947     2528110     3.18     0.6     22814     2541780     3.29     0.6     0.58     0.11     *     0.19       PRIVATE     63     14993     3.12     0.6     823     98946     3.33     0.6     0.60     0.21     0.35       CATHOLIC     966     222262     3.19     0.6     2538     188735     3.29     0.6     0.59     0.11 *     0.19       SOUTH     966     222262     3.19     0.6     2538     188735     3.29     0.6     0.59     0.10 *     0.16       SECOGRAPHIC REGION:     9007     80171     3.13     0.6     7526     810254     3.27     0.6     0.57     0.11 *     0.19       SOUTH     \$1337     742617     3.22     0.6     6357     64762     3.30     0.6     0.57     0.11 * </td <td>PUERTO RICAN</td> <td>86</td> <td>8848</td> <td>3.28</td> <td>0.7</td> <td>275</td> <td>15798</td> <td>3.19</td> <td>0.7</td> <td>0.73</td> <td>-0.09</td> <td>-0.12</td>	PUERTO RICAN	86	8848	3.28	0.7	275	15798	3.19	0.7	0.73	-0.09	-0.12
SCHOOL TYPE:   13947   2528110   3.18   0.6   22814   2541780   3.29   0.6   0.58   0.11 *   0.19     PRIVATE   63   14993   3.12   0.6   823   98946   3.33   0.6   0.60   0.21   0.35     CATHOLIC   966   222262   3.19   0.6   2538   188735   3.29   0.6   0.59   0.10 *   0.16     GEOGRAPHIC REGION:	OTHER HISPANIC	111	17113	3.20	0.6	880	59325	3.30	0.6	0.61	0.09	0.15
PUBLIC   13947   2528110   3.18   0.6   22814   2541780   3.29   0.6   0.58   0.11 *   0.19     PRIVATE   63   14993   3.12   0.6   823   98946   3.33   0.6   0.60   0.21   0.35     CATHOLIC   966   222262   3.19   0.6   2538   188735   3.29   0.6   0.59   0.10 *   0.16     GEOGRAPHIC REGION:	SCHOOL TYPE:											
PRIVATE   63   14993   3.12   0.6   823   98946   3.33   0.6   0.60   0.21   0.35     CATHOLIC   966   222262   3.19   0.6   2538   188735   3.29   0.6   0.59   0.10   *   0.16     GEOGRAPHIC REGION:     NORTHEAST   3357   751776   3.14   0.6   5330   657742   3.28   0.6   0.58   0.14   *   0.24     NORTH CENTRAL   4270   859323   3.16   0.6   7526   810254   3.27   0.6   0.57   0.11   *   0.19     SOUTH   5137   742617   3.22   0.6   8537   847062   3.30   0.6   0.59   0.07   *   0.12     WEST   2792   494448   3.20   0.6   4782   514401   3.32   0.6   0.57   0.10   *   0.17     ACADEHIC   6485   1324975   3.23   0.6   10044   1090398   3.38   0.6   0.59   0.11   *   0.18 <t< td=""><td>PUBLIC</td><td>13947</td><td>2528110</td><td>3.18</td><td>0.6</td><td>22814</td><td>2541780</td><td>3.29</td><td>0.6</td><td>0.58</td><td>0.11 ×</td><td>0.19</td></t<>	PUBLIC	13947	2528110	3.18	0.6	22814	2541780	3.29	0.6	0.58	0.11 ×	0.19
CATHOLIC   966   222262   3.19   0.6   2538   188735   3.29   0.6   0.59   0.10 *   0.16     GEOGRAPHIC REGION: NORTHEAST   3357   751776   3.14   0.6   5330   657742   3.28   0.6   0.59   0.10 *   0.16     SOUTH NORTH CENTRAL   4270   859323   3.16   0.6   7526   810254   3.27   0.6   0.57   0.11 *   0.19     SOUTH NEST   2792   494448   3.20   0.6   4782   514401   3.32   0.6   0.57   0.11 *   0.12     GENERAL   5214   892139   3.14   0.6   9452   1023734   3.24   0.6   0.57   0.10 *   0.17     GENERAL   5214   892139   3.14   0.6   9452   1023734   3.24   0.6   0.57   0.10 *   0.17     ACADEMIC   6485   1324975   3.23   0.6   10344   1090398   3.38   0.6   0.59   0.11 *   0.18     CURRICULM:   GENERAL   5214   8924975   3.23 <th< td=""><td>PRIVATE</td><td>63</td><td>14993</td><td>3.12</td><td>0.6</td><td>823</td><td>98946</td><td>3.33</td><td>0.6</td><td>0.60</td><td>0.21</td><td>0.35</td></th<>	PRIVATE	63	14993	3.12	0.6	823	98946	3.33	0.6	0.60	0.21	0.35
GEOGRAPHIC REGION:       NORTHEAST     3357     751776     3.14     0.6     5330     657742     3.28     0.6     0.58     0.14     0.24       NORTHEAST     4270     859323     3.16     0.6     7526     810254     3.27     0.6     0.57     0.11     0.19       SOUTH     5137     742617     3.22     0.6     8537     847062     3.30     0.6     0.59     0.07     w     0.12       WEST     2792     494448     3.20     0.6     4782     514401     3.32     0.6     0.57     0.12     w.22       CURRICULMM:       GENERAL     5214     892139     3.14     0.6     9452     1023734     3.24     0.6     0.57     0.10     ×     0.17       ACADEMIC     6485     1324975     3.23     0.6     10044     1090398     3.38     0.6     0.58     0.14     *     0.25       VOCATIONAL     3856     <th colspan="</td> <td>CATHOLIC</td> <td>966</td> <td>222262</td> <td>3.19</td> <td>0.6</td> <td>2538</td> <td>188735</td> <td>3.29</td> <td>0.6</td> <td>0.59</td> <td>0.10 *</td> <td>0.16</td>	CATHOLIC	966	222262	3.19	0.6	2538	188735	3.29	0.6	0.59	0.10 *	0.16
NORTHEAST   3357   751776   3.14   0.6   5330   657742   3.28   0.6   0.58   0.14 *   0.24     NORTH CENTRAL   4270   859323   3.16   0.6   7526   810254   3.27   0.6   0.57   0.11 *   0.19     SOUTH   5137   742617   3.22   0.6   8537   847062   3.30   0.6   0.59   0.07 *   0.12     WEST   2792   494448   3.20   0.6   4782   514401   3.32   0.6   0.57   0.12 *   0.22     CURRICULUM:	GEOGRAPHIC REGION:											
NORTH CENTRAL   4270   859323   3.16   0.6   7526   810254   3.27   0.6   0.57   0.11 *   0.19     SOUTH   5137   742617   3.22   0.6   8537   847062   3.30   0.6   0.59   0.07 *   0.12     WEST   2792   494448   3.20   0.6   4782   514401   3.32   0.6   0.57   0.12 *   0.22     CURRICULUM:   GENERAL   5214   892139   3.14   0.6   9452   1023734   3.24   0.6   0.57   0.12 *   0.22     CURRICULUM:   GENERAL   5214   892139   3.14   0.6   9452   1023734   3.24   0.6   0.57   0.11 *   0.17     ACADEMIC   6485   1324975   3.23   0.6   10044   1090398   3.38   0.6   0.58   0.14 *   0.25     VOCATIONAL   3856   630747   3.12   0.6   6312   674922   3.23   0.6   0.59   0.11 *   0.18     COMPLINITY   TYPE:   URBAN   4285   7	NORTHEAST	3357	751776	3.14	0.6	5330	657742	3.28	0.6	0.58	0.14 ×	0.24
SOUTH NEST   5137   742617   3.22   0.6   8537   847062   3.30   0.6   0.59   0.07 *   0.12     WEST   2792   494448   3.20   0.6   4782   514401   3.32   0.6   0.57   0.12 *   0.22     CURRICULUM: GENERAL   5214   892139   3.14   0.6   9452   1023734   3.24   0.6   0.57   0.10 *   0.17     ACADEMIC   6485   1324975   3.23   0.6   10044   1090398   3.38   0.6   0.58   0.14 *   0.25     VOCATIONAL   3856   630747   3.12   0.6   6312   674922   3.23   0.6   0.59   0.11 *   0.18     COMPLINITY TYPE:   URBAN   4285   744759   3.21   0.6   5958   561741   3.33   0.6   0.60   0.12 *   0.20     SUBURBAN   7445   1443463   3.18   0.6   12666   1403528   3.29   0.6   0.58   0.12 *   0.20     RURAL   3444   598592   3.15   0.6   <	NORTH CENTRAL	4270	859323	3.16	0.6	7526	810254	3.27	0.6	0.57	0.11 ×	0.19
WEST   2792   494448   3.20   0.6   4782   514401   3.32   0.6   0.57   0.12 *   0.22     CURRICULUM: GENERAL   5214   892139   3.14   0.6   9452   1023734   3.24   0.6   0.57   0.10 *   0.17     ACADEMIC   6485   1324975   3.23   0.6   10344   1090398   3.38   0.6   0.58   0.14 *   0.25     VOCATIONAL   3856   630747   3.12   0.6   6312   674922   3.23   0.6   0.59   0.11 *   0.18     COMPLINITY TYPE:   URBAN   4285   744759   3.21   0.6   5958   561741   3.33   0.6   0.60   0.12 *   0.20     SUBURBAN   7445   1443463   3.18   0.6   12666   1403528   3.29   0.6   0.58   0.12 *   0.20     RURAL   3444   598592   3.15   0.6   7551   864192   3.26   0.6   0.57   0.11 *   0.19	SOUTH	5137	742617	3.22	0.6	8537	847062	3.30	0.6	0.59	0.07 ×	0.12
CURRICULUM:     GENERAL   5214   892139   3.14   0.6   9452   1023734   3.24   0.6   0.57   0.10   *   0.17     ACADEMIC   6485   1324975   3.23   0.6   10044   1090398   3.38   0.6   0.58   0.14   0.25     VOCATIONAL   3856   630747   3.12   0.6   6312   674922   3.23   0.6   0.59   0.11   *   0.18     COMMUNITY TYPE:   uRBAN   4285   744759   3.21   0.6   5958   561741   3.33   0.6   0.60   0.12   *   0.20     SUBURBAN   7445   1443463   3.18   0.6   12666   1403528   3.29   0.6   0.58   0.12   *   0.20     RURAL   3444   598592   3.15   0.6   7551   864192   3.26   0.6   0.57   0.11   *   0.19	WEST	2792	494448	3.20	0.6	4782	514401	3.32	0.6	0.57	0.12 *	0.22
GENERAL   5214   892139   3.14   0.6   9452   1023734   3.24   0.6   0.57   0.10 *   0.17     ACADEMIC   6485   1324975   3.23   0.6   10044   1090398   3.38   0.6   0.58   0.14 *   0.25     VOCATIONAL   3856   630747   3.12   0.6   6312   674922   3.23   0.6   0.59   0.11 *   0.18     COMPLINITY TYPE:   URBAN   4285   744759   3.21   0.6   5958   561741   3.33   0.6   0.60   0.12 *   0.20     SUBURBAN   7445   1443463   3.18   0.6   12666   1403528   3.29   0.6   0.58   0.12 *   0.20     RURAL   3444   598592   3.15   0.6   7551   864192   3.26   0.6   0.57   0.11 *   0.19	CURRICULUM:											
ACADEMIC   6485   1324975   3.23   0.6   10044   1090398   3.38   0.6   0.58   0.14 *   0.25     VOCATIONAL   3856   630747   3.12   0.6   6312   674922   3.23   0.6   0.59   0.11 *   0.18     COMPLINITY TYPE:   URBAN   4285   744759   3.21   0.6   5958   561741   3.33   0.6   0.60   0.12 *   0.20     SUBURBAN   7445   1443463   3.18   0.6   12666   1403528   3.29   0.6   0.58   0.12 *   0.20     RURAL   3444   598592   3.15   0.6   7551   864192   3.26   0.6   0.57   0.11 *   0.19	GENERAL	5214	892139	3.14	0.6	9452	1023734	3.24	0.6	0.57	0.10 ×	0.17
VOCATIONAL     3856     630747     3.12     0.6     6312     674922     3.23     0.6     0.59     0.11     *     0.18       COMPLINITY     TYPE:     URBAN     4285     744759     3.21     0.6     5958     561741     3.33     0.6     0.60     0.12     *     0.20       SUBURBAN     7445     1443463     3.18     0.6     12666     1403528     3.29     0.6     0.58     0.12     *     0.20       RURAL     3444     598592     3.15     0.6     7551     864192     3.26     0.6     0.57     0.11     *     0.19	ACADEMIC	6485	1324975	3.23	0.6	10044	1090398	3.38	0.6	0.58	0.14 ×	0.25
COMPLINITY TYPE:     URBAN     4285     744759     3.21     0.6     5958     561741     3.33     0.6     0.60     0.12 *     0.20       SUBURBAN     7445     1443463     3.18     0.6     12666     1403528     3.29     0.6     0.58     0.12 *     0.20       RURAL     3444     598592     3.15     0.6     7551     864192     3.26     0.6     0.57     0.11 *     0.19	VOCATIONAL	3856	630747	3.12	0.6	6312	674922	3.23	0.6	0.59	0.11 *	0.18
URBAN     4285     744759     3.21     0.6     5958     561741     3.33     0.6     0.60     0.12     *     0.20       SUBURBAN     7445     1443463     3.18     0.6     12666     1403528     3.29     0.6     0.58     0.12     *     0.20       RURAL     3444     598592     3.15     0.6     7551     864192     3.26     0.6     0.57     0.11     *     0.19	CONNUNITY TYPE:											
SUBURBAN     7445     1443463     3.18     0.6     12666     1403528     3.29     0.6     0.58     0.12 *     0.20       RURAL     3444     598592     3.15     0.6     7551     864192     3.26     0.6     0.57     0.11 *     0.19	URBAN	4285	744759	3.21	0.6	5958	561741	3.33	0.6	0.60	0.12 ×	0.20
RURAL 3444 598592 3,15 0.6 7551 864192 3,26 0.6 0.57 0.11 * 0.19	SUBURBAN	7445	1443463	3.18	0.6	12666	1403528	3.29	0.6	0.58	0.12 *	0,20
	RURAL	3444	598592	3.15	0.6	7551	864192	3.26	0.6	0.57	0.11 *	0.19

\*SIGNIFICANT AT .05 OR LESS



#### I TAKE A POSITIVE ATTITUDE TOWARD MYSELF (1=DISAGREE STRONGLY; 4=AGREE STRONGLY)

	NLS 1972			HSB 1980							
	SAMPLE N	MEIGHTED N	MEAN	\$.D.	SAMPLE N	WE IGHTED N	MEAN	\$.D.	POOLED S.D.	1980-1972 DIFFERENCE	EFFECT SIZE
TOTAL	14808	2712597	3.13	0.7	25980	2799578	3.25	0.6	0.65	0.12 *	0.18
SEX:											
MALE	7317	1347297	3.19	0.7	12078	1311224	Z 7.,	0.6	0 43		A 94
FEMALE	7487	1364534	3.06	0.7	13210	1417984	3.16	0.7	0.66	0.03 *	0.24
SES:											-
LON	4101	<b>632748</b>	3.13	Ô.7	76.02	720481	R 95	• 4	A 4 3	A A A Y	
MIDDLE	7082	1389360	3 11	0 7	11051	730401	3.66	U.6	0.67	0.09 #	0.14
HIGH	3578	682515	3.15	0.7	5867	686428	3.23	U.6 0.6	0.65 0.65	0.12 ¥ 0.14 *	0.19 0.21
RACE :							-				
MHITE	11551	2274517	3 30	07	10525	2266107		• •			
BLACK	1855	225008	J.10 1 17	0.7	10365	2204187	3.21	0.6	0.65	0.11 ×	0.16
ASTAN-AMERICAN	1055	223000	3.3/ 7 Al	V./	740	309097	3.51	0.6	0.64	0.14 ×	0.22
AMEDICAN TANTAN	137 158	66077 94451	2.01 2.01	V.O	540	36213	3.27	0.6	0.61	0.26 *	0.43
MEXTCANLAMEDICAN	150	2003J 41502	5.00	U./	170	19814	3.23	0.7	0.70	0.15	0.22
DIFOTA STCAN	737	013UC	3.21	U.O	1/11	926 36	3.23	0.6	0.63	0.02	0.03
ATTER TABANT	/7	0244	3.21	0.7	277	16008	3.28	0.6	0.66	0.08	0.12
VINER RESPANCE	109	16005	3.17	0.7	881	58913	3.32	0.6	0.60	0.15	0.25
SCHOOL TYPE:											
PUBLIC	13247	2399313	3.13	0.7	22626	2514594	1 25	0.4			
PRIVATE	61	15115	3.05	0.6	A24	67449	3.63	0.0	0.05	0.12 *	0.18
CATHOLIC	948	218672	3.09	0.7	2510	7/40L 107E91	3.20	U.6	U.64	0.21	0.33
	·	Levvit	3.97		C33V	10/361	3.22	0.0	0.66	0.12 *	9.19
GEOGRAPHIC REGION:											
NORTHEAST	3212	716589	3.06	0.7	5252	644872	3.22	0.6	0.67	0.16 *	n 94
NORTH CENTRAL	4093	824457	3.09	0.7	7466	801589	3.21	0.6	0.45	0.10 -	0.24
SOUTH	4841	700556	3.20	0.7	8490	841279	3.30	0.6	0.45	0.10 *	0.11
WEST	2662	470995	3.17	0.7	4772	511837	3.25	0.6	0.64	0.08 *	0.13
CURRICULUM:											
GENERAL	4866	832166	3.10	9.7	8368	1000804	7 23	<b>A</b> 4	A 4E	A y	
ACADEMIC	6324	1291562	3.16	0.7	10010	1007074	J.CI 7 20	U.0	V.05	U.11 #	0.17
VOCATIONAL	3617	588567	3.09	0.7	4214	1000027	J.67 7 91	U.0	0.65	0.14 #	0.21
		200207	3.47	V. /	0614	004341	3.21	U.0	0.65	0.13 *	0.19
COMPLINITY TYPE:											
URBAN	4105	712219	3.17	0.7	5976	560015	3.31	0.6	0.66	0.14 #	6 91
- SUBURBAN	7136	1384280	3.11	0.7	12558	1389267	3.24	0.6	0 65	0.17 -	0.21
RURAL	3211	558300	3.10	0.7	7446	850296	3.22	0.6	0.05	0 19 ×	0.17
				~ • • •	• • •	0	3.66	V.V	V.0-7	A.TC #	n.10

\*SIGNIFICANT AT .05 OR LESS



## ON THE NHOLE, I'M SATISFIED WITH MYSELF (1=DISAGREE STRONGLY; 4=AGREE STRONGLY)

	NLS 1972				HSB 1980						
	SAMPLE N	WEIGHTED	MEAN	S.D.	SAMPLE N	WEIGHTED N	MEAN	S.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	15223	2785681	2.90	0.7	26078	2810192	3.04	0.7	0.71	0.14 *	0.20
SEX:											
MALE	7549	1386188	2.87	0.8	12090	1313281	3.05	0.7	0.72	0 18 #	0 25
FEHALE	7670	1398728	2.93	0.7	13330	1429573	3.03	0.7	0.70	0.11 *	0.15
SES:											
LOH	4348	667800	2.90	0.8	7676	736826	3.00	0.7	0.74	0.10 #	0 13
MIDDLE	7219	1420398	2.90	0.7	12029	1337762	3.03	0.7	0.49	0.10 #	0.13
HIGH	3606	689400	2.90	0.7	5840	682299	3.10	0.7	0.70	0.20 *	0.29
RACE:											
MHITE	11828	2326669	2.90	0.7	18684	2221160	3.03	0.7	0.69	0.14 *	0 20
BLACK	1893	229745	2.90	0.9	3355	304796	3.10	0.8	0.84	0.20 #	0.24
ASIAN-AMERICAN	166	24174	2.87	0.7	334	36310	3.00	0.7	0.68	0.13	0.19
AMERICAN INDIAN	169	27733	2.89	0.8	192	19544	2.96	0.7	0.78	0.07	0 08
MEXICAN-AMERICAN	493	65004	2.94	0.8	1733	92157	3.03	0.7	0.72	0.09	0.13
PUERTO RICAN	86	8861	2.99	0.9	2.76	15865	3.07	0.8	0.82	0.08	0.10
OTHER HISPANIC	111	17503	3.01	0.8	879	58773	3.10	0.7	0.69	0.09	0.13
SCHOOL TYPE:											
PUBLIC	13638	2470362	2.90	0.7	22714	2524554	3.04	0.7	0.71	0.14 ×	0.20
PRIVATE	59	14237	2.97	0.7	814	96670	3.08	0.7	0.71	0.11	0.15
CATHOLIC	955	219149	2.93	0.7	2550	188968	3.05	0.7	0.69	0.12 *	0.17
GEOGRAPHIC REGION:											
NORTHEAST	3313	739286	2.91	0.7	5295	651196	3.03	0.7	0.70	0.12 *	0.18
NORTH CENTRAL	4150	837622	2.85	0.7	7494	805513	3.02	0.7	0.69	0.17 ×	0.24
SOUTH	5052	730416	2.93	0.8	8533	844575	3.06	0.7	0.74	0.13 *	0.18
HEST	2708	478357	2.93	0.8	4756	508909	3.05	0.7	0.71	0.12 *	0.17
CURRICULUM:											
GENERAL	5075	868250	2.88	0.8	9424	1018599	3.00	0.7	0.70	0.12 *	0.17
ACADEMIC	6342	1298832	2.90	0.7	9994	1079933	3.09	0.7	0.72	0.19 *	0.27
VOCATIONAL	3805	6182%	2.93	0.7	6298	672296	3.02	0.7	0.71	0.09 *	0.13
CONTLNITY TYPE:											
URBAN	4191	727557	2.92	8.0	5947	557778	3.07	0.7	0.74	0.15 ¥	0.20
SUBURBAN	7288	1412000	2.90	0.7	12625	1395685	Z.04	0.7	0.71	0.14 ×	0.20
RURAL	3373	586311	2.88	0.7	7506	856730	3.03	0.7	0.69	0.15 *	0.21

\*SIGNIFICANT AT .05 OR LESS

149

•



#### GOOD LUCK IS MORE IMPORTANT THAN HARD WORK FOR SUCCESS (1=AGREE STRONGLY; 4=DISAGREE STRONGLY)

	NLS 1972				HSB 1980						
	SAMPLE N	NE IGHTED	MEAN	S.D.	SAMPLE N	MEIGHTED N	MEAN	s.D.	POOLED S.D.	1980-1972 Difference	EFFEC SIZE
TOTAL	15652	2862222	3.30	0.7	25749	2782175	3.19	0.7	0.71	-0.11 *	-0.25
SEX:											
MALE	7668	1406585	3.25	07	11857	1298472	7 34	0 7	A 77	A 11 ×	• • •
FEMALE	7981	1455081	3.36	0.7	13226	1425314	3.25	0.7	0.73	-0.11 *	-0.16 -0.15
SES:											
LOW	6687	602069	T 21	0.7	7671	707511					
MTODI F	74 71	3444959	3.21	0.7	7531	/2/511	3.10	0.8	0.76	-0.il *	-0.15
NTCH	7471	1400232	3.33	0.7	11940	1323412	3.21	0.7	0.68	-0.12 <b>*</b>	-0.18
11200	3045	074/13	3.34	U. /	5799	679646	3.29	0.7	0.66	-0.06 ¥	-0.09
RACE:											
MHITE	12196	2397367	3.34	0.7	18510	2203419	3.24	0.7	0.67	-0.09 #	-0 14
BLACK	1920	232008	3.10	0.8	3237	295468	2.96	0.8	0.83	-0.14 #	-0.14
ASIAN-AMERICAN	172	24769	3.22	0.7	335	36173	3.14	0.7	0.72	-0.09	-0.17
AMERICAN INDIAN	173	28675	3.17	0.7	192	19373	2.98	0.8	0.79	-0.19	-0.12
MEXICAN-AMERICAN	513	67493	3.19	0.8	1704	91728	3.00	0.8	0 79	-0.19 #	-0.24
PUERTO RICAN	82	8575	3.10	0.7	273	15864	3.00	0.8	0.76	-0.10	-0.24
OTHER HISPANIC	114	17576	3.13	0.8	872	58728	3.02	0.8	0.78	-0.11	-0.15
SCHOOL TYPE:											
PUBLIC	14028	2530862	3 30	07	224 74	2409047	7 30	• •			
PRIVATE	65	15986	3.50	0.7	20434	2470007	3.10	0.7	0.71	-0.12 *	-0.16
CATHOLIC	969	222104	3.37	0.7	2510	187208	3.26	0.7	0.70	-0.20 -0.11 #	-0.28
											•••
MODTHEAST	7700	75/770									
NORTH CENTRAL	3302	/50/10	3.32	0.7	5244	646114	3.20	0.7	0.70	-0.12 ¥	-0.17
CONTRAL	4310	866280	3.30	0.7	7376	795518	3.23	0.7	0.68	-0.07 ¥	-0.11
LEET	51/0	/48243	3.29	0.7	8406	834346	3.14	0.8	0.73	-0.15 *	-0.21
ME31	2776	490982	3.31	0.7	4723	506197	3.22	0.7	0.71	-0.08 *	-0.12
CURRICULUN:											
GENERAL	5259	900149	3.24	0.7	9278	1004497	3.14	0.7	0.71	-0 11 #	-0.15
ACADEMIC	6468	1322351	3.38	0.6	9922	1074079	3.31	0.7	0.66	-0.07 #	-0.13
VOCATIONAL	3924	639419	3.23	0.7	6195	664893	3.10	0.8	0.75	-0.13 *	-0.17
CONTUNITY TYPE:											
URBAN	6257	738134	3.30	07	6817	647957	7 15	• •	A 77	<b>A B</b> 4 <b>M</b>	
SUBURBAN	7523	1456060	3.31	0 7	12670	1101111	J.13	0.0	0.75	~0.14 #	-0.20
RURAL	TARA	606319	3 30	0.7	7633	+ 30++++	3.66	U./	0.67	-0.09 *	-0.13
	3400	UVU J4 7	3.30	v./	/433	023611	3.1/	U./	0.71	-0.13 ¥	-0.19

**#SIGNIFICANT AT .05 OR LESS** 

- -

PLANNING ONLY MAKES A PERSON UNHAPPY SINCE PLANS HARDLY EVER WORK OUT ANYWAY (1=AGREE STRONGLY; 4=DISAGREE STRONGLY)

	NLS 1972 HSB 1980										
	SAMPLE	WEISHTED		e n	SAMPLE	WEIGHTED			POOLED	1980-1972	EFFECT
	N N	2	TIEAN	5.0.	N	N	MEAN	S.D.	S.D.	DIFFERENCE	SIZE
TOTAL	15470	2831431	3.04	0.8	25699	2776320	3.06	0.8	0.79	0.02	0.02
SEX:											
MALE	7557	1387505	2.97	0.8	11793	1291296	T 01				
FENALE	7909	1443160	3.11	0.8	13275	1429278	3.11	0.8	0.79	0.04	0.05
SES:											
LOW	4704	477790	2 00								
MTDDLF	4370	1466313	2.07	<b>J.8</b>	7550	728086	2.88	0.8	0.83	-0.01	-0.01
MTCH	7377	1499213	3.05	0.8	11857	1320766	3.08	0.8	0.78	0.02	0.03
NIGH	20 35	641410	3.16	0.7	5786	678094	3.25	0.7	0.73	0.08 ×	0.11
RACE:											
NHITE	12073	2 _ 29	3.08	0.8	18504	2204262	<b>X 10</b>		A 77		
BLACK	1881	227391	2.83	n 9	10207	202403	2 00	0.0	0.77	0.02	0.03
ASIAN-AMERICAN	181	26080	3 09	0.7	3223	272473	2.07	0.9	0.8/	0.06	0.06
AMERICAN INDIAN	175	28880	2 96	0.0	332	30097	3.14	0.8	0.79	0.06	0.08
MEXICAN-AMERICAN	501	46091	2 03	0.7	192	19447	2.95	0.8	0.86	0.11	0.12
PLIFOTO OTCAN		03701	2.02	0.0	1/14	91364	2.84	0.8	0.81	0.02	0.02
OTHED HYSDANTC	100	16(1)	2.00	0.8	273	15546	2.88	0.8	0.80	0.20	0.25
	100	12010	2.00	0.8	848	56625	2.87	0.9	0.85	0.01	0.01
SCHOOL TYPE:											
PUBLIC	13854	2509719	3.04	0.8	22403	2404423	7 AE				
PRIVATE	64	15627	2 00 2	0.0	004	C4740CJ	3.03	0.0	0.79	0.01	0.01
CATHOLIC	976	224097	3 08	0.0	2400	90U/1 30E494	3.22	0.7	0.75	0.12	0.17
		224077	5.00	0.0	2490	193050	3.13	0.7	0.75	0.04	0.06
GEOGRAPHIC REGION:											
NORTHEAST	3339	746860	3.00	0.8	5220	642042	3.03	0.8	0 86	F0 0	0 04
NORTH CENTRAL	4258	855717	3.06	0.8	7393	797902	3.08	0.0	0.00	0.03	0.04
SOUTH	5101	738083	3.02	0.8	1359	822605	3 00	0.0	0.77	~0.02	0.03
NEST	2772	490772	3.09	0.8	4727	506771	3.15	0.8	0.32	0.02	-0.03
									••••	0.00	0.00
CORRICULUM											
GENERAL	5187	889815	2.95	0.8	9266	1001187	2.97	0.8	0.80	0.02	0.03
ACADEMIC	6404	1309442	3.17	0.7	9879	1072654	3.23	3.7	0.74	0.06 *	0.05
VOCATIONAL	3878	631872	2.90	0.8	6194	662682	2.93	0.8	0.83	0.03	0.04
CONTUNITY TYPE:											
LIRBAN	4227	719190	7 65		5040						
SUBUDRAN	766/	/ JCJCU 1/376/F	3.05	v.a	5848	551031	5.02	0.8	0.62	-0.03	~0.03
	7414	143/545	3.05	8.0	12430	1376717	3.09	0.8	0.78	0.04 *	0.05
RUNAL	3435	000968	5.03	0.8	7421	₹`3 <b>5</b> 72	3.04	0.8	0.79	0.01	0.01

\*SIGNIFICANT AT .05 OR LESS



÷---

## EVERY TIME I TRY TO GET AHEAD, SOMETHING OR SOMEBODY STOPS ME (1=AGREE STRONGLY; 4=DISAGREE STRONGLY)

	NLS 1972				HSB 1980				_		
	SAMPLE N	NEIGHTED N	MEAN	\$.D.	SAMPLE N	NEIGHTED N	MEAN	\$.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	15017	2745586	2.92	0.7	25163	<u>.</u> 716731	2.86	0.7	0.72	-0.06 *	-0.08
SEX:											
MALE	7362	1352359	2.84	0.7	11596	1260280	2.81	0.7	0.73	-0.03	~0.04
FEMALE	7651	1392462	2.99	0.7	12939	1392621	2.90	0.7	0.69	-0.08 *	-0.12
SES:											
LOW	4269	657310	2.75	0.7	7386	711515	2.69	0.8	0.75	-0.07 ×	-0.09
NIDDLE	7162	1404285	2.92	0.7	11601	1292252	2.86	0.7	0.70	-0.06 *	-0.09
HIGH	3544	677157	3.06	0.6	5664	662532	3.06	0.7	0.66	-0.00	-0.00
RACE :											
MHITE	11683	2296381	2.95	0.7	10057	2152527	2.89	9.7	0.69	-0.06 *	-0.08
BLACK	1846	223168	2.74	0.8	3214	239883	2.70	0.8	0.82	-0.03	-0.04
ASIAN-AMERICAN	163	23549	2.91	0.8	326	35680	2.83	0.8	0.78	-0.08	-0.11
AMERICAN INDIAN	164	27286	2.70	0.8	187	18750	2.63	0.7	0.79	-0.07	-0.09
MEXICAN-AMERICAN	505	66762	2.74	0.7	1667	88913	2.77	0.7	0.73	0.03	0.04
PUERTO RICAN	79	8174	2.65	.3.0	270	15145	2.71	0.7	0.77	0.07	0.09
OTHER HISPANIC	106	16411	2.95	¢.7	848	57134	2.74	0.8	0.76	-0.21	-0.27
SCHOOL TYPE:											
PUBLIC	13451	2433444	2.91	0.7	21926	2440326	2.85	0.7	0.72	-0.07 ×	~0.09
PRIVATE	62	15251	3.08	0.8	783	93284	3.01	0.7	0.72	-0.07	-0.10
CATHOLIC	941	216036	2.98	0.7	2454	183121	2.95	0.7	0.68	-0.03	~0.05
GEOGRAPHIC REGION:											
NORTHEAST	3252	725437	2.96	0.7	5135	631351	2.90	0.7	0.71	-0.06 *	-0.08
NORTH CENTRAL	4140	829031	2.89	0.7	7209	777248	2.85	0.7	0.70	-0.05	-0.07
SOUTH	4933	713591	2.87	0.7	8208	811098	2.80	0.7	0.74	-0.07 *	-0.10
WEST	2692	477528	2.95	0.7	4611	497 <b>035</b>	2.92	0.7	0.70	-0.04	-0.05
CURRICULUM:											
GENERAL	4987	853575	2.81	0.7	9069	980741	2.78	0.7	0.72	-0.03	-0.04
ACADEMIC	6249	1277974	3.04	0.7	9690	1051863	3.02	0.7	0.67	-0.02	-0.03
VOCATIONAL	3781	614038	2.80	0.7	6 0 5 5	647116	2.72	0.8	0.75	-0.08 *	-0.11
CONHUNITY TYPE:											
URBAN	4118	710736	2.93	0.7	5751	539017	2.86	0.8	0.74	-0.07 ×	-0.09
SUBURBAN	7231	1402141	2.94	0.7	12194	1350614	2.89	0.7	0.70	-0.05 *	-0.07
RURAL	3308	574503	2.85	0.7	7218	827100	2.80	0.7	0.71	-0.04	-0.06

\*SIGNIFICANT AT .05 OR LESS

152



P - -

	NLS 1972				HSB 1980						
	SAMPLE	WEIGHTED	MEAN	S.D.	SAMPLE N	WEIGHTED N	MEAN	\$.D.	POOLED S.D.	1980-1972 DIFFERENCE	EFFECT SIZE
TOTAL	15041	2756993	2.80	0.9	24643	2662116	2.66	0.9	0.91	-0.14 *	-0.15
SEX:											
MALE	7359	1356289	2.79	0 0	11366	1234816	2 4 7		n 00	_0 17 #	-0.14
FEMALE	7678	1399938	2.80	0.9	12682	1365149	2.66	0.9	0.92	-0.14 *	-0.14
SES:											
LOW	4295	666903	2.61	0.0	7966	700840	2 47		0 01	. 0 34 #	
MIDDLE	7177	1411715	2 80	0.7	11 7 8 4	1946876	2.4/ 9 4E	0.9	0.91	~0.14 *	-0.16
HIGH	3525	673433	2,97	0.9	5511	647048	2.89	0.9	0.87	-0.09 *	-0.16
RACE:											
MHITE	11751	233 36AA	2 81		17708	23 00540	2 70	• •			
BLACK	1803	217484	2.03	3.0	1//00	2100500	2.70	0.9	0.90	-0.13 *	-0.15
ASTAN-AMERICAN	1005	217004	2.03	1.0	2107	204013	2.51	1.0	0.97	-0.12 *	-0.12
AMERICAN INDIAN	105	29543	2.07	0.7	215	33422	2.52	0.9	0.90	-0.17	-0.19
MEXICAN_AMERICAN	1/1	20301	2.41	0.9	180	18524	2.51	0.8	0.89	0.10	0.11
DUEDTO DTCAN	407	04010	2.3/	0.9	1647	87866	2.48	0.9	0.87	-0.10	-0.11
ATHED UTEDANTC	107	04/5	2.4/	1.0	261	15020	2.59	0.9	0.96	0.13	0.13
WINER HISPANIC	107	10122	2.70	1.0	839	56384	2.49	0.9	0.90	-0.21	-0.23
SCHOOL TYPE:											
PUBLIC	13464	2441235	2.79	0.9	21488	2390733	2.64	0.9	FP 0	_0 16 #	-0.14
PRIVATE	63	15660	2.84	1.0	781	946.90	2 82	n ¢	0 02	~0.15 ~	-0.10
CATHOLIC	953	219709	2.94	0.9	2374	176695	2.79	0.9	0.89	-0.15 *	-0.03
GEOGRAPHIC REGION:											
NOPTHEAST	3250	726530	2.84	0.9	4004	419195	2 72	n 9	n on	-0 17 *	0 14
NORTH CENTRAL	4179	840298	2.82	0.9	7043	764979	5.7E	0.7	0.70	-0.13 *	-0.14
SOUTH	4964	721911	2.71	3 0	8107	130717 001705	2.00 9 E7	0.7	0.07	-0.10 *	-0.18
HEST	2648	468254	2.82	0.9	4501	483237	2.73	0.9	0.89	-0.14 ×	-0.15
CUPPICULUM:											
GENERAL	5038	845103	2 68	• •		043747	9 E4				
ACADEMIC	4998	1977119	2.00	0.7	0090	701202	2.30	0.9	0.90	-0.12 *	-0.13
VOCATIONAL	3774	614474	2.58	0.9	5955	638917	2.87	0.9	0.88 0.91	-0.10 * -0.11 *	-0.12 -0.12
COMMINITY TYPE:										_	
LIRBAN	6120	716608	2 82	<b>n</b> o	EE4A	534454		• •			
SUBURBAN	7214	140010	2.0C 9 81	V.7	3300	762072	2.04	0.9	U. 92	-0.18 #	-0.19
RURAL	7610	1700JU/ E8187E	2.0J 9 47	0.7	77,400	1210001	2./1	0.9	0.90	-0.13 *	-0.14
	JJ46	3010/3	2.0/	<b>V.</b> 7	/1/7	824218	2.57	0.9	0.91	-0.08 ¥	-0.09

## PEOPLE WHO ACCEPT THEIR CONDITION IN LIFE ARE HAPPIER THAN THOSE WHO TRY TO CHANGE THINGS (1=AGREE STRONGLY; 4=DISAGREE STRONGLY)

\*SIGNIFICANT AT .05 OR LESS

153



Table 6-36

#### I. TIME SPENT ON HOMEWORK

The students also indicated on the questionnaires the amount of time they spent each week doing homework. The scale was from 1 = 0 to 5 hours a week to 3 = more than 10 hours a week.

The amount of homework done showed a decrease, with small effect size, from 1972 to 1980. As can be seen in Table 6-37, the mean went from 1.41 in 1972 to 1.31 in 1980, representing a decline from about 4.55 hours of homework per week in 1972 to  $4.0^{-1}$  hours of homework per week in 1980.

Females showed a greater decrease in amount of homework than males, although the effect size was small. There was no decrease in the amount of homework done by students from high SES backgrounds, but the effect size for low and middle SES students indicates a small but significant decrease. White, Other Hispanic, and Black students were the racial/ ethnic groups with significant decreases in the amount of homework done; there was a very slight but not significant increase in the amount of homework done by Asian-American students. Students in public schools and those in the general and vocational curricula also had homework decreases of moderate effect size. Students from the South and those from rural communities showed moderate decreases in the amount of homework.

Examination of the interaction between sex and curriculum type showed that females in the general and vocational curricula had small decreases in the amount of homework done. The cross-tabulations by socioeconomic status and race showed moderate decreases for all low SES students and moderate decreases for White and Black middle SES students. There were also moderate decreases in the amount of homework done by low and middle SES students in public schools. The interactions between SES and region showed moderate decreases in homework for low SES students in all regions but the West, a moderate decrease for middle SES students from the South, and small decreases for middle SES students from all other regions. There were small but not significant increases in the amount of homework done by high SES students in the Northeast and the West. There was a molerate decrease in the homework done by low and middle SES students from suburban and rural communities and small decreases for low and middle SES students in urban communities. High SES students from rural communities showed a very slight nonsignificant increase in the amount of homework done. Similar results can be seen in the interaction of SES and curriculum. Low SES students in all curricula and middle SES students in the general and vocational curricula showed a homework decrease with a moderate effect size. There was a small, nonsignificant increase in the amount of homework done by high SES students in the academic curriculum.

In sum, although the amount of homework done by students showed a small decrease between 1972 and 1980, this decrease varied across groups. It was greatest among low SES students, students in the South, and females in the general or vocational curriculum. The effect that decreasing homework has on test scores will be explored in the relational analysis.



#### TIME PER WEEK SPENT ON HOMEWORK (1=0-5 HOURS; 3=MORE THAN 10 HOURS)

	NLS 1972			HSB 1980							
	SAMPLE N	NEIGHTED N	MEAN	\$.D.	SAMPLE	WE IGHTED N	MEAN	 S.D.	POOLED S.D.	1980-1972 DIFFERENCE	EFFECT SIZE
TOTAL	16602	3028771	1.41	0.6	28051	3020945	1.31	0.6	0.59	-0.10 *	-0.17
SEX:									••••		-0.17
MALE	8230	1508450	3 70	A E							
FEMALE	8368	1519556	1.30	0.5	12846	1393588	1.25	0.5	0.54	-0.05 *	-0.09
	0000	2327339	1.31	V.0	14063	1510853	1.37	0.6	0.62	-0.15 *	-0.24
SES:											
LON	4801	738100	1 70	• •							
MIDDLE	7895	1548400	1.30	0.6	8357	806239	1.23	0.5	0.54	-0.15 *	-0.29
HIGH	3944	711724	1.30	0.6	12758	1418705	1.27	0.5	0.56	-0.11 *	-0.20
		/31/50	1.40	0.6	6150	719551	1.48	0.7	0.68	-0.00	-0.00
RACE:											
MHITE	12804	2517005									
BLACK	2333	231/703	1.40	0.6	19763	2353204	1.31	0.6	0.59	-0.09 *	-0.16
ASTAN-AMERICAN	202	23400/	1.45	0.6	3748	341773	1.31	C.6	0.59	-0.15 *	-0.25
AMEDICAN TNOTAN	172	2/629	1.71	0.8	363	39273	1.74	0.8	0.79	0.03	0.04
MEYTCAN_AMEDTCAN	100	30947	1.34	0.6	214	21978	1.24	0.5	0.55	-0.10	-0.17
DIEDTA DICAN	221	/2506	1.32	0.5	1881	101390	1.24	0.5	0.52	-0.08	-0.16
ATHED MICAN	75	9684	1.48	0.6	305	17986	1.30	0.6	0.59	-0.19	-0.32
OTTER HISPANIC	121	18672	1.45	0.6	968	66590	1.23	0.5	0.54	-0.23 *	-0 43
SCHOOL TYPE :											
			_								
	14884	2687839	1.39	0.6	24508	2718145	1.28	0.6	0.57	-0.11 *	-0.20
	66	16256	1.63	0.7	866	103557	1.66	0.8	0.78	0.03	-0.20
CATHOLIC	1026	235579	1.54	0.6	2677	199243	1.47	0.7	0.66	-0.07	-0.30
GEOGRAPHIC DECIMU											-0.10
MOTURART											
NORTH CENTRAL	3597	799732	1.44	0.6	5653	692380	1.38	0.6	0.63	~0.06 #	-0.10
PORTA CENTRAL	4553	914850	1.40	0.6	8053	8645 <b>86</b>	1.31	0.6	0.58	-0.09 #	-0.10
SUUTH	5485	792387	1.42	0.6	9244	917812	1.26	0.5	0.56	-0 14 #	-0.15
REGI	2967	521803	1.35	0.6	5101	546166	1.30	0.6	0.58	-0.05 *	-0.27
									0.50	-0.03 *	-0.00
GENERAL	5636	961920	1.27	0.5	10233	1106102	1.17	0.4	0 44	-0.10 #	
ACADEMIC	6790	1386730	1.55	0.7	10495	1134561	1.53	0.7	0.40	-0.10 *	-0.22
JUCATIONAL	4175	679819	1.31	0.5	6902	735086	1.18	0.4	0.67	-0.02	-0.03
								•••	0.47	-0.13 *	-0.27
LUTWINIT TTPE:											
UKBAN	4547	784745	1.42	0.6	6461	604459	1.33	0.6	0 40	-0.00 *	
JUBURBAN	7924	1533221	1.41	0.6	13494	1492707	1.32	0.6	0.00	-0.07 *	-0.14
RURAL	3673	637907	1.39	0.6	8096	923780	1.27	0.5	0.00	-0.09 #	-0.15
								4.3	v.35	~0.12 #	~0.23

\*SIGNIFICANT AT .05 OR LESS



#### J. EXTRACURRICULAR ACTIVITIES

One factor that might affect the amount of time high school students spend on homework is the amount of time they spend on other activities, both in-school extracurricular activities and other out-of-school activities. The student questionnaire provides a clue to how students use their time by asking whether or not the student participated in each of nine extracurricular activities. Table 6-38 shows the results in summary form. There were minor changes between 1972 and 1980 in the questions about extracurricular activities. For example, all athletic teams were combined in 1972 but, in 1980, varsity teams were separated from other athletic teams. These changes may have created minor differences in the responses.

#### Table 6-38

#### Percentage of Students Participating in Extracurricular Activities

	<u>1972</u>	1980	Difference
Athletics	44 <b>.9%</b>	51 <b>.8%</b>	6.9*
Debating and/or Music	33.1	36.4	3.3*
Subject Matter Clubs	25.6	23.9	-1.8*
Vocational Education Clubs	22.3	23.2	0.9
Newspaper and/or Yearbook	20.2	19.7	-0.5
Student Government	19.4	18.3	-1.1*
Hobby Clubs	18.7	22.9	4.1*
Cheerleading	17.3	15.0	-2.3*
Honorary Clubs	14.4	16.8	2.4*

\*Significant at .05 or less

As can be seen, athletics was the most popular extracurricular activity involving 44.9 percent of the students in 1972 and 51.8 percent in 1980. Thus, the percentage of students participating in athletics increased 6.9 percent in this period. The change in athletics participation rates was higher for females than for males, but the proportion of males taking part in athletics continues to exceed that of females. (See Table 6-39.) Participation in athletics increased more for high than for low SES students, thus adding to the already existing differential participation rates in these groups. Asian-American, American Indian, and Mexican-American students showed larger increases than did Whites, Blacks, or Puerto Ricans. Non-Catholic private school students showed a greater increase than public school students while there was a small decrease in athletic participation rates for Catholic school students. The increase was lower for academic curriculum students than for those in other curricula.



## Table 6-39 PERCENTAGE PARTICIPATING IN ATHLETICS

## 

	NLS 1972						
	SAMPLE N	WEIGHTED N	PERCENT	SAMPLE	WEIGHTED N	PERCENT	1980-1972 DIFFERENCE
TOTAL	16453	3006842	44.9	27779	2993578	51.8	6.9 <b>*</b>
SEX:							
MALE	8186	1501415	58.2	12774	1 107100	44 .	<b>–</b> • •
FEMALE	8262	1504410	31.7	13949	1498306	40.6	5.8 * 9.0 *
SES:							
LON	4733	728902	74 7				
MIDDLE	791	1539394	30.7	8281	798574	43.2	4.5 ¥
HIGH	3830	1330304	44.5	12654	1407505	52.1	7.6 *
	5050	/ 27121	51./	6124	716791	61.7	9.9 ¥
RACE:							
WHITE	12718	2502673	45.1	19646	2740470	F] (	<b>1 - - -</b>
BLACK	2058	249083	49.6	17040	2340437	51.0	0.5 *
ASIAN-AMERICAN	190	27464	36.0	300 7	334311	54.5	4.8 *
AMERICAN INDIAN	186	31153	43.6	217	30204 22254	40.0	12.8 *
MEXICAN-AMERICAN	545	71623	38.6	1944	00040	02.2	18.6 *
PUERTO RICAN	95	9659	41.1	1004	77040	47.0	11.0 *
OTHER HISPANIC	120	18471	36.7	955	65391	47.9	0.8 12.8 ¥
SCHOOL TYPE:							
PIBLIC	34750	0//0000					
PRIVATE	14/30	2669209	43.6	24241	2690539	50.6	7.1 ×
	1010	15919	57.2	864	103673	73.1	16.0 ¥
GAINOLIG	1019	233649	58.1	2674	199366	57.1	-1.1
GEOGRAFHIC REGION:							
NORTHEAST	3582	797729	47.2	5408	697463	E/ E	
NORTH CENTRAL	4518	907712	46.7	7976	959306	59.0	/.3 *
SOUTH	5402	781336	42.2	0770	006210	26.0	0.1 *
WEST	2951	520064	42.3	5065	541511	52.9	0.0 <del>*</del> 10.6 *
CURRICULUM:							
GENERAL	55 01	055420	47.8				
ACADEMIC	2371	777020	41.0	10142	1096489	49.9	8.9 ×
VOCATIONAL	6744	13/9024	53.4	10449	1130050	60.1	6.7 ×
TOTALIONAL	4117	6/198/	32.9	6782	723242	42.4	9.5 ×
COMMUNITY TYPE:							
URBAN	4496	776847	43.0	6367	596028	49 2	F 0 -
SUBURBAN	7886	1527429	44.6	13381	1491555	TU.L E9 4	5.6 *
RURAL	3627	631401	48.6	8029	916995	52.0	0.U *
				OVE 7	743773	36 . 7	4.3 #

\*SIGNIFICANT AT .05 OR LESS



•

Debating and musical activities, such as band, chorus or orchestra, were combined in the second most frequent form of extracurricular participation. There was a slight increase in participation between 1972 and 1980. This increase was greater for females than males, greater for high than for low SES students, and greater for Asian-American, American Indian, and Puerto Rican students than for other students. (See Table 6-40.) The change in participation rate varied little across curricula.

Subject matter clubs, the third most frequent type of extracurricular activities, showed a small but significant decrease in participation rates between 1972 and 1980. This suggests that the 1930 students had less opportunity to acquire additional knowledge through non-formal learning in a subject matter club than did the 1972 students. As shown in Table 6-41, the decrease was greater for females than for males. Although students from the White, Black, and Other Hispanic racial/ethnic groups showed decreased participation in subject matter clubs, participation increased for other minority groups. Participation in subject matter clubs declined more in Catholic schools than in public schools while non-Catholic private schools showed an increase.

Participation rates in vocational education clubs and in school newspapers and/or yearbooks were relatively stable between 1972 and 1980. There was a rise in participation in vocational clubs for males and a decline for females. (See Table 6-42.) There was a decrease in participation in vocational clubs for students in the academic and general curricula and an increase for students in the vocational curriculum. This suggests that vocational curriculum students are using these clubs as a way to increase their knowledge of a vocational field, while students in other curricula see the content of these vocational clubs as less relevant. In newspaper and yearbook activities, males showed a slight increase while female participation declined. Participation in this activity also decreased for low and middle SES students and for students in the nonacademic curricula; participation in these writing-related extracurricular activities increased among high SES students and among students in the academic curriculum. These findings, presented in Table 6-43, suggest that college-bound students may be using writing-related extracurricular activities to enhance their writing skills.

Participation in student government decreased slightly but significantly. This was due primarily to a decrease in the participation of male students, a decrease in the participation of students not enrolled in the academic curriculum, and a decrease in the participation of students from suburban schools. (See Table 6-44.)

Participation in hobby clubs, however, showed an increase. The increase was greater for females than for males. There is very little variation in the participation rates in these clubs across SES groups or across school curricula. (See Table 6-45.)



## PERCENTAGE PARTICIPATING IN DEBATING AND/OR MUSIC

	NL\$ 1972						
	SAMPLE N	WEIGHTED N	PERCENT	SAMPLE	MEIGHTED N	PERCENT	1980-1972 DIFFERENCE
TOTAL	16202	2962931	33.1	27596	2972184	36.4	3.3 #
SEX:		,					••••
MALE	7987	1466013	24 B	10/45			
FEMALE	8210	1495901	39.2	12045	1373401 1495066	28.4	
SES:						4317	7.0 *
LON	44.30	715004					
MIDDLE	4037	715224	30.1	8221	792425	31.1	1.0
HIGH	7703	1213012	32.1	12598	1400831	35.8	3.8 *
	3002	123902	38.0	6081	711065	43.9	5.8 ×
RACE:							
MHITE	12566	2472337	32 A	10554	9393394		
BLACK	1999	241621	40.8	7550	2367764	35.0	3.0 ¥
ASIAN-AMERICAN	189	27303	21 9	3021	330035	43.3	2.4
AMERICAN INDIAN	183	30862	23.0	337	21402	30.6	14.7 *
MEXICAN-AMERICAN	528	69002	27 1	1963	51045	37.0	13.9 <b>*</b>
PUERTO RICAN	92	9232	30 2	2020	70043	29.9	2.9
OTHER HISPANIC	118	18011	26.9	939	17269 64410	50.6 30.8	8.4 7 9
SCHOOL TYPE:							3.7
PUBLIC	34570	0/ 30343					
PRIVATE	14330	20 30/4/	33.2	24080	2671586	35.9	2.8 *
CATHOLIC	100	06201	59.7	858	102641	53.0	-6.7
041110220	1004	231100	32.2	2658	197957	34.0	1.8
GEOGRAPHIC REGION:							
NORTHEAST	3525	785111	31.7	5558	480717	14 E	
NORTH CENTRAL	4449	894966	35.5	7939	251950	37.3	2.0
SOUTH	5319	769762	32.9	9062	899768	37.1	1.0
WEST	2909	513091	30.9	5037	537842	35.2	5.0 <del>*</del> 4.3 *
CURRICULUM:							
GENERAL	5504	642071	70 .			_	
ACADEMIC	5500	742071	30.8	10061	1087093	34.5	3.7 <b>*</b>
VOCATIONAL	4040	1324040	39.4	10398	1123523	44.1	4.7 ×
	7047	00000/	23.1	6738	718585	27.7	4.6 *
CONTRACTLY TYPE:							
URBAN	4429	767396	30.8	6328	591725	35 6	A B 🛏
SUBURBAN	7758	1502946	32.4	13290	147021	34 1	4.0 *
RURAL	3581	623516	38.0	7978	910247	40 4	1./
				1710	72464/	-U.O	2.0

\*SIGNIFICANT AT .05 OR LESS



## PERCENTAGE PARTICIPATING IN SUBJECT MATTER CLUBS

		NLS 1972			H <b>SB</b> 1980		
	SAMPLE N	WEIGHTED N	PERCENT	SAMPLE N	WEIGHTED N	PERCENT	1980-1972 DIFFERENCE
TOTAL	16246	2970934	25.6	27421	2953602	23.9	-1.8 #
REV:							
MALE	8037	1475854	20.3	12570	1365377	19.0	-1.3
FEMALE	8205	1494314	30.9	13844	1486403	28.3	-2.6 *
4F5:							
104	4641	71 5930	24.4	8165	786461	22.6	-1.8
MTIMI E	7762	1520270	25 2	12517	1392631	23.8	-1.4
HIGH	3405	724844	27.7	6054	707907	25.7	-2.0
PACE :							
WHITE	12605	2479184	25.0	19466	2317478	22.9	-2.1 *
BLACK	1997	242039	33.1	356 3	323655	28.8	-4.4 #
ASIAN-AMERICAN	191	27418	26.6	355	38254	29.6	3.0
AMERICAN INDIAN	181	30424	27.9	213	21967	28.2	0.3
MEXICAN-AMERICAN	533	69748	23.8	1840	98521	25.2	1.4
PUERTO PICAN	93	9401	16.2	296	16880	18.1	1.9
OTHER HISPANIC	115	17689	30.1	931	63545	25.7	-4.3
SCHOOL TYPE:							
PUBLIC	14568	2637940	24.7	23928	2655250	23.8	-1.0
PRIVATE	66	16256	20.5	845	101056	24.6	4.1
CATHOLIC	1010	231203	35.6	2648	197296	25.2	-10.5 *
GEOGRAPHIC REGION:							
NORTHEAST	3533	787928	21.8	5525	676181	19.6	-2.3
NORTH CENTRAL	4466	897079	26.8	7901	850823	21.3	-5.5 *
SOUTH	5330	772652	31.0	8993	892017	30.8	-0.3
WEST	2917	51 <b>3</b> 27 <b>5</b>	21.3	5002	534581	21.9	0.6
CURRICULUM:							
GENERAL	5521	944723	21. <b>2</b>	10008	1081657	19.6	-1.6
ACADEMIC	6670	1364719	29.6	10330	1115939	28.7	-0.9
VOCATIONAL	4054	661189	23.8	66 <b>86</b>	713020	23.1	-0.7
COMMINITY TYPE:							
URBAN	4436	767 <b>96</b> 9	25.9	6285	<b>58696</b> 2	24.4	-1.6
SUBURBAN	7783	1508449	24.8	13201	1460755	<b>2</b> 2.1	-2.7 *
RURAL	3598	625789	27.5	7935	905885	26.5	-1.0

\*SIGNIFICANT AT .05 OR LESS

160



\_

## PERCENTAGE PARTICIPATING IN VOCATIONAL EDUCATION CLUBS

		NL\$ 1972					
	SAMPLE N	WEIGHTED N	PERCENT	SAMPLE	WE IGHTED	PERCENT	1980-1972 Difference
TOTAL	16314	2980737	22.3	27481	2960166	23.2	0.9
SEX:							
MALE	8064	1478928	1E E	19560			_
FEMALE	8246	1501044	29.0	12309	1367074	19.1	3.7 #
050				23003	1470000	20.7	-2.3 *
SES:							
LOW	4684	721685	30.9	8181	788057	30.6	-04
MIDDLE	7759	1522925	22.6	12559	1396712	24.1	-0.4
HIGH	3813	726 381	13.1	6054	708619	13.4	0.3
RACE:							
MHITE	12450	2494937	21 7				
BLACK	2014	243040	21.3	19502	2322012	22.3	1.0
ASIAN-AMERICAN	102	27607	32.7	3581	325843	30.1	-2.6
AMERICAN INDIAN	192	2/00/	12.1	353	38220	9.6	-2.5
MEXICAN-AMERICAN	E 74	30/1/	27.7	213	21892	22.2	-5.5
PUERTO RICAN	230	/0200	51.1	1846	98117	30.2	-1.0
OTHER HISPANIC	76	9325	7.6	293	16820	9.0	1.4
	113	1/040	17.1	940	64419	27.6	10.5
SCHOOL TYPE:							
PUBLIC	146 35	2647460	28.3	22082	2440047		
PRIVATE	66	16256	9.7	23902	2000003	24.9	1.8 *
CATHOLIC	1009	231139	11 7	040	101480	9.0	-0.7
				2651	197822	0.0	-5.1 *
GEOGRAPHIC REGION:							
NORTHEAST	3546	791116	13.1	5525	474883	10 4	
NORTH CENTRAL	4469	896302	22.0	7915	070331	20.0	-2.5 *
South	5372	777535	36.3	9036	807808	20.0	-2.0
NEST	2927	515785	15.7	5005	534578	15.8	3./*
					304210	23.0	4.1
CENEDAI							
ACADENTO	5590	9486 34	23.3	10026	1084132	22.6	-0.6
VOCATIONAL	6679	1365810	14.7	10344	1116869	13.1	-1.6 *
TOCATIONAL	4088	665992	36.3	6718	716626	39.3	3.0 *
COMMUNITY TYPE:							
URBAN	4447	769146	16.6	4084	<b>FA</b> (000		
SUBURBAN	7810	1513109	10.0	0200	500092	17.9	1.3
RURAL	2619	628569	10.7	13237	1464753	19.5	0.9
	2027	VLUJV 7	30.3	/958	908521	32.5	-5.8 ×

\*SIGNIFICANT AT .05 OR LESS



.

--

## PERCENTAGE PARTICIPATING IN NEWSPAPER AND/OR YEARBOOK

		NLS 1972					
	SAMPLE N	WEIGHTED	PERCENT	SAMPLE	WE IGHTED N	PERCENT	1980-1972 DIFFERENCE
TOTAL	16233	2967880	20.2	27327	2 <b>9</b> 43729	19.7	-0.5
SEX:						_	
MALE	8023	1472907	14.7	12528	1361630	15.4	0.7
FEMALE	8206	1494208	25.5	13792	1480026	24.0	-1.5
SES:							
LON	4638	716223	17.6	8127	782543	15.7	-1.9 *
MIDDLE	7742	1519896	20.2	12498	1390922	19.3	-1.0
HIGH	3796	722152	22.8	6028	704672	25.3	2.5 *
RACE:							
WHITE	12600	2477707	20.5	19414	2311043	20.1	-0.3
BLACK	1996	241317	20.7	3546	322032	17.7	-3.0 *
ASIAN-AMERICAN	191	27517	16.2	353	37982	21.4	5.2
AMERICAN INDIAN	182	30659	19.1	211	21613	25.6	6.5
MEXICAN-AMERICAN	527	69368	14.9	1831	97581	14.7	-0.2
PUERTO RICAN	91	9200	16.0	293	17105	20.4	4.4
OTHER HISPANIC	117	17871	21.1	929	63671	16.2	-4.9
SCHOOL TYPE:							
PUBLIC	14554	2634258	19.4	23825	2644387	18.1	-1.3 *
PRIVATE	66	16256	41.7	854	102276	45.0	3.3
CATHOLIC	1007	231166	27.8	2648	197066	28.0	0.2
GEOGRAPHIC REGION:							
NORTHEAST	3527	786314	22.0	5503	673674	24.6	2.6 #
NORTH CENTRAL	4450	894270	21.0	7879	848049	18.4	-2.6 *
SOUTH	5332	772241	20.1	8962	8890 <b>01</b>	18.8	-1.3
WEST	2924	515056	16.0	4983	533005	16.9	0.9
CURRICULUM:							
GENERAL	5512	942680	17.1	9972	1078551	16.7	-0.4
ACADEMIC	6665	1363662	24.3	10299	1112468	26.9	2.6 *
VOCATIONAL	4055	661236	16.0	6662	710661	13.2	-2.8 *
CONTUNITY TYPE:							<b>.</b> -
URBAN	4427	766 <b>558</b>	17.8	6264	585839	18.0	0.2
SUBURBAN	7787	1508164	19.4	13163	1455563	17.5	-2.0 *
RURAL	3585	623822	25.2	79 <b>00</b>	902327	24.3	-0.9

\*SIGNIFICANT AT .05 OR LESS

162



#### PERCENTAGE PARTICIPATING IN STUDENT GOVERNMENT

		NLS 1972					
	SAMPLE	WEIGHTED		SAMPLE WEIGHTED			1980-1972
	N	N	PERCENT	N	N	PERCENT	DIFFERENCE
TOTAL	16163	2958781	19.4	27289	2941468	18.3	-1.1 *
SEX:							
MALE	8012	1473300	18.1	12497	1357916	15.8	-2.3 #
FEMALE	8147	1484715	20.8	13791	1482345	21.0	0.2
SES:							
LON	4610	711990	14.0	8112	781007	17 7	_0 7
MIDDLE	7695	1512739	18 5	12448	1707075	13.3	-0.7
HIGH	3801	724236	26.6	6032	705906	25.2	-1.4
PACF:							
WHITE	12573	2474043	19 2	10101	2300480	177	-1 6 4
BLACK	1956	234548	25 3	17373	2307400	1/./	-1.5 *
ASIAN-AMERICAN	191	27522	24 9	354	38072	23.1 97 4	-1.2
AMERICAN INDIAN	182	30701	16.3	208	21387	20 7	-1.2
MEXICAN-AMERICAN	527	68800	15.2	1820	97177	14 4	1 5
PUERTO RICAN	91	9256	18.3	2020	16678	10.0	1.5
OTHER HISPANIC	115	17647	17.7	930	63875	16.2	-1.6
SCHOOL TYPE:							
PUBLIC	14482	2625107	19.1	23802	2643036	17 8	-1 7 #
PRIVATE	65	16049	25.9	25002	101052	29 7	-1.3 *
CATHOLIC	1013	232220	20.8	2643	197400	19.9	-1.0
GEOGRAPHIC REGION:							
NORTHEAST	3530	786974	18.6	5505	674525	18 4	-0.2
NORTH CENTRAL	4451	894588	18.9	7879	847733	16.2	-0.L -2.7 ×
SOUTH	5275	765259	20.6	8926	886799	19.2	-14
WEST	2907	511959	19.9	4979	532411	20.3	0.4
CURRICULUM:							
GENERAL	5493	941318	15.1	9950	1075878	14 4	-0.7
ACADEMIC	6658	1362957	26.0	10313	1114637	26 7	-0.7
VOCATIONAL	4011	654203	11.9	66 37	708807	11.5	-0.4
COMMUNITY TYPE:							
URBAN	4408	763416	19.1	6260	581877	19.4	0 4
SUBURBAN	7753	1504429	20.0	12144	303077	17.0	.9 0 ×
RURAL	3572	622085	19.0	7881	900279	19.6	-6.7 *
				,	799617	- / • V	V.V

\*SIGNIFICANT AT .05 OR LESS



.

#### PERCENTAGE PARTICIPATING IN HOBBY CLUBS

		NLS 1972					
	CAMPI C	LETCUTED					
	N	N	PERCENT	SAMPLE N	NEIGHTED	PERCENT	1980-1972 DIFFERENCE
TOTAL	16289	2977075	18.7	27447	2956163	22.9	4.1 *
SEX:							
MALE	8064	1479779	24.3	12592	1367247	26.6	2.3 #
FEMALE	8221	14%531	13.2	13842	1486419	19.3	6.0 *
SES:							
LON	4664	719199	16 5	<b>8170</b>	787482	20 4	A 1 A
MIDDLE	7760	1522608	19.9	32544	170/402	20.0	4.1 *
HIGH	3808	725659	20.6	10944	202020	23.4	4.5 *
	5000	/2303/	20.0	0431	/0/920	24.2	3.0 *
RACE:							
WHITE	12622	2482342	18.4	19472	2317722	22.5	4.1 *
BLACK	2013	243535	20.0	3580	325915	23.0	3.0
ASIAN-AMERICAN	189	27302	16.8	354	38247	26.6	9.8
AMERICAN INDIAN	184	30886	29.5	213	21887	31.5	2.1
MEXICAN-AMERICAN	538	70541	18.6	1843	98402	24.6	6.0 #
PUERTO RICAN	93	9401	21.5	295	16864	21.5	0.0 *
OTHER HISPANIC	117	17905	14.2	936	64315	23.0	8.8
SCHOOL TYPE:							
PIBLIC	14609	2444840	10.8	27054	9483043		<i>4</i> • •
POTVATE	14007	2044007	10.0	23754	205/941	22.8	4.0 *
	1000	220880	19.0	040	100943	20.9	8.0
CAMOLIC	1007	230009	10.2	2647	197279	22.1	3.9
GEOGRAPHIC REGION:							
NORTHEAST	3544	789571	18.5	5529	676606	24.0	5.5 ¥
NORTH CENTRAL	4471	898904	18.0	7906	851160	21.0	3.0 *
SOUTH	5347	773280	17.9	9006	893628	22.0	4.1 *
WEST	2927	515320	21.6	5006	534769	25.8	4.2 #
CURRICULUM:							
GENERAL	5533	946972	18 2	10014	1002745		4 <b>T</b> 4
ACADEMIC	6677	1744478	18.0	10014	1002/45	22.5	4.3 *
VOCATIONAL	4080	665162	19.2	6703	714422	22.9	9.U # 3.7 #
CONTUNITY TYPE:							
URBAN	4452	770780	18.0	6287	587276	23.3	5.3 ×
SUBURBAN	7803	1511316	18.4	13223	1462677	22.7	4.3 ×
RUKAL	3600	625654	20.5	7937	906210	22.8	2.3 ¥
				1	64		
				1	<b>1'U</b>		

\_\_\_\_



\_\_\_\_\_

Participation in cheerleading declined between 1972 and 1980. As shown in Table 6-46, decline was greater for females than for males, for students in Catholic schools than for those in public or other non-Catholic private schools, and for students in the vocational curriculum.

Participation in honorary societies increased between 1972 and 1980. It is likely that this increase is linked to the increase in grades, since most honorary societies specify a grade point average for entrance. The increase was greater for males than females, and was larger for middle and high SES students, for students in nonpublic schools, and for academic curriculum students. (See Table 6-47.)

These data, taken together, suggest that student participation in extracurricular activities tended to increase between 1972 and 1980 in the recreational types of activities, such as athletics, hobby clubs and music groups, and showed a decline in extracurricular activities that provide an opportunity for non-formal learning, such as subject matter clubs, the newspaper, or yearbook.

#### K. COURSES TAKEN

The amount of instruction which students receive in a subject is usually closely related to their achievement in that subject. The number of courses taken in relevant subjects provide one indicator of the opportunity which a student has to learn the information and skills covered in the NLS and HS&B tests.

Information on the number of semesters of instruction in mathematics, English, science, social studies, and foreign languages were obtained from the 1972 and the 1980 seniors. The results, by classification groups, are shown in Tables 6-48 through 6-52.

There was a small but significant increase, between 1972 and 1980, in the number of semesters of mathematics which students reported taking. The mean rose from 3.93 semesters in 1972 to 4.06 semesters in 1980. The increase in the amount of mathematics taken was due primarily to females and to minorities. However, females continued to average fewer semesters of mathematics than males. Blacks, however, increased the amount of mathematics taken to an extent that, by 1980, they surpassed Whites. Asian-Americans and Puerto Ricans took more mathematics than Whites both in 1972 and in 1980. Mathematics course taking increased significantly in all curricula. Academic students continued to report taking more mathematics than general or vocational curriculum students, but vocational students showed the greatest increase.

Cross-tabulations show a significant increase in the number of mathematics courses taken by females in all curricula and by males in the academic and vocational curricula.



#### Table 5-46

٠

#### PERCENTAGE PARTICIPATING IN CHEERLEADING

		NLS 1972					
	SAMPLE N	MEIGHTED N	PERCENT	SAMPLE N	WEIGHTED N	PERCENT	1980-1972 Difference
TOTAL	16247	2971262	17.3	27397	2951820	15.0	-2.3 *
SEX:							
MALE	7985	1468523	5.3	12522	1360890	4.5	-0.8
FEMALE	8258	1501974	29.1	13869	1489181	24.8	-4.3 *
SES :							
LOH	4651	717578	14.8	8168	786419	13.0	-1.7 *
MIDDLE	7738	1519618	18.4	12519	1392897	15.9	-2.5 ×
HIGH	3800	724058	17.7	6034	706699	15.6	-2.1
RACE							
NHITE	12609	2480327	17.3	19447	2315618	14.9	-2.4 *
BLACK	1991	240287	21.2	3576	325305	17.6	-3.6 *
ASIAN-AMERICAN	190	27413	13.1	355	38397	14.6	1.4
AMERICAN INDIAN	182	30689	14.8	212	21868	17.0	2.2
MEXICAN-AMERICAN	537	70520	16.1	1831	97395	13.2	-2.9
PUERTO RICAN	92	9346	7.6	295	16859	12.1	4.5
OTHER HISPANIC	116	17731	13.5	929	63472	11.0	-2.5
SCHOOL TYPE:							
PUBLIC	14565	2637516	17.1	23906	2653800	14.9	-2.2 *
PRIVATE	65	16042	20.3	845	101225	17.3	-3.0
CATHOLIC	1011	231557	19.8	2646	196794	14.7	-5.1 *
GEOGRAPHIC REGION:							
NORTHEAST	J5 7	784345	14.0	5504	673936	11.5	-2.6 *
NORTH CENTRAL	4467	897791	20.0	7902	851132	15.3	-4.7 *
SOUTH	5333	773183	18.7	9002	894008	18.0	-0.7
WEST	2930	515944	15.6	4989	532744	14.0	-1.6
CURRICULUM:							
GENERAL	5515	943933	15.9	9990	1080015	14.0	-1.9 *
ACADEMIC	6670	1364595	19.0	10333	1116458	17.1	-1.8 *
VOCATIONAL	4061	662431	15.9	6682	712897	13.3	-2.6 *
CONTUNITY TYPE:							
URBAN	4437	768540	16.1	6274	586385	14.3	-1.9
SUBURBAN	7788	1508765	16.7	13187	1459166	13.1	-3.6 *
RURAL	3592	625c37	20.9	7936	906268	18.6	-2.3 <b>*</b>
				16	l R		
1				T0			

HATCHITFICANT AT .05 OR LESS

ERĬC Full Text Provided by ERIC

#### PERCENTAGE PARTICIPATING IN HONORARY CLUBS

------

	NLS 1972					
SAMPLE N	WE IGHTED N	PERCENT	SAMPLE N	WEIGHTED N	PERCENT	1980-1972 Difference
16235	2970780	14.4	27339	2946629	16.8	2.4 *
8030	1475398	10.7	12530	1361947	13.9	3.2 *
8201	1494617	18.1	13810	1483601	20.1	1.9 *
46 39	716781	10.3	A124	782597	11.1	0.8
7733	1519824	13.2	12514	1393046	16.3	3.1 #
3806	724572	21.2	6024	705028	25.1	3.9 *
12606	2480023	15.1	19434	2336673	17.7	2.7 *
1981	240071	11.7	3546	322714	13.7	2.0
189	27344	23.6	353	38336	23.4	-0.2
183	30798	6.6	210	21629	13.5	6.9
534	70168	9.5	1828	97028	12.0	2.4
93	9401	11.8	292	16902	14.5	2.8
117	17905	12.0	931	63665	11.0	-1.0
14559	2637995	14.5	23861	2649219	16.5	2.1 ×
66	16256	13.3	843	100705	17.2	3.9
1009	231005	14.6	2635	196705	20.4	5.9 ¥
3535	788542	12.7	5507	674570	16.0	3.3 *
4449	894709	12.1	7889	850062	15.2	3.1 *
5327	771839	18.0	8961	889602	19.6	1.6
2924	515690	15.7	4982	532395	15.7	-0.0
5518	945351	8.1	9979	1079274	9.6	1.5 ×
6666	1363662	22.3	10315	1114777	29.1	6.8 *
4050	661464	7.3	6658	710560	8.9	1.6 *
4434	767569	12.6	6261	585327	15.6	3.0 *
7791	1510073	14.6	13167	1457804	16.5	1.9 ×
3584	624597	16.9	7911	903498	18.1	1.3
	SAMPLE N 16235 8030 8201 4639 7733 3806 12606 1981 189 183 534 93 117 14559 66 1009 3535 4449 5327 2924 5518 6666 4050 4434 7791 3584	NLS 1972       SAMPLE     WEIGHTED       N     N       16235     29707A0       8030     1475398       8201     1494617       4639     716781       7733     1519824       3806     724572       12606     2480023       1981     240071       189     27344       183     30798       534     70168       93     9401       117     17905       14559     2637995       66     16256       1009     231005       3535     788542       4449     894709       5327     771839       2924     515690       5518     945351       6666     1363662       4050     661464       4434     767569       7791     1510073       3584     624597	NLS 1972       SAMPLE     WEIGHTED N     PERCENT       16235     29707A0     14.4       8030     1475398     10.7       8201     1494617     18.1       4639     716781     10.3       7733     1519824     13.2       3806     724572     21.2       12606     2480023     15.1       1981     240071     11.7       189     27344     23.6       183     30796     6.6       534     70168     9.5       93     9401     11.8       117     17905     12.0       14559     2637995     14.5       66     16256     13.3       1009     231005     14.6       3535     788542     12.7       4449     894709     12.1       5327     771839     18.0       2924     515690     15.7       5518     945351     8.1       6666     1363662     22.3 <td>NLS 1972     SAMPLE     WEIGHTED     SAMPLE       N     N     PERCENT     N       16235     29707A0     14.4     27339       8030     1475398     10.7     12530       8030     1475398     10.7     12530       8201     1494617     18.1     13810       4639     716781     10.3     8124       7733     1519824     13.2     12514       3806     724572     21.2     6024       12606     2480023     15.1     19434       1981     240071     11.7     3546       189     27344     23.6     353       183     30796     6.6     210       534     70166     9.5     1828       93     9401     11.8     292       117     17905     12.0     931       14559     2637995     14.5     23861       66     16256     13.3     843       1009     231005     14.6</td> <td>NLS 1972     HSB 1980       SAMPLE     KEIGHTED     SAMPLE     MEIGHTED       N     N     PERCENT     N     N       16235     2970740     14.4     27339     2946629       8030     1475398     10.7     12530     1361947       8201     1494617     18.1     13810     1483601       4639     716781     10.3     8124     782597       7733     1519824     13.2     12514     1393046       3806     724572     21.2     6024     705028       12606     2480023     15.1     19434     2314473       1981     240071     11.7     3546     322714       189     27344     23.6     210     21629       534     70168     9.5     1828     97028       193     30798     6.6     210     21629       534     70168     9.5     1828     97028       117     17905     12.0     931     63665  1</td> <td>NLS 1972     HSB 1980       SAMPLE     WEIGHTED     SAMPLE     WEIGHTED       N     N     PERCENT     N     N     PERCENT       16235     29707A0     14.4     27339     2946629     16.8       8030     1475398     10.7     12530     1361947     13.9       8201     1494617     18.1     13610     1483601     20.1       4639     716781     10.3     8124     782597     11.1       7733     1519824     13.2     12514     1393046     16.3       3806     724572     21.2     6024     705028     25.1       12606     2480023     15.1     19434     2314473     17.7       1981     240071     11.7     3546     322714     13.7       189     27344     23.6     33     3836     23.4       103     30798     6.6     210     21629     13.5       117     17905     12.0     931     63655     11.0</td>	NLS 1972     SAMPLE     WEIGHTED     SAMPLE       N     N     PERCENT     N       16235     29707A0     14.4     27339       8030     1475398     10.7     12530       8030     1475398     10.7     12530       8201     1494617     18.1     13810       4639     716781     10.3     8124       7733     1519824     13.2     12514       3806     724572     21.2     6024       12606     2480023     15.1     19434       1981     240071     11.7     3546       189     27344     23.6     353       183     30796     6.6     210       534     70166     9.5     1828       93     9401     11.8     292       117     17905     12.0     931       14559     2637995     14.5     23861       66     16256     13.3     843       1009     231005     14.6	NLS 1972     HSB 1980       SAMPLE     KEIGHTED     SAMPLE     MEIGHTED       N     N     PERCENT     N     N       16235     2970740     14.4     27339     2946629       8030     1475398     10.7     12530     1361947       8201     1494617     18.1     13810     1483601       4639     716781     10.3     8124     782597       7733     1519824     13.2     12514     1393046       3806     724572     21.2     6024     705028       12606     2480023     15.1     19434     2314473       1981     240071     11.7     3546     322714       189     27344     23.6     210     21629       534     70168     9.5     1828     97028       193     30798     6.6     210     21629       534     70168     9.5     1828     97028       117     17905     12.0     931     63665  1	NLS 1972     HSB 1980       SAMPLE     WEIGHTED     SAMPLE     WEIGHTED       N     N     PERCENT     N     N     PERCENT       16235     29707A0     14.4     27339     2946629     16.8       8030     1475398     10.7     12530     1361947     13.9       8201     1494617     18.1     13610     1483601     20.1       4639     716781     10.3     8124     782597     11.1       7733     1519824     13.2     12514     1393046     16.3       3806     724572     21.2     6024     705028     25.1       12606     2480023     15.1     19434     2314473     17.7       1981     240071     11.7     3546     322714     13.7       189     27344     23.6     33     3836     23.4       103     30798     6.6     210     21629     13.5       117     17905     12.0     931     63655     11.0

\*SIGNIFICANT AT .05 OR LESS



## SEMESTERS (HALF-YEARS) OF MATHEMATICS

\_\_\_\_\_\_

		NLS 1	972		HSB 1980						
	SAMPLE	WEIGHTED N	MEAN	\$.D.	SAMPLE N	WEIGHTED	MEAN	<b>\$</b> .D.	POOLED S.D.	1980-1972 Difference	EFFEC SIZE
TOTAL	11771	2173309	3.93	1.8	27928	301 7150	4.06	1.9	1.89	0.14 *	0.07
SEX:											
MALE	5897	1087004	6 22	1.0	39756	1745476	4		• • •		
FEMALE	5871	1085602	3.63	1.0	13980	13050/5	9.31 7.87	1.9	1.89	0.09	0.05
					23700		3.07	1.7	1.00	U.24 *	9.13
363:											
	3276	509101	3.45	1.9	8289	799547	3.60	1.9	1.91	0.16 ¥	8.08
AIDOLE	5628	1114947	3.88	1.8	12693	1412714	3.99	1.9	1.89	0.11 ¥	0.06
HICH	2830	543308	4.47	1.7	6155	720884	4.73	1.7	1.70	0.27 *	0.16
RACE:											
MHITE	9228	1826746	3.97	1.8	19495	2747008	A 84				
BLACK	1402	170819	3.86	1.0	17075	2347000	4.04	1.9	1.90	0.07	0.04
ASIAN-AMERICAN	144	21438	4 28	1.0	3707	337703	9.20	1.0	1.77	0.42 #	0.24
AMERICAN INDIAN	117	19854	2 47	1.0	210	39200	4.71	1.8	1.78	0.63 #	0.35
MEXICAN-AMERICAN	341	45473	1 10	1.7	213	22132	3.52	1.9	1.86	0.85 *	0.46
PUERTO RICAN	62	43073	6 69	2.0	10/3	101049	3.75	1.8	1.79	0.43 *	0.24
OTHER HISPANIC	77	11792	4.12	1.7	303	1/903	9.26	Z.1	2.11	0.18	0.08
		220 /2			737	09032	4.07	1.4	1.91	-0.05	-0.03
SCHOOL TYPE:											
PUBLIC	10595	1936725	3.86	1.9	24393	2707514	3.98	1.9	1 90	A 12 #	
PRIVATE	37	10207	5.26	1.8	862	103179	4.64	1.7	1.70	_0 57	-0.77
CATHOLIC	750	171402	4.50	1.5	2673	199457	4.90	1.7	1.70	0.40 ¥	0.23
GEOGRAPHIC REGION:											
NORTHEAST	2765	636287	4 48	17	E4 7 8	(	A / 7				
NORTH CENTRAL	3173	634607	1 42	2.0	9022	003251	4.0/	1.8	1.79	0.18 *	0.10
SOUTH	3815	551044	3.42	2.0	0102	662126	3.75	2.0	1.99	0.33 *	0.16
MEST	2018	351044	7.40	1./	9192	913255	4.13	1.8	1.78	0.05	0.03
	2010	333344	3.00	1.0	5000	545248	3.70	1.9	1.84	0.10	0.05
CURRICULUM:											
GENERAL	3994	679408	3.34	1.8	10176	100078	7 64	1 0	1 65		
ACADEMIC	5023	1042789	4.72	1.6	10491	1135281	5.50	1.7	1.05	U.23 #	0.13
VOCATIONAL	2753	450809	2.97	1.7	6850	730207	3.36	1.8	1.60	0.28 #	0.18
CRADE AITTY TYDE -											<b>v.</b>
IDRAU											
	3015	520626	5.87	1.8	6432	663016	4.21	1.8	1.83	0.34 *	0.19
BUDAL	5/51	1129230	4.11	1.8	13458	1490643	4.17	1.9	1.87	0.05	0.03
RURAL	2686	475034	3.58	1.9	8038	9164')1	3.80	2.0	1.95	0.22 *	0.11

#SIGNIFICANT AT .05 OR LESS



There was no significant change in the amount of instruction in English. The mean number of semesters of English was 5.83 in 1972 and 5.86 in 1980. This was not surprising since most high schools require students to take a course in English each semester. Females, high SES students, Blacks, Catholic school students, students in urban schools, and students in the academic curriculum showed small but significant increases between 1972 and 1980 in the amount of English taken. Crosstabulations show significant increases in the amount of English taken by males and females in the academic curriculum.

The amount of science taken by students decreased slightly but significantly between 1972 and 1980. In 1972 students reported taking an average of 3.71 semesters but, by 1980, this had declined to 3.46 semesters. The decline was greater for males than for females and occurred primarily among middle and low SES students, White students, public school students, and students in the general and vocational curriculum. Cross-tabulations show significant decreases for both males and females in the general curriculum for low and middle SES students in the general curriculum, and for low SES students in the academic curriculum. High SES students in the academic curriculum showed a significant increase in the number of semesters of science taken.

There was a larger increase in number of courses which students reported taking in the social sciences, declining from 5.21 semesters in 1972 to 4.64 semesters in 1980. This decrease was significant for all classification groups. The decreases are also persistent across all cross-classifications.

The largest change in course-taking behavior, however, occurred in foreign languages where there was a decline from 2.64 semesters of instruction in 1972 to 1.65 semesters in 1980. Again, this decrease was consistent across classification groups and across cross-classifications. Despite this decline academic curriculum students continued to take more semesters of foreign language instruction than general or vocational curriculum students.

In summary, students showed a small but significant increase in the number of courses taken in mathematics, no change in the amount of coursework in English, and significant decreases in the amount of coursework in science, social science, and foreign languages. The decrease in foreign languages was the greatest, equivalent to a full semester of instruction.

Taken together, the results in this chapter show several major changes in students' attitudes, values and behaviors between 1972 and 1980. Educational aspirations increased for females and for students in the academic curriculum. More females in 1980 than in 1972 planned to enter a 4-year college after high school. There was an increase in the percentage of students who believed they had the ability to complete college. Students placed higher value on job security and pay in 1980 than in 1972. Concern with social issues decreased markedly while interest in making money rose.



#### SEMESTERS (HALF-YEARS) OF ENGLISH

	NLS 1972				HSB 1980						
	SAMPLE N	WEIGHTED N	MEAN	5.D.	SAMPLE N	MEIGHTED N	MEAN	s.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
TOTAL	12286	2272928	5.83	0.9	27864	30J5326	5.86	1.2	1.09	0.03	0.03
SEX:											
MALE	6091	1124264	5.82	1.6	19797	1 787974	E 47			• • •	
FEMALE	6192	1147961	5.84	0.9	13954	1500031	5.92	1.2	1.13	0.02 0.0A ¥	0.01
SES:										0100 *	•
LOM	3444	670409		• •				_			
MINDIF	5444	237072	₽.80	1.0	8254	797188	5.78	1.3	1.20	-0.02	-0.02
HTCH	5074	110///0	5.81	1.0	12682	1411330	5.85	1.1	1.08	0.05	0.04
N100	2909	559051	5.91	0.9	6141	719608	6.01	0.9	0.92	0.11 *	0.12
RACE:											
MHITE	9641	1010475	E 84		104.84						
BLACK	1445	377666	5.04	0.7	14004	2345967	5.87	1.1	1.04	0.03	0.03
ASTAN-AMERICAN	1445	1//440	<b>9.02</b>	0.7	3660	335855	5.95	1.3	1.18	0.13 *	0.11
ANERTCAN THREAM	741	2000/	5.0/	0.8	364	39346	5.83	1.2	1.08	~0.04	-0.04
MEXTCAN_AMENTCAN	133	22300	5.50	1.0	211	21611	5.79	1.4	1.26	0.22	0.18
	352	46 43	5.04	1.0	1861	100116	5.64	1.5	1.42	-0.01	-0.01
OTHER HIGHNIC	60	6663	6.01	1.1	302	17840	6.06	1.2	1.20	0.04	0.04
OTHER HISPARIC	81	12262	5.74	1.0	955	65423	5.79	1.4	1.35	0.05	0.04
SCHOOL TYPE:											
PUBLIC	11045	2025354	E 23	1 A	54177				• • • •	•	
PRIVATE	35	QAST	4 44	1.V	24333	6/02//9	5.03	1.2	1.12	0.01	0.01
CATHOLIC	778	174968	5 8G	0.5	00 3	103300	6.12	0.8	0.78	-0.36 *	-0.46
		1/0/40	3.07	v.o	2000	TAATOO	6.20	0.8	0.76	0.31 *	0.41
GEOGRAPHIC REGION:											
NORTHEAST	2942	670711	5.97	0.8	5417	480051	4 16			<b>•</b> •• •	
NORTH CENTRAL	3339	667607	5 74	1 1	9017	007791	0.15	0.9	0.84	0.18 #	0.21
SOUTH	3932	569633	5 \$7	0.0	0145	037401	5.04	1.3	1.25	-0.10 *	-0.08
NEST	2073	345177	5.07	1 1	7103 6074	911014	6.U1	1.0	0.96	0.13 *	0.14
	2073	303277	3.00	4.1	5470	544079	5.02	1.5	1.24	-0.04	-0.03
CURRICULUM:											
GENERAL	4178	716287	5 A1	1.0	10150	1000100					
ACADEMIC	5083	1057683	5.01		10137	1077177	5.79	1.2	1.15	-0.02	-0.02
VOCATIONAL	3024	499655	5 45	1 0	104/0	1133744	0.10	0.8	0.84	0.17 *	0.20
	3024	477033	9.09	1.0	0023	/28008	5.03	1.4	1.29	-0.02	-0.02
COMPRINITY TYPE:											
URBAN	3155	544501	5.78	11	6408	403408					
SUBURBAN	5975	1176146	5 45	ñ 0	11495	301000	2.00 F 44	1.2	1.1/	0.10 *	0.08
RURAL	2828	502039	5.84	0.7	13463	1400200	5.00	1.1	1.05	0.03	0.03
	2020	245431	2.04	V. 7	0431	41/154	5.03	1.2	1.11	-0.01	-0.01
						1	70 -				
							• •				

#SIGNIFICANT AT .05 OR LESS



# SEMESTERS (HALF-YEARS) OF SCIENCE

	NLS 1972				HSB 1980						
	SAMPLE N	ME IGHTED N	MEAN	\$.D.	SAMPLE	WE IGHTED N	MEAN	\$.D.	POOLED S.D.	1980-1972 DIFFERENCE	EFFECT SIZE
TOTAL	12002	2212239	3.71	1.8	27482	2963192	3.46	1.9	1.90	-0.25 #	-0.13
SEX:											
MALE	6000	1105236	3.93	1.8	12570	1364606	3.68	2.0	1.93	-0.26 *	-0.13
FEMALE	6000	1106511	3.48	1.7	13753	1479646	3.29	1.9	1.84	-0.19 #	-0.10
SES:											
LON	3372	52 3686	3.30	1.7	8123	785221	2.98	1.8	1.81	-0.33 *	-0.18
MIDDLE	5730	1132824	3.67	1.8	12518	1391644	3.39	1.9	1.89	-0.28 *	-0.15
HIGH	2860	549435	4.16	1.7	6084	712532	4.18	1.9	1.82	0.02	0.01
RACE:											
NHITE	9397	1857346	3.77	1.8	19465	2317289	3.48	2.0	1.91	-0.29 #	-0.15
BLACK	1449	178724	3.52	1.7	3604	328674	3.45	1.9	1.80	-0.07	-0.04
ASIAN-AMERICAN	138	20447	3.82	1.8	357	38220	4.12	1.9	1.90	0.30	0.16
AMERICAN INDIAN	120	19788	2.75	1.5	212	21783	3.02	1.7	1.62	0.27	0.17
MEXICAN-AMERICAN	350	46627	2.96	1.6	1828	98836	3.05	1.6	1.64	0.09	0.06
PUERTO RICAN	66	6750	3.67	1.7	290	17517	3.57	2.0	1.97	-0.10	-0.05
OTHER HISPANIC	81	12236	3.80	1.9	945	64933	3.31	1.8	1.84	-0.48	-0.26
SCHOOL TYPE:											
PUBLIC	10782	1968412	3.67	1.8	23988	2663244	3.39	1.9	1.90	-0.27 <b>#</b>	-0.14
PRIVATE	37	10261	4.71	1.5	858	102478	4.01	1.8	1.84	-0.70	-0.38
CATHOLIC	755	172780	4.12	1.7	26 <b>36</b>	197471	4.05	1.8	1.78	-0.57	-0.04
GEOGRAPHIC REGION:											
NORTHEAST	2796	637836	4.27	1.7	5532	680186	4.02	2.0	1.93	-0.25 ¥	-0.13
NORTH CENTRAL	3268	652119	3.39	1.9	7896	848245	3.23	2.0	1.97	-0.16 <b>*</b>	-0.08
SOUTH	3910	564813	3.59	1.6	9027	898224	3.42	1.8	1.77	-0.26 #	-0.15
HEST	2028	357471	3.30	1.7	5027	536538	3.16	1.8	1.75	-0.14	-0.08
CURRICULUM:											
GENERAL	4065	689721	3.22	1.7	10004	1082888	3.01	1.8	1.73	-0.21 ¥	-0.12
ACADEHIC	5085	1058080	4.44	1.7	10353	1120754	4.45	1.8	1.77	0.01	0.00
VOCATIONAL	2852	464438	2.75	1.6	6716	715584	2.62	1.7	1.67	-0.13 *	-0.08
CONTINITY TYPE:											
URBAN	3098	532901	3.53	1.8	6296	590089	3.50	1.9	1.87	-0.04	-0.02
SUBURBAN	5823	1144546	3.85	1.8	13250	1466021	3.50	1.9	1.90	-0.35 *	-0.19
RURAL	2763	485944	3.60	1.8	7936	907082	3.38	1.9	1.91	-0.22 #	-0.12

\*SIGNIFICANT AT .05 OR LESS

•



## SEMESTERS (HALF-YEARS) OF SOCIAL STUDIES

	NLS 1972				HSB 1980						
	SAMPLE N	MEIGHTED N	HEAN	S.D.	SAMPLE N	NE IGHTED N	MEAN	S.D.	POOLED S.D.	1980-1972 D1FFERENCE	EFFECT SIZE
TOTAL	12256	2268019	5.21	1.3	27724	2987813	4.64	1.6	1.48	-0.58 *	-0.39
SEX:											
MALE	6076	1121728	5.26	1.3	12679	1 276 254	A 47	14	3 40		
FEMALE	6177	1145588	5.17	1.3	13877	1491507	4.62	1.5	1.47	-0.55 *	-0.40
SES:											
LON	3432	538240	F 21	1 1		707746	4 47				
MIDDLE	5883	1146225	5.LL 5.91	1.3	12620	793705	9.9/	1./	1.55	-0.74 #	-0.47
HIGH	2902	557348	5.CJ 5.20	1.3	12022	1404463	9.00	1.5	1.46	-0.57 #	-0.39
		337340	9.20	1.3	0110	/15208	4.79	1.5	1.41	-0.41 *	-0.29
RACE											
MITE	9629	1987892	5 21	3 3	10685		A 11				
BLACK	1439	177009	5 25	1.5	3469	2334037	4.00 4.54	1.5	1.45	-0.55 *	-0.38
ASIAN-AMERICAN	148	20717	5.25	1.6	3036	333225	4.34	1./	1.56	-0.71 *	-0.46
AMERICA'I INDIAN	131	22091	5 04	1 1	212	30907	4.71	1.5	1.30	-0.67 #	-0.49
HEXICAN-AHERICAN	348	46352	5 19	1.3	1840	C100/	4.3/	1./	1.57	-0.67 #	-0.43
PUERTO RICAN	65	4683	5 57	1.6	207	77030	4.31	1./	1.62	-0.88 *	~0.55
OTHER HISPANIC	81	12695	4.92	1.2	956	64905	4.51	1.7	1.65	-0.83 # -0.41	~0. <b>50</b> -0.25
SCHOOL TYPE:											
PUBLIC	11025	2021952	5 22	1 1	24282	24 87487	A 41	14			<b>.</b>
PRIVATE	35	9883	▲ <u>2</u> 2	1 7	841	107525	4.01	1.0	1.40	-0.61 *	-0.41
CATHOLIC	770	175405	5.12	1.2	2659	196880	4.90	1.5	1.55	-U.24 -0.22 #	-0.16 -0.15
GEOGRAPHIC REGION:											
NORTHEAST	2928	667293	5 27	1 3	6678	401057	A 00				
NORTH CENTRAL	3344	669039	5.23	1 3	7957	854840	¥.77 A 47	1.5	1.39	-0.28 #	-0.20
SOUTH	3924	568847	5.05	1 1	0141	900058	4.03	1.0	1.51	-0.60 #	-0.40
MEST	2060	362842	5.35	1.2	5048	540057	4.57	1.5	1.50	~0.64 # -0.78 #	-0.43
CURRICULUM:											
GENERAL	<b>4171</b>	714397	E 20		10117						_
ACADEMIC	5069	3054420	3.67 E 94	1.3	10113	109401/	9.59	1.6	1.51	-0.70 #	-0.46
VOCATIONAL	3015	408800	5.20	1.2	10421	1120445	9.89	1.5	1.39	-0.42 #	-0.30
	3413	470077	3.41	1.3	0/04	123291	4.38	1.6	1.54	-0.63 #	-0.41
CONTRACTLY TYPE:											
URBAN	3131	540625	5.14	1 3	6374	508853	A 40	14	1 40		• • •
SUBURBAN	5958	1172238	5.22	1.2	11169	370031	4.00	1.0	1.47	-0.54 *	-0.36
RURAL	2839	504697	6 2A	1.6	7084	012027 T4/3431	4.00	1.9	1.46	-0.56 #	-0.38
		244077	J. LU	2.3	1770	713431	4.02	1.0	1.52	-0.66 #	-0.44
						a and (***					

\*SIGNIFICANT AT .05 OR LESS



#### SEMESTERS (HALF-YEARS) OF FOREIGN LANGUAGES

	NLS 1972				HSB 1980						
	SAMPLE	ME IGHTED N	HEAN	S.D.	SAMPLE N	WEIGHTED N	MEAN	.υ.	POOLED S.D.	1980-1972 DIFFERENCE	EFFECT SIZE
TOTAL	<b>99</b> 76	1868395	2.64	2.2	27573	2971555	1.65	2.1	2.10	-0.99 *	-0.47
SEX:											
MALE	4858	<b>90</b> 9112	2.38	2.1	12560	1363884	1.48	2.0	2.03	-0.90 ×	-0.44
FEMALE	5114	958485	2.88	٤.2	13874	1490834	1.84	2.1	2.14	-1.04 *	-0.48
SES:											
LON	2557	400436	1.79	2.1	8169	788875	1.13	1.8	1.88	-0.65 <b>*</b>	-0.35
MIDDLE	4770	956400	2.56	2.2	12553	1395654	1.52	2.0	2.02	-1.04 *	-0.52
HIGH	2622	506190	3.47	2.0	6094	713085	2.52	2.3	2.18	-0.94 *	-0.43
RACE:											
MITE	7923	1591648	2.70	2.2	19488	2321438	1.65	2.1	2.09	-1.05 *	-0.50
BLACK	1062	127404	2.07	2.1	3631	331761	1.47	2.0	2.03	-0.61 *	-0.30
ASIAN-AMERICAN	133	19111	3.19	1.9	360	38682	2.34	2.1	2.06	-0.86 *	-0.42
AMERICAN INDIAN	102	16122	1.37	2.0	208	21394	0.86	1.7	1.78	-0.51	-0.29
MEXICAN-AMERICAN	322	42867	2.23	2.1	1854	99596	1 80	2.0	2.04	-0.44 #	-0.22
PUERTO RICAN	56	6273	3.24	2.3	296	17688	2.78	2.5	2.46	-0 46	-0.19
OTHER HISPANIC	68	10464	3.87	1.9	955	65113	1.80	2.3	2.28	-2.07 *	-0.91
SCHOOL TYPE:											
PIBLIC	8930	1650286	2.50	2.2	24064	2672068	1.51	2.0	2 04	-0.99 #	-0.69
PRIVATE	40	11105	4.75	2.4	851	101193	2.94	2.5	2 49	-1 61 #	-0.73
CATHOLIC	717	166596	3.88	1.9	2658	198294	2.82	2.1	2.05	-1.06 *	-0.52
REACHADUTE DECTAN:											
NOTHFAST	2312	540588	3.70	2.0	5540	679630	2. 36	23	2 21	-1 34 #	-8 60
NORTH CENTRAL	2827	570554	2.00	2.0	7910	850526	1.37	2.0	1.98	-0.63 #	-0.32
SOUTH	2973	436253	2.62	2.1	9082	902326	1.31	1.9	1.93	-1.11 #	-0 57
NEST	1864	320999	2.27	2.1	5041	539275	1.74	2.0	2.05	-0.54 *	-0.26
CENEDAL	3241	549218	1.75	2.0	10049	1086832	1.15	1.8	1.82	-0 60 #	-0.33
ACADEMIC	4802	998674	3.53	2.0	10435	1126264	2.68	2.2	2.15	-0.85 #	-0.40
VOCATIONAL	1%3	320503	1.35	1.8	6710	715600	0.81	1.5	1.58	-0.54 *	-0.34
COMMINITY TYPE:											
IDRAN	2642	661884	2.76	2.2	6344	R04833	1.80	2.1	2.15	-0.96 #	-0 45
SIRIPAN	AGAR	GAGARA	2 90	2.1	13974	1469579	1.43	2.1	2.13	-1 06 #	-0.49
RURAL	2122	378412	1.82	2.0	7953	907143	1.24	1.9	1.89	-0.58 #	-0.31

\*SIGNIFICANT AT .05 OR LESS



Students became more self-confident but less sure of their ability to control their own lives. The amount of homework done decreased. There was a decrease in the amount of coursework taken in science, social studies and foreign languages but an increase in the amount of coursework in mathematics.

-63-



#### CHAPTER VII

#### PARTITIONING MEAN TEST SCORE CHANGES

As was indicated in Chapter II, the relational analysis in this study utilizes three methods: (1) partitioning of mean test score changes, (2) analysis of covariance partitioning, and (3) path analysis. This chapter utilizes the first method to partition the total score change into the part due to population shifts and the part due to mean changes within the classification groups. This type of analysis provides considerable detail about how classifying an individual on or or two variables at a time relates to test score change between 1972 and 1980.

#### A. METHODOLOGY

The overall mean for two or more subgroups can be viewed as the sum of the subgroup means when each subgroup mean is weighted in accordance with its proportion of the total group. In other words,

$$\overline{X}_{T} = (p \overline{x})_{1} + (p \overline{x})_{2} + \dots$$

Where  $\overline{X}_{T}$  is the mean for total sample and p is proportional size of each subgroup and  $\overline{X}$  is the mean of each subgroup.

Furthermore, a decline or gain in some time period--in the present case, from 1972 to 1980--is as follows:

$$\overline{x}_{T80} - \overline{x}_{T72} = (p_{80} \ \overline{x}_{80} - p_{72} \ \overline{x}_{72})_1 + (p_{80} \ \overline{x}_{80} - p_{72} \ \overline{x}_{72})_2 + \dots$$

Thus, the total mean can change as a result of either a change in the mean of a subgroup or a change in the proportional representation of that subgroup. We will refer to the first component as the group mean change or G and the second as the population shift change or P. To estimate the magnitude of these components, two different calculations are made. The first component (G) is calculated by applying, for each subgroup, the group's proportion in 1972 to the group's mean in the 1980 population, and then summing over all subgroups. The result is the mean score in 1980 that would have been expected if each subgroup's representation in the population had not changed but its mean had changed. The difference between this number and the observed 1972 mean is the change due to subgroup mean changes, or G. The change due to population shift, or P, is calculated by applying, for each subgroup, the group's proportion in the 1980 population to the group's mean score in 1972, and then summing over all subgroups and proceeding as with G.

The sum of P and G is not equal to the total mean change for the grouping variable, for there may be an interaction between the two components. This interaction component (I) is calculated simply by subtracting the sum of P and G from the total mean change. In most cases, the term is a negligibly small number, and even where it is not its meaning usually is so difficult to interpret that the authors have not attempted to do so.

#### 1. <u>Moment of Group Mean Change</u>

When, in a given time period, the means for two or more subgroups change and, in addition, each subgroup's proportion in the population changes, it is not obvious how much each subgroup may have contributed to the total mean change. For this reason, a fourth statistic was computed that is referred to as the <u>moment</u> of the subgroup mean (M) because of its similarity to the familiar moment or torque in physics. This is the difference between the group's "weighted distance" in 1980 and in 1972. This distance is computed as the product of the group's proportional representation in its own year and the deviation of the group's mean from the 1972 grand mean. The moment, M, or, more precisely, the change in moment may be expressed as

$$\Delta M = p_2(\overline{X}_2 - \overline{X}_{1T}) - p_1(\overline{X}_1 - \overline{X}_{1T})$$

where  $p_2$  and  $p_1$  are the proportions at time 2 and time 1,  $\overline{X}_{1T}$  is the total mean at time 1, and  $\overline{X}_2$  and  $\overline{X}_1$  are the subgroup means at time 2 and time 1. The sum of the  $\Delta M$ 's or the change in the weighted distances over all groups is the distance of the whole population from the original mean, i.e., the difference between the 1980 and 1972 grand mean. In the tables to be presented shortly, the  $\Delta M$  for each subgroup was divided by the total mean change from 1972 to 1980 and multiplied by 100. This quantity is referred to as partition due to group. A large number in the partition due to group column can be a result of a population change for a highly deviant group (an increased proportion for a lowscoring group or decreased representation of a high-scoring group) and/or a score decline for a group that is unusually large relative to other groups. Also note that if, for a particular group,  $p_2 = p_1$ , then  $\Delta M = p(\overline{X}_2 - \overline{X}_1)$ . Interpretation of the score partitions must be in conjunction with score mean and proportion statistics, as well as the change due to population shift index.

In discussing the results for a particular grouping variable, e.g., race/ethnicity, we frequently will refer to the contribution of a



particular category or subgroup, e.g., White students, to the overall decline, judging from the change in the subgroup's moment. It should be understood in all cases that the apparent causation may well be spurious; that is, the change in the variable in question may be a reflection of a change in some more fundamentally causative variable with which it is correlated. In the interest of brevity, we usually will omit the proper qualifications.

#### 2. Some Comments on Mean Score Change

To assist the reader in interpreting mean score changes for a group from 1972 to 1980, the following comments are offered. The mean score for a particular subgroup of the total sample can change from 1972 to 1980 for at least four possible reasons:

1. The members of the category may in 1980 represent a smaller more selected sample (or a larger, less selected sample) of the population than in 1972. For example, since the proportion of students in the academic curriculum decreased from 1972 to 1980, it is reasonable to expect less score decline or even a score gain in this subgroup from 1972 to 1980 if the students remaining in this curriculum represent a higher ability stratum of the population.

2. A difference in the mean score for a subgroup may reflect a change in the personal characteristics of the population of students represented by the subgroup category. For example, in 1972 the category "Suburban students in academic programs" may have been predominantly White students, whereas in 1980 the category may have included a significant fraction of non-White students.

3. From 1972 to 1980, the classroom, school, or community environment may have changed in such a way as to change the behavior of the students at the two times.

4. The meaning of the category may have changed. For example, in 1972 the category "Less than 5 hours of homework" may have meant that the average category member did 4 hours of homework each week, whereas in 1980 the members of that category did an average of 2 hours of homework each week. In other words, the authors may have been only partially successful in holding certain independent variable constant by subdividing the total group into categories. Each of these possible interpretations are exemplified in the results reported in the next section.

#### **B. VARIABLES**

The dependent variables for the relational analyses are the three IRT scores already described in the descriptive analyses: the Vocabulary, Reading, and Mathematics scores. The classification or grouping variables for the score change partitioning are:



-66-

- 1. Total
- 2. Sex: male, female
- 3. Socioeconomic status (SES) (high, middle, low)
- 4. Race/ethnicity (White, Asian-American and American Indian; Black; Mexican-American; other Hispanic)
- 5. School type (public vs. nonpublic)
- 6. Geographical region (Northeast, North Central, South, West)
- 7. Curriculum (academic vs. general and vocational)
- 8. Community type (urban, rural, suburban)
- 9. Homework time/week (5 or more hours vs. less than 5 hours)
- 10. Level of education planned (less than high school, high school, vocational/junior college, college, graduate/professional school)
- 11. Number of "study aids" in home (2 or fewer vs. 3 or more)
- 12. Attitude towards more academic emphasis (agree vs. disagree)
- 13. Projects/labs used in courses (never-seldom vs. fairly oftenfrequently)
- 14. Essays used in courses (never-seldom vs. fairly often-frequently)
- 15. Percent of teachers with M.S., Ph.D. (0-49% vs. 50-100%)
- 16. Percent of White students in school (0-89% vs. 90-100%)
- 17. Percent college-bound in school (0-29%, 30-49%, 50-69%, 70-100%)
- 18. High school offers Advanced Placement courses (Yes/No)
- 19. Semesters of matuematics (4 or fewer vs. 5 or more)
- 20. Semesters of science (4 or fewer vs. 5 or more)
- 21. Semesters of foreign language (3 or fewer vs. 4 or more)
- 22. Athletic participation (Yes vs. No)
- 23. Mother's educational aspirations for son or daughter (No 4-year college vs. 4-year college)



#### C. RESULTS

-68-

The results are presented separately for each of the three tests-vocabulary, reading, and mathematics. In each case the test score is the IRT (Item Response Theory) scaled score. (See Rock et al., 1984a, for a description of the psychometric procedures.)

#### 1. Vocabulary

Vocabulary test results were determined for each of the subcategories of the 23 classification variables. The results are shown in Table D-1 of Appendix D.

That the score of the average female declined more than that of the average male is a fact that must be dealt with. But whether the difference contributed to the mean change for the total sample depends on the proportional size of the female sample in 1972 and 1980. As shown in Section A of Table 7-1, the estimated size of the male population decreased by 207,000 and the female population decreased by 78,000, leading to changes in the proportions from .50 to .47 for males and from .50 to .53 for females. (These estimates are based on actual data cases. In 1972, the sex identification item of the Student Questionnaire was omitted by 5 sample members or 1,017 in the weighted sample, and in 1980 the parallel item was omitted by 1,247 sample members or 127,739 in the weighted sample.) In accordance with the procedure described earlier we can estimate that these changes in population proportions would have resulted in an overall score gain of .01, as shown in Section F of Table 7.1. This is a negligible amount.

We also can estimate how much the total mean would have changed as a result of subgroup mean changes (See Section E in Table 7-1). This is -.76 or practically all of the observed change, since the interaction (see Section G in Table 7-1) is negligible. In other words, the total change did not result from a population shift or from an interaction of the shift and the subgroup means, but from a change in the subgroup means themselves.

How much each subgroup contributed to the 1972-1980 change is indicated by Section D in Table 7-1. This change is  $\Delta M/total$  difference. The results indicate that the change in score means and change in population proportions for the women accounted for twice as much of the change as that accounted for by the men. Since the change in the population



-	Heat INI VOCEDUIATY S	cores in 1972 a	nd 1980 by Sex, w	ith Partitioning
		Ye	ar	
		1050		1980-1972
		<u>1972</u>	<u>1980</u>	Difference
A.	Estimated populatio size	n		
	Male	1,426,000	1 218 000	-207 000
	Female	1,434,000	1,356,000	- 78,000
	Total	2,859,000	2,574,000	-285,000
B.	Proportion of high school senior population			
	Male	0.50	0.47	03
	Female	0.50	0.53	.03
c.	Mean Vocabulary sco	re		
	Male	6.44	5,90	- 54
	Female	6.67	5.69	^8
	Total	6.55	5.79	77
D.	Partition of 1972-19 difference due to g (the change in mome	980 roup nt)		
	Male			337
	Female			67%
E.	Expected change with	n no popul <b>at</b> ion	shift (G)	-0.76
F.	Change due to popula	ation shift (P)		0.01
G.	Change due to intera	action (I)		-0.01

-69-

Table 7-1



proportions was slight, the contribution of each subgroup was approximately proportional to the score decline of each group.

In order to examine the effect of grouping by sex when other variables were held roughly constant, the sex categories were subdivided. The result, when cross-classified by socioeconomic status, was that middle and high SES women displayed the largest scor? declines. When Vocabulary scores were cross-classified by sex and by race/ethnicity, the White females showed the greatest decline (-1.00), and contributed by far the most to the total decline (54%). They were followed by the White males (-.57) and the Black females (-.33). The Black males actually gained in IRT Vocabulary score (.47), but, because their proportional cell size was relatively minor, they had a negligible effect on the overall decline.

The scores of females in each curriculum subgroup declined more than the males, with the scores of females in academic programs declining the most (-1.00). But the women in the general and vocational programs contributed more to the total decline since their population proportion increased more. The differing patterns of change in school program enrollments by males and females was one of the more important educational trends from 1972 to 1980, as was shown in the descriptive analyses. In number, the males in academic programs decreased the most while the number of females in vocational and general programs increased the most. When the regions were cross-classified by se., women in the Northeast emerged with the largest decline and they also contributed the most to the overall decline.

The scores of women who reported they devoted less than five hours each week to homework declined appreciably more than the scores of women who reported five hours or more and contributed much more to the total decline, apparently because the proportional number in the category increased substantially. Students who felt that their schools should have more academic emphasis tended to have lower vocabulary test scores than students who disagreed, and the number of such students increased from 1972 to 1980, especially among women. One of the largest vocabulary score declines for any subgroup was the decline of 1.30 scaled score points for the women who disagreed that their schools needed more academic emphasis. This category also declined dramatically in number, from 616,000 to 347,000. As was noted in the descriptive analysis, satisfaction with the academic program on the part of students decreased markedly from 1972 to 1980. Why women who were satisfied with the academic emphasis in their school should display such a large score decline is not clear. What we do know is that in 1972 they were high scoring (7.20) and in 1980 they were near the middle of the distribution (5.90). The largest score decline was for woman in "mostly White" schools, and these women also contributed most to the overall decline.

b. <u>Socioeconomic Status</u>. Next the total sample was divided into three socioeconomic status (SES) groups, and the changes were partitioned in the same way as for sex. As shown in Table D-1 of the Appendix, the



score decline is correlated with level  $\supset$ f SES; the low group displayed the least decline (-.53) and the high group the greatest (-.93). As with sex, the component attributable to population shift is small, but this result is meaningless since the SES categories for 1972 and 1980 were created by standardizing within year.

Cross-classification of score changes by SES and by race/ethnicity and curriculum yielded results that tended to be similar to the main effects.

Cross-classification of SES by region and by amount of homework indicated that high SES students in the South displayed the greatest decline. Population shifts for combinations of SES and amount of homework contributed appreciably to the overall decline, apparently because of increases in the proportion of low and middle SES students reporting they did less than five hours of homework each week.

Cross-classification of SES by attitude towards academic emphasis and by percentge of White students in the school showed that high SES students who did not agree that their schools should have more academic emphasis--a group with high mean scores--declined appreciably (-.96) and contributed the most to the total decline.

Middle SES students in mostly White schools (90-100% White), which is the largest category by far in size, contributed the most to the total score decline simply because of its size.

c. <u>Race/Ethnicity</u>. In the third major analysis of test score changes by race/ethnicity, the change due to population shifts is non-trivial (Table D-1). As might be expected since the student population is predominantly White, the partitioning of the total change indicates that the White students contributed by far the most to the score decline. The Other Hispanic group showed approximately the same score decline as the White students, but their numbers are so cmall that their contribution to the grand mean is slight.

As indicated above, White female students showed the largest decline and contributed the most to the overall decline.

Cross-classification of race/ethnicity by curriculum yielded one of the largest population shift effects observed (-.39) as a result of a huge decline in the number of a high-scoring subgroup, the White students in academic programs, and because of the simultaneous slight increase in the proportion of a low-scoring group, the White students in vocational and general programs. Apparently, the decline in high school enrollments from 1972 to 1980 occurred largely among White students in academic programs, while White students in vocational and general programs increased slightly, presumably because of shifts of White students from the academic program to the vocational and general program. A small share of the score decline in this table (8%) was attributable to a small increase in the proportion of an extremely low-scoring group, the Black students in vocational and general programs.


Cross-classification of race/ethnicity by geographical region, and by hours of homework indicated that shifts in the proportion of racial groups in various geographical regions may have accounted for as much as .20 points of the total decline, with decreasing numbers of White students in the Northeast and in the North Central the leading contributors.

White students who reported less than five hours of homework each week showed the greatest decline and the largest contribution to the total decline, primarily because of the large size of the subgroup.

Racial/ethnic cross-classification by attitude towards academic emphasis and by the percentage of White students in the school indicated that shifts in these categories may have contributed to much of the total score decline. White students who did not wish more academic emphasis in their school showed the greatest score decline and contributed the most to the total decline, apparently because they represent an above average group that decreased by almost 50 percent as a proportion of the population. The more important fact, however, is that from 1972 to 1980 the number of students who agreed that there should be more emphasis on academics increased by nearly 500,000.

Students in the Other Hispanic subgroup in predominantly White schools declined the most, but the White students in such schools contributed the most because of their larger numbers.

d. <u>School Type</u>. Students attending nonpublic schools tended to decline in IRT vocabulary by the same amount as pub. c school students, and controlling for sex, SES, race, curriculum, geographical region and hours of homework did not make an appreciable difference (Table D-1). Because of the relatively large size of the public school groups, they accounted for most of the total decline. The largest decline was shown by nonpublic students in the South. Their number was too small to result in a significant contribution to the total decline.

e. <u>Geographic Region</u>. Classification of the data by geographical region revealed few insights not reported already. Within each region higher mean declines tended to be displayed by females, high SES students, Whites, students enrolled in the academic curriculum, students reporting less homework, students disagreeing that their school should place more emphasis on academics, and by students in predominantly White schools. The substantial decrease in the number of White students in the Northeast and academic students in the Northeast, both of which are high-scoring groups, resulted in significant contributions to the overall score decline.

f. <u>Curriculum</u>. Grouping by curriculum indicated that shifts in the numbers enrolled in each curriculum category resulted in major contributions to the score decline. The academic students displayed the greatest mean vocabulary test score decline (-.67 vs. -.50 for the general and vocational students) and also contributed to the overall decline since they constituted a high-scoring group that decreased in number. However, the vocational and general category contributed slightly more since they were a low-scoring category that increased in number.

The cross-classifications yielded no new findings except for attitudes toward academic emphasis and percentage of White students in the school. The striking <u>decline</u> in the number of academic students who did not think there should be more academic emphasis resulted, along with concomitant changes in other categories, in an unusually high figure for the total change due to population shift, namely, .45. In other words, over half of the total decline could be attributable to this effect alone.

The large decline in the number and proportion of academic students in predominantly White schools also resulted in a large estimate of total change due to population shift (-.34).

g. <u>Community Type</u>. Classification by community type generated results which define further the schools which were the large contributors to the decline. Students in urban schools declined the most, but, because of their larger number, suburban schools contributed most to the decline. The cross-classifications point to the following categories within community type as significant contributors to the total decline:

> Suburban females Suburban high-SES students Suburban White students Suburban academic students Suburban students reporting less than five hours of homework Suburban students disagreeing that there should be more academic emphasis

Suburban students in predominantly White schools

Urban students in the Northeast displayed an unusually large decline (-1.43) but, because of their relatively small number, they did not contribute appreciably to the decline.

h. <u>Homework per Week</u>. The number of hours that the students reported they devoted to homework each week resulted in a number of substantial differences when the students were divided into those groups that reported less than five hours each week and those that reported five hours or more. Hours of homework clearly is correlated with IRT Vocabulary score. Students reporting less than five hours each week had mean scores in 1980 that were nearly one-half a standard deviation less than those reporting five hours or more. Females who reported less than five hours declined 1.09 on the average versus a decline of .22 for those who reported five hours or more. The comparable figures for men were -.60 and +.03. Since the number of women reporting less than five hours of homework increased from 28 percent of the total sample in 1972 to 37 percent in 1980, the contribution of the group to the total decline amounted to -.44 or 56 percent of the total change.



Similarly, the SES groups that reported less homework showed substantially greater declines than those reporting more homework, and the same held for the other cross-classifications. The cross-tabulation of "academic emphasis" and hours of homework resulted in another large estimated population shift, apparently attributable primarily to the massive increase (from 790,000 to 1,300,000) in the number of students who reported less than five hours homework and who agreed that their schools should have placed more emphasis on basic academic subjects.

i. <u>Education Planned</u>. Unfortunately, a large number of students failed to answer this item in the 1972 survey, which introduces added uncertainty in the results. Of most interest is the decrease in the proportion planning to attend college. As a result, this category contributed more than the others to the overall decline.

j. <u>Study Aids in Home</u>. Sample members who reported three or more study aids in their home had higher mean vocabulary scores than students who reported two or less aids, but both groups showed approximately equal declines. However, because the proportion of students reporting three or more study aids declined (from .8 to .7), that category contributed appreciably more to the overall decline than the "two or less" category. Furthermore, the cross-classifications by sex, SES, race, and curriculum indicate that the females who identified themselves as being White, Asian-American or American Indian, contributed the most to the decline.

k. <u>Attitude towards Academic Emphasis</u>. Students who disagreed with the proposition that there should be more emphasis on academics in the curriculum had higher Vocabulary scores than those who agreed with the statement and also showed greater declines; the women, White, Asian-American, and American Indian, and academic curriculum categories again contributed the most to the decline.

1. <u>Instructional Variables</u>. Two measures were selected as important dimensions of instruction. The first measure, whether projects and labs were used in the courses taken by the student during "this year," generated no relevant differences. The second measure, whether essays were used, is more interesting. Those who reported essays were used fairly often or frequently had higher mean scores in both 1972 and 1980. Women who reported they never or seldom wrote essays declined more than women who wrote essays "fairly often or frequently." "Other Hispanic" students who never or seldom wrote essays declined more than other racial/ ethnic groups and students who disagreed that there should be emphasis on academic matters and who never or seldom wrote essays declined markedly (-1.34). They also declined in number.

m. <u>School Variables</u>. Of five measures selected to reflect possibly important school variables, three provided noteworthy results. Students in schools in which 90 to 100 percent were White displayed greater declines than students in schools with less than 90 percent Whites, and in these predominantly White schools, the White students declined most and also accounted for the largest share of the total decline.

Secondly, schools which did not offer Advanced Placement courses displayed greater declines and contributed most to the overall decline; within this group, the higher SES, White women in academic programs declined the most and contributed most to the overall decline.

Thirdly, students in schools with 50 percent to 100 percent of full-time teachers with advanced degrees declined more than students in schools in which less than 50 percent of the teachers had advanced degrees; they also accounted for a larger share of the overall decline. In addition, the following categories displayed high declines and large shares of the total decline:

- High percentage schools in the South
- High percentage schools in which students did less homework
- Low percentage schools in which students disagreed with more academic emphasis

n. Exposure to Subject Matter. The number of semesters in which the student enrolled in mathematics, science, and foreign language seems to have played an important role. Students enrolled in 4 or fewer semesters of mathematics or 4 or fewer semesters of science displayed greater declines and contributed more to the total decline than students taking 5 or more semesters. The increase in the proportion of students taking more mathematics and science could have produced a gain in overall mean score; however, the group means decreased enough to offset the population increase.

The results in the foreign languages are somewhat different from those for mathematics and science. Students who reported they took 4 or more foreign language courses had relatively high Vocabulary scores (an interesting finding in itself), and the proportion of students taking fewer language courses increased. However, when the sex of the student is taken into consideration, the absolute number of the men taking fewer language courses decreased by approximately 130,000 and the number of women by 15,000. Thus, there was a loss from the sample, from 1972 to 1980, of a relatively large number of men who took fewer language courses, and a corresponding increase in the proportion of women taking fewer language courses. Nevertheless, the mean decline in vocabulary score was greater for the women than for the men, so the overall decline cannot be said to result simply from a population shift. Indeed, only 7/78ths or 9 percent was attributable to a population shift. But it was the men who "caused" the population shift in terms of their decrease in absolute number. This effect demonstrates the complexity of attributing causation to subgroups of the total sample.

o. <u>Participation in Athletics</u>. This measure, which could have been discussed as an "exposure" variable, was included in this analysis because of the changes in rate of participation from 1972 to 1980. Specifically, this rate increased from 45 percent to 52 percent. The gain was largely



attributable to the women athletes, who increased in number by 103,000 or 23 percent while the men athletes decreased by 47,000 or 6 percent. The female nonathletes declined most in score and contributed most to the overall decline, presumably because of their relatively large number.

As far as the other cross-classifications are concerned, the pattern followed the main effects already reported.

P. <u>Mother's Educational Aspirations for Student</u>. In both 1972 and 1980 the mean Vocabulary scores of students for whom mothers had aspirations of four-year college attendance were substantially higher than thore of the balance of the students, but the mean scores of the women in this category declined substantially from 1972 to 1980 and contributed most to the total change. "Other Hispanics" in the "No four-year college" category also declined substantially but, because of their relatively small numbers, did not contribute appreciably to the total decline.

### 2. Reading

For this test and the Mathematics test to follow, we will focus primarily on results that either strongly reinforce conclusions suggested by the Vocabulary results or that are at odds with them.

As shown in Table 7-2, the results for IRT Reading, when classified by sex, are similar to those of IRT Vocabulary, with the exception that the change in moment of the female category is less than it was for the Vocabulary score, apparently because the males declined almost as much as the females. Between 1972 and 1980 the scores of the males and females in Reading converged to a point of equality.

The results for other grouping variables (shown in Appendix Table D-2) are similar except that mean Reading scores for students in the South declined more than for other regions and more than the Vocabulary scores declined. The cross-classifications indicate that males in the South and middle-class students in the South declined the most.

The cross-classification of community type and SES is of particular interest in that the results appear to document the migration of higher SES subgroups from urban areas. The middle and high SES urban and suburban categories decreased in number while the rural categories all increased in number. The largest decline was for high SES students in the suburbs, followed by low SES students in rural areas. Whether the "movers" or the "stayers" changed most in mean score would have required longitudinal data, but the suggestion is that the movers declined least since the mean for the rural category, which increased the most in size, declined the least in score.

The analysis by educational aspirations revealed unusually large declines, especially for students with low educational aspirations. However, the relatively large amount of missing data for this particular measure requires that the results be interpreted with caution.

ą,

# Table 7-2

-77-

# Mean IRT Reading Scores in 1972 and 1980 by Sex with Partitioning

		Ye		
		1972	<u>1980</u>	1980-1972 Difference
A.	Estimated population size			
	Male Female	1,427,000 1,435,000	1,215,000 1,352,000	-212,000 - 83,000
	Total	2,862,000	2,567,000	-295,000
B.	Proportion of population			
	Male Female	0.50 0.50	0.47 0.53	03 .03
c.	Mean Reading score			
	Male Female	9.83 9.95	8.95 8.96	-0.88 -0.99
	Total	9.89	8.95	-0.93
D.	Partition of total change due to group (M)			
	Male Female			44% 56%
E.	Expected change with	no population	shift (G)	-0.93
F.	Change due to populat	ion shift (P)		0.00
G.	Change due to interac	tion (I)		-0.00

.

•

, 34

The measure of study aids in the home also revealed large declines-larger for comparable groups than for the Vocabulary score. Those students reporting two or fewer study aids declined more than the "three or more" category. High SES students reporting two or fewer aids showed a high decline (-2.21), but because they were a small proportion of the population their contribution to the overall decline was slight.

### 3. Mathematics

Generally, the grouping variables that seem to be implicated in the declines of the Vocabulary and Reading scores are also implicated in the decline of the IRT Mathematics score. (See Table 7-3.) However, the magnitude of the changes due to population shifts tends to be greater than in the case of Vocabulary and Reading. Furthermore, the major grouping variables generated a number of noteworthy differences. (Appendix Table D-3.) Students in the South declined the most (-1.88) and contributed most to the overall decline. Students who reported less than five hours homework each week declined considerably and accounted for most of the total decline (99%). Other subgroups that declined appreciably and contributed to the total decline were:

Students who disagreed that there should be more emphasis on academics

Males (unlike vocabulary and reading)

White students

When the sample was grouped by curriculum, neither group (i.e., neither the academic group nor the vocational/general group) declined appreciably, but because the academic group represented a high scoring group that declined substantially in proportion and the vocational and general group represented a low scoring group that gained in proportion, the value of P, the total change due to population shifts, was substantial (-.53). In addition, the cross-classifications pointed to certain subgroups as possibly having an important role:

Males who disagreed that there should be more academic emphasis

Low SES White students

Low SES students reporting less than 5 hours of homework

White students reporting less than 5 hours homework



# Table 7-3

-79-

<u>Me</u>	an IRT Mathematics Sco	ores in 1972 an	nd 1980 by Sex wit	<u>h Partitioning</u>
		Yea	ar	1080-1072
		<u>1972</u>	1980	Difference
<b>A.</b>	Estimated population size			
•	Male	1,426,000	1,214,000	-213,000
	Female	1,435,000	1,346,000	- 89,000
	Total	2,861,000	2,560,000	-301,000
В.	Proportion of population			
	Male	0.50	0.47	03
	Female	0.50	0.53	.03
C.	Mean Mathematics score			
	Male	13.79	12.83	-0.95
	Female	12.09	11.39	-0.70
	Total	12.94	12.07	92
D.	Partition of total change due to group (M)			
	Male Female			55% 45%
E.	Expected change with	no population	shift (G)	-0.83
F.	Change due to populat	ion shift (P)		0.04
G.	Change due to interac	tion (I)		-0.01



.

.

Public school students reporting less than 5 hours homework

Students in the South reporting less than 5 hours homework

- Academic students who disagreed that there should be more emphasis on academics
- Vocational and general curriculum students who reported 2 or fewer study aids in the home (low scoring students whose number nearly doubled)
- Students who never or seldom had projects and labs and who reported less than 5 hours homework (large group with large decline who increased in proportion)
  - C. SUMMARY OF MEAN SCORE CHANGE PARTITIONING

The most relevant statistics concerning the score decline for vocabulary, Reading, and Mathematics are summarized in Tables 7-4, 7-5, and 7-6. The first conclusion one can draw from these tables concerns the similarity of the numbers in them and the agreement across the three tests with respect to the subgroups that may have contributed the most to the total decline, as reflected in the quantity we have chosen to call  $\Delta$  M, the change in the moment of the subgroup mean. This quantity might be described as the weighted distance of the 1980 group mean from the 1972 grand mean minus the weighted distance of the group's 1972 mean from the 1972 grand mean. Thus, it reflects (1) any change in the size of the group (as a proportion of the total sample), (2) any change in the mean of the group, and (3) the distance between the mean of the group and the grand mean.

Because of the many and complex reasons why a subgroup may have contributed to the overall decline, we have prepared Table 7 which lists the major reasons for the contribution of each of the subgroups that contributed the most.

Table 7-7 highlights a major conclusion from the score change partitioning analysis, namely that we must recognize three major categories of groups that contributed to the total score declines. The first category is those large groups that contributed a large amount to the decline simply because they represented a large number of individuals whose scores declined, but no more or less than students in other groups. The second category comprises population shifts--either increases in the proportional representation of low-scoring groups or decreases in the proportional representation of high-scoring groups. The third category comprises groups that exhibited large score declines and were substantial in size, for if they were small their contribution to the total decline was negligible.



	 _
<b>-</b>	
4	
C	

### Table 7-4

Variables	Population Shift	Group Mean Change	Largest Change in Moment (percent)	Subgroup With Largest Change in Moment
Demographic				
Sex	.01	76	67	Females
SES	06	74	46	Middle
Race/ethnicity	17	73	80	White students
Geographical region	06	79	32	South
Community type	09	88	55	Suburban
Student behaviors				
Currieu <sup>1</sup> um	25	58	52	Vocational, general
Hours of homework	12	66	85	Less than 5 hours
Educational plans	.00	-1.45	35	College
Semesters, mathematics	.14	-1.01	71	4 or fewer
Semesters, science	.02	88	91	4 or fewer
Semesters, language	07	78	71	3 or fewer
Athletic participation	.00	87	55	No participation
Attitude, academics	24	70	58	Disagree with more emphasis
School characteristics				
Public, nonpublic	.02	92	92	Public
Projects used	06	79	57	Never, seldom
Essays used	01	83	57	Fairly often, frequently
Teachers ed.	.12	93	53	50-100% advanced
Percentage White	11	74	66	90-100%
Advanced placement	.12	-1.00	58	No advanced placement
Percentage, college	-,16	72	32	50-69%
Home support				
Study aids	20	68	62	3 or more
Mothers aspirations	.10	-1.05	69	4-year college

# Population Shift Effect, Subgroup Mean Change Effect, and Change in Moment<sup>1</sup> of Grouping Variables by IRT Vocabulary Score

<sup>1</sup>Moment refers to the extent to which a subgroup may have contributed to the total decline in comparison to the other subgroups of that variable.

Variables	Population Shift	Group Mean Change	Largest Change in Moment (percent)	Subgroup With Largest Change in Moment
Demographic				
Sex	.00	93	55	Females
SES	07	90	44	Middle
<b>Ra</b> ce/ethnicity	21	89	75	White students
Geographical region	05	99	40	South
Community type	06	-1.11	54	Suburban
Student behaviors				
Curriculum	31	69	59	Vocational, general
Hours of homework	14	80	89	Less than 5 hours
Educational plans	01	-1.86	35	Vocational, Junior College
Semesters, mathematics	.19	-1.25	71	4 or fewer
Semesters, science	.02	-1.08	91	4 or fewer
Semesters, language	09	96	71	3 or fewer
Athletic participation	.03	-1.07	50	Yes, participates
Attitude, academics	30	83	58	Disagree with more emphasis
School characteristics				
Public, nonpublic	.03	-1.12	92	Pu <b>b</b> lic
Projects used	07	96	62	Never, seldom
Essays used	02	-1.03	53	Fairly often, frequently
Teachers ed.	.12	-1.12	52	50-100% advanced
Percent <b>ag</b> e White	13	91	58	90-100%
Advanced placement	.12	-1.20	56	No advanced placement
Percentage, college	18	89	36	30-49%
Home support				
Study aids	21	82	56	3 or more
Mothers aspirations	.12	-1.32	65	4-year college

# Population Shift Effect, Subgroup Mean Change Effect, and Change in Moment<sup>1</sup> of Grouping Variables by IRT Reading Score

<sup>1</sup>Moment refers to the extent to which a subgroup may have contributed to the total decline in comparison to the other subgroups of that variable.



Table 7-5

# Table 7-6

# Population Shift Effect, Subgroup Mean Change Effect, and Change in Moment<sup>1</sup> of Grouping Variables by IRT Mathematics Score

			Largest	
		Group	Change in	Suogroup With
	Population	Mean	Moment	Largest Change
<u>Variables</u>	Shift	Change	(percent)	in Moment
Demographic				
Sex	04	83	55	Male
SE S	11	81	39	Middle
Race/ethnicity	34	81	74	White students
Geographical region	08	91	59	South
Community type	07	-1.09	51	Suburban
Student behaviors				
Curriculum	53	45	51	Vocational, general
Hours of homework	25	59	99	Less than 5 hours
Educational plans	09	-2.21	38	College
Semesters, mathematics	.44	-1.73	91	4 or fewer
Semesters, science	.05	-1.10	101	4 or fewer
Semesters, language	13	90	72	3 or fewer
Athletic participation	.14	-1.15	58	Yes, participates
Attitude, academics	33	74	61	Disagree with more emphasis
School characteristics				
Public, nonpublic	.04	-1.15	97	Public
Projects used	11	89	68	Never, seldom
Essays used	02	99	56	Never, seldom
Teachers ed.	.16	-1.19	51	0-49% advanced
Percentage White	20	80	59	90-100%
Advanced placement	.22	-1.32	61	No advanced placement
Percentage, college	29	77	32	30-49%
Home support				
Study aids	34	68	50	2 or fewer
Mothers aspirations	.21	-1.48	73	4-year college

<sup>1</sup>Moment refers to the extent to which a subgroup may have contributed to the total decline in comparison to the other subgroups of that variable.



# Table 7..7

# Largest Contributors for All Three Tests

Subgroup	Apparent Reason
Middle SES	Largest category and equal decline
White students	Largest category and largest decline
Schools in South	Largest category in 1980 and largest decline in reading and mathematics
Suburban students	Largest category and equal decline
Vocational and general students	Large low-scoring group that increased in size
Students doing less than 5 hours homework	Increasingly large group with larger decline
Students disagreeing with more academics	High-scoring group declining in number and larger decline
Students taking fewer math courses	Overall increase in enrollments should have increased scores but "4 or fewer" group showed large score decline
Students taking fewer science courses	Same as mathematics
Students taking fewer foreign language courses	Increase in proportion of a low-scoring group
Students in public schools	Gain in proportion of high-scoring group (nonpublic students) should have in- creased scores but both public and nonpublic students declined
Students never or seldom doing projects or lab work	Increase in low-scoring group
Students in schools without Advanced Placement	Increase in schools offering AP should have increased scores but "No AP" group declined in scores



.

# Table 7-7 (continued)

<del>-</del>85-

# Largest Contributors for All Three Tests

Subgroup	Apparent Reason
Students with mother's wanting them to attend 4-year colleges	Increase in proportion should have increased scores but large score decline for 4-year college group
Schools predominantly White	High-scoring group declining in number and showing larger score decline
Largest Contributors for Vocabulary	and Reading Only
Females	Larger decline for larger group (but declined less in math)
Students with more study aids in home	High-scoring group declining in proportion
Schools used essays often or frequently	High-scoring group declining in pro- portion (and in all tests the "Never or seldom" group declined more in score)
Teachers' education more advanced	Increase in proportion should have raised scores but students with more educated teachers declined more. Slight reversal in mathematics
Other Possibly Large Contributors	
Educational plans	Small decrease in proportion of high- scoring college-bound group and small increase in lower scoring junior college group (many 1980 subjects omitted this item in the base year survey)
Athletic participation	Vocabulary and Reading: Nonparticipants declined more. Math: Participants declined more
Percentage of graduates attending college	In general, gain in proportion of low- scoring schools with 30-49% attending college



•

In the first category of large groups there were the following:

Middle SES students

Suburban schools

Public schools

The results for these groups are of little interest. Of much more interest is the second category of groups that contributed to the score decline primarily because of a change in their proportional size from 1972 to 1980. This category includes:

Vocational and general students (increased greatly in number)

Students never or seldom doing projects or laboratory work (increased in number)

Students wich more study aids in home (decreased in number)

- Students planning college attendance (high-scoring group decreasing moderately in number)
- Students in school with lower proportion attending college (lower scoring group increasing in number)
- Students taking fewer foreign language courses (increased in number)
- Students in predominantly White schools (high-scoring group decreasing in number)

The third category of large groups--some of which got larger-that exhibited large score declines includes:

Students doing less than 5 hours of homework each week

Students disagreeing that there should be more academic emphasis (but number of students wanting more academic emphasis increased in number)

Students taking fewer math courses

Students taking fewer science courses

White students (higher scoring group declining in number and also displaying larger declines)

Students in schools in the South

Students not taking Advanced Placement courses



-87-

Students in schools "never or seldom" using essays

Students with mothers wanting them to attend four-year colleges

Females (for vocabulary and reading only)

Males (for mathematics only)

Obviously, these observations suggest some general conclusions about the score decline, but we will withhold comment until after the results of the covariance analysis and the path analysis are presented.

In summary, score declines on all three tests were primarily the result of declines in subgroup mean scores, but, in some cases, population shifts contributed to or helped to resist these declines.

Score declines on all three tests were associated with population shifts in the curriculum, academic attitudes, study aids, and race/ ethnicity variables. Population shifts associated with the semesters of mathematics, percentage of teachers with advanced degrees, availability of advanced placement courses, and the amount of education mothers wanted the students to obtain variables helped to resist score declines.

The largest changes in subgroup mean affecting all three tests were associated with the educational plans, semesters of mathematics, mother's educational aspirations for the student, and advanced placement variables. No variable had group mean changes which resisted score decline.

The subpopulations with the largest score changes on all three tests were White students, students doing less than five hours of homework a week, students taking four or fewer semesters of mathematics and of science, students who felt their schools should not have more academic emphasis, students in schools where projects/labs/essays were seldom or never used, students in schools 90 percent or more White and those which sent 50 to 69 percent of their students to college, students in schools that did not offer advanced placement courses, and students whose mothers wanted them to attend a four-year college.



### CHAPTER VIII

### PARTITIONING TEST SCORE CHANGES BY BLOCKS OF VARIABLES USING ANALYSIS OF COVARIANCE

Chapter VII describes the extent of the relationship between selected population and behavioral shifts to score decline. This type of analysis provides considerable detail about how classifying an individual on one or two variables at a time affects test score changes from 1972 to 1980. This procedure, however, does not lend itself to evaluating the impact of any one given variable or a set of variables while controlling for the effects of other numerous confounding variables. This section attempts to look at the relative impact of selected blocks of variables on the 1972-1980 mean score changes both before and after controlling for other confounding "blocks" of variables. The blocks differ with respect to whether they consist of variables that are: (1) responsive to educational policy (e.g., student behaviors and school characteristics), and/or (2) descriptive of either students, schools, or the home.

The four blocks of variables in this analysis include:

race, sex, family SES, region of the country and com-Demographics: munity type. amount of homework, athletic participation, study Student Behaviors habits, plans for higher education, number of and Attitudes: semesters in selected subject matter areas, and type of program, e.g., academic, vocational or general. School student self-reports yielded school means based on: Characteristics: condition of the buildings, library, quality of academic instruction, school reputation, teacher interest, amount of homework done, school's emphasis on academics, labs in courses, essays in courses, courses in subject matter areas, age of seniors, and

courses in subject matter areas, age of seniors, and ratings on employment counseling. School questionnaire data included percent of White teachers, percent of teachers with Master's or Ph.D., teacher turnover, dropout rate, percent of White students, percent of students in academic program, studentteacher ratio, availability of bilingual education, availability of advanced placement courses, school type, and size of senior class.

<u>Home Educational</u> <u>Support System</u>: student self-report of parental influence on plans, father's educational level, mother's educational level, mother's educational plans for student, study aids available in the home.



199

-88-

Similar to the previous section we would like to examine the relative influence of each of the above blocks of variables on test score decline. However, unlike the previous analysis, we would also like to get a "gross" estimate of each block's influence while controlling for the remaining blocks. The procedure used here will be a "step down" analysis of covariance where the primary outcome is the difference between covariate adjusted means. For example, if when controlling for the demographic block alone one can significantly reduce the difference between the two cohort means (i.e., reduce the mean score decline), then one could reasonably argue that changes in demographic makeup of students between 1972 and 1980 may indeed be contributing to the score decline. In a sense, the estimated reduction in the spread between the 1972 and 1980 means following adjustment for differences in demographic makeup is an estimate of the maximum impact of demographics on score decline since the remaining covariate blocks are not being considered. In order to arrive at an estimate of the unique contributions of the demographic block to 1972 to 1980 mean test score changes, one first compares the spread between adjusted means when all blocks are controlled for (full model) and then subsequently recomputes the adjusted means when the one block of interest is removed from the full model yielding a reduced model. Assuming that the assumptions of the analysis of covariance (ANCOVA) are reasonably met, then the differences between the two spreads, i.e., the difference between the spread of the reduced model and that of the full model, is an estimate of the block of interest's relative influence on the observed cohort mean differences while controlling for the other blocks.

This procedure is similar to defining unique partitions of the predictable variance in commonality analysis, but since it estimates adjusted means, it partitions score differences rather than variances. This approach also gives an indication of the direction of influence of the covariate blocks. That is, some blocks might be characterized by variables that may help resist the drop in scores. That is, if the behavior included in the block had not changed in the direction that they did, the score decline would have been even greater.

Table 8-1 below presents estimates of the maximum potential contributions of each block to either the decline or resistance to the decline in each of the three test score areas. These estimates of the reduction in mean spreads are maximum potential contributions since each block is treated separately without controlling for the remaining blocks.

Inspection of Table 8-1 indicates that controlling separately for the school characteristic and student behavior blocks leads to the larger reduction in the spreads between the 1972 and 1980 means. This suggests that other things being constant, if student behaviors or school characteristics had not changed between 1972 and 1980, the decline would be considerably smaller. This finding is particularly true for school characteristics.



### Table 8-1

-90-

# Differences between 1972-80 Senior Test Score Means When Each Block Is Separately Held Constant

	Vocabulary	Reading	<u>Math</u>
Observei Mean Decline (d)	.85	1.04	1.04

Mean Decline or Increase after Adjusting for:

	Change		Change		Change	
	<u>d</u> 1	( <u>d-d</u> 1)	$\underline{d^1}$	( <u>d-d</u> 1)	<u>d1</u>	$(\underline{d-d^1})$
Demographics Only	.63	. 22	. 79	.25	.60	.44
Student Behaviors Only	.28	. 57	.48	. 56	. 56	.48
School Characteristics Only	.21	.64	.24	.80	.23	.81
Home Support Only	1.06	21	1.31	27	1.45	41

Note: d<sup>1</sup> = Expected decline after adjusting only for changes in individual block of variables.

d-d<sup>1</sup> = Net change, i.e., that part of the mean decline due to an individual block while ignoring the effects of the remaining blocks.

For a block to have the potential to contribute to the score decline, it must consist of variables that are primarily positively (negatively) related to tested achievement also positively (negatively) related to a "dummy" coded indicator of whether an individual is a member of the 1972 senior cohort (coded "1") or a member of the 1980 senior cohort (coded "0"). Significant variables in the student behavior block regression equation that fit this pattern are reported below.

That is, changes from 1972 to 1980 in individual student school behaviors that contributed most to the score decline were:

- o Taking less semesters of foreign language courses. This reduction was proportionately greater for females.
- o Spending less hours on homework. This reduction was proportionately greater for females.
- o Taking less semesters of science courses.
- o Less likely to be in the academic curriculum.



Similarly, changes from 1972 to 1980 in individual school characteristics that contributed most to the score decline, were:

- o An <u>increase</u> in the proportion of students rating the school as needing more academic emphasis.
- A <u>decrease</u> in school means with respect to homework done by students.
- A <u>decrease</u> in school means with respect to semesters of foreign language courses taken by students.
- o An increase in schools with a high dropout rate.
- A <u>decrease</u> in school means with respect to laboratory courses taken by students.
- A <u>decrease</u> in students' rating of their school's reputation in the community.
- A <u>decrease</u> in students' rating of the quality of their academic instruction.
- A <u>decrease</u> in students' rating of the physical condition of their school buildings.

Significant variables in the demographic block regressions were being White and coming from the Northeast.

It is interesting to note that when variation in the home educational support block is held constant, the spread between the adjusted means increases. This finding is suggesting that if the home environment had not changed between 1972 and 1980 in a positive direction the declines would have been even greater. That is, the means on those variables in the home educational support block that were most highly correlated with achievement tended to go up from 1972 to 1980. Both parents' education and mother's educational plans for the child's education tended to increase from 1972 to 1980. Among the variables in this block, these three had the highest relationship with the achievement test scores.

Another way of looking at this approach to estimating the relative influence of blocks as either contributing or resisting score decline is to think of it as predicting achievement (e.g., vocabulary) from the variables within a block within each cohort using the pooled within group regression weights and then getting the 1972 and 1980 senior cohort means on the predicted scores. If the 1972 mean is higher than the 1980 mean and the variables that get the biggest positive weights in the regressions are positive student behaviors (e.g., in the student block) or positive school characteristics (in the school block), then it would appear that changes from 1972 to 1980 in student or school behaviors have contributed to the decline. If the predicted cohort means are in the opposite direction (i.e., 1980 higher than 1972), then we would expect this direction of



change in behavior to resist any decline (e.g., the home support block). The covariance adjustment in Table 8-1 for the home educational support block suggests that if there had not been a change from 1972 to 1980 in the direction of increased home educational support, the decline would have been greater than the observed .85; more like 1.06.

As indicated above, the estimates of the relative contribution of each block to the decline (or to resisting the decline) are likely to be overstated unless one estimates the impact of a given block on the tested achievement while controlling for the remaining blocks.

Figure 8-1 presents an estimate of the relative contributions of each block to the decline while controlling for the remaining blocks. These estimates are simply the difference between the covariance adjusted means for the full model (all blocks as covariates) and that of the reduced model where one block has been removed. For example, the impact of the demographic block on the vocabulary score ducline is estimated in the following way. The difference between adjusted means for the full model (all blocks as covariates) is subtracted from the difference between adjusted means for the reduced model. In the case of vocabulary we have the difference between adjusted means for the reduced model (.16) minus the difference between the adjusted means for the full model (.09) yielding .07 of a test score point as an estimate of the impact of changes in demographics on the score decline while controlling for the remaining blocks of variables. As in commonality analysis, the partitions of score point differences attributable to each block will not, in general, add up to the observed mean differences. The partitions by block do, however, give one a rough estimate of the relative importance of each block in contributing or resisting the score decline. A technical note describing the computations in more detail may be found in Appendix E. Appendix E also shows the structure coefficients indicating the important variables in each block.

The negative numbers associated with the home educational support system suggest that changes in variables in this block from 1972 to 1980 are in the direction of resisting score decline. This effect is relatively constant for all three tested achievement areas. The effect of changes in demographics from 1972 to 1980 have a consistent but relatively small impact in the direction of contributing to the score decline. Compared to changes in demographics and home educational support systems, changes from 1972 to 1980 in student and school characteristics have considerable impact in the direction of contributing to score decline.

This is true for reading and even more so for mathematics. In the case of mathematics, school characteristics and student behaviors are roughly seven to eight times more important than are changes in demographics with respect to contributing to score decline.

The relatively greater impact of student academic behavior and school characteristics on reading and especially mathematics is consistent



203



Figure 8-1

204



with what one would expect since gains or losses in these achievement areas would appear to be more sensitive to formal schooling.

While the above approach does yield rough but seemingly not unreasonable estimates of the relatively unique impact (and direction) of logical blocks of explanatory variables on 1972-1980 achievement changes, one must be cautious when interpreting the adjusted mean differences from any analysis of covariance in a nonexperimental situation. In particular, the assumptions of homogeneity of achievement covariate regressions probably do not hold, but it is expected that the main effects are large compared to the interactions. The next section which attempts to estimate and compare path models underlying achievement separately for 1972 and 1980 suggests that indeed there is some interaction between 1972 and 1980 and achievement-covariate relationships. These interactions are, however, generally small compared to the main effect and are always ordinal.

Another potential pitfall of the above methodology is the reliance on differences between adjusted means for interpretation of the relative sizes of the effect associated with a particular block. The difference between adjusted means is, of course, simply estimated by a partial raw score regression coefficient while holding constant numerous other variables in the system. All the problems of colinearities and the accompanying instability in the parcicular regression coefficient defining the adjusted means are also present. It is hoped that these types of problems are somewhat alleviated by the large sample size within each cohort.



### CHAPTER IX

-95-

#### PATH ANALYSIS

The earlier section dealing with population shifts and mean changes within populations indicated that while the score decline cut across almost all groups of individuals, it appeared to be larger for White individuals in vocabulary, reading, and mathematics and females in vocabulary and reading. In an effort to see if race, sex and SES subgroups appeared to go through a somewhat different educational process in 1980 from their counterparts in 1972, three path mouels were constructed--one for each outcome variable. The three final outcome variables were vocabulary, reading, and mathematics scores. The three path models shared the same assumed causal chain but differed in the final dependent variable and one intermediate educational process variable. The intermediate process variable that was varied depending on the outcome was "number of semesters of language courses," which was used for the vocabulary and reading outcomes and "number of semesters of mathematics," which served the same role for the mathematics achievement outcome.

Figure 9-1 presents the path model that was assumed to underly the process in vocabulary achievement. The path models investigate how membership in those groups showing the greatest declines may differ in selected educational processes from those who showed lesser declines. It also yields the net effect of being in these groups on the achievement outcomes. That is, the effect of group membership on the achievement outcomes while controlling for all variables in the model. The finding of a diminishing effect on achievement outcomes in 1980 compared to 1972 for minority, SES, or sex group membership could point to possible change in the educational process towards emphasizing equality of educational outcomes.

The path coefficients or regression weights in Figures 9-1 - 9-3 are raw score regression weights since the model compares path coefficients across populations. The usual standardized path coefficients would be inappropriate for cross population comparisons because of their sensitivity to sample differences in variability. The 1972 and 1980 estimates of the effect of the variable at the tail of the arrow on the variable at the head of the arrow are shown as pairs with the 1980 estimates in parentheses. Arrows are only present in the Figure if either the 1972 or 1980 coefficient or both are at least twice their standard errors. Where the comparable 1972 and 1980 path coefficients differ by as much as four pooled standard errors an asterisk is placed along side. This conservative approach reduces the possibility of presenting many relatively trivial effects and is consistent with the average design effects for the 72 and 80 cohorts.

Two of the intermediate dependent variables, attendance at a private school and membership in the academic curriculum, are dichotomous and as a result do not meet the assumptions of ordinary least squares (OLS)



Figure 9-1. Hypothesized Path Model to Explain Vocabulary Achievement

# BEST COPY AVAILABLE





BEST COPY AVAILABLE

# BEST COPY AVAILABLE

Figure 9-3: Hypothesized Path Model to Explain Mathematics Achievement



**BEST COPY AVAILABLE** 

estimation. Estimates based on ordinary least squares were used, however, since the computer programs available for the preferred technique (logistic regression) do not readily handle either missing data or sample weights. It has been our experience that in the case of large samples the OLS estimates of effect size are reasonably close to those of the logistic regression and the potential for bias lies in the tendency of OLS to underestimate the size of the coefficient's standard error. With that in mind we will be cautious in our interpretations of effect sizes of variables assumed to be explanatory variables for these two outcomes.

### A. DETERMINANTS OF MOTHERS' EDUCATIONAL ASPIRATION FOR THE CHILDREN

Mothers' educational aspiration was predicted reasonably well by the demographic explanatory variables. The multiple correlation was .42 in 1972 and .39 in 1980. Inspection of Figure 1 indicates that after controlling for SES, mothers of non-White students tend to have considerably more ambitious educational plans for their daughter or son. In addition, there is a suggestion that this differential aspiration in favor of non-Whites increases from 1972 to 1980. In 1972 mothers tended to have significantly higher educational aspirations for their sons than for their daughters as indicated by the sign and relative size of the path coefficients. However, in 1980 the mothers' aspirations were essentially independent of the child's sex. The asterisk indicates that the 1972 and 1980 path coefficients are at least four standard errors apart for the sex group effect on mothers' aspirations. The relationship of SES to mothers' plans remained high but relatively stable from 1972 to 1980.

### B. LETERMINANTS OF PRIVATE SCHOOL ATTENDANCE

The hypothesized model was unable to predict with any accurancy whether or not the student attended a private school. That is the multiple R was .20 in HS&B and .18 in NLS. The inability to predict attendance at a private school was probably due to the fact that private schools were a relatively heterogeneous lot, being a mixture of both private non-Catholic and Catholic schools.

### C. CHOOSING THE ACADEMIC CURRICULUM

Choosing the academic curriculum was predicted reasonably well, R = .47 and .52 in HS&B and NLS cohorts, respectively. Explanatory variables that were positively related to selecting the academic curriculum were attendance at a private school, mothers' educational plans for the offspring, family SES and living in the Northeast region of the country. There is some indication that being in a private school had a stronger relationship with being in the academic curriculum in 1980 than it did in 1972. This is probably not the whole story, however, since the earlier analysis of population shifts suggest that there were proportionally bigger shifts of students from academic to vocational and general curricula in the public schools than in the private schools



during this period. It is interesting to note that sex group membership was independent of choice of the academic curriculum in both 1972 and 1980. This is indicated by the lack of arrows from sex group membership to curriculum choice.

### D. NUMBER OF SEMESTERS OF LANGUAGE COURSES

This process variable was chosen for inclusion in the model because of its potential relationship with achievement in both verbal skills-vocabulary and reading. In addition, because of its potential critical nature in the development of the verbal skills, it might help to explain the greater vocabulary score decline that was found for females than for males. The multiple correlation between the hypothesized explanatory variables and semesters of language courses was .57 in 1972 and .48 in 1980. The path coefficients defining female vs. male contrasts seem to support the notion that females were less likely in 1980 than in 1972 to be taking language courses. The female-male contrast path coefficient indicates that on the average, females still take more semesters of language courses than do males, but the difference appears to be diminishing. This finding may reflect a new awareness among female students and a "turning away" from traditional female courses.

There is an indication that SES's relationship with number of language courses is somewhat more attenuated in 1980 than in 1972. Similarly, students from the Northeast region showed a considerable drop in their relative advantage over the other regions in their study of language. Significant drops between 1972 and 1980 in the relationship between being in the academic curriculum and taking language courses also occurred. Mothers' educational plans also had a smaller albeit still positive effect on the taking of language courses.

### E. AMOUNT OF HOMEWORK

The amount of homework done reported by the student was only modestly explained by the model. The multiple correlation was .36 and .32 in the HS&B and NLS cohorts, respectively. With the exception of the female-male contrast, the remaining explanatory variables had stable, although relatively weak relationships with the amount of homework they reported doing. Consistent with the pattern of some of the earlier reported female-male contrasts, the females advantage over males with respect to the amount of homework done decreased significantly from 1972 to 1980. It would appear that the greater observed score decline among women as compared to men is consistent with the school behavior patterns highlighted in the path models.

### F. SEMESTERS OF MATHEMATICS COURSES

The more relevant academic process variable for mathematics achievement is number of semesters of mathematics courses. Thus, in Figure 9-3 where mathematics achievement is the primary outcome, we have replaced number of language courses with number of mathematics courses.

ERIC Fuil Eaxt Provided by ERIC

The non-White students, compared to Whites, increased the differential in their favor with respect to number of semesters of mathematics when going from 1972 to 1980. Although women are still taking significantly less math courses than men, they showed a slight decrease in this differential when going from 1972 to 1980. All the other explanatory variables had relatively stable relationships with this outcome when going from 1972 to 1980.

### G. DETERMINANTS OF TESTED ACHIEVEMENT

Because the contrasts between what are the important determinants of achievement in each test are of considerable interest, all three achievement outcomes and their determinants will be discussed together. Table 9-1 summarizes the direct effects of major explanatory variables on test outcomes.

Inspection of the determinants of vocabulary, reading, and mathematics achievement (Figure 9-1) shows a significant reduction in the gap between White and non-White achievement, while controlling for all other variables in the model as one goes from 1972 to 1980. This is consistent with other recent findings (Jones, 1984). It is also consistent with the direction of changes in other educational process variables. While the 197? to 1980 measured changes in the educational behavior pattern of the non-Whites are not statistically significant by the four standard error criteria, their pattern taken as a whole is certainly consistent with the finding that the White/non-White achievement gap is narrowing. For example, the differential in favor of Whites decreased with respect to attendance at private schools (differences in favor of whites went from 4 percent in 1972 to 1 percent in 1980) when controlling for other confounding variables. The path coefficient indicates that the gap in favor of non-Whites with respect to mothers' educational plans increased in 1980. Conversely, the gap in favor of Whites with respect to the selection of academic curriculum also was reduced by 1980. The nonwhites also increased the differential in their favor with respect to the number of language courses taken as one goes from 1972 to 1980. This increase in differential is even more notable in the case of self-report of the number of mathematics courses (see Figure 9-3). There was, however, no difference in the amount of homework reported by Whites and non-Whites. One should keep in mind that these changes in a positive direction, both in intermediate and final outcomes, are (path coefficients) in a model where other demographic variables are being held constant. That is, when we hold constant critical variables such as SES we notice that non-Whites are more likely than Whites to change their educational behavior in a direction to resist score decline.

An important determinant of achievement in all three areas was being in the academic curriculum. That is, when controlling for all other variables in the model being in the academic curriculum leads to the greatest increment (with the exception of race) in one's tested achievement. In addition, the impact of being in the academic curriculum increases as one goes from 1972 to 1980. Membership in the academic curriculum not only has both this large direct effect on achievement, but



# -99-

### Table 9-1

1972-80 Contra	asts of the	Direct E	ffects of	Major	Explanatory
	Variabl	es on Tes	t Outcome	8	

	Vocab <u>1972</u>	ul <b>ary</b> <u>1980</u>	Rea <u>1972</u>	ding <u>1980</u>	<b>Mat</b> he <u>1972</u>	matics <u>1980</u>
Race/Ethnicity (White = "1" Others = "0")	2.74	(2.32)*	3.71	(3.41)*	5.75	(4.94)*
Curriculum (Academic = "1" Others = "0")	1.31	(1.49)*	1.63	(2.14)*	3.07	(3.54)*
Mother's Educational Plans for Child	.91	(.63)*	1.33	(.92)*	1.80	(1.15)*
SES	.80	(.70)	.59	(.71)*	.96	(1.22)*
Sex (Female = "1" Male = "0")	.44	(20)*	.42	(.04)*	41	(94)*
Number of Related Courses	.34	(.28)	.43	(.30)	1.05	(.79)*
Amount of Homework ("1" = 5 or more hours per week, "0" = less than 5 hours per week)	.16	(.47)*	.20	(.68)*	.80	(1.42)*

\*Indicates that the 1972 to 1980 comparison of direct effects differ by four pooled standard errors.



also has an indirect effect on achievement in that those students in the academic curriculum take more critical subject matter courses (math and languages) and do more homework both of which, in turn, have significant impacts on all three tested achievement areas.

Achievement mean score differences between males and females also changed between 1972 and 1980. Accompanying these female-male changes in achievement were some sex group-related changes in their schoolrelated behavior. That is, while females showed greater decline than did males in both the vocabulary and reading achievement areas, they also showed a greater decline than males in the amount of homework that they reported doing. Females showed a proportionately greater decline than males in the number of language courses taken. Curiously enough, the gap between males and females is somewhat reduced with respect to the number of mathematics courses taken as one goes from 1972 to 1980. This may reflect a general trend in that females appear to be moving from the historically more female-oriented courses (e.g., literature and languages) to the more male-oriented science and mathematics courses.

The male/female gap in mathematics achievement grew larger in 1980, however, even after controlling for the number of courses taken. While females report taking more courses, the data úo not allow one to determine the level and/ or quality of these courses. It may be that females are either enrolling in the lower level mathematics courses and/or are taking the more occupationally-related math courses. It also may be partly artifactual in that coursework was based on student responses in 1980 and the Survey Administrator's Report in 1972. As indicated earlier, Fetters et al. (1984) has shown that students tend to over-report their coursework.

The fact that the path coefficient from number of semesters of mathematics courses to mathematics achievement shows a significantly smaller effect in 1980 suggests that while students may be taking more math courses in 1980 than in 1972, they may be indeed nonacademically-oriented math courses and therefore the finding of less relationship with tested math achievement.

While the average amount of homework done decreased from 1972 to 1980, its impact on achievement increased. It would seem that those seniors in 1980 who were willing to put in the effort were more than repaid for their effort.

The question as to whether the educational system was more or less equitable in 1980 compared to 1972 is somewhwat of a "mixed bag." That is, while the White/non-White gap decreased one's family SES level had a bigger effect on tested achievement in both reading and mathematics achievement in 1980 compared to 1972. Also, other things being equal, the gap in tested achievement between students in the academic curriculum and those in the other curricula increased from



1972 to 1980 in all three basic skills areas. This increase in the gap is particularly noteworthy in mathematics and reading which, in turn, are more likely to be sensitive to formal educational training. Are those individuals in vocational and general programs in 1980 receiving either less or inferior coursework in the basic skills areas than they did in 1972? If so, then the 1972-1980 population shift from academic to vocational and general curricula has very serious consequences with respect to allowing young adults to achieve their full potential in reading and computation. Another possibility is that the marginal ability students have a greater likelihood of being in the general and vocational programs in 1980 than in 1972.

One other curious result in the path analysis is that mothers' educational plans for the child are less related in 1980 than in 1972 to the normal consequences, e.g., being in the academic curriculum, number of language courses, and tested achievement in all three basic skills areas. It may be that mothers in 1980 are less likely to view the traditional academic "paths" as necessary for continuing on in higher education. Part of this may reflect the fact that post-high school educational institutions, in particular junior colleges, provide an increasingly greater array of non-academic curriculum choices. Similarly, the course requirements of many four-year colleges may have "softened." It also may reflect the general economic climate and consequently the view that the college graduate who isn't trained in specific job relevant skills may be in less demand.



### -102-

### CHAPTER X

### SUMMARY AND POLICY IMPLICATIONS

In 1983, eight major national studies reported on the status of public education in the United States. These reports sounded a common theme: The American educational system is in trouble. The major evidence cited in support of this claim was that academic achievement, as measured by performance on the College Board's Scholastic Aptitude Tests and the National Assessment of Educational Progress, had declined. This situation was attributed to demographic changes, lower standards, lower expectations for students, a less rigorous curriculum, and the poor academic preparation of new teachers.

This study documented a similar decline in the academic achievement of high school seniors between 1972 and 1980. It found, however, that the major factor contributing to changes in the vocabulary, reading and mathematics test scores of these students was a decreased academic emphasis in the educational process. Shifts in population demographics from 1972-1980 played a minor role in explaining .est score decline.

### A. SUMMARY OF STUDY FINDINGS

The study findings show that there were significant changes in test scores, in high schools, and in student behavior. They also show that these changes were interrelated.

### 1. Test Score Change

There were declines on all three achievement tests between 1972 and 1980. The largest declines occurred in vocabulary and reading. The average senior in 1980 (a student at the 50th percentile in 1980 in vocabulary and reading achievement) would rank at about the 41st percentile among the 1972 seniors in both vocabulary and reading. Similarly, a 1980 senior with average mathematics achievement in 1980 would be at the 45th percentile when computed with the 1972 seniors. When these changes are measured in standard deviation units, the declines are .22 for Vocabulary, .21 for Reading, and .14 for Mathematics, indicating a greater decline in verbal than in quantitative skills.

a. <u>Yocabulary</u>. Females showed a greater decline than males on the Vocabulary test. Whites showed greater declines than Blacks and Mexican-Americans, but these comparisons may be confounded by test score floor effects. The decline fcr Whites was relatively pervasive, cutting across SES levels, geographic regions, curriculum type, and school type (public vs. Catholic).

b. <u>Reading</u>. The decline is Reading test scores tended to be somewhat more consistent across subpopulations than the Vocabulary test scores. Declines were relatively consistent across sex, SES, curriculum

type, and school type, but Whites showed a greater decline than Blacks or Mexican-Americans. The Reading test score declines were found primarily among students who reported doing less than five hours of homework per week.

c. <u>Mathematics</u>. The decline in Mathematics test scores was slightly larger for males than for females and larger for Whites than for other racial/ethnic groups. Blacks showed a small, but not statistically significant, increase in Mathematics test scores. The Mathematics test score declines were found primarily among students who reported doing less than five hours of homework per week. Females and Blacks who reported doing more than five hours of homework per week st wed significant increases. The largest score decline was among male students who took four or fewer semesters of mathematics and/or science.

d. <u>Comparison with SAT Score Decline</u>. SAT scores for men and women declined in a somewhat similar pattern in the same time period. As shown in Table 10-1 below, men had a slight lead in SAT verbal scores (454 vs. 452) and a large lead in SAT math scores (505 vs. 461) in 1972. Between 1972 and 1980, the women declined somewhat more than the men on both scales, increasing the discrepancy between men and women.

### Table 10-1

		Verbal				Mathematics			
		<u>1972</u>	<u>1980</u>	<u>Diff</u> .	Change in S.D. Units	<u>1972</u>	<u>1980</u>	<u>Diff</u> .	Change in S.D. <u>Units</u>
Male		454	428	-26	23	505	491	-14	12
Femal	e	452 <sup>°</sup>	420	-32	29	461	443	-18	17
2. C	hanges	<u>in</u> th	e Chara	cterist	ics of High	n School	Seniors	and	Their Schools

### SAT Test Score Changes, 1972-1980

There were significant changes from 1972-1980 in the characteristics of high school seniors, their homes and families, the schools they attended, and their attitudes and behaviors.

- o The 1980 seniors were more likely to be members of a minority group and from the South than were the 1972 seniors. Females constituted a slightly larger proportion of the 1980 seniors than the 1972 seniors.
- o There was a significant increase in the mean level of purental education from 1972 to 1980, but there was relatively little difference in fathers' occupations in 1972 and 1980.



- Mothers of 1980 senior women had higher educational aspirations for their daughters than did the mothers of 1972 senior women. Aspirations for senior men remained unchanged.
- o There were fewer study aids in the homes of 1980 seniors than in the homes of 1972 seniors.
- o More seniors were enrolled in the general or vocational curriculum in 1980 than in 1972, while fewer students were enrolled in the academic curriculum. In addition, seniors took fewer semesters of social studies, science and foreign languages in 1980 than in 1972, but more semesters of mathematics.
- o Seniors reported doing less homework in 1980 than in 1972. The estimated decline was from approximately 4.55 hours of homework per week in 1972 to 4.05 hours in 1980.
- The attitudes and values of high school seniors also changed between 1972 and 1980. Interest in correcting social and economic inequities declined, while interest in making money and in job success increased. Students became more self-confident between 1972 ary 1980 but less sure of their ability to control the course of their own lives.
- There were also increases, between 1972 and 1980, in the number of schools with high dropout rates and in the proportion of schools reporting a majority of their students in the general curriculum. Students' ratings of the quality of their schools' facilities, academic instruction and reputation dropped.

### 3. <u>Factors Affecting Test Score Change</u>

A "step down" analysis of covariance was used to estimate how 1972-1980 changes in population demographics, student behavior, school characteristics and home educational support systems separately affected the average score decline. Path analysis was employed to ascertain if certain subpopulations, which were characterized by larger declines, might have experienced different educational processes in 1972 than their counterparts in 1980.

The results show that:

- o Shifts in population demographics from 1972 to 1980 were a minor factor contributing to the score decline.
- Changes from 1972 to 1980 in student school behaviors and in school characteristics played the major roles in the score decline. This finding was consistent across all three tested achievement areas.


- o Student's school behaviors and school policies contributed equally to the vocabulary score decline.
- o School characteristics played a somethat larger role than did students' school behaviors in the reading score decline.
- o Conversely, students' school behaviors played a slightly greater role than did school characteristics in the mathematics score decline.
- Changes from 1972 to 1980 in the home educational support system were in a direction that would resist score decline. However, the magnitude of the effects of changes in home educational support were small compared to student and school characteristics.

Changes from 1972 to 1980 at both the school level and student level that seem to have contributed most to the decline were: (1) a greater likelihood of being in the general or vocational curriculum rather than the academic curriculum, (2) a drop in the frequency with which students report taking "traditional" college preparation core courses such as foreign languages, science and/or courses requiring laboratory work, (3) a decrease in the amount of homework done, and (4) an increasing dissatisfaction among the students with the lack of emphasis on academics in the schools.

#### **B. POLICY IMPLICATIONS**

The national reports issued last year contained a wide range of recommendations designed to strengthen the educational process. They addressed school curriculum, programs for special populations, college entrance requirements, performance standards for students, training of teachers, administrative leadership, fiscal support, and the proper role of local, state and federal governments. Many states have responded to these recommendations by raising high school graduation requirements and/or college admission standards, requiring more testing of students, and changing policies on the preparation and licensing of teachers. The findings of this study support the appropriateness of many of these policy recommendations.

#### 1. Curriculum

The National Commission on Excellence in Education recommended that students devote significantly more time to the "New Basics"--English, mathematics, science, social studies, computer sciences and, for the college-bound, a foreign language. This study suggests that more coursework in science, foreign languages and mathematics contributes to higher scores on the kinds of vocabulary, reading and mathematics skills measured by the NLS/HS&B test battery. However, policymakers must be sensitive to differences in course-taking behavior among curricula. The course-taking patterns of students in the academic curriculum showed little change between 1972 and 1980. But a significant number of students shifted into



the general and vocational curricula, where they took fewer New Basics than their academic counterparts. Therefore, we suggest that schools increase the required number of courses in the New Basics for non-academic students. However, these courses should not be provided at the expense of remedial work in reading and mathematics when it is needed. In addition, schools should provide more frequent opportunities for students to write and more opportunities for students to participate in laboratory courses.

#### 2. Course Content

Several commissions recommended that the content of textbooks and instructional materials be made more academically demanding. Since students in all curricular areas in this study were critical of the lack of academic emphasis in their coursework and of the quality of their academic instruction, we suggest that course content and instructional methods be reviewed and upgraded, as necessary, to insure more rigorous content.

#### 3. Homework

Several reports recommended that homework be required regularly of all students, and many school districts have implemented homework policies. Students in this study reported doing less homework in 1980 than in 1972, but students in the general and vocational curricula did less homework and showed greater test score declines than those in the academic program. Since there was a strong positive relationship between homework and achievement in all three tested areas, we concur that appropriate amounts of homework be assigned in all courses.

#### 4. Programs for Special Populations

Many of the national studies stated that the Federal government, in cooperation with the states and local school districts, should continue to provide special programs for educationally and economically disadvantaged students, the handicapped, and language minority students. This recommendation is critical since a growing body of data have documented the relative improvement in achievement of Blacks and other disadvantaged students over the last decade. For example, test score declines for Blacks in this study were considerably below those of White students. This was particularly true of low SES Black students, the most likely beneficiaries of federal and state compensatory education programs. Similar findings were reported by the National Assessment of Educational Progress in an analysis of changes in the reading and mathematics achievement of 17-year-old students during the 1970s.



#### References

- Austin, G. R., & Garber, H. (Eds.) The rise and fall of national test scores. New York: Academic Press, 1982.
- Boyer, E. L. <u>High school: A report on American secondary education</u>. New York: Harper and Row, 1983.
- Carnegie Corporation (James B. Hunt, Jr. & David Hamburg, Chairmen). Education and economic progress: Toward a national education policy. New York: Carnegie Corporation, 1983.
- Cohen, J. <u>Statistical power analysis for the behavioral sciences</u>. New York: Academic Press, 1969.
- The College Board. <u>Academic preparation for college: What students need to</u> know and be able to do. New York: The College Board, May 1983.
- Das Gupta, A. A general method of decomposing a difference between two rates into several components. <u>Demography</u>, 15, 1978, 99-112.
- Donlon, T. F., Hilton, T. L. & Schroeder, W. B. <u>Designing a Test Plan for the</u> <u>1980 National Longitudinal Study "High School and Beyond</u>." Princeton, NJ: Educational Testing Service, October 1978.
- Fetters, W. D., Stowe, P. S., & Owings, J. A. Quality of responses of high school students to questionnaire items. Washington, D.C.: National Center for Education Statistics Report, September 1984.
- Frankel, M. R., Kohnke, L., Buonanno, D., & Tourangeou, R. <u>Sample Design</u> <u>Report</u>. Report to the National Center for Education Statistics by the National Opinion Research Center, 1981.
- Glass, G. V., & Smith, M. L. Meta analysis of research on the relationship of class size and achievement. San Francisco, CA: Far West Laboratories, 1978.
- Haertel, E. H., & Wiley, D. E. <u>Rebalancing the Test Battery for "High School</u> <u>and Beyond": Analyses and Recommendations</u>. Paper prepared for the joint meeting of the National Planning Committee and the Test Advisory Panel for "High School and Beyond," December 1978. Chicago, IL: CEMREO, Inc., ML-Group for Policy Studies in Education.
- Hilton, T. L., & Rhett, H. <u>The base-year survey of the national longitudinal</u> <u>study of the high school class of 1972</u>. (Final report to the U.S. Department of Health, Education, and Welfare, National Center for Education Statistics, Contract No. OEC-0-72-0903). Princeton, NJ: Educational Testing Service, 1973.



- Jencks, C. S. What's behind the drop in test scores? <u>Working Papers</u>, <u>6</u>(4), 1978, 29-41.
- Jones, L. White-Black achievement differences: The narrowing gap. <u>American</u> <u>Psychologist</u>, <u>39</u>, 1984, 1207-1213.
- National Center for Education Statistics. <u>The Condition of Education</u>, 1984. Washington, D.C.: National Center for Education Statistics, 1984.
- National Commission on Excellence in Education (David P. Gardner, Chair). A <u>nation at risk: the imperative for educational reform</u>. Washington, D.C.: U.S. Department of Education, 1983.
- National Task Force on Education for Economic Growth (James B. Hunt, Jr., Chair). <u>Education for a high technology economy</u>. Denver, CO: Education Commission of the State, 1983. Draft Document.
- Peng, S. S., Fetters, W. B., & Kolstad, A. J. <u>A capsule description of high</u> <u>school students</u>. Washington, D.C.: National Center for Education Statistics, 1981.
- Riccobono, J., Henderson, L. B., Burkheimer, G. J., Place, C., & Levinsohn, J. R. <u>National longitudinal study: Base year (1972) through fourth</u> <u>follow-up (1979) data file users manual</u>. Report to the National Center for Education Statistics by the Research Triangle Institute, 1981.
- Rock, D. A., et al. <u>Psychometric Analysis</u>. Unpublished report under review by the National Center for Education Statistics, 1985.
- Sizer, T. H. <u>Horace's compromise: The dilemma of the American high school</u> today. Boston, MA: Houghton-Mifflin, 1984.
- Twentieth Century Fund. <u>Report of the Task Force on Federal Elementary and</u> <u>Secondary Education Policy</u>. New York: Twentieth Century Fund, 1983.
- Wirtz, W. <u>The rise and fall of national test scores</u>. (Austin, G. R., & Garber, H., Eds.), 1982.



### APPENDIX A

Items Common to the 1972 and 1980 Student Questionnaires



APPENDIA A

### Table 1

ITEMS COMMON TO THE 1972 AND 1980 STUDENT QUESTIONNAIRES

VAR	LABEL	DESCRIPTION	RESPONSE CODES (CONVERTED)		NL S	HS8 00
2	HSGRADES	GRADES IN HIGH SCHOOL	8=MOSTLY A'S; 7=A/B; 6=MOST B; 5=B/C; 4=MOST C; 3=C/D; 2=D; 1=BELOW 0; MISSING=BLANK	229	B05	85 88 <b>00</b> 7
3	HOMEWORK	TIME/WEEK SPENT ON HOMEWORK	1=0-5 HRS.; 2=5-10 HRS; 3=10+ HRS; MISSING=BLANK	235	BQ 7	128 88015
		HAVE YOU PARTICIPATED IN THE F	OLLOWING ACTIVITIES			
٩	PART-ATH	PARTICIPATE IN ATHLETICS	0=NO: 1=YES: ACTIVE OR LEADER	241	6010A	143 E8032A 144 880328
5	PART-CHR	PARTICIPATE IN CHEERLEADING	O-NO: 1-YES: ACTIVE OR LEADER	242	99108	145 886320
•	PAR T-DE B	PARTICIPATE IN DEBATING, MUSIC	0=NO: 1=YES: ACTIVE /R LEADER	243	891 <b>0C</b>	166 88 <b>6329</b> 167 800328 168 88032F
7	PART-HOS	PARTICIPATE IN HOBSY CLUBS	0-ND: 1-YES: ACTIVE OR LEADER	244 (	9100	167 888328
•	PAR T-HON	PARTICIPATE IN HONORARY CLUBS	O-NO; 1-YES: ACTIVE OR LEADER	245	9010E	170 E8632W
•	PART-NWS	PARTICIPATE IN NEWSPAPER, YEARBOOK	0-NO: 1-YES: ACTIVE OR LEADER	246	9010 <b>F</b>	171 200321
10	PART-SBJ	PARTICIPATE IN SUBJECT MATTER CLUBS	0-NOI 1-YES: ACTIVE GR LEADER	247	89106	172 89632J
11	PART-GOV	PARTICIPATE IN STUDENT GOVERNMENT	O-ND: 1-YES: ACTIVE OR LEADER	248	801 <b>0</b> H	173 8 <b>903</b> 28
12	PART-VOC	PARTICIPATE IN VOC ED CLUBS	G-NO: 11005: ACTIVE OR LEADER	249 (	60101	174 88032L
		HOW NUCH HAS EACH OF THE FOLLOWING IN	FLUENCED YOUR PLANS FOR AFTER H.S.			
13	INF-PRNT	INFLUENCE ON PLANS-PARENTS	1-NO: 2-SOMENHAT: 3-GREAT DEAL	261 (	9914A	233 E8049A 234 E80498
15	INF-COUN	INFLUENCE ON PLANS-GUIDANCE COUNSELOR	1-NO; 2-SOMEWHAT; 3-GREAT DEAL	263 (	014C	235 E8049C
36	INF-TCHR	INFLUENCE ON PLANS-TEACHER	1-NO: 2-SOMEWHAT: 3-GREAT DEAL	264 (	99140	236 E80490
21	INF-FRNO	INFLUENCE ON PLANS-FRIENOS & RELATIVES	1-NO: 2-SOMEWHAT: 3-GREAT DEAL	269 ( K(14) 262 ( 2	0148 0148 228	237 E80498

A-1



ERIC

### STUDENT RATINGS OF SCHOOL

23	BUILDING	STU/SCHUOL RATING: CONDITION OF BUILDING	1=POOR;2=FAIR;3=GOOD;4=EXC (DK=BLANK)	291 BQ18A	283 88053A
27	LIBRARY	STU/SCHOOL RATING: LIBRARY	1=POOR;2=FAIR;3=GODD;4=EXC (DK=BLANK)	295 8018E	284 880538
28	AC INSTR	STU/SCHOOL RATING: ACADEMIC INSTRUCTION	1=POOR : 2=FA IR : 3=GOOD : 4=E XC ( DK=BLANK )	296 801 <i>8</i> F	285 860530
30	REPUTATN	STU/SCHOOL RATING: REPUTATION IN COMMUNITY	1=P00R;2=FAIR;3=G000;4=EXC (DK=BLANK)	298 8018H	286 880530
32	ICHR INT	STU/SCH RATING: TEACHER INTEREST IN STUDENTS	1=POOR:2=FAIR:3=GOOD:4=EXC (DK=BLANK)	300 BQ18J	287 8805¥
		HON IMPORTANT IS EAC	N OF THE FOLLOWING TO YOU IN YOUR LIFE		
33	INP-WK S	IMPORTANCE OF SUCCESS IN WORK	1-NOT IMP: 2-SOMEWHAT IMP: 3-VERY IMP	313 8020A	294 BB057A
34	INP-HARR	INFORTANCE OF MARRIAGE, FAMILY LIFE	1-NOT IMP: 2-SOMEWHAT IMP; 3-VERY IMP	314 80208	295 880578
35	INP-HONY	INFORTANCE OF NONEY	1=NOT IMP: 2=SOMEWHAT IMP: 3=VERY IMP	315 BQ20C	296 880570
36	INP-FRND	INPORTANCE OF STRONG FRIENOSHIPS	1-NOT IMP; 2-SOMEWHAT IMP; 3-VERY IMP	316 80200	297 880570
37	INP-STWK	INPORTANCE OF STEADY NORK	1-NOT IMP: 2=SOMEWHAT IMP: 3-VERY IMP	317 8020E	298 880578
36	INP-LEAD	INFORTANCE OF SEING COMMUNITY LEADER	1-NOT IMP: 2-SOMEWHAT IMP: 3-VERY IMP	318 8020F	299 88057F
39		INPORTANCE OF OPPORTUNITIES FOR MY CHILDREN	I-NOT INP: 2-SONEWHAT INP: 3-VERY INP	319 80206	300 880576
40	INP-CLOS	INPORTANCE OF LIVING CLOSE TO RELATIVES	1-NOT IMP: 2-SOMEWHAT IMP: 3-VERY IMP	320 8920H	301 88057N
41	INP-AHAY	INPORTANCE OF GETTING AWAY FROM THIS AREA	1-NOT INP: 2-SONEWHAT INP: 3-VERY INP	321 89201	392 B80571
42	INP-SOCL	INPORTANCE OF WORK TO CORRECT SOCIAL PROBLEMS	1-NOT IMP: 2-SOMEWHAT IMP: 3-VERY IMP	322 BQ20J	303 88057J
	F	EELINGS ABOUT SELF: SCALE REVERSED TO MAKE HIG	HER NUMBER CORRESPOND TO POSITIVE FEELIN	6	
43	POS ATT	FEELINGS ABOUT SELF: POSITIVE ATTITUDE	REVERSED SCALE: 1=DISAGREE STRONGLY: 2=DISAGREE: 3=AGREE: 4=AGREE STRONGLY	323 8021A	306 BB058A
44	GOODLUCK	FEELINGS ABOUT SELF: LUCK MOST INPORTANT	ORIGINAL SCALE: 1=AGREE STRONGLY: 2=AGREE:3=DISAGREE:4=DISAGREE STRONGLY	324 89218	307 880588
45	WORTH	FEELINGS ABOUT SELF: WORTH EQUAL TO OTHERS	REVERSED SCALE: 1-DISAG. 4-AGREE STR	325 8921C	308 88058C
44	DO NELL	FEELINGS ABOUT SELF: CAN DO THINGS WELL	REVERSED SCALE: 1-0ISAG. 4-AGREE STR	326 80210	309 880580
47	GETAHEAD	FEELINGS ABOUT SELF : CAN'T GET AHEAD	ORIGINAL SCALE: 1=AGREE: 4=DISAGR STR	327 8021F	310 BR058F

4-D19404 314 361 DATTE 31A DOA34E PLANNING FEELINGS ABOUT SELF: PLANS DON'T WORK OUT 48 DRIGINAL SCALE: 1-AGREE: 4-DISAGR STR 320 BO21F 311 88058F 229 40 ACCEPT FEELINGS ABOUT SELF: ACCEPT COND. IN LIFE ORIGINAL SCALE: 1=AGREE: 4=DISAGR STR 329 BQ21G 312 88050G SATISF FEELINGS ABOUT SELF: BATISFIED WITH MYSELF REVERSED SCALE: 1-DISAG. 4-AGREE STR 330 8021H 313 88058H BEST COPY AVAILABLE

50

## A-2

10=PROPRIETOR: 11=PROTECTIVE: 12=SALES:

13-SERVICE: 14-TECHNICAL

49	WK PLAN WORK STUDENT WANTS TO DO	1=CLERICAL: 2=CRAFTSNAN: 3=FARMER: 367 BQ25A 345 BB062 4=HONENAKER: S=LABORER: 6=MANAGER:
	( PLEASE NOTE: SEE 119-121 FOR OCCUPATIONS	7=MILITARY; 8=OPERATIVE; 9=PROFESSIONAL;
	RECODED TO DUNCAN SEI SCALE +}	10=PROPRIETOR: 11=PROTECTIVE: 12=SALES:
		13-SERVICE: 14-TECHNICAL
71	FATH OCC FATHER'S OCCUPATION	1=CLERICAL; 2=CRAFTSMAN; 3=FARMER; 368 89258 208 88038
		7-NULLTART, S-CABURER, G-ARNER,
		1/magropaleTrp: 11-ppntective, 12-sales,
		13-FERVICE, 14-FRUIELIVE; 12-JALEJ;
		13-3EKVILET 14-TECHNICAL
73	NOTH OCC NOTHER'S OCCUPATION	1=CLERICAL; 2=CRAFTSMAN; 3=FARMER; 369 B025C 211 BB041
		4=HONE MAKER: S=LABORER: 4=MANAGER:
		7=MILITARY: 0=OPERATIVE: 9=PROFESSIONAL:

#### INPORTANCE OF FACTORS IN DETERMINING PLANS FOR LIFE WORK

75	I-PREVWK	INPORTANCE TO CAREER PLANS: PREVIGUS WORK	1=NOT IMPORTANT: 2=SOMEWHAT: 3=VERY	370 BQ26A	346 EB063A
74	I-INCOME	INPORTANCE TO CAREER PLANS: GOOD INCOME	1-NOT INPORTANT; 2-SOMEWHAT; 3-VERY	374 8Q26E	347 EB0638
71	I-SECURT	INPORTANCE TO CAREER PLANS: JOB SECURITY	1-NOT INPORTANT: 2-SOMEWHAT: 3-VERY	375 8Q26F	348 E8063C
78	I-INTWRK	INPORTANCE TO CAREER PLANS: INTERESTING WORK	1-NOT INPORTANT: 2-SOMEWHAT: 3-VERY	376 BQ26G	349 E80630
79	I-DEC I SN	INPORTANCE TO CAREER PLANS: FREE DECISIONS	1-NOT INPORTANT: 2-SOMEWHAT: 3-VERY	377 BQ26H	350 E8063E
80	I-PEOPLE	INPORTANCE TO CAREER PLANS: FRIENDLY PEOPLE	1-NOT INPORTANT; 2-SOMEWHAT; 3-VERY	379 BQ26J	351 E8063F
81	ABILITY	DO YOU HAVE ABILITY TO COMPLETE COLLEGE	REVERSED SCALE: 1=DEF NOT: 2=DOUBT IT: 3=Not sure: 4=Probably: 5=DEFINITELY	381 8928	362 88069
82	ED PLAN	HIGHEST LEVEL OF EDUCATION PLANNED	1=LT HS: 2=GRAD HS: 3=VDC/JR COLL.; 4=COLLEGE: 5=GRAD/PROF	383 BQ298	353 88065
83	IYR PLAN	PLANS FOR FIRST YEAR AFTER HIGH SCHOOL	1=WORK FT: 2=APPRENT: 3=HILITARY: 4=HOMENAKER: 5=VOCTECH: 6=JR COL ACAD; 7=JR COL VOCTECH: 8=4YR COLLEGE: 9=WORK PT: 10=OTHER:11=JR/4YR COL	385 8Q31	364 88071

232

A-3

231

99	FATH ED	FATHER'S HIGHEST LEVEL OF EDUCATION	1=LESS THAN HS;2=HS/EQUIV. ADULT ED; 3=BUSINESS/TRADE OR SOME COL.;4=CCLLEGE; 5=GRAD OR SOME GRAD SCHOOL; BLK=NISSING	396 ;	BQ9DA	209	88039
100	NOTH ED	MOTHER'S HIGHEST LEVEL OF EDUCATION	1-LESS THAM > 2-HS/EQUIV ADULT ED.; 3-BUSINES (C	397	80908	212	88042
101	N ED PLN	SCHOOLING MOTHER (PARENT) WANT FOR STUDENT	1=QUIT HS: 2=HS ONLY; 3=VOC./JR. COLL; 4=COLLEGE: 5=GRAD/PROF	400	80918	354	BB 069
103	PL STUDY	HAVE IN HOME: PLACE TO STUDY	1-HAVE: 0-DO NOT HAVE	403	8094A	435	88104A
104	NEWSPAPR	HAVE IN HOME: DAILY NEWSPAPER	1-HAVE: O-DO NOT HAVE	404	BQ 948	436	881048
105	REF BKS	HAVE IN HOME: ENCYCLOPEDIA/REFERENCE BOOKS	1-HAVE: 0-DO NOT HAVE	406	80940	437	88104C
106	TYPEWRTR	HAVE IN HOME: TYPEWRITER	1-HAVE: D DO NOT HAVE	411	80941	438	881040
107	STUDYAID	NUMBER OF STUDY AIDS AVAILABLE IN HOME ( Counts of Original 103 - 106)	0 - 4: BLK(ONLY IF 103-106 ARE BLK)				

#### COMMENTS AND CRITICISM ON SCHOOL PROGRAMS

33

- # -

				est copy available				
33	116	STUDY (	DID POOR STUDY MABITS INTERFERE WITH ED. Reversed Scale)	1=A GREAT DEAL:2=SOMEWHAT: 3=NOT AT All: MISSING=BLANK	288	8917	201 680520	
0.0	115	TEACHER	DID POOR TEACHERS INTERFERE WITH EDUC. Reversed scale)	1=A GREAT DEAL;2=SOMEWHAT; 3=NOT AT All; MISSING=BLANK	285	89171	260 E8052C	234
	414	TZULDA	OID SCHOOL ADJUSTMENT INTERFERE Reversed scale)	1=A GREAT DEAL;2=SOMEWHAT; 3=NOT AT All; MISSING=BLANK	284	8Q1 7H	279 E80528	
	113	TOO HARD	DID HARD COURSES INTERFERE WITH EDUC. Reversed scale?	1=A GREAT DEAL:2=SONEWHAT: 3=NOT AT All: MISSING=BLANK	277	8017A	278 E8052.3	
	111	WAK COU	SCHOOL PROVIDED MORE NORK COUNSELING	- J=DJSAGREE-STR-J-2=DJSAGREEJ-3=AGREE 4=AGREE STRONGLY; BLANK=NOT APPLY/ MISSING	. <b>_309</b> .	80191 -	<b>.185</b> . E8035E	
	111	ED-COUN	SCHOOL PRUVIDED EDUCATIONAL COUNSL. Reversed scale)	1=DISAGREE STR.; 2=DISAGREE; 3=AGREE 4=AGREE STRONGLY; BLANK=NOT APPLY/ MISSING	306	8019F	184 EB0350	
	1 10	VOC-ED	SCHOOL SHOULD HAVE MORE VO-TECH COURSES	1=AGREE STRONGLY; 2=AGREE; 3=DISAGREE 4=DISAGREE STRONGLY; BLANK=NDT APPLY/ HISSING	304	80190	182 E80358	
	109	PRACHOR	SCHOOL OIDN'T HAVE ENDIGH PRACTICAL WORK EXP	1=AGREE STRONGLY; 2=AGREE; 3=DISAGREE 4=DISAGREE STRONGLY; BLANK=NOT APPLY/ M1SS1NG	305	8Q19E	183 E8035C	4

## BEST COPY AVAILABLE

**A-**4

#### MISC. VARIABLES

117 NOVEJOB (FORMERLY 808)	WILLING TO MOVE TO GET JOB YOU WANT	REVERSED SCALE: 1=NO; 2=YES BUT RATHER STAY; 3=YES; MISSING=BLANK	417 8034	352 EB064
11\$ ACADEMP	SHOULD HAVE MORE ACADENIC EMPHASIS/COURSES	1=AGREE STRONGLY:2=AGREE; 3= DISAGREE: 4=DISAGREE STRONGLY BLK=MISSING/NA	301 BQ19/	181 EB035A
119 STUDSEI	STUDENT OCCUPATIONAL ASPIRATIONS (DUNCAN SEL)	DERIVED FROM X(69)		
120 FATHSEI	FATHERS OCCUPATIONAL (DUNCAN SEI)	DERIVED FROM X(71)		
LAL NOTHSEL	MOTHERS OCCUPATIONAL (DUNCAN SEI)	DERIVED FROM X(73)		
	TEACHING TECHNIQUES USED IN COURSES THIS Y	EAR		
122 LECTURE	LECTURES USED IN COURSES THIS YEAR	1=NEVER: 2=SELDON; 3=FAIRLY OFTEN; 4=FREQUENTLY; BLK=MISSING	221 BQ4A	19 E8003A
123 DI SCUSS	STUDENT DISCUSSION USED THIS YEAR	SANE	222 8048	20 E80038
124 LAB	PROJECT/LABS USED IN COURSES THIS YEAR	SANE	223 BQ4C	21 E8003C
125 ESSAYS	ESSAYS USED IN COURSES THIS YEAR	SANE	224 8940	22 EB0030
126 INDIVID	INDIVIOUAL INSTRUCTION USED IN COURSES	SAME	226 804F	23 E8003F
127 MACHINE	TEACHING MACHINE USED THIS YEAR	SAME	227 BQ46	24 E8003G
	CLASSIFIER VARIABLES			
150 SEX	STUDENT'S SEX	1=MALE; 2=FEMALE; BLANK=MISSING	1626 CSEX	404 88083
151 SES	SES COMPUSITE	1=LOW; 2=MEOIUM; 3=HIGH; BLK=MISSING	1070 SE <b>S</b>	512 BBSES
152 RACE	STUDENT'S RACE COMPOSITE	1=WHITE: 2=BLACK: 3=ASIAN-AMER.: 4=AMER. INDIAN: 5=MEXICAN-AMER:	1625 CRACE	416 88089 417 88090

6=PUERTO RICAN: 7=CTHER HISPANIC:

1=PUBLIC: 2=PRIVATE: 3=CATHOLIC

1=GENERAL: 2=ACADEMIC: 3=VOCTECH

1=URBAN: 2=SUBURB: 3=RURAL:

8=OTHER/NA

BLANK=MISSING

BLK=MISSING

4=WEST: BLANK=MISSING

153 SCHTYPE STUDENT'S SCHUOL TYPE

235

154 GEUREG GEOGRAPHIC REGION OF STUDENT'S SCHOOL

155 HSPROG STUDENT'S PRESENT HIGH SCHOOL PROGRAM

COMMUNITY TYPE

ERIC Pruit Ext Provided by ERIC

156 CONTYPE



236

17 BB002

(FROM SCH. TAPE) 2 SCHTYPE

= SCHVAR(2)

1073 HSPGM

414 8095

1=NORTHEAST; 2=NORTH CENTRAL; 3=SOUTH: 1066 REGION 6 CENREG

. ~

237

### APPENDIX A - Table 2

### ITEMS COMMON TO 1972 AND 1980 SCHOOL QUESTIONNAIRES

V/ 	LABEL	DESCRIPTION	RESPONSE CODES (CONVERTED)	NLS	NS8 00
2	ENH STA	S OF CURRENT FACULTY WHO ARE WHITE GI	0-1008 MISSING=BLANK	141 SCH0180	58075F
	485/PHD	<b>X OF FT TEACHERS WITH MASTERS OR DOCTORATE G</b>	2 1=0-29%, 2=30-49% 3=50-69%, 4=70+ %, #Issing=Blank	179 SCHQ30	58042
4	STURNOVI	8 OF FT TEACHERS LEAVING SINCE LAST YEAR G3	1=LT 5%, 2=5-9%, 3=10-19%, 4=20-100%, MISSING=BLANK	177 SCHQ28	58043
5	SATTENO	APPROX. AVERAGE DAILY PERCENTAGE ATTENDANCE G4	1=95+8. 2=90-958. 3=85-898. 4=LT 858. MISSING-8LANK	124 SCHQ16	58048
6	SOROPOUT	I ENTERING 10TH GRADE BUT NOT GRADUATING G3	0-100: MISSING-BLANK AVG OF: (NLS: AVG OF BOYS AND GIRLS)	125 SCHQ17A	<b>50</b> 014
7	SWHITE	T OF CURRENT STUDENTS WHO ARE WHITE GI	0-1001 MISSING-BLANK		
•	TBLACK	S OF CURRENT STUDENTS WHO ARE BLACK 65	0-1001 MISSINGERI ANK		389733
•	THISPAN	S OF CURRENT STUDENTS WHO ARE HISPANIC 65	0-100: MISSING-BLANK SUM OF (SUM OF MEXICAN, P.R., OTHER LATIN)	128 SCHOIDD 129 SCHOIDC 130SCHOIDD 131SCHOIDF	380943 580935
10	ECOL BND	8 OF LAST YEARS GRADS NOW IN 7/4 YR COLLEGE G2	1=LT 30%, 2=30-49%, <u>3=50-69%</u> , 4=70+%, MISSING=BLANK	157 SCH022	58011
11	SACAD	S ENROLLEJ IN COLLEGE PREPARATORY CURRICULUM G2	0-100: MISSING-BLANK	53 SCHQO3T	WEIGHTED AVG OF SAG178Y + 1785
12	<b>SGENERAL</b>	8 ENROLLED IN GENERAL CURRICULUM G2	0-100: MISSING-BLANK	52 SCHQ03S	WEIGHTED AVG OF
13	SVOCTECH	TENROLLED IN VOC-TEGH CURRICULUM G6	0-100: MISSING=BLANK SUN OF # THROUGH	54 <b>SCHQO3U</b> 59 SCHQO3Z	SUM OF WTD AVE
16	STUO/TCH	Ø OF STUDENTS PER CLASROM TEACHER	ØSTUDENTS/Ø TEACHERS: NISSING-BLANK (RANGE CHECK: 0-100)	176 SCHQ27 / 33 Schq20	58039C/58042A
14	S HAND	T OF STUDENTS CLASSIFIED AS HANDICAPPED	100+ #HANDICAPPED/# STUDENTS (Range Check of 7 ) Missing-Blank	70 SCHQ8 /33 SCHQ20	SB0 34/ SB30 2A
237	SPEC ED	TYPE OF SPECIAL EO TREATMENT	NEAN OF ALL HANDICAPPEO VARS. 1=REGULAR CLASSES: 2=NIX OF 1 & 3 3=SPECIAL CLASSES	MEAN OF 71-81 Scho9a-k	MEAN OF SB035A SB035K
20 FRIC	S DISADV	T OF STUDENTS CLASSIFIED AS DISADVANTAGED	LOO+ #0ISAOVANTAGED/#STUDENTS (RANGE CHECK 0- 7 } HISSING=BLANK	87 SCH011 /33 SCH020	58037/58002A
Full Text Provided by ERIC		BEST COPY A			

## BEST COPY AVAILABLE

24	TITLE I	NIGH SCHOOL PARTICIPATES IN TITLE I (ESS ED. ACT. LOW INCOME)	1=YES; 0=ND; MISSING-BLANK	166 SCH926C	SB032C1
25	TITLE 7	HIGH SCHOOL PARTICIPATES IN TITLE VIS (ESS BILINGUAL)	1=YES; 0=ND; MISSING-BLANK	169 SCHQ26F	\$803265
26	TITLE IS	HS PART. IN TITLE I-8: VOC EDUC-BASIC (VOCATIONL EDUACTION)	1-YES: 0-HD: MISSING-BLANK	171 SCH026H	S8032H2
27	TITLE IF	HS PART. IN TITLE I-F: VOC EDUC-HOMEMAKING	1=YES; O=ND; MISSING=BLANK	173 SCHQ26J	58032H1
28	ABIL GRP	HIGH SCHOOL USES ABILITY GROUPING	1=ANY YES: O=NO: MISSING =BLANK	121 SCH013 (ANY YES)	SB019 OR SB020 (ENG10 OR ENG12)
29	OFFER AP	HIGH SCHOOL OFFERS AP COURSES	1-YES: 0-ND: MISSING -BLANK	160 SCH025A	58029AD
		CLASSIFIERS			
35	SCH SES	NEAN SES OF STUDENTS IN SCHOOL	1=LOW QUAP.FILE: 2=MIDDLE Two quartiles: 3=HIGH Q.	MEAN OF CONTIN Student files	NUDUS SES FROM
36	SCH TYPE	SCHOOL TYPE OF CONTROL	1=PUBLIC; 2=PRIVATE; 3=CATHOLIC	16	
37	REGION	REGION	1-NORTHEAST: 2-NORTH CENTRAL; 3-SOUTH: 4-WEST	502 (STATE) Converted	
38	CON TYPE	COMMUNITY TYPE (URBANIZATION)	1=URBAN; 2=SUBURBAN; 3=RURAL	240	
39	WEIGHT	SCHOOL SAMPLE WEIGHT		02	
40	SCHL ID	SCHOOL ID NUMBER		01	
41	#STUDENT	# OF STUDENTS IN THE SCHOOL		33 SCHQ020	COLS (117.121)
42	# SENTORS	# OF SENIORS IN THE SCHOOL		32 SCHO2N	COLS (127.131)
43	STATUS	SURVEY STATUS	NLS: 1=BYP; 2=Extra by; 3=resurvey	03	

240

## BEST COPY AVAILABLE

Full Text Provided by ERIC

239

A-7

### APPENDIX B

Definitions of Variables and Coding Procedures



#### APPENDIX B

Some of the population classification variables used in this report are self-explanatory while others need additional definition. Definitions and coding procedures for the latter are presented below.

SES - SES scores are based on an equally weighted composite consisting of father's occupation, family income and selected household items. LOW SES--is the lower quartile of the SES composite scores. MIDDLE SES--is the two middle quartiles of the SES composite scores. HIGH SES--is the upper quartile of the SES composite scores. RACE - In 1972, individuals who responded as Blacks were coded as Blacks. Similar coding was carried out for Mexican-Americans and Puerto Ricans. "Other Hispanics" were other Latin Americans. In selected cross-tabs, Mexican-Americans, Puerto Ricans and Other Hispanics were combined into a "Hispanic" category. - In 1980, individuals who responded Black but who also responded that their place of origin was one of the Hispanic countries were coded to reflect place of origin. With this exception, the coding was the same as 1972.

GEOGRAPHIC

- REGION In 1972 and 1980 the four regions consisted of the following: Northeast--New England and Middle Atlantic
  - North Central--East North Central and West North Central
  - South--South Atlantic, East South Central and West South Central

West--Mountain and Pacific

CURRICULUM - Self-report in both 1972 and 1980.

URBAN,

SUBURBAN,

RURAL - 1972: Rural is rural only.

Suburban is small city or town, suburb of a medium or large city.

Urban is medium-sized city, large city or very large city.

- 1980: Reported by school administrators in response to a question having the above three categories.



### APPENDIX C

Cross-Tabulations of Test Scores



#### 5-1.1

IRT VOCABULARY FORMULA SCORE (SCALED TO NLS VOCABULARY TEST)

### SEX X CURRICULUM

#### -----

			NLS 19	972			H38 1	968				
SEX	CURRICUUM	SANFLE	HEIGHTED N	HEAN	<b>S.</b> D.	SAMPLE N	NE IGHTED N	NEAN	s.D.	POOLED S.O.	1980-1972 DIFFERENCE	EFFECT SIZE
MALE	GENERAL	2785	448512	5 97	3.4	4184	A					
MALF	ACADEMIC	7788	400000		3.0	4100	150.66	3.45	3.Z	3.37	-0.23	-0.07
	ACADENIC	20CC	895507	8.04	3.9	4496	470223	7.81	3.6	3.73	-0.23	_0 66
MALE	VOCATIONAL	1637	261824	4.27	3.4	<b>25</b> 51	275329	4.16	2.9	3.12	-0.11	-1.14
PENALE	GENERAL	2519	432012	5.36	3.6	4514	681037	4.76		7 74	-8 48 3	
FEMALE	ACADEMIC	3897	626809		17	4779	F14470			3.34	-0.00 #	-0.19
FEMALE	MOCATIONAL	8078	774464	0.30	3.7	4//2	219434	/.30	3.6	3.66	-1.00 <b>*</b>	-0.27
	TURAL	22/0	374954	5.01	3.4	31%	338336	4.25	3.0	3.15	-0.75 *	-0.24

#### 5-1.2

## SES X RACE

			NL\$ 19	72			HS8 19	980				
SES	RACE	SAMPLE N	WEIGHTED	MEAN	S.D.	SAMPLE N	WEIGHTED N	MEAN	S.D.	PUOLED S.D.	1980-1972 Difference	EFFECT SIZE
LOH	WHITE	2494	440597	5.52	3.7	3891	449362	6 71	39	1 10		
LOH	BLACK	1230	146835	2.76	2.7	1603	147551	7.7±	J.L 3 E	3.37	-0.01 #	-0.24
LOW	HISPANIC	456	5735A	3 26	2 9	1694	247332	2.75	2.5	2.59	-0.01	-0.01
			21324			1964	/0204	5.07	2.7	Z.74	-0.18	-0.07
MIDDLE	WHITE	6273	1283402	6.80	3.7	9033	1068234	6.09	34	1 57	-0.72 *	
MIDDLE	BLACK	585	74623	3.91	3.1	1115	97817	3 67	2 9	2.33	-0.76 *	-0.20
MIDDLE	HISPANIC	206	29568	4.14	3.2	909	5764	A 34	2.0	2.70	-0.24	-0.08
					3.2	700	3/440	4.20	3.1	3.09	0.03	0.01
HIGH	WHITE	3383	653922	8.70	3.8	4709	567221	7.88	3.6	3.65	-0.82 #	-0.23
NTCH	BLACK	102	12332	5.76	4.1	276	22306	5.37	3.6	3.72	-0 39	-0.11
HIGH	HISPANIC	47	6777	5.79	3.2	226	14503	4.90	3.2	3.19	-0.89	-0.11

#SIGNIFICANT AT .05 OR LESS

BEST COPY AVAILABLE



#### 5-1.3

#### IRT VOCABULARY FORMULA SCORE (SCALED TO NLS VOCABULARY TEST)

\*\*==\*\*\*==

#### SES X SCHOOL TYPE

#### ------

		NLS 19			HSB 19	80					
SCHOOL TYPE	SAMPLE N	WEIGHTED N	MEAN	s.D.	SAMPLE N	WEIGHTED N	MEAN	s.D.	POOLED S.D.	1980-1972 DIFFERENCE	EFFECT SIZE
PUBLIC	4112	630281	4.48	3.6	6775	670332	3.95	3.1	3.27	-0.53 *	-0.16
PRIVATE	9	1940	8.19	3.2	57	9351	4.69	3.3	3.33	-3.51	-1.05
CATHOLIC	138	29087	7.63	3.7	537	27671	6.27	3.5	3.53	-1.37 *	-0.39
PUBLIC	6672	1285659	6.40	3.7	10121	1152308	5.66	3.4	3.55	-0.74 *	-0.21
PRIVATE	33	7890	7.10	3.0	178	26930	6.55	3.5	3.45	-0.55	-0.16
CATHOLIC	549	132729	7.87	3.7	1175	89768	6.94	3.4	3.53	-0.93 *	-0.26
PUBLIC	3261	615445	8.58	3.8	4198	517537	7.57	3.6	3.69	-1.01 *	-0.28
PRIVATE	23	6068	8.97	4.7	490	47134	9.24	3.8	3.86	0.27	0.07
CATHOLIC	303	62517	9.29	3.5	769	62715	7.69	3.5	3.51	-1.61 *	-0.46
	SCHOOL TYPE PUBLIC PRIVATE CATHOLIC PRIVATE CATHOLIC PUBLIC PRIVATE CATHOLIC	SCHOOL TYPE SCHOOL TYPE SCHOOL TYPE N PUBLIC 4112 PRIVATE 9 CATHOLIC 138 PUBLIC 6672 PRIVATE 33 CATHOLIC 549 PUBLIC 3261 PRIVATE 23 CATHOLIC 303	NLS 19           SAMPLE         WEIGHTED           SCHOOL TYPE         N         N           PUBLIC         4112         630281           PRIVATE         9         1940           CATHOLIC         138         29087           PUBLIC         6672         1285659           PRIVATE         33         7890           CATHOLIC         549         132729           PUBLIC         3261         615445           PRIVATE         23         6068           CATHOLIC         303         62517	NLS 1972           SAMPLE         WEIGHTED           SCHOOL TYPE         N         N         MEAN           PUBLIC         4112         630281         4.48           PRIVATE         9         1940         8.19           CATHOLIC         138         29087         7.63           PUBLIC         6672         1285659         6.40           PRIVATE         33         7890         7.10           CATHOLIC         549         132729         7.87           PUBLIC         3261         615445         8.58           PRIVATE         23         6068         8.97           CATHOLIC         303         62517         9.29	NLS 1972           SAMPLE         WEIGHTED           SCHOOL TYPE         N         N         MEAN         S.D.           PUBLIC         4112         630281         4.48         3.6           PRIVATE         9         1940         8.19         3.2           CATHOLIC         138         29087         7.63         3.7           PUBLIC         6672         1285659         6.40         3.7           PUBLIC         549         132729         7.87         3.7           PUBLIC         549         132729         7.87         3.7           PUBLIC         3261         615445         8.58         3.8           PRIVATE         23         6068         8.97         4.7           CATHOLIC         303         62517         9.29         3.5	NLS 1972           SAMPLE         WEIGHTED         SAMPLE           SCHOOL TYPE         N         N         MEAN         S.D.         N           PUBLIC         4112         630281         4.48         3.6         6775           PRIVATE         9         1940         8.19         3.2         57           CATHOLIC         138         29087         7.63         3.7         537           PUBLIC         6672         1285659         6.40         3.7         10121           PRIVATE         33         7890         7.10         3.0         178           CATHOLIC         549         132729         7.87         3.7         1175           PUBLIC         3261         615445         8.58         3.8         4198           PRIVATE         23         6068         8.97         4.7         490           CATHOLIC         303         62517         9.29         3.5         769	NLS 1972         HSB 1972           SAMPLE         WEIGHTED         SAMPLE         WEIGHTED           SCHOOL TYPE         N         N         MEAN         S.D.         N         N           PUBLIC         4112         630281         4.48         3.6         6775         670332           PRIVATE         9         1940         8.19         3.2         57         9351           CATHOLIC         138         29087         7.63         3.7         537         27671           PUBLIC         6672         1285659         6.40         3.7         10121         1152308           PRIVATE         33         7890         7.10         3.0         178         26930           CATHOLIC         549         132729         7.87         3.7         1175         89768           PUBLIC         3261         615445         8.58         3.8         4198         517537           PRIVATE         23         6068         8.97         4.7         490         47134           CATHOLIC         303         62517         9.29         3.5         769         62715	NLS 1972         HSB 1980           SAMPLE         WEIGHTED         SAMPLE         WEIGHTED           SCHOOL TYPE         N         N         NEAN         S.D.         N         N         MEAN           PUBLIC         4112         630281         4.48         3.6         6775         670332         3.95           PRIVATE         9         1940         8.19         3.2         57         9351         4.69           CATHOLIC         138         29087         7.63         3.7         537         27671         6.27           PUBLIC         6672         1285659         6.40         3.7         10121         1152308         5.66           PRIVATE         33         7890         7.10         3.0         178         26930         6.55           CATHOLIC         549         132729         7.87         3.7         1175         89768         6.94           PUBLIC         3261         615445         8.58         3.8         4198         517537         7.57           PRIVATE         23         6068         8.97         4.7         490         47134         9.24           CATHOLIC         303         62517	NLS 1972         HSB 1980           SAMPLE WEIGHTED         SAMPLE WEIGHTED           SCHOOL TYPE         N         N         MEAN         S.D.         N         N         MEAN         S.D.           PUBLIC         4112         630281         4.48         3.6         6775         670332         3.95         3.1           PRIVATE         9         1940         8.19         3.2         57         9351         4.69         3.3           CATHOLIC         138         29087         7.63         3.7         537         27671         6.27         3.5           PUBLIC         6672         1285659         6.40         3.7         10121         1152308         5.66         3.4           PRIVATE         33         7890         7.10         3.0         178         26930         6.55         3.5           CATHOLIC         3261         615445         8.58         3.8         4198         517537         7.57         3.6           PRIVATE         23         6068         8.97         4.7         490         47134         9.24         3.8           CATHOLIC         303         62517         9.29<	NLS 1972         HSB 1980           SAMPLE         WEIGHTED         SAMPLE         HEIGHTED         PODLED           SCHOOL TYPE         N         N         MEAN         S.D.         N         N         NEAN         S.D.           PUBLIC         4112         630281         4.48         3.6         6775         670332         3.95         3.1         3.27           PRIVATE         9         1940         8.19         3.2         57         9351         4.69         3.3         3.33           CATHOLIC         138         29087         7.63         3.7         537         27671         6.27         3.5         3.53           PUBLIC         6672         1285659         6.40         3.7         10121         1152308         5.66         3.4         3.55           PRIVATE         33         7890         7.10         3.0         178         26930         6.555         3.5         3.45           CATHOLIC         549         132729         7.87         3.7         1175         89768         6.94         3.4         3.53           PUBLIC         3261         615445         8.58         3.8         4198         517537	NLS 1972         HSB 1980           SAMPLE         WEIGHTED         SAMPLE         WEIGHTED         POOLED         1980-1972           SCHOOL TYPE         N         N         MEAN         S.D.         N         N         MEAN         S.D.         N         POOLED         1980-1972           PUBLIC         4112         630281         4.48         3.6         6775         670332         3.95         3.1         3.27         -0.53         *           PRIVATE         9         1940         8.19         3.2         57         9351         4.69         3.3         3.33         -3.51           CATHOLIC         138         29087         7.63         3.7         537         27671         6.27         3.5         3.53         -1.37         *           PUBLIC         6672         1285659         6.40         3.7         10121         1152308         5.66         3.4         3.55         -0.74         *           PUBLIC         6672         1285659         6.40         3.7         10121         1152308         5.66         3.4         3.53         -0.55           CATHOLIC         549         132729         7.87         3.7         117

#### 5-1.4

## SES X GEOGRAPHIC REGION

			NLS 19	972			HSB 19	980				
<b>1</b> FS	BECTON	SAMPLE	WEIGHTED	ME ALI	S D	SAMPLE	WEIGHTED	ME ANI		POOLED	1980-1972	EFFECI
				HEAN	J.V.			FIEAN	<b>J.</b> U.	5.0.	DIFFERENCE	312E
LOH	NORTHEAST	836	157542	5.57	3.9	1457	145742	4.72	3.3	3.53	-0.85 ×	-0.24
LOH	NORTH CENTRAL	987	187162	5.03	3.6	1735	179518	4.52	3.2	3.34	-0.51 *	-0.15
LOW	SOUTH	2045	257851	3.74	3.2	2968	282557	3.42	2.9	3.04	-0.32 ¥	-0.11
LOH	HEST	623	91290	4.37	3.5	1209	99538	4.03	3.1	3.27	-0.34	-0.11
MIDDLE	NORTHEAST	1786	422549	7.14	3.8	2314	288557	6.30	3.5	3.63	-0.83 *	-0.23
MIDDLE	NORTH CENTRAL	2147	454048	6.47	3.7	3754	414256	5.82	3.4	3.52	-0.65 *	-0.18
TODLE	SOUTH	2197	341416	5.77	3.6	3396	346648	5.17	3.4	3.45	-0.59 *	-0.17
TODLE	WEST	1356	242788	6.59	3.8	2010	219546	5.92	3.5	3.62	-0.67 *	-0.18
ICH	NORTHEAST	855	194505	9.25	3.6	1119	162116	8.66	3.7	3.67	-0.59 *	-0.16
IGH	NORTH CENTRAL	946	187723	8.46	3.8	1632	175383	7.50	3.5	3 61	-0.96 #	-0.26
IGH	SOUTH	1080	172449	8.24	3.9	1606	153689	7 07	3.6	3 74	-1 18 #	_0 11
IGH	WEST	782	141085	8.49	3.8	1100	136197	7.56	3.5	3.62	-0.93 *	-0.26

CANT AT .05 OR LESS

Full Text Provided by ERIC

----

245

#### 5-1.5

IRT VOCABULARY FORMULA SCORE (SCALED TO NLS VOCABULARY TEST)

#### SES X CURRICULUM

			NLS 19	72			HSB 19	980				
SES	CURRICULUM	SAMPLE	WEIGHTED N	MEAN	\$.D.	SAMPLE N	WEIGHTED N	MEAN	\$.D.	POOLED S.D.	1980-1972 DIFFERENCE	EFFECT SIZE
LON	GENERAL	1792	275447	4.09	3.3	3124	303653	3.85	2.9	3.05	-0.24	-0.08
LOH	ACADEMIC	1049	176784	6.46	4.0	1636	147855	5.69	3.6	3.77	-0.77 ×	-0.20
LOW	VOCATIONAL	1650	241613	3.79	3.2	2476	243811	3.39	2.7	2.92	-0.40 *	-0.14
MIDDLE	GENERAL	2585	46 3462	5.47	3.5	4305	481705	5.07	3.2	3.28	-0.40 *	-0.12
MIDDLE	ACADEMIC	2998	663687	7.95	3.6	4253	459862	7.36	3.5	3.53	-0.58 *	-0.17
MIDDLE	VOCATIONAL	1902	333350	5.13	3.4	2794	312810	4.59	2.9	3.14	-0.55 *	-0.17
HIGH	GENERAL	905	157840	7.06	3.7	1419	170255	6.17	3.2	3.42	-0.89 *	-0.26
HIGH	ACADEMIC	2422	479275	9.46	3.6	3448	387614	8.79	3.5	3.51	-0.67 *	-0.19
HIGH	VOCATIONAL	336	58648	6.11	3.5	549	64281	5.45	3.0	3.21	-0.66	-0.21

#### 5-1.6

## SES X COMMUNITY TYPE

			NLS 19	72			HSB 19	<b>0</b> 80				
SES	CONTUNITY TYPE	SAMPLE N	WEIGHTED N	MEAN	S.D.	SAMPLE N	WEIGHTED N	MEAN	S.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
LOH	URBAN	1192	173605	4.48	3.6	2059	164566	3.75	3.1	3.25	-0.73 *	-0.22
LOH	SUBURBAN	1668	271149	4.78	3.7	2697	264518	4.22	3.2	3.39	-0.57 *	-0.17
LON	RURAL	1489	231322	4.59	3.6	2613	278270	4.08	3.1	3.28	-0.51 *	-0.16
MIDDLE	URBAN	1985	364152	6.28	3.7	2296	227999	5.46	3.5	3.61	-0.82 *	-0.23
MIDDLE	SUBURBAN	3633	751303	6.88	3.8	5633	625854	6.01	3.4	3.57	-0.87 *	-0.24
MIDDLE	RURAL	1669	311812	6.12	3.7	3545	415153	5.58	3.4	3.50	-0.55 *	-0.16
HIGH	URBAN	903	163506	8.22	3.8	950	101713	7.53	3.8	3.83	-0.69 *	-0.18
HIGH	SUBURBAN	2328	454622	8.91	3.7	3323	386381	7.86	3.6	3.66	-1.05 *	-0.29
HIGH	RURAL	372	66836	8.06	3.9	1184	139292	7.40	3.5	3.61	-0.67	-0.18

**\*SIGNIFICANT AT .05 OR LESS** 



### 5-2.1

IRT READING FORMULA SCORE (SCALED TO NLS READING TEST)

#### SEX X CURRICULUM

#### 

SEX			NLS I	972			HSB 1	980				
	CURRICULUM	SAMPLE N	WEIGHTED	MEAN	s.D.	SAMPLE N	NEIGHTED N	MEAN	\$.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
MALE	GENERAL	2789	469294	8.48	4.8	4181	457068	7.79	4.9	4.85	-0.68 *	-0.14
TALE	A' EMIC	3386	696241	11.77	4.4	4486	468792	11.52	4.7	4.61	-0.25	-0.05
RALE	VOCATIONAL	1636	261880	7.09	4.8	2548	274626	6.62	4.8	4.80	-0.47	-0.10
FEMALS	GENERAL	2520	432074	8.49	4.8	4494	4790 98	7.78	4.7	4.72	-0 71 +	-0 15
FEMALE	ACADEMIC	3098	627198	12.23	4./	4755	517360	11.31	4.6	4.57	-0 92 #	-0.19
FEMALE	VOCATIONAL	2278	375477	7.81	4.5	3192	337661	7.20	4.5	4.54	-0.60 *	-0.13

#### 5-2.2

## SES X RACE

BES			NLS 1	972			HSB 1	980				
	RACE	SAMPLE N	WEIGHTED	MEAN	S.D.	SAMPI.E N	WEIGHTED N	MEAN	s.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
LOH	WHITE	2498	441177	8.90	4.9	3887	448611	7.71	4.8	6 AN	-1 10 #	-0.25
LOH	BLACK	1237	147590	5.25	4.2	1591	146527	4	4.1	4.00	-0 35	-0.29
LOH	HISPANIC	457	57591	5.84	4.3	1522	78299	5.11	4.2	4.22	-0.83 *	-0.20
HIDDLE	WHITE	6275	1284218	10.30	4.6	1000	1064311	9 47	<b>4 8</b>	A 74	-0.07 ×	
HIDDLE	BLACK	586	74714	6.88	4.6	1117	98049	6.42	4.0	4.70	-0.03 *	-0.1.
HIDDLE	HISPANIC	205	29511	6.80	4.9	904	57212	6.44	4.6	4.70	- 9.36	-0.10
HICH	WHITE	3383	653764	12.23	4.4	4701	566634	11.47	47	<u>4 60</u>	-0 76 #	-0.14
HIGH	BLACK	102	12332	8.80	4.9	274	22217	7.84	4.9	4.94	-0.96	-0.10
HIGH	HISPANIC	48	7002	8.67	4.9	226	14527	7.04	4.7	4.78	-1.63	-0.34

\*SIGNIFICANT AT .05 OR LESS

247



#### 5-2.3

#### IRT READING FORMULA SCORE (SCALED TO NLS READING TEST)

-----

#### SES X SCHOOL TYPE

-----

			NLS 1	972			HSB 1	980				
\$ES	SCHOOL TYPE	SAMPLE N	WEIGHTED	MEAN	\$.D.	SAMPLE N	WEIGHTED N	MEAN	s.p.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
LOH	PUBLIC	4124	631877	7.52	4.9	6758	668490	6.63	4.7	4.81	-0.89 *	-0.19
LON	PRIVATE	9	1940	12.37	5.5	57	9351	7.18	5.0	5.13	-5.20	-1.01
LOH	CATHOLIC	138	29087	11.19	4.2	535	27659	8.87	4.6	4.51	-2.32 *	-0.51
MIDDLE	PUBLIC	6675	1286706	9.82	4.8	10089	1147909	8.95	4.9	4.87	-0.87 *	-0.18
MIDDLE	PRIVATE	33	7890	10.13	4.3	178	27117	10.27	5.4	5.25	0.14	0.03
MIDDLE	CATHOLIC	548	132557	11.20	4.4	1171	89847	10.01	4.6	4.54	-1.19 *	-0.26
HIGH	PUBLIC	3262	6155]1	12.08	4.5	4196	517043	11.29	4.8	4.71	-0.99 *	-0.21
HIGH	PRIVATE	23	6068	12.92	5.4	488	47122	12.89	4.3	4.38	-0.03	-0.01
HIGH	CATHO <sup>1</sup> .IC	30 3	62517	12.65	4.2	764	62685	10.77	4.8	4.65	-1.89 *	-0.41

### 5-2.4

## SES X GEOGRAPHIC REGION

			NLS 1	.972			HSB 1	.980				
SES	REGION	SAMPLE N	WEIGHTED	MEAN	S.D.	SAMPLE	WE IGHTED N	MEAN	S.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
LOW	NORTHEAST	837	157764	8.57	4.9	1446	144489	7.53	4.7	4.79	-1.04 *	-0.22
LOW	NORTH CENTRAL	992	187797	8.20	4.9	1732	179111	7.46	4.8	4.84	-0.74 *	-0.15
LON	SOUTH	2049	258334	6.73	4.9	2965	282458	5.89	4.7	4.76	-0.84 #	-0.18
LOH	WEST	625	91545	7.49	4.9	1207	99442	6.63	4.5	4.67	-0.86 *	-0.18
MIDDLE	NORTHEAST	1787	422789	10.35	4.7	2295	286207	9.48	4.8	4.75	-0.86 *	-0.18
MIDDLE	NORTH CENTRAL	2150	454846	9.94	4.7	3748	413330	9.31	4.9	4.81	-0.63 #	-0.13
MIDDLE	SOUTH	2196	341285	9.47	4.8	3393	346732	8.37	5.1	4.95	-1.09 #	-0.22
MIDDLE	WEST	1356	242962	9.75	4.9	2002	218604	9.08	4.9	4.92	-0.67 *	-0.14
HIGH	NORTHEAST	853	193977	12.68	4.5	1115	161665	11.93	4.6	4 56	-0.75 *	-0 16
HIGH	NORTH CENTRAL	948	188091	11.86	4.5	1633	175487	11.14	4 A	4.68	-0.52 #	-0 15
HIGH	SOUTH	1081	172597	12.15	4 4	1606	1518407	10 74	F 0	4.00	-1 61 #	-0.19
HIGH	WEST	782	141164	11.73	4.8	1096	135836	10.92	4.8	4.78	-4.81 *	-0.17

\*SIGNIFICANT AT .05 OR LESS

BEST COPY AVAILABLE



#### 5-2.5

#### IRT READING FORMULA SCORE (SCALED TO NLS READING TEST)

#### SES X CURRICULUM

#### -----

			NLS 1	972			HSB 1	980				
<b>3</b> E <b>3</b>	CURRICULUM	SAMPLE N	WEIGHTED N	MEAN	S.D.	SAMPLE N	WEIGHTED N	MEAN	s.D.	POOLED S.D.	1980–1972 Difference	EFFECT SIZE
LOH	GENERAL	1795	275873	7.00	4.8	3112	302403	6.35	4.5	4 59	-0 45 *	- 0 14
LON	ACADEMIC	1050	176902	10.21	4.9	1632	147683	9.17	4.9	4.97	-1 05 *	-0.14
LON	VOCATIONAL	1658	242666	6.51	4.6	2472	243219	5.84	4.5	×.51	-0.67 *	-0.21 -0.15
MIDDLE	GENERAL	2585	46 35 76	8.79	4.6	4290	480026	8.11	4.8	4 71	-0 49 *	-0.34
MIDDLE	ACADEMIC	3001	664544	11.67	4.3	4236	458151	11.18	4.6	4.72	-0.00 *	-0.14
MIDDLE	VOCATIONAL	1902	333460	7.99	4.6	2793	312351	7.45	4.6	4.58	-0.54 *	-0.11
HIGH	GENERAL	907	158144	10.25	4.6	1419	170498	9.37	4.7	4.64	-0 88 #	-0.19
HIGH	ACADEMIC	2423	479422	13.11	4.1	3440	386868	12.52	4.4	1 27	-0 60 #	-0.17
HICH	VOCATIONAL	334	58262	9.19	4.7	548	64248	8.38	4.9	4.85	-0.81	-0.14

#### 5-2.6

## SES X COMMUNITY TYPE

			NLS 1	972			HS8 1	980				
SES	CONTUNITY TYPE	SAMPLE N	WEIGHTED N	MEAN	S.D.	SAMPLE N	WEIGHTED N	MEAN	S.D.	POOLED S.D.	1980–1972 Difference	EFFEC1 SIZE
LOH LOH	URBAN SUBURBAN	1198 1673	174221 271990	7.39 7.79	4.8 4.9	2055 2684	164204 203083	6.35	4.6	4.66	-1.04 *	-0.22
LOH	RURAL	1492	231638	7.89	5.2	2611	278214	6.75	4.8	4.96	-1.14 *	-0.18
MIDDLE MIDDLE MIDDLE	URBAN Suburban Rural	1985 3633 1671	364226 751486 312429	9.45 10.31 9.83	4.8 4.7 4.8	2288 5608 3542	226736 623296 414842	8.64 9.23 9.01	5.0 4.8 5.0	4.91 4.76 4.96	-0.81 * -1.08 * -0.82 *	-0.17 -0.23 -0.17
HIGH HIGH HIGH	URBAN Suburban Rural	904 2328 372	163697 454498 66836	11.70 12.42 11.51	4.7 4.4 4.6	953 3315 1180	101670 386174 139006	11.06 11.27 11.11	4.9 4.8 4.8	4.79 4.65 4.78	-0.63 -1.16 * -0.41	-0.13 -0.25 -0.09

\*SIGNIFICANT AT .05 OR LESS

249



1. 1

#### 5-3.1

IRT MATHEMATICS FORMULA SCORE (SCALED TO NLS MATHEMATICS TEST)

#### SEX X CURRICULUM

----

			NLS 1	972			HSB 1	980				
SEX	CURRICULUM	SAMPLE N	IGHTED N	MEAN	S.D.	SAMPLE N	WEIGHTED	MEAN	S.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
MAIS	GENEDAL	2785	468685	11.17	6.8	4164	455165	10.72	6.7	6.70	-0.45	-0.07
MALE	ACADEMIC	3386	696179	17.26	6.2	4491	470390	17.18	6.1	6.18	-0.08	-0.01
MALE	VOCATIONAL	1636	261450	9.26	6.4	2528	272996	9.06	6.6	6.56	~0.20	-0.03
FEMALE	GENERAL	2517	432063	9.58	6.7	4457	475102	9.30	6.3	6.42	-0.28	-0.04
MALE	ACADEMIC	3099	627279	16.00	6.2	4748	516868	15.47	6.0	6.08	-0.53 <b>*</b>	-0.09
PEHALE	VOCATIONAL	2276	375276	8.45	6.0	3172	336 380	8.28	5.8	5.90	-0.17	-0.03

#### 5-3.2

## SES X RACE

			NLS 1	972			HSB 1	980				
SES	RACE	SAMPLE	WEIGHTED	MEAN	\$.D.	SAMPLE N	WEIGHTED N	MEAN	\$.D.	POOLED S.D.	1980–1972 Difference	EFFECT SIZE
LOH	WHITE	2495	440916	11.16	7.0	3858	445567	9.70	6.4	6.64	-1.47 *	-0.22
LOW	BLACK	1233	147062	5.68	5.8	1592	147685	5.76	5.7	5.76	0.09	0.01
LOH	HISPANIC	453	57164	7.41	6.2	1506	76832	6.46	6.5	6.42	-0.95	-0.15
MIDDLE	WHITE	6275	1284140	13.47	6.8	8986	1064151	12.77	6.6	6.69	-0.70 ¥	-0.11
MIDDLE	BLACK	586	74716	7.54	6.5	1114	97737	7.58	6.4	6.45	0.04	0.01
NIDDLE	HISPANIC	206	29568	8.08	7.0	895	56474	8.95	6.9	6.94	0.86	0.12
HIGH	WHITE	3385	654070	16.82	6.1	4688	564825	16.14	6.3	6.20	-0.68 *	-0.11
HIGH	BLACK	102	12332	10.21	6.3	272	22047	10.88	7.3	7.05	0.67	0.10
HIGH	HISPANIC	48		· 1º 🖧	NIL'ABLI	E <sup>224</sup>	14298	11.44	6.5	6.59	0.50	0.08

\*SIGNIFICANT AT .05 OR LESS



.

### 5-3.3

IRT MATHEMATICS FORMULA SCORE (SCALED TO NLS MATHEMATICS TEST)

### DEDI COPT AVAILABLE

SES X SCHOOL TYPE

--------------

			NLS 1	972			HSB 1	980				
SES	SCHOOL TYPE	SAMPLE N	WEIGHTED N	MEAN	S.D.	SAMPLE N	WEIGHTED N	) MEAN	S.D.	POOLED S.D.	1980-1972 DIFFERENCE	EFFECT SIZE
Lon	PUBLIC	4114	6 <b>30718</b>	9.27	7.0	6708	664658	8.27	6.5	6.70	-0.99 *	-0.15
Lon	PRIVATE	9	1940	13.86	6.4	57	9351	8.64	6.5	6.58	-5.22	-0.79
Lon	CATHOLIC	138	29087	13.81	7.0	538	27694	12.37	6.7	6.77	-1.44	-0.21
MIDDLE	PUBLIC	6675	1286515	12.76	7.1	10058	1146465	11.98	6.8	6.92	-0.78 *	-0.11
MIDDLE	PRIVATE	33	7890	14.36	5.6	179	27304	13.04	6.9	6.73	-1.32	-0.20
MIDDLE	CATHOLIC	549	132729	14.70	6.3	1173	89868	14.26	6.2	6.20	-0.44	-0.07
hich	PUBLIC	3264	615817	16.53	6.3	4175	514958	15.63	6.6	6.46	-0.90 *	-0.14
Hich	PRIVATE	23	6068	18.05	5.3	488	47122	18.35	5.2	5.21	0.30	0.06
Hich	CATHOLIC	303	62517	17.50	5.8	765	62555	15.56	5.8	5.77	-1. <del>9</del> 4 *	-0.34

5-3.4

SES X GEOGRAPHIC REGION

			NLS ]	972			HS <b>B</b> 1	980				
SES	REGION	SAMPLE N	MEIGHTED N	MEAN	\$.D.	SAMPLE N	MEIGHTED N	MEAN	s.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZF
lon Lon Lon	NORTHEAST NORTH CENTRAL SOUTH NEST	836 992 2043 622	157716 187797 257616 91153	10.47 10.41 8.16 8.91	7.2 7.0 6.8 7.0	1430 1724 2953 1196	143718 178420 281934 97631	9.62 9.74 7.04 8.33	6.7 6.6 6.1 6.7	6.91 6.74 6.39 6.77	-0.85 -0.67 -1.11 #	-0.12 -0.10 -0.17
MIDOLE MIDDLE MIDDLE MIDDLE	NORTHEAST NORTH LENTRAL SOUTH MEST	1788 2150 2196 1356	422848 454942 341237 242837	13.58 13.09 12.44 12.04	6.9 6.9 7.0 7.3	2297 3749 3389 1975	287033 413646 346872 216085	13.38 12.67 10.76 11.82	6.7 6.7 6.7 6.8	6.81 6.78 6.83 7.02	-0.20 -0.42 -1.68 * -0.22	-0.03 -0.06 -0.25 -0.03
HIGH FIGH FIGH	NORTHEAST NORTH CENTRAL SOUTH HEST	854 947 1081 784	194231 187807 172597 141500	17.48 16.71 16.72 15.21	5.9 6.2 6.2 6.7	1110 1628 1600 1090	161056 175098 153784 134697	17.24 15.81 14.75 15.39	6.0 6.4 6.6 6.4	5.96 6.36 6.42 6.56	-0.24 -0.90 * -1.97 * 0.18	-0.04 -0.14 -0.31 0.03

\*SIGNIFICANT AT .05 OR LESS

۰.

251



#### 5-3.5

### IRT MATHEMATICS FORMULA SCORE (SCALED TO NLS MATHEMATICS TEST)

#### SES X CURRICULUM

-----

			NLS 1	972			HSB 1	.980				
SES	CURRICULUM	SAMPLE N	NEIGHTED N	MEAN	\$.D.	SAMPLE N	WEIGHTED N	MEAN	S.D.	POOLED S.D.	1980-1972 DIFFERENCE	EFFECT SIZE
1.04	GENERAL	1787	274915	8.31	6.5	3085	299906	7.76	6.1	6.23	-0.55	-0.09
L CH	ACADEMIC	1050	176935	14.03	7.4	1627	147350	12.44	6.9	7.09	-1.59 *	-0.22
LON	VOCATIONAL	1656	242432	7.24	5.8	2458	242398	6.99	5.9	5.88	-0.24	-0.04
HTDDLE	GENERAL	2584	463435	10.61	6.6	4280	479249	10.36	6.3	6.44	-0.24	-0.04
MEDOLE	ACADENIC	3002	664530	16.19	6.1	4239	459415	16.00	6.0	6.06	-0.19	-0.03
HEDOLE	VOCATIONAL	1903	333596	9.56	6.3	2771	310480	9.38	6.1	6.17	-0.18	-0.03
isten	GENERAL	989	158623	13.57	6.4	1407	168699	12.87	6.4	6.38	-0.70	-0.11
ATCH	ACADENTC	2423	479422	18.32	5.4	3437	386931	18.01	5.4	5.37	-0.31	-0.06
NICH	VOCATIONAL	334	58089	10.94	5.1	544	6 3884	10.82	6.5	6.37	-0.12	-0.02

#### 5-3.6

## SES X CONTIUNITY TYPE

			NLS 1	972			H58 1	98 <b>Q</b>				
SES	COMMUNITY TYPE	SAMPLE N	NEIGHTED	MEAN	<b>S</b> .D.	SAMPLE N	WEIGHTED N	MEAN	\$.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
1.014	LIDRAM	1195	173866	8.67	6.8	2025	161943	7.99	6.7	6.71	-0.67	-0.10
LON		1440	271500	9 60	7.1	2678	263153	8.76	6.5	6.75	-0.84 #	-0.12
LOH	RURAL	1489	231 345	9.99	7.2	2600	276607	8.39	6.5	6.75	-1.60 *	-0.24
	URBAN	1987	364531	12.13	7.0	2282	226895	11.59	7.0	6.97	-0.54	-0.08
MITODIF	SIRUBBAN	3633	751238	13.43	7.0	5591	622240	12.51	6.7	6.83	-0.92 *	-0.13
MIDOLE	RURAL	1671	312429	12.97	7.0	3537	414501	11.94	6.8	6.86	-1.02 *	-0.15
MTCH	LIPBAN	903	163413	16.01	6.5	946	101386	15.51	6.7	6.60	-0.51	-0.08
MTCH	SI RI DRAN	2331	455088	16.99	6.1	3305	384616	16.06	6.4	6.25	-0.93 *	-0.15
MICH	RURAL	372	66836	16.01	6.4	1177	138632	15.42	6.4	6.40	-0.59	-0.09

**#SIGNIFICANT AT .05 OR LESS** 



#### 5-4.1

GRADES IN HIGH SCHOOL (1=BELOW D; 8=MOSTLY A)

#### SEX X CURRICULUM

-----------------

4 , ;			NLS 1	972			HSB 1	980				
SEX Mai f	CURRICULUM	SAMPLE N	WEIGHTED	MEAN	S.D.	SAMPLE N	WEIGHTED N	MEAN	S.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
MALE	GENERAL	2944	498317	4.77	1.3	4730	519772	5.00	1.4	1.35	0.23 *	0.17
MALE	ACADEMIC	3540	728838	5.79	1.3	5004	535880	6.17	1.3	1.32	0.38 *	0.29
MALE	VOCATIONAL	1737	279414	4.77	1.3	2908	314921	5.06	1.3	1.29	0.29 *	0.22
FEMALE	GENERAL	2691	463502	5.43	1.4	4993	532192	5.57	1.4	1.38	0.14 *	0.10
FEMALE	ACADEMIC	3230	654107	6.34	1.2	5231	571689	6.56	1.2	1.22	0.23 *	0.18
FEMALE	VOCATIONAL	2428	398889	5.50	1.3	3581	379345	5.66	1.3	1.31	0.15 *	0.12

#### 5-4.2

## SES X RACE

			NLS 19	72			HSB 19	780				
SES	RACE	SAMPLE N	WEIGHTED N	MEAN	S.D.	SAMPLE N	WEIGHTED N	MEAN	S.D.	POOLED S.D.	1980-1972 Difference	EFFECT SIZE
loh	NHITE	2622	464 <b>3</b> 95	5.36	1.5	4276	494242	5.45	1.4	1.45	0.10	0.07
Loh	Black	1347	159372	5.07	1.3	1896	178039	5.25	1.3	1.33	0.18 *	
Loh	Hispanic	489	61050	5.10	1.4	1751	93565	5.15	1.4	1.38	0.05	
MIDDLE	MHITE	6586	1356371	5.55	1.4	99 <b>33</b>	1781692	5.77	1.4	1.40	0.22 <b>*</b>	0.16
MIDDLE	Black	635	80621	5.16	1.3	1297	114810	5.28	1.3	1.32	0.12	0.09
MIDDLE	Hispanic	223	31815	5.21	1.3	1034	68013	5.24	1.4	1.38	0.03	0.02
high	HHITE	3547	687 <b>301</b>	6.00	1.4	5278	645596	6.18	1.4	1.38	0.19 *	0.13
Nigh	Black	111	1 <b>33</b> 42	5.11	1.2	318	26785	5.57	1.4	1.35	0.46	0.34
Nigh	Hispanic	49	7282	5.50	1.4	272	18636	5.56	1.3	1.31	0.06	0.05

. .

\*SIGNIFICANT AT .05 OR LESS



### 5-4.3

GRADES IN HIGH SCHOOL (1=BELOW D; 8=MOSTLY A)

#### SES X SCHOOL TYPE

-----

			NLS 19	972			HSB 19	980				
SES	SCHOOL TYPE	SAMPLE N	WEIGHTED N	MEAN	\$.D.	SAMPLE N	NEIGHTED H	MEAN	\$.D.	POOLED S.D.	1980–1972 DIFFERENCE	EFFECT SIZE
LON	PUBLIC	4347	665677	5.22	1.4	7694	763634	5.34	1.4	1.43	0.12 *	0.09
LON	PRIVATE	9	1940	6.42	1.0	75	11381	5.63	1.3	1.30	-0.79	-0.61
LOH	CATHOLIC	142	30552	5.91	1.3	566	29165	5.86	1.4	1.36	-0.05	-0.03
NICOLE	PUBLIC	7033	1362498	5.47	1.4	11285	1287288	5.67	1.4	1.41	0.20 *	0.14
MIDDLE	PRIVATE	33	7890	5.45	1.3	221	33625	6.06	1.3	1.34	0.61	0.46
MIDDLE	CATHOLIC	565	137933	5.85	1.3	1236	96100	5.93	1.3	1.30	0.08	0.06
HICH	PUBLIC	3424	646266	5.%	1.4	4764	591150	6.12	1.4	1.41	0.16 *	0.11
HIGH	PRIVATE	24	6382	5.62	1.1	560	57216	6.29	1.3	1.25	0.67	0.53
HIGH	CATHOLIC	316	66323	6.21	1.1	826	71040	6.14	1.3	1.25	-0.06	-0.05

#### 5-4.4

#### SES X GEOGRAPHIC REGION

-----

			NLS 19	972			HSB 19	980				
SES	REGION	SAMPLE N	WEIGHTED N	MEAN	\$.D.	SAMPLE H	WEIGHTED N	MEAN	\$.D.	POOLED S.D.	1980-1972 DIFFERENCE	EFFECT SIZE
LOH	NORTHEAST	863	162829	5.38	1.4	1655	166638	5.55	1.4	1.38	0.17	0.12
LOH	NORTH CENTRAL	1160	210689	5.06	1.5	1948	200398	5.22	1.5	1.46	0.16	0.11
LOH	SOUTH	2099	263362	5.35	1.4	3341	319533	5.39	1.4	1.42	0.04	0.03
LOH	WEST	667	99957	5.13	1.5	1391	117611	5.27	1.4	1.41	0.14	0.10
MIDDLE	NORTHEAST	1832	433713	5.56	1.3	2557	321711	5.83	1.3	1.31	0.26 *	0.20
MIDDLE	NORTH CENTRAL	2 345	496047	5.36	1.4	4105	453061	5.59	1.5	1.45	0.23 *	0.16
MIDDLE	SOUTH	2256	349749	5.63	1.4	3773	388208	5.76	1.4	1.43	0.13	0.09
MIDDLE	NEST	1451	266371	5.50	1.4	2307	254032	5.64	1.4	1.40	0.14	0.10
HIGH	NORTHEAST	871	196885	6.03	1.3	1253	184337	6.27	1.3	1.34	0.23 *	0.17
HIGH	NORTH CENTRAL	1032	204171	5.87	1.3	1797	192736	6.05	1.4	1.39	0.18	0.13
HIGH	SOUTH	1109	176926	6.08	1.4	1825	181383	6.17	1.4	1.44	0.09	0.06
HIGH	WEST	829	152998	5.94	1.4	1275	160951	6.06	1.4	1.37	0.12	0.09

\*SIGNIFICANT AT .05 OR LESS

Full Text Provided by ERIC

254

.

### 5-4.5

#### GRADES IN HIGH SCHOOL (1=BELOW D; 8=HOSTLY A)

#### SES X CURRICULUM

#### -----

			NLS 19	972			HSB 19	98 <b>0</b>				
SES	CURRICULUM	SAMPLE	MEIGHTED N	MEAN	\$.D.	SAMPLE N	NEIGHTED N	MEAN	<b>S</b> .D.	POOLED \$.D.	1980-1972 Difference	EFFEC' SIZE
lon Lon Lon	GENERAL ACADEMIC VOCATIONAL	1905 1100 1784	291672 185782 259383	4.97 5.90 5.08	1.4 1.3 1.4	<b>35</b> 14 1839 2819	<b>341454</b> 166 <b>887</b> 279206	5.12 6.00 5.29	1.4 1.4 1.4	1.39 1.36	0.15 * 0.09 0.21 *	0.11 0.07
MIDOLE MIDDLE MIDDLE	GENERAL ACADEMIC VOCATIONAL	2741 3140 2002	496541 696326 352709	5.08 5.91 5.30	1.4 1.3 1.3	4803 4649 3152	538090 508929 353541	5.30 6.32 5.43	1.4 1.3 1.3	1.39 1.28 1.33	0.23 * 0.41 *	0.16
high High High	GENERAL ACADEMIC VOCATIONAL	968 2516 357	170186 497701 63092	5.33 6.31 5.13	1.4 1.2 1.2	1613 3854 634	1 <b>96081</b> 442 <b>733</b> 74778	5.48 6.56 5.40	1.4 1.2 1.3	1.39 1.24 1.28	0.14 6.25 <b>*</b> 0.27	0.10 0.20 0.21

#### 5-4.6

#### SES X COMMUNITY TYPE

			NLS 19	972			HSB 19	980				
SES	COMMUNITY TYPE	SAHPLE N	WEIGHTED N	MEAN	<b>S</b> .D.	SAMPLE N	WEIGHTED N	MEAN	<b>S</b> .D.	POCLED S.D.	1980-1972 Difference	EFFECI SIZE
loh	URBAN	1372	197577	5.19	1.4	2419	196817	5.26	1.4	1.42	0.07	0.05
Loh	Suburban	1728	280161	5.19	1.4	31 <b>03</b>	305581	5.28	1.4	1.39	0.09	
Loh	Rural	1535	239407	5.40	1.5	2813	301782	5.51	1.4	1.45	0.11	
MIDDLE	URBAN	2174	4 <b>05151</b>	5.41	1.4	2663	266 <b>85</b> 9	5.52	1.4	1.41	0.10	0.07
MIDDLE	Suburban	3752	776124	5.50	1.4	6274	703130	5.67	1.4	1.40	0.17 *	0.12
MIDDLE	Rural	1742	32 <b>7138</b>	5.68	1.4	<b>3805</b>	447024	5.85	1.4	1.39	0.17 *	0.12
HI <del>gh</del>	URBAN	966	17 <b>7444</b>	5.94	1.3	1082	116816	6.21	1.4	1.38	0.27 *	0.20
Hi <del>gh</del>	Suburban	2427	472603	6.01	1.4	3779	447999	6.07	1.4	1.40	0.06	0.04
Hi <del>gh</del>	Rural	389	70032	5.94	1.3	1289	1 <b>545</b> 92	6.26	1.3	1.35	0.32 *	0.23

\*SIGNIFICANT AT .05 OR LESS

.

255



APPENDIX D

Test Score Partitioning

.

4



.

### Table D-1

### IRT VOCABULARY

## Grouping Variable: Total

			91					1000-1072	% OF TOTAL Change	TOTAL CHANGE DUE TO	TOTAL CHANGE DUE TO SUBGROUP HEAN	TOTAL CHANGE DUE TO INTER-
				PROP	DRILON		SCOOL	1700-17/2 Atecedence	CONTRO	SHIFT	CHANGES	ACTION
		1972	1980	1972	1980	1972	1980	ATLA ENCINCE	GROOP			
	BY: TOTAL			_								
TOTAL	TOTAL	<u>2860438.</u>	2666481.	1,000	1.000	<u>6.55</u>	<u> </u>	-0.05*	100.00	-0.00	-0.85	0 00
	SUBGROUPS COMBINED:	2860438.	2666481.			6.55	5.70	-0.05*	100.00	-0.00	-0.03	0.00
	RY: SFY											
TOTAL	HALF	1425843.	1218450.	8.499	0.473	6.44	5.90	-0.54*	33.16			
TOTAL	FEMALE	1433577.	1355969.	0.501	0.527	6.67	5.69	-0,98*	66.84		_	
	SUBGROUPS CONBINED:	2859421.	2574419.			6.55	5.79	-0.77*	100.00	0.01	-0.76	-0.01
TOTAL	DI: 353	AGTAAK	787354	0.243	0.272	4.59	4.05	-0.53*	25.35			
TOTAL	2000 2010	1460802	1269007.	0.512	0.487	6.52	5.77	-0.75*	45.70			
TOTAL	NTCH	695762	627386.	0.244	0.241	8.63	7.71	-0.93*	28.95			
TUIAL	SUBGROUPS CONBINED	2850409.	2603746.			6.56	5.77	-0.79#	100.00	-0.06	-0.74	0.01
TOTAL	OI · KALL	74 1844 9	2166617.	0.881	0.831	7.04	6.22	-0.82*	80.44			
TOTAL	BI ACY	234726	285008.	0.085	0.109	3.28	3.20	-0.08	10.31			
TOTAL	HEYTCAM.AMER	68274	83936.	0.025	0.032	3.47	3.50	0.03	2.59			
TOTAL	OTH HISPANIC	25698.	71148.	0.009	0.027	4.36	3.71	-0.65	6.66			
	SUBGROUPS CONBINED	2767167.	2604709.			6.60	5.73	-0.87*	100.00	-0.17	-0.73	0.03
TOTAL	ACADEMIC	1 199765	1009104.	9.462	0.304	8.29	7.62	-9.67#	48.04			
TOTAL	VOCACENERAL	1537350	1619997.	0.538	0.616	5.06	4.56	-0,50*	51.96			
TOTAL	SUBGROUPS CONBINED	2060135.	2629101.			6.55	5.73	-0.82*	100.00	-0.25	-0.58	0.01
	AV: SCHOOL TYPE											
PUBLIC	TOTAL	2540625.	2399504.	0.914	8.900	6.44	5.52	-0.92*	92.19			
NON-PUB	IC TOTAL	_240567.	266977.	0.086	0.100	8.21	7.30	-0.92*	7.81			
	SUBGROUPS COMBINED:	2781191.	2666481.			6.60	5.70	-0.90*	100.00	0.02	-0.92	0.00
		<b>5</b>										
	BT: CONTUNITY IT	789481	E1 1004		701.8	6.28	5.20	-1.07#	20.76			
	TUTAL	/UCDE3. 1480061	313708.	0.530	0.489	7.11	6.14	-0.97*	54.90			
JUDUKOA	TOTAL	A11217	849066	0.214	0.318	5.75	5.32	-0.42*	24.34			c.
RUMAL	SUBGROUPS COMBINED	2793902.	2666481.			6.61	5.70	-0.90*	100.00	-0.09	-0.88	0.06

ERIC 257

D**-1** 

er.

ERIC <sup>A</sup>Full Rox Provided by ERIC

Y: TOTAL TOTAL GROUPS COMBINED: Y: GEOB. REGION NORTHEAST N. CENTRAL SOUTH NEST GROUPS COMBINED: Y: HOMENORK/NK L.T. 5 HOURS S HOURS, HORE GROUPS COMBINED: Y: MORE AC EMPHAS AGREE DISAGREE GROUPS COMBINED:	1972 2860438. 2860438. 777868. 831402. 773440. 477727. 2860438. 1845836. 1000873. 2846709. SIS 1254152. 1239234. 2493386.	1980 <u>2666481.</u> 2666483. <u>609829.</u> 785432. 806215. <u>465005.</u> <u>2666481.</u> <u>2666481.</u> <u>2001304.</u> <u>655264.</u> <u>2655264.</u> <u>2655663.</u> <u>689406.</u> <u>2445268.</u>	1972 1.000 0.272 0.291 0.270 0.167 0.648 0.352 0.503 0.497	1980 1,000 0.229 0.295 0.302 0.174 0.753 0.247 0.718 0.202	1972 6.55 6.55 7.33 6.58 5.63 6.55 6.55 6.17 7.29 6.56 5.93 7.04	1980 5.70 5.70 6.48 5.85 4.84 5.94 5.70 5.26 7.08 5.71 5.64	-0.85* -0.86* -0.74* -0.74* -0.80* -0.77* -0.85* -0.91* -0.20 -0.86* -0.28*	<u>100.00</u> 100.00 26.97 25.52 31.81 <u>15.70</u> 100.00 85.20 <u>14.80</u> 100.00 42.19	-0.06 -0.12	-0.85 -0.79 -0.66	0.00 0.00 -0.07
Y: TOTAL TOTAL GROUPS COMBINED: Y: GEOB. REGION NORTHEAST N. CENTRAL SOUTH MEST GROUPS COMBINED: Y: HOMENORK/MK L.T. 5 HOURS S HOURS, HORE GROUPS COMBINED: Y: MORE AC EMPHAS AGREE DISAGREE GROUPS COMBINED:	2860438. 2860438. 31402. 773440. 477727. 2860438. 1845836. 1000873. 2846709. SIS 1254152. 1239234. 2493386.	2666481. 2666483. 785432. 806215. 465005. 2666481. 2001304. 655264. 2656567. 1755863. 689406. 2445268.	1.000 0.272 0.291 0.270 0.167 0.648 0.352 0.503 0.497	0.229 0.295 0.302 0.174 0.753 0.247 0.718 0.202	6.55 6.55 7.33 6.58 5.63 6.71 6.55 6.17 7.29 6.56 5.93 7.04	5.70 5.70 6.48 5.85 4.84 5.94 5.70 5.26 7.08 5.71 5.64	-0.85* -0.86* -0.74* -0.74* -0.80* -0.77* -0.85* -0.91* -0.20 -0.86*	<u>100.00</u> 100.00 26.97 25.52 31.81 <u>15.70</u> 100.00 85.20 <u>14.80</u> 100.00 42.19	-8.00 -0.06 -0.12	-0.85 -0.79 -0.66	0.00 0.00 -0.07
TOTAL GROUPS COMBINED: Y: GEOG. REGION NORTHEAST N. CENTRAL SOUTH NEST GROUPS COMBINED: Y: HOMENORK/MK L.T. 5 HOURS S HOURS, HORE GROUPS COMBINED: Y: HORE AC EMPHAS AGREE DISAGREE GROUPS COMBINED:	2860438. 2860438. 31402. 773440. 477727. 2860438. 1845836. 100873. 2846709. SIS 1254152. 1239234. 2493386.	2666481, 2666483, 785432, 806215, 465005, 2666481, 26566481, 2001304, 655264, 2656567, 1755863, 689406, 2445268,	1.000 0.272 0.291 0.270 0.167 0.648 0.352 0.503 0.497	0.229 0.295 0.302 0.174 0.753 0.247 0.718 0.282	6.55 6.55 7.33 6.58 5.63 6.55 6.17 7.29 6.56 5.93 7.04	5.70 5.70 6.48 5.85 4.84 5.94 5.70 5.26 7.08 5.71 5.64	-0,85* -0.86* -0.74* -0.80* -0.77* -0.85* -0.91* -0.20 -0.86*	100.00 100.00 26.97 25.52 31.81 15.70 100.00 85.20 14.80 100.00 42.19	-0.00 -0.06 -0.12	-0.85 -0.79 -0.66	0.00 0.00 -0.07
GROUPS COHBINED: Y: GEOB. REGION NORTHEAST N. CENTRAL SOUTH MEST GROUPS COMBINED: Y: HOMENORK/MK L.T. 5 HOURS S HOURS, HORE GROUPS COMBINED: Y: HORE AC EMPHAS AGREE DISAGREE GROUPS COMBINED:	2860438. 777868. 831402. 773440. 477727. 2860438. 1845836. 1000873. 2846709. SIS 1254152. 1239234. 2493386.	2666483. 609829. 785432. 806215. 465005. 2666481. 2001304. 655264. 2655667. 1755863. 689406. 2445268.	0.272 0.291 0.270 0.167 0.648 0.352 0.503 0.497	0.229 0.295 0.302 0.174 0.753 0.247 0.718 0.282	6.55 7.33 6.58 5.63 <u>6.71</u> 6.55 6.17 <u>7.29</u> 6.56 5.93 7.04	5.70 6.48 5.85 4.84 5.94 5.70 5.26 7.08 5.71 5.64	-0.85* -0.86* -0.74* -0.80* -0.77* -0.85* -0.91* -0.20 -0.86*	<u>100.00</u> 100.00 26.97 25.52 31.81 <u>15.70</u> 100.00 85.20 <u>14.80</u> 100.00 42.19	-0.06 -0.12	-0.85 -0.79 -0.66	0.00 0.00 -0.07
Y: GEOB. REGION NORTHEAST N. CENTRAL SOUTH NEST SROUPS COMBINED: Y: HOMENORK/NK L.T. 5 HOURS 5 HOURS, HORE SROUPS COMBINED: Y: MORE AC EMPHAS AGREE DISAGREE ROUPS COMBINED:	777868. 831402. 773440. 477727. 2860438. 1845836. 1000873. 2846709. SIS 1254152. 1239234. 2493386.	609829. 785432. 806215. 465005. 2666481. 2001304. 655264. 2656567. 1755863. 689406. 2445268.	0.272 0.291 0.270 0.167 0.648 0.352 0.503 0.497	0.229 0.295 0.302 0.174 0.753 0.247 0.718 0.202	6.33 7.33 6.58 5.63 <u>6.71</u> 6.55 6.17 <u>7.29</u> 6.56 5.93 7.04	5.70 6.48 5.85 4.84 5.94 5.70 5.26 7.08 5.71 5.64	-0.85* -0.86* -0.74* -0.80* -0.77* -0.85* -0.91* -0.20 -0.86* -0.28*	100.00 26.97 25.52 31.81 15.70 100.00 85.20 14.80 100.00 42.19	-0.00 -0.06 -0.12	-0.85 -0.79 -0.66	0.00 0.00 -0.07
Y: GEOG. REGION NORTHEAST N. CENTRAL SOUTH NEST SROUPS COMBINED: Y: HOMENORK/NK L.T. 5 HOURS 5 HOURS, HORE SROUPS COMBINED: AGREE DISAGREE ROUPS COMBINED:	777868. 831402. 773440. <u>477727.</u> 2860438. 1845836. <u>1000873.</u> 2846709. SIS 1254152. <u>1239234.</u> 2493386.	609829. 785432. 806215. 465005. 2666481. 2001304. 655264. 2656567. 1755863. 689406. 2445268.	0.272 0.291 0.270 0.167 0.648 0.352 0.503 0.497	0.229 0.295 0.302 0.174 0.753 0.247 0.718 0.202	7.33 6.58 5.63 6.71 6.55 6.17 7.29 6.56 5.93 7.04	6.48 5.85 4.84 <u>5.94</u> 5.70 5.26 <u>7.08</u> 5.71 5.64	-0.86+ -0.74+ -0.80+ -0.77+ -0.85+ -0.91+ -0.20 -0.86+ -0.28+	26.97 25.52 31.81 15.70 100.00 85.20 <u>14.80</u> 100.00 42.19	-0.06 -0.12	-0.79 -0.66	0.00 -0.07
NORTHEAST N. CENTRAL SOUTH NEST SROUPS COMBINED: I HOMENORK/MK L.T. 5 HOURS 5 HOURS, HORE SROUPS COMBINED: I HORE AC EMPHAS AGREE DISAGREE ROUPS COMBINED:	777868. 831402. 773440. <u>477727.</u> 2860438. 1845836. <u>1000873.</u> 2846709. SIS 1254152. <u>1239234.</u> 2493386.	609829. 785432. 806215. 465005. 2666481. 2001304. 655264. 2656567. 1755863. 689406. 2445268.	0.272 0.291 0.270 0.167 0.648 0.352 0.503 0.497	0.229 0.295 0.302 0.174 0.753 0.247 0.718 0.202	7.33 6.58 5.63 6.71 6.55 6.17 7.29 6.56 5.93 7.04	6.48 5.85 4.84 <u>5.94</u> 5.70 5.26 <u>7.08</u> 5.71 5.64	-0.86* -0.74* -0.80* -0.77* -0.85* -0.91* -0.28*	26.97 25.52 31.81 15.70 100.00 85.20 <u>14.80</u> 100.00 42.19	-0.06 -0.12	-0.79 -0.66	0.00 -0.07
N. CENTRAL SOUTH HEST GROUPS COMBINED: /: HOMENORK/MK L.T. 5 HOURS HOURS, HORE GROUPS COMBINED: /: HORE AC EMPHAS AGREE DISAGREE GROUPS COMBINED:	831402. 773440. <u>477727.</u> 2860438. 1845836. <u>1000873.</u> 2846709. SIS 1254152. <u>1239234.</u> 2493386.	785432. 806215. 465005. 2666481. 2001304. 655264. 2656567. 1755863. 689406. 2445268.	0.291 0.270 0.167 0.648 0.352 0.503 0.497	0.295 0.302 0.174 0.753 0.247 0.718 0.202	6.58 5.63 6.71 6.55 6.17 7.29 6.56 5.93 7.04	5.85 4.84 5.94 5.70 5.26 7.08 5.71 5.64	-0.28+	25.52 31.81 15.70 100.00 85.20 <u>14.80</u> 100.00 42.19	-0.06 -0.12	-0.79 -0.66	0.00 -0.07
SOUTH NEST SROUPS COMBINED: 1: HOMEHORK/MK L.T. 5 HOURS 5 HOURS, HORE SROUPS COMBINED: 1: HORE AC EMPHAS AGREE DISAGREE SROUPS COMBINED:	773440. <u>477727.</u> 8860438. 1845836. <u>1000873.</u> 2846709. SIS 1254152. <u>1239234.</u> 2493386.	806215. 465005, 2666481. 2001304. 655264. 2656567. 1755863. 689406. 2445268.	0.270 0.167 0.648 0.352 0.503 0.497	0.302 0.174 0.753 0.247 0.718 0.202	5.63 6.71 6.55 6.17 7.29 6.56 5.93 7.04	5.26 5.26 5.70 5.26 5.71 5.64	-0.91* -0.85* -0.91* -0.91* -0.86* -0.28*	25.52 31.81 <u>15.70</u> 100.00 85.20 <u>14.80</u> 100.00 42.19	-0.06	-0.79 -0.66	0.00 -0.07
HEST SROUPS COMBINED: 1: HOMENORK/HK L.T. 5 HOURS 5 HOURS, HORE SROUPS COMBINED: 1: HORE AC EMPHAS AGREE 0 ISAGREE SROUPS COMBINED:	477727. 2860438. 1845836. 1000873. 2846709. SIS 1254152. 1239234. 2493386.	<u>465005</u> , 2666481. 2001304. <u>655264</u> , 2656567. 1755863. <u>689406</u> , 2445268.	0.167 0.648 0.352 0.503 0.497	0.174 0.174 0.753 0.247 0.718 0.202	6.71 6.55 6.17 <u>7.29</u> 6.56 5.93 7.04	5.94 5.70 5.26 7.08 5.71 5.64	-0.20 -0.28*	31.81 <u>15.70</u> 100.00 85.20 <u>14.80</u> 100.00 42.19	-0.06	-0.79 -0.66	0.00 -0.07
ROUPS COMBINED: /: HOMEMORK/MK L.T. 5 HOURS 5 HOURS, HORE ROUPS COMBINED: /: HORE AC EMPHAS AGREE DISAGREE ROUPS COMBINED:	2860438. 1845836. 1000873. 2846709. SIS 1254152. 1239234. 2493386.	26666481. 2001304. 655264. 2656567. 1755863. 689406. 2445268.	0.648 0.352 0.503 0.497	0.753 0.247 0.718 0.262	6.17 7.29 6.55 5.93 7.04	5.26 <u>7.08</u> 5.64	-0.91* -0.85* -0.20 -0.86*	100.00 85.20 14.80 100.00 42.19	-0.06	-0.79 -0.66	0.00 -0.07
<ul> <li>HOMEHORK/IAK</li> <li>L.T. 5 HOURS</li> <li>5 HOURS, HORE</li> <li>ROUPS COMBINED:</li> <li>1' HORE AC EMPHAS</li> <li>AGREE</li> <li>DISAGREE</li> <li>ROUPS COMBINED:</li> </ul>	1845836. <u>1000873.</u> 2846709. SIS 1254152. <u>1239234.</u> 2493386.	2001304. 655264. 2656567. 1755863. 689406. 2445268.	0.648 0.352 0.503 0.497	9.753 0.247 0.718 0.282	6.17 7.29 6.56 5.93 7.04	5.26 <u>7.08</u> 5.71	-0.91* -0.20 -0.86*	85.20 <u>14.80</u> 100.00 42.19	-0.12	-0.79	0.00 -0.07
<ul> <li>HOHEHORK/MK</li> <li>L.T. 5 HOURS</li> <li>5 HOURS, HORE</li> <li>5 HOURS, HORE</li> <li>3 ROUPS COMBINED:</li> <li>A GREE</li> <li>D ISAGREE</li> <li>3 ROUPS COMBINED:</li> </ul>	1845836. 1000873. 2846709. SIS 1254152. 1239234. 2493386.	<b>2001304</b> . <u>655264</u> . 2656567. <b>1755863</b> . <u>689406</u> . 2445268.	0.648 0.352 0.503 0.497	0.753 0.247 0.718 0.282	6.17 7.29 6.56 5.93 7.04	5.26 <u>7.08</u> 5.71 5.64	-0.91* -0.20 -0.86*	85.20 <u>14.80</u> 100.00 42.19	-0.12	-0.66	-0.07
L.T. 5 HOURS 5 HOURS, MORE ROUPS COMBINED: 1: MORE AC EMPHAS AGREE DISAGREE ROUPS COMBINED:	1845836. <u>1000873.</u> 2846709. SIS 1254152. <u>1239234.</u> 2493386.	2001304. <u>655264.</u> 2656567. 1755863. <u>689406.</u> 2445268.	0.648 0.352 0.503 0.497	0.753 0.247 0.718 0.282	6.17 7.29 6.56 5.93 7.04	5.26 7.08 5.71 5.64	-0.91* -0.20 -0.86*	85.20 <u>14.80</u> 100.00 42.19	-0.12	-0.66	-0.07
5 HOURS, HORE ROUPS COMBINED: 1: HORE AC EMPHAS AGREE DISAGREE ROUPS COMBINED:	<u>1000873.</u> 2846709. <b>SIS</b> 1254152. <u>1239234.</u> 2493386.	<u>655264.</u> 2656567. 1 <b>755863.</b> <u>689406.</u> 2445268.	0.352 0.503 0.497	0.247 0.718 0.282	7.29 6.56 5.93 7.04	<u>7.08</u> 5.71	-0.20 -0.86* -0.28*	<u>14.80</u> 100.00 42.19	-0.12	-0.66	-0.07
ROUPS COMBINED: ': MORE AC EMPHA: AGREE DISAGREE ROUPS COMBINED:	2846709. SIS 1254152. <u>1239234.</u> 2493386.	2656567. 1755863. 689406. 2445268.	0.503 0.497	0.718 0.282	6.56 5.93 7.04	5.64	-0.28*	42.19	-0.12	-0.66	-0.07
(: HORE AC EMPHA: Agree DISAGREE ROUPS COMBINED:	<b>515</b> 1254152. <u>1239234.</u> 2493386.	1 <b>755863</b> . <u>689406</u> 2445268.	0.503 0.497	0.718 0.282	5.93	5.64	-0.28*	42.19	-0.12	-0.66	-0.07
AGREE DISAGREE DISAGREE ROUPS COMBINED:	515 1254152. <u>1239234.</u> 2493366.	1 <b>755863</b> . <u>689406</u> . 2445268.	0.503 0.497	0.718 0.282	5.93	5.64	-0.28*	42.19			
AGREE DISAGREE ROUPS COMBINED:	1254152. <u>1239234.</u> 2493386.	1755863. <u>689406.</u> 2445268.	0.503 0.497	0.718 <u>0.282</u>	5.93 7.04	5.64	-0.28*	42.19			
DISAGREE ROUPS COMBINED:	<u>1239234.</u> 2493386.	<u>689406.</u> 2445268.	0.497	0.282	7.04	E 01					
WOUPS CONSINED:	2493386.	2445268.						57.81			
					6.48	5.72	-0.76*	100.00	-0.24	-0.70	0 18
										-0.70	V.10
· WHITE TH SCH	OOL										
U-07%	771957.	1101375.	0.356	0.434	5.62	5.14	-0.48*	34.08			
	1/91035.	1435905.	<u>    9.644                               </u>	0.566	<u> </u>	6.17	-0.88*	65.92			
WOULS COURTHER!	2783099.	253/300.			6.54	5.73	-0.82*	100.00	-0.11	-0.74	0.03
S TOTAL	47186.	106.98.	8.022			• **	-1 4.0%	7			
TOTAL	362347.	502502.	0.167	0.193	5 44	3 81	-1.02*	-3.07			
TUTAL	661843.	907915.	0.305	0.348	6.00	4 71	-1.03*	24./1 34.95			
TOTAL	820637.	642089.	0.379	8.254	A 30	A 83	-1.00	34.63			
TUTAL	275284.	524265.	0.127	0.201	9.60	8.07	-1.4/*	33.13			
GROUPS COMBINED:	\$167297.	2607470.			7.20	5.74	-1.46#	100.00	-0.00	-1 45	
					,	2	-2.70*	100.00	-0.00	-1.43	v.uv
Y: STUDY AIDS IN	N HOME										
TOTAL	565540.	820351.	0.178	0.308	5.12	4.59	-0.53*	37.64			
TOTAL	<u>2294897.</u>		208.0	0.692	6.91	6.19	-0.71*	62.36			
	2860438.	2666481.			6.55	5.70	-0.85*	100.00	-0.20	~0.68	50.0
GROUPS CONDINED:											
•	TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL TOTAL	TOTAL 820637. TOTAL <u>275284.</u> ROUPS CONBINED: 2167297. STUDY AIDS IN NOME TOTAL 565540. TOTAL <u>2294897.</u> ROUPS CONBINED: 2860438.	T(TAL 820637. 662089. TOTAL <u>275284. 524265.</u> STOUPS CONDINED: 2167297. 2607470. (* STUDY AIDS IN NOME TOTAL 565540. 820351. TOTAL <u>2294897. 1846130.</u> STOUPS CONDINED: 2860438. 2666481.	T(TAL 820637. 642089. 0.379 TOTAL <u>275204. 524265. 0.127</u> ROUPS COMBINED: 2167297. 2607470. TOTAL 565540. 820351. 0.198 TOTAL <u>2294097. 1046130. 0.802</u> ROUPS COMBINED: 2860438. 2666481.	T(TAL       820637.       662089.       0.379       0.254         TOTAL       275284.       524265.       0.127       0.201         ROUPS COMBINED:       2167297.       2607470.         **       STUDY AIDS IN HOME         TOTAL       565540.       820351.       0.176       0.308         TOTAL       565540.       820351.       0.176       0.308         TOTAL       565540.       820351.       0.176       0.308         TOTAL       2294897.       1096130.       0.602       0.692         WOUPS COMBINED:       2860430.       2666481.       0.002       0.692	T(TAL       820637.       642089.       0.379       0.254       8.30         TUTAL       275284.       524265.       0.127       0.201       9.60         SROUPS CONBINED:       2167297.       2607470.       7.20         7:       STUDY AIDS IN HOME       7.20         TOTAL       565540.       620351.       0.178       0.308       5.12         TOTAL       565540.       620351.       0.278       0.692       6.91         WOUPS COMBINED:       2860430.       2666481.       6.55	T(TAL       820637.       642089.       0.379       0.254       8.30       6.83         TUTAL       275284.       524265.       0.127.       0.201       9.60       8.07         WOUPS COMBINED:       2167297.       2607470.       7.20       5.74         **       STUDY AIDS IN HOME       7.20       5.74         TOTAL       565540.       620351.       0.198       0.308       5.12       4.59         TOTAL       2294897.       1046130.       0.802       0.692       6.91       6.19         WOUPS COMBINED:       2060438.       2666481.       6.55       5.70	T(TAL       820637.       642089.       0.379       0.254       8.30       6.83       -1.47H         T(TAL       275204.       524265.       0.127.       0.201       9.60       8.07       -1.54H         T(TAL       275204.       524265.       0.127.       0.201       9.60       8.07       -1.54H         WOUPS COMBINED:       2167297.       2607470.       7.20       5.74       -1.46H         /*       STUDY AIDS IN HOME       7.20       5.74       -1.46H         /*       STUDY AIDS IN HOME       7.20       5.74       -1.46H         /*       STUDY AIDS IN HOME       820351.       0.198       0.306       5.12       4.59       -0.53H         TOTAL       565540.       620351.       0.198       0.306       5.12       4.59       -0.53H         TOTAL       2294897.       1846130.       0.802       0.692       6.91       6.19       -0.71H         WOUPS COMBINED:       2860438.       2666481.       6.55       5.70       -0.85H	T(TAL       820637.       642089.       0.379       0.254       8.30       6.83       -1.47#       35.13         T(TAL       275284.       524265.       0.127       9.201       9.60       8.07       -1.54#       9.01         MOUPS COMBINED:       2167297.       2607470.       7.20       5.74       -1.46#       100.00         /*       STUDY AIDS IN HOME       7.20       5.74       -1.46#       100.00         /*       STUDY AIDS IN HOME       7.20       5.74       -0.53#       37.64         TOTAL       565540.       820351.       0.198       0.308       5.12       4.59       -0.53#       37.64         TOTAL       2294897.       1046130.       0.802       0.692       6.91       6.19       -0.71#       62.36         MOUPS COMBINED:       2860430.       2666481.       6.55       5.70       -0.85#       100.00	TOTAL       820637.       64289.       0.379       0.254       8.30       6.83       -1.47#       35.13         TOTAL       275284.       524265.       0.127.       0.201       9.60       8.07       -1.54#       9.01         WOUPS CONDINED:       2167297.       2607470.       7.20       5.74       -1.46#       180.00       -0.00         /*       STUDY AIDS IN HOME       7.20       5.74       -1.46#       180.00       -0.00         /*       STUDY AIDS IN HOME       565540.       820351.       0.178       0.308       5.12       4.59       -0.53#       37.64         TOTAL       565540.       820351.       0.402       0.692       6.91       6.19       -0.71#       62.36         WOUPS CONSINED:       2860438.       2666481.       6.55       5.70       -0.85#       100.00       -0.20	TOTAL       820637.       642089.       0.379       0.254       8.30       6.83       -1.47#       35.13         TOTAL       275204.       524265.       0.127.       0.201       9.60       8.07       -1.54#       9.01         WOUPS CONDINED:       2167297.       2607470.       7.20       5.74       -1.46#       180.00       -0.00       -1.45         /*       STUDY AIDS IN HOME       7.20       5.74       -1.46#       180.00       -0.00       -1.45         /*       STUDY AIDS IN HOME       7.20       5.74       -1.46#       180.00       -0.00       -1.45         /*       STUDY AIDS IN HOME       565540.       820351.       0.178       0.308       5.12       4.59       -0.53#       37.64         TOTAL       565540.       820351.       0.178       0.308       5.12       4.59       -0.53#       37.64         TOTAL       8294097.       1046130.       0.402       0.692       6.91       6.19       -0.71#       62.36         WOUPS COMBINED:       2660438.       2666481.       6.55       5.70       -0.85#       100.00       -0.20       -0.68

BEST COPY AVAILABLE

# BEST COPY AVAILABLE

D-1

			1TED .N	FROPOI <u>OF POFUI</u>	RTION LATION	MEAN_S	SCORE	1980-1972 <u>DIFFERENCE</u>	X OF TOTAL CHANGE DUE TO GROUP	TOTAL CHANGE DUE TO POP. SHIFT	TOTAL CHANGE DUE TO SUBGROUP MEAN <u>CHANGES</u>	TOTAL CHANGE DUE TO INTER- ACTION
		1972	1980	1972	1980	1972	1980					
		INSTR TH CO	PSES THIS	YEAR								
MEVER . SET	IN TOTAL	1437820.	1464632.	0.514	9.566	6.08	5.27	-0.81*	57.16			
OFTENSER	Q. TOTAL	1356871.	1122084.	0.406	0.434	7.17	6.40	-0.78*	42.84		- 0 70	-0.00
5	UEIGROUPS COMBINED :	2794691.	2586716.			6.61	5.76	-0.85*	100.00	-0.00	-0.77	-0.00
		N COLIRSES	THIS YEAR									
NEVED. SE	TOTAL	995.896	949652			K 07	A. 98	-8 97#	41 19			
OFTEN, FR	EQ. TOTAL	1805717.	1621616.	0.645	0.631	6.97	6.22	-0.75*	56.61			
	SUDGROUPS COMBINED:	2801613.	2571268.			6.60	5.75	-0.85*	100.00	-0.01	-0.83	-0.00
		FT TEACHE	RS NITH AD	VANCED DE	GREES							
0-49%	TOTAL	1904970	1359848.	8.676	8.526	6.29	5 45	-8 864	44 94			
50-100%	Te-JAL	.921880.	1224012.	0.326	0.474	7.10	5.97	-1.13*	53.06			
	SUBBROUPS COMBINED:	2826850.	2583852.			6.56	5.70	-9.86*	100.00	0.12	-0.93	-0.04
		E 1497 YFA		TES NON I	N COLLEG	E						
9-302	TOTAL	320994	477376	0.113	0.183	4.96	4.57	-0.39#	<b>21.57</b>			
30-49%	TOTAL	1072007	. 1012141.	0.378	0.389	6.05	5.38	-9.67#	31.35			
50-69%	TOTAL	937672	. 697691.	0.331	0.268	6.86	5.93	-0.93*	31.63			
70-100%	TOTAL	502310	416185.	<u>0.177</u>	0,160	8.08	7.43	-0.65*	15.45			• • •
	SUBGROUPS COMBINED	2832982	. 2603393.	•		6.55	5.71	-0.85*	100.00	-0.16	-0.72	0.04
	BY: HIGH SCHOOL S	FFERS ADVA		MENT COLR	SES							
YES AP	TOTAL	814253.	1205570.	0.318	0.467	7.13	6.18	-0.95*	41.73			
NO AP	TOTAL	1746361	1378247.	0.682	0.533	6.31	5.29	-1.02*	58.27	_		
	SUBEROUPS CONSINED	2560614.	2583816.			6.57	5.71	-0.87*	100.00	0.12	-1.00	0.01
	BY: SEHESTERS OF	HATHEMATI	ICS TAKEN S	OPH-SENIC	DR YEAR							
4 OR FE	WER TOTAL	2030138	1655776.	0.710	0.621	6.08	5.04	-1.04*	70.69			
s or ho	NE TOTAL	_830300	1010706.	0,290	0.379	<u>7.71</u>	6.78	0.93*	29.31			
	SUBBROUPS CONSINED	2860438.	2666481.			6.55	5.70	-0.85*	100.00	0.14	-1.01	0.01
	BY: SEMESTERS OF	SCIENCE T	AKEN SOPH-	SENIOR YE	AR		<i></i>					
	NED TOTAL	Z1713A2.	1995678.	8.759	<b>I. 748</b>	6.16	5.12	-1.04#	91.24			

Q	4 OR FENER 5 OR MORE	TOTAL TOTAL	2171382. 689056.	1995678. 670804	0.759 0.241	0.748 0.252	6.16 7,79	5.12	-1.04*	91.24 <u>8.76</u>				262
Full Text Provided by ERIC	1 sue	GROUPS COMBINED:	2860438.	2666481.			6.55	5.70	-0.85*	100.00	30.0	-9.68	0.01	

	<u> </u>	FROPORTION <u>OF POPULATION</u> 1972 1980	<u>MEAN SCORE</u> 1972 1980	X OF Total Change 1980-1972 Oue to <u>Difference</u> Group	TOTAL Change Due to Pop. Shift	TOTAL CHANGE TOTAL DUE TO CHANGE SUBGROUP DUE TO MEAN INTER- CHANGES ACTION
BY: SEMESTE	RS OF FOREIGN LANGUAGE TAK	EN SOPH-SR YEAR				
3 OR FENER TOTAL 4 OR MORE TOTAL	2090521. 2025039. 	0.731 0.759	5.85 5.08 <u>8.46</u> 7.66	- <b>0.7</b> 7# 71.36 -0.80# 28.94		
SUBGROUPS COM	INED: 2860438. 2666481.		6.55 5.70	-0.85* 100.00	-0.07	-0.78 <b>-</b> .00
BY: PARTICIP NO TOTAL YES TOTAL SUBGROUPS COND:	ATE IN ATHLETIC <b>5</b> 1559175. 1273380. <u>1266320. 1363250.</u> INED: 2825495. 2636630.	0.552 0.483 	6.55 <b>5.57</b> <u>6.62 5.88</u> 6.58 5.73	-0.98# 55.42 	0.00	-0.87 0.02
BY: SCHOOLIN NO 4-YR COL. TOTAL 4-YR COLLEGE TOTAL	6 NOTHER NANTS FOR STUDEN 1005835. #31633. 1307015. 1269422.	T	5.25 4.37 8.14 <u>6.96</u>	-0.89# 31.07 -1.19# 68.93		
SUBGROUPS COMB	INED: 2312850. 2121054.		6.89 5.92	-0.97* 100.00	0.10	-1.05 -0.01

D-1

263



BEST COPY AVAILABLE

### Table D-2

### IRT READING Grouping Variable: Total

		<u>NE I</u> 1972	<del>GNTED N</del> 1980	PROF <u>OF POF</u> 1972	ORTION ULATION 1980	<u> </u>	<u>SCORE</u> 1980	1980–197; <u>Differenci</u>	2 OF TOTAL CHANGE DUE TO E GROUP	TOTAL CHANGE DUE TO POP. Shift	TOTAL CHANGE DUE TO SUBGROUP MEAN <u>Changes</u>	TOTAL CHANGE DUE TO INTER- ACTION
	BY: TOTAL											
TOTAL	TOTAL	2863482.	2658958.	1.000	1.000	<u> </u>	8,84	-1.05*	100.00	-0.00	-1 05	
	SUBGROUPS COMBINED:	2863482.	2658958.			9.09	0.04	-1.05*	100.00	-0.00	-1.03	•
	RY: SFY											
TOTAL	MALE	1427414.	1215335.	0.499	0.473	9.53	8.95	-0.88*	44.53			
TOTAL	FEMALE	<u>1435051.</u>	1352068.	0.501	0.527	9.95	8.%		<u>55.47</u>			
	SUBGROUPS CONSINED:	2862465.	2567403.			9.89	8.95	-0.93*	100.00	0.00	-0.93	-0.00
	AY: SFS											
TOTAL	LON	695440.	705500.	0.244	0.272	7.65	6.73	-0.92*	32.37			
TOTAL	MIDDLE	1461882.	1264873.	0.512	9.487	9.92	9.05	-0.86*	43.64			
TOTAL	HIGH	695829.	626850.	0.244	0.241	12.13	11.20	-0,94*	23.99			
	SUBGROUPS COMBINED:	2853152.	2597223.			9.90	8.94	-0.97#	100.00	-0.07	-0.90	-0.00
	RY: DACE											
TOTAL	MITE+AS+IND	2439707.	2159015.	0.881	0.831	10.51	9.57	-0.94*	74.69			
TOTAL	BLACK	235572	283823.	0.085	0.109	5.94	5.56	-0.38	12.92			
TOTAL	HEXICAN-AHER	68498.	83914.	0.025	•.032	6.28	5.60	-0.69	4.63			
TOTAL	OTH HISPANIC	25874	71074.	0.009	0.027	6.49	5.72	0.77	7.76			
	SUBGROUPS CONDINED:	2769051	2577020.			9.98	8.89	-1.084	100.90	-0.21	-0.89	0.02
	BY: CIERTCINIE											
TOTAL	ACADEMIC	1323908.	1006476.	9.462	0.304	11.99	11.33	-0.66*	41.41			
TOTAL	VDC+GENERAL	1539272	1615794.	0.538	0.616	8.08	7.35	-0.73*	<u>58,59</u>			<b>-</b>
	SUBGROUPS CONSINED	2863189	2622270.			9.89	8.66	-1.01*	100.00	-0.31	-9.69	-0.01
	TOTAL	2543636	2391769.	0.914	0.900	9.78	8.66	-1.12#	91.74			
NON-PUBLI	C TOTAL	240395.	267189.	0.086	0.100	11.59	10.46	-1.144	8.26			
5	UDEROUPS CONDINED:	2784031.	2658958.			9.93	8.84	-1.10#	100.00	0.03	-1.12	-0.00
		_										
	BY: CONTINITY TYP	72	811844		. 1.89		8.21	-1.259	18.74			
	TUTAL	703309.	911044. 1904217	V.692 8 Ein	U.176	10.49	9.29	-1.20*	53.95			
SUBURDA		1901609. 419181	A48277	W.73V 8.210	0.110	9.97	8.52	-0.74*	27.31			
NUNAL	TUTAL	\$706010	26580F#	<u>¥1647</u>		40.0	8.84	-1.12*	100.00	-0.06	-1.11	0.05
	andaking roublike.	6/70719.	2030730.			,,,,,				-		

BEST COPY AVAILABLE

265

ERIC Full fact Provided by ERIC

D--2

#### INT READING

### GROUPING VARIABLE: TOTAL

			<u>HE I</u> 1972	<u>GHTED N</u> 1980	<b>PROF</b> <u>OF POP</u> 1972	ORTION ULATION 1980	<u>MEAN</u> 1972	<u>SCORE</u> 1980	1 <b>980-1</b> 97 DIFFERENCI	2 OF TOTAL CHANGE 2 DUE TO 5 SROUP	TOTAL CHANGE DUE TO POP. 	TOTAL CHANGE DUE TO SUBGROUP HEAN <u>CHANGES</u>	TOTAL CHANGE DUE TO INTER- ACTION
	-	BY: GEOG. REGION											
	TUTAL	NORTHEAST	777802.	605302.	0.272	855.0	10.55	9.57	-8.984	23 88			
	TOTAL	M. CENTRAL	833505.	78397 <b>0</b> .	0.291	0.295	9.97	9.21	-0.764	21.34			
	TOTAL	JUUIN	773939.	806228.	0.270	0.303	9.14	7.83	-1.31*	40.27			
			<u>478236.</u>	463458	0.167	0.174	9.88	9.01	87*	14.40			
		CONSTINCT.	<003402.	2058958.			9.89	8.84	-1.05*	100.00	-0.05	-8.99	-8.01
		BY: HOMELORK/AK											
	TOTAL	L.T. 5 HOLRS	1847371.	1004447									
	TOTAL	5 HOURS, HORE	100, 420.	655167	0.040	0.753	7.42	8.25	-1.17*	88.59			
	<b>SU</b>	BGROUPS CONBINED:	284 75 91.	2649615.		<u>29/_</u>	10.78	10.66		<u>11.41</u>			
							7.70	0.0>	-1.05*	100.00	-0.14	-0.80	-0.11
		BY: MORE AC EMPHAS	515										
	TOTAL	AGREE	1256334.	1750236.	0.503	· 0.718	9.12	A.79	-0 19#	A1 / 7			
	TOTAL	DISAGREE	<u>1240246.</u>	687661	0.497	0.282	10.49	9.15	-1 34#	<b>41.0/</b>			
	50	DEROUPS COMBINED:	24 <b>96580.</b>	2437897.			9.80	5.89	-0.91*	100 00	-0.30		
									V. 74	100.00	-0.30	-0.83	52.0
	TOTAL	DI: ZHHITE IN SCH	00L										
	TOTAL	U-54%	993257.	1096594.	0.357	0.433	8.83	8.03	-0.80*	42.03			
			1792723.	1433466.	9.643	<b>0.56</b> 7_	10.47	9.50	-9.96*	57.97			
	30	servers constrates:	2/05988.	2530061.			9.88	8.86	-1.02#	100.00	-0.13	-8.91	8.81
												••••	
	LESS THAN	R TATAI											
	NIGH SCHOOL	TÖTAL	47273.	10698.	9.022	0.004	6.20	3.36	-9 814	-7 40			
	VOC/JR COL	TOTAL	362691.	500240.	0.167	0.192	8.55	6.01	-2.0J+	-3.67			
	COLLEGE	TOTAL	<b>0</b> 02633.	906129.	0.305	0.348	9.48	7.73	-1.75#	20.00			
	GRAD/PROF	TOTAL	0613V3. 476164	659693.	0.379	0.254	12.22	10.60	-1.428	33.23			
	SUE	GROUPS COMBINED:	<u>-6/2120.</u> 914 9254	<u>523756</u> ,	0.127	0.201	13.48	11.70	-1.78#	8.41			
			CI07230.	2000330.			10.80	8.91	-1.89*	100.00	-8.61		
											-0.01	-1.00	-0.02
		BY: STUDY AIDS IN	N HOME										
	Z OR FENE	R TOTAL	566472										
	3 ON MORE	TOTAL	2297010	· • • • · / · • • / ·	V.176	0.307	8.34	7.40	-8.94#	41.40			
	5	USGROUPS COMBINED	286 3482	24540CA	<u>v. 802</u>	<u> </u>	10.27	9.47	-0.79*	56.40			
2	67						9.89	8.84	-1.05*	100.00	-8.21	- 0.00	
~	· ·										V. L.1	-4.02	-0.02

BEST COPY AVAILABLE



## BEST COPY AVAILABLE

D	-2
_	_

				STER M	PROPI	WTICH			1980-1972 Bresserie	% OF TOTAL CHAIRGE DUE TO	TOTAL CHANGE DUE TO POP.	TOTAL CHANGE DUE TO SUBGROUP MEAN	TOTAL CHARGE DUE TO INTER-
			1972	1989	1972	1000	1972	1088		DRUOT_		CHARGE 3	_ACTION
								470 <b>0</b>					
	BY:	PROJECTS/LABS	USED IN CO	URSES THIS	VEAR								
	HEVER, SELDUN	TOTAL	1438607.	1450696.	0.514	0.555	9.32	8.24	-1.08#	61.62			
	UTIENSIKEY.	IUIAL	1350603.	_1121409	486	0.435	_10.66	9.83	-9,84#	<u>38,38</u>			
	3000M	NL2 COUDTHED:	2/9/490.	2580015.			9.97	8.93	-1.04*	100.00	-0.07	-0.96	-0.01
	8Y :	ESSAYS USED I	N COURSES	THIS TEAR									
	NEVER , SELBOH	TOTAL	997056.	947335.	0.35*	0.369	9.17	7.87	-1.30#	46.90			
	OFTEN, FREQ.	TOTAL	<u>1007355.</u>	1617189.	9.644	<u> </u>	10.41	9.53	-0.87*	53.10	• ••		
	SUDGIN	OUPS COMBINED:	2804411.	2564524.			9.97	8.92	-1.05*	100.00	-9.92	-1.03	-4.41
	BY :		FT TEACH	ERS METH AD		EGNEES							
	8-49%	TOTAL	1907481.	1357989.	8.474	8 827	9.43		-1 814	44 12			
	50-100%	TOTAL	922414.	1210510.	9.326	0.473	18.42	9.11	-1.32#	51.68			
	SUDGA	OUPS CONDINED:	2029095.	2576507.			9.89	8.84	-1.05*	100.00	0.12	-1.12	-0.04
					-	-							
	671 671	PROPORTION OF	LAST TEAR	S GRADUATE	2 NON IN	COLLEGE							
	V-3V/. 30-407	TOTAL	322141.	475223.	0.114	0.183	8.01	7 <b>.55</b>	-9.46*	20.75			
	JV~47% 88_497	TOTAL	10/3570.	1011020.	0.379	0.389	9.45	8.51	-0.93#	35.59			
	70-3007	TOTAL	73//30. E02127	<b>072040</b> .	0.331	0.267	10.26	9.21	-1.05#	29.27			
			<u></u>	9504954		0.161	_11.35_	_10.57	-0.78*	14.39			
			2033002.	£370£3 <b>0</b> .			7.87	a. 85	-1.04#	100.00	-0.18	-0.89	0.04
	BA:	HIGH SCHOOL OF		LEU PLALEN	CHI COURS	63							
	TES AP	TUTAL	315417.	1200793.	0.310	9.466	10.47	9.28	-1.19#	43.75			
		IVIAL	1747498.	1375595.	<u> </u>	0.534	9.66	8.45		<u>_56.25</u>			
	automic	NA2 CONDINCO:	2202913.	23/0300.			9.92	8.84	-1.08*	100.00	0.12	-1.20	0.00
	BY:	SEMESTERS OF M	ATHEMATICS	TAKEN SOP	N-SENICH	YEAR							
	4 OR FEWER	TOTAL	2032255.	1650791.	9.719	0.621	9.27	7.98	-1.29#	71.26			
	S OR HORE	TOTAL	_831227.	1008166.	0,290	0.379	_11.38	10.24	-1.15#	28.74			
	SUDGR	OUPS COMBINED:	286 3482.	2658958.			9.89	8.84	-1.05*	100.00	0.19	-1.25	0.01
				KEN SOPH-SI	ENTOR TEA								
		TUTAL	9171EA4	1064019	8.7EA	8 748	. 14	8 10	-]. #A#	91.01			
0		TOTAL	489974	67002A	0.241	8.252	11.47	11.02	-8.44#	8.99			
EDIC		OUPS COMBINED:	206 3482 .	2658958.			9.89	8.84	-1.05*	100.00	9.92	-1.08	0.01
EKIC	69		3000.000	2000.200									
Full Text Provided by ERIC													

BY: SEMESTERS OF FOREIGN LANGUAGE TAKEN SOPH-SR YEAR	
3 ON FENER TOTAL 2093906. 2019138. 0.731 0.759 9.04 8.09 -0.95# 71.13 4 OR MORE TOTAL	
SUBGROUPS CONDINED: 2863482. 2658958. 9.89 8.84 -1.05% 100.00 -0.09 -0.76 0	ð. <b>9</b> 0
BY: PARTICIPATE IN ATHLETICS	
NO TOTAL 1560039. 1269913. 0.552 0.403 9.76 0.66 -1.094 49.56	
TES TOTAL <u>1267496, 1359451, 0,448 0,517 10,13 9,09 -1,04# 50.44</u> Subgroups Combined: 2828335, 2629364, 9,92 8,88 -1.04# 100.00 0.03 -1.07 0.00	0
NO 4-YR COL. TOTAL 1007236. 849489. 0.435 0.461 8.33 7.11 -1.22# 34.69	
SUBGROUPS COMBINER: 2315443, 2115440	
	D. <b>01</b>

D-2

₹.



272
## BEST COPY AVAILABLE

### D-3

### IRT MATHEMATICS Grouping Variable: Total

		<u>NE 16</u>	HTED N	<b>PROPO</b> <u>OF POPU</u> 1972	RTION LATION		SCORE	1980-1972 <u>DIFFERENCE</u>	% OF TOTAL Change Due to <u>Group</u>	TOTAL CHANGE DUE TO POP. SHIFT	TOTAL Change Due to Subgroup Mean <u>Changes</u>	TOTAL CHANGE DUE TO INTER- ACTION
		- // -	2/00	-//-	2700	177L	1700					
	BY: TOTAL											
TUTAL	JUBGROUPS COMBINED:	2862252.	<u>   2050440                              </u>	1.000	1.000	12.94	<u>11.90</u> 11.90	<u>-1.03</u> *	100.00	-0.00	-1.03	0.00
	RY: CFY											
TOTAL	HALF	1426314.	1213609.	0.408	0 474	13.79	12 81	-0 04#	65 A1			
TOTAL	FEMALE	1434921	1346152.	0.502	0.526	12.09	11.39	-0.704	55.01 66 00			
	SUBGROUPS COMBINED:	2861235.	2559761.			12.94	12.07	-0.87*	100.00	-0.04	-0.83	0.01
	BY: SES											
TOTAL	LOW	694282.	701703.	0.243	0.271	9.39	8.44	-0.95*	38.74			
TOTAL	MIDDLE	1461863.	1263636.	0.513	0.488	12.90	12.16	-0.74#	39.32			
TOTAL	HIGH	<u>696135</u>	<u>624635.</u>	0.244	0.241	16.62	<u>    15.83  </u>	-0,79#	21.94			
	SUBGROUPS COMBINED:	2852280.	2589974.			12.96	12.04	-0.92*	100.00	-0.11	-0.81	-0.01
	BY: RACE											
TOTAL	MHITE+AS+IND	2439673.	2153015.	0.881	0.831	13.90	12.97	-0.93*	74.28			
TOTAL	BLACK	235045.	284281.	0.085	0.110	6.50	6.69	0.19	13.01			
TOTAL	MEXICAN-AMER	68165.	82650.	0.025	0.032	8.02	7.54	-0.48	4.77			
TOTAL	OTH HISPANIC		<u>69596</u>	0.009	0.027	7.48	7,90	0.41	7,94			
	SUBGROUPS COMBINED:	2768720.	2589543.			13.07	11.97	-1.09#	100.00	-0.34	-0.81	0.05
	BY: CURRICULUM											
TOTAL	ACADEMIC	1323927.	1007295.	0.463	0.385	16.66	16.17	-8.49#	48.81			
TOTAL	VUC+GENERAL	1538022	_1606553	0.537	0.615	<u> </u>	9.33		_51.19			
	SUBGROUPS COMBINED:	2861949.	2613849.			12.94	11.96	-0.98*	100.00	-0.53	-0.45	0.01
		98499		7 .	-	<b>eo</b> 12	.79 11.	59 -1.20	# 97.01			
		2405	A7. 260310	<b>9</b> 0.0	A6 0.1	01 15	.37 14.	71 -0.67	2.99			
-		ED: 27628	00. 265044	6.		13	.01 11.	90 -1.11	# 100.00	. 0.04	-1.15	0.01
	BT: CUTTUNITY	11PE			-	182 12		98				
0	NDAN IUIAL	743	LOG. 90910 198 199299	C/. V.) 19 A 4	LJL V.J	L7C 10		.70 -1.1	2# 61.03			
5		1401.	RRA. 8441	07. 0.9	19 0.1	<b>110</b> 12	.15 11	.24 -0.9	1# 33.56			
		FD: 2796	151. 26504	46.		13	5.04 11	.90 -1.1	3* 100.00	0.07	-1.09	0.02

273

ERIC<sup>®</sup>

BEST COPY AVAILABLE

274

# BEST COPY AVAILABLE

## SEST COPY AVAILABLE

# D-3 COPY AVAILABLE

			WE10	HTED N	PROP OF POP	DRTION	MFAN	SCOPE	1980-1972 Diffedire	X OF TOTAL CHANGE OUE TO	TOTAL CHANGE DUE TO POP.	TOTAL CHANGE DUE TO SUBGROUP MEAN	TOTAL CHANGE DUE TO INTER-
			1972	1980	1972	1980	1972	1980	VALLENCINGE	OROOP	<u></u>	CHANGES	ACTION
	BY:	GEOG. REGION											
TOTAL		NORTHEAST	778067.	604620.	0.272	0.228	13.90	13 34	-0 67	15			
TOTAL		N. CENTRAL	833316.	783062.	0.291	0.295	13.29	12.50	-0.55	15.72			
TOTAL		SOUTH	773173.	805015,	0.270	0.304	11.95	10.07	-1.884	17.73			
TOTAL		NEST	477695.	<u>457749.</u>	0.167	0.173	12.37	12.04	-0.34	50.51			
	SUDER	UPS COMBINED:	2862252.	2650446.			12.94	11.90	-1.03#	100 00	-0.00	- 8 . 07	
										100.00	-0.00	-0.91	-0.04
TOTAL	D1:	TUTCHURK/NK											
TOTAL		E Martine Mone	1097193.	1967737.	0.648	0.753	12.11	10.86	-1.25#	99.03			
TUTAL			1001427.	<u>653119</u>	<u> </u>	0,247	14.52	<u>15.14</u>	9.62*	0.97			
	JODORI	OLD COUDINED.	20905/2.	2090050.			12.96	11.92	-1.04*	100.00	-0.25	-0.59	-0.19
	BY:		TE										
TOTAL		ACDEF	1955178	1 74 785 7									
TOTAL		DISACOFF	1255170.	1/43033.	7.373	0.717	12.06	11.83	-0.23	38.72			
	SUBGRO	UPS COMPTNEN:	2405274	2430848	<u></u>	<u></u>	13.60	12.35	<u> </u>	61.28			
			2973270.	2430000.			12.82	11.98	-0.85*	100.00	-0.33	-0.74	0.22
	BY:	ZOHITE IN SCHOOL	31										
TOTAL		8-89%	99984.8	1004210									
TOTAL		70-106/	1792958.	1427348	9 444		11.20	10.65	-0.634	42.21			
	SUBGRO	UPS COMBINED:	2784997.	2521344		<u><u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u><u></u></u></u>	12.00	12.11	0,89#	<u>_58.79</u>			
							12.94	11.95	-9.98*	100.00	-0.20	-9.80	9.0Z
	BY:												
LESS T	HAN HS	TOTAL	47971	18408			-						
HIGH S	CHOOL	TOTAL	362642	497081	0.022		7.00	5.26	-1.80	-3.24			
VOC/JR	COL	TOTAL	662941	003180	8 304	0.196	10.10	7.36	-2.74*	27.37			
COLLEG	8	TOTAL	821320.	459451	8.178	0.340 8 264	12.42	9.03	-2.19#	37.25			
GRAD/PI	ROF	TOTAL	275284	522490	0 127	0.234	17.00	14.87	-2.13*	37.98			
	SUDGR	OUPS COMBINED:	2169461.	2593643.			14 30			<u>₹.69</u>			
							14.30	15.00	-2.3.7	100.00	-0.09	-2.21	-0.00
	-												
	C C C L C C	TT STUDY AIDS	TH MOLE										
1,700		TUTAL	56548	6. 813344	. 0.19	0.30	7 10.0	66 B.A					
3 08		IUIAL	229676	5. 1037100	0.80	2.6.	313.	55 12.9	1 -0.01	- 34.44			
	300	CURSINE	DI 286225	2. 2650446			12.9	94 11.9	0 _1 01	- <u></u>			
										- TAA*AA	-4.34	-9.68	-0.02



BEST COPY AVAILABLE

# BEST COPY AVAILABLE

D-3

		<u>WEI(</u>	<u>SHTED N</u>	PROP <u>OF POP</u> 1972	DRTION ULATION 1980	<u></u>	<u>SCORE</u> 1980	1980–1972 <u>DIFFERENCE</u>	Z OF TOTAL CHANGE DUE TO <u>GROUP</u>	TOTAL CHANCE DUE TO POP. SHIFT	TOTAL ~"ANGE DUE TO SUBGROUP MEAN <u>CHANGES</u>	TOTAL CHANGE DUE TO INTER- ACTION
			COLESES THI	S YEAR		-//-	_/					
MEVER. C	ELECTRA TOTAL	1439303	1466106			19 63	10.00	-1 148	68.18			
NETEN.FI		1357758	1117818	0.313	0 414	16 16	11.51	-1.14*	33 00			
	SUBGROUPS CONBINED:	2797061.	2573005.			13.06	12.04	-1.03*	100.00	-0.11	-0.89	-0.03
	NY FRAYS USED I	N COURSES 1	THIS YEAR									
	HOON TOTAL	997045.	944795.	0.356	0.369	12.10	10.58	-1.52*	56.20			
OF TEN. FI	FQ. TOTAL	1806937.	1612411.		0.631	13.58	12.00	-0.70*	43.80			-0.01
	SUBGROUPS COMBINED:	2803982.	2557206.		-ŋ-	13.06	12.03	-1.02*	100.00	-9.92	-0.77	-0.01
	BY: MOROWIICH C	F FT TEACHE	RS NITH AD	ANCED DE	GREES							
0-49%	TOTAL	1906576.	1349955.	0.674	0.526	12.59	11.47	-1.12#	50.87			
50-100%	TOTAL	_922088.	1217193.	0.326	0.474	13.69	12.37	<u>-1.32*</u>	49,13		_	
	SUBGROUPS COMBINED	: 2828644.	2567148.			12.95	11.89	-1.05*	100.00	0.16	-1.19	-0.03
	BY: PROPERTION O	F LAST YEAR		ES NON IN		!						
0-30X	TOTAL	321401.	470968.	0.113	0.182	9.97	9.58	-0.39	26.92			
30-49%	TOTAL	° 972530.	1008366.	0.378	0.390	12.21	11.39	-0.82#	32.08			
50-69%	TOTAL	938296.	695090.	6.331	8.269	13.45	12.41	-1.04*	30.55			
70-100%	TOTAL	<u>_502581</u>	413334.	0.177	0.160	15.44	15.07		10.45			
	SUBGROUPS CONBINED	: 2834808.	2587758.			12.94	11.92	-1.02*	100.00	-0.29	-0.77	0.04
				IENT COUR	585							
	DI: TOTAL	A1617A	1197455	0.318	0.444	14.00	12.78	-1.22#	39.05			
163 AP	TOTAL	1744474	1370544	0.682	0.534	12.51	11.15	-1.36#	60.95			
NV AF	SUBGROUPS CONSINED:	2561814.	2568199.			12.99	11.91	-1.08*	100.00	9.22	-1.32	<b>U.</b> UZ



# BEST COPY AVAILABLE

D-3

	<u>NE16</u> 1972	<u>MTED N</u> 1980	PROP( <u>OF_POP(</u> 1972	DRTION <u>JLATION</u> 1980	<u>MEAN</u> 1972	<u>SCORE</u> 1980	1980-1972 <u>DIFFERENCE</u>	X OF TOTAL CHANGE DUE TO <u>GROUP</u>	TOTAL CHANGE DUE TO POP. <u>Shift</u>	TOTAL CHANGE DUE TO SUBGROUP MEAN <u>Changes</u>	TOTAL CHANGE DUE TO INTER- ACTION
BY: SEMESTERS CF	FOREIGN LA	NGUABE TAKE	N 90PH-9	R YEAR							
3 OR FENER TOTAL 4 OR MORE TOTAL	2092335. 769917.	2011342. 639104.	0.731 0.269	0.759 0.241	11.65 16.43	10.71 15.65	-8.94# -8.78#	72.40 27.60			
SUBGROUPS COMBINED:	2862252.	2650446.			12.94	11.90	-1.03*	100.00	-0.13	-0.90	-9.00
BY: PARTICIPATE	IN ATHLETIC	:5									
ND TOTAL	1559989.	1265014.	0.552	0.483	12.09	11.06	-1.02*	41.73			
SUBGROUPS COMPANIES	<u>1267227.</u> 1 <b>26</b> 27216		0,448	9.517	14.13	12.82	-1.31*	<u>58.27</u>			
		• ••••••••••			13.00	11.97	-1.03#	100.00	0.14	-1.15	-0.02
BY: SCHOOLINE HOT	HER NANTS	FOR STUDENT	,								
NO 4-YR COL. TOTAL	1006432.	845980.	0.435	0.401	10.23	9.07	-1.16*	26.99			
9-TH COLLEGE TUTAL	1308145.	1264195.	0.565	0.599	<u>16.37</u>	19.64		73.01		-1 48	-0.02
STORAALS FOUDTIED.	63143//.	CIIVI/3.			13.70	16.41	-7.5.4	100.00	W. C.I	-2.40	-0.05

250

279

٠



BEST COPY AVAILABLE

## APPENDIX E

Analysis of Covariance Partition of Score Change



#### Appendix E

### Analysis of Covariance Partition of Score Change

The analysis of covariance model can be written in the following computational form:

$$\alpha = (\overline{\mathbf{Y}}_{j} - \overline{\mathbf{Y}}_{k}) - \mathbf{b}_{p}(\overline{\mathbf{X}}_{j} - \overline{\mathbf{X}}_{k})$$

where

 $\alpha = \text{the treatment effect or equivalently the} \\ \text{difference between adjusted means.} \\ \mathbf{b}_p = \text{pooled within group regression coefficient.} \\ \overline{\mathbf{Y}_j} = \text{observed mean on the dependent variable in} \\ \text{group j.} \end{cases}$ 

 $\overline{X}_j$  = observed mean on the covariate in group j. Now more generally but in terms of the NLS and HSB cohort means we have:

$$\alpha = (\bar{Y}_{72} - \bar{Y}_{80}) - (\hat{\bar{Y}}_{72} - \hat{\bar{Y}}_{80})$$

where  $\overline{\bar{Y}}_{72}$  and  $\overline{\bar{Y}}_{80}$  are the estimated within cohort means where the estimation is from the vector of covariates. In the "step down" procedure used here the  $\overline{\bar{Y}}$ 's are first estimated using the covariate vector based on all the blocks (the full model) then reestimated leaving out the block of interest. The "net" affect of the block left out is the difference between the estimated mean in the reduced model and those in the full model or

$$\alpha_{\rm R} - \alpha_{\rm F} = (\hat{\bar{{\bf Y}}}_{72\,,{\rm R}} - \hat{\bar{{\bf Y}}}_{80\,,{\rm R}}) - (\hat{\bar{{\bf Y}}}_{72\,,{\rm F}} - \hat{\bar{{\bf Y}}}_{80\,,{\rm F}})$$



where

 $\alpha_R - \alpha_F =$  "net" effect of the block left out. If the difference based on the reduced model is large compared to the full model then the block left out has a relatively strong positive relationship with score decline. That is, it is a potential contributor to score decline. If the "net" effect of a block that is left out is negative then  $\hat{\overline{Y}}_{80,R} > \hat{\overline{Y}}_{72,R}$  indicating that changes in the covariate means in the "left out" block were in the direction of resisting score decline.

Inspection of either the standardized partial regression weights and/or the structure coefficients (i.e., the correlation between variables in a block and the covariate composite score for the block) help to identify which variables in the block contribute the most to the blocks' net effect. This information along with the variables correlation with the "dummy" code for group membership pretty much tell the story.



### IMPORTANCE OF INDIVIDUAL DEMOGRAPHIC, SCHOOL, AND STUDENT VARIABLES IN EXPLAINING SCORE DECLINES AS INDICATED BY STRUCTURE COEFFICIENTS GREATER THAN ± 20<sup>b</sup> (CORRELATIONS WITH COVARIATE COMPOSILE)

	VOCABULARY R=.61	<u>READ</u> R=.56	MATH <b>R-6</b> 9	r <sub>xd</sub> a
DENOGRAPHICS				
WAITE	.49	- 50	45	07
HICH SES	-48	-43	42	.07
SUBURBAN	.21			.00
NORTHEAST	•20			.05
STUDENT SCHOOL BEHAVIORS				
NUMBER OF HOMEWORK HOURS	.30	. 29	30	11
STUDENT EDUCATIONAL PLANS	.61	• 64	•30	- 02
NUMBER OF LABS	.28	.29	- 28	02
NUMBER OF ESSAYS	.26	-24		-01
SEVESTERS MATH	.39	-41	- 59	03
SEMESTERS SCIENCE	-48	-48	-54	.15
SEMESTERS FOREIGN LANGUAGE	.62	.59	-54	.23
ACADEMIC CURRICULUM	.67	.67	-68	-08
VOCATIONAL CURRICULUM	41	41	42	03
SCHOOL CHARACTERISTICS				
BUILDING CONDITIONS	-20			10
QUALITY OF ACADEMIC INSTITUTION	.31	.29	26	.10
REPUTATION	.31	.29	- 40	12
VOCATIONAL COUNSELOR	41	36	- 32	- 34
ACADEMIC EMPHASIS	.41	- 38	.32	-,,,,
NUMBER OF LABS (MEAN)	.25	.24	20	19
X WHITE TEACHERS	.31	-29	26	.10
2 DROPOUTS	28	24	23	- 14
2 WHITE STUDENTS	.38	-36	.33	-06
ACADEMIC CURRICULUM	.36	.29	-30	-04
SCHOOL TYPE	.21			02
Homework (mean)	.26	.24	.23	.33
SEMESTERS FOREIGN LANGUAGE	.33	.28	.23	.32
HOME EDUCATIONAL SUPPORT				
MOTHERS EDUCATIONAL LEVEL	-48	.43	.42	12
MOTHERS EDUCATIONAL LEVEL	.43	.38	.36	10
STUDY ATDS	.26	•57	.55	06
TOTAL REPO		.28	•28	.13

 $a_{r_{xd}}$  is the correlation between a "dummy" code ("1" = 72 cohort); "0" = 80 cohort and a given demographic, school, or atudent variable. For example a positive correlation between the number of homework hours and "dummy" code indicates that 1972 seniors on average raport doing more homework than 1980 seniors.

<sup>b</sup>Variables that are positively (negatively) related to the covariate composits for predicting achievement -- vocabulary, raeding, or mathematics and positively (negatively) related to the "dummy" cohort code indicates changes in demographics, student behaviors, or school characteristics that contribute to the acore decline. Variables that are positively related to the covariate composite but negatively related to the "dummy" cohort code are changing in a direction that resists the acore decline.

284

+U.S. GOVERNMENT PRINTING OPPICE: 1985-496-308

